

University of Huddersfield Repository

Mahmud, Hoger

Modelling and Developing Virtual Collaboration for Healthcare

Original Citation

Mahmud, Hoger (2019) Modelling and Developing Virtual Collaboration for Healthcare. Doctoral thesis, University of Huddersfield.

This version is available at http://eprints.hud.ac.uk/id/eprint/35243/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/



MODELLING AND DEVELOPING VIRTUAL COLLABORATION

FOR HEALTHCARE

Hoger Mahmud

A thesis submitted to the University of Huddersfield in partial fulfilment of the requirements for the degree of Doctor of Philosophy

The University of Huddersfield

December 2019

Copyright Statement

- The author of this thesis (including any appendices and/or schedules to this thesis) owns any copyright in it (the "Copyright") and s/he has given The University of Huddersfield the right to use such copyright for any administrative, promotional, educational and/or teaching purposes.
- II. Copies of this thesis, either in full or in extracts, may be made only in accordance with the regulations of the University Library. Details of these regulations may be obtained from the Librarian. This page must form part of any such copies made.
- III. The ownership of any patents, designs, trademarks and any and all other intellectual property rights except for the Copyright (the "Intellectual Property Rights") and any reproductions of copyright works, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property Rights and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property Rights and/or Reproductions

Abstract

The demand for healthcare services are rising as the world's population is increasing, and on average, people are living longer. As a result, healthcare services are becoming more complex to organise and expensive to provide. Researchers in the field are arguing that the current centralised model of healthcare provision cannot address challenges relating to service cost, quality and availability and there is a need to decentralise it. To decentralise, virtual collaboration systems where healthcare organisers, providers and receivers can work together and share resources across time and space are seen as the future of healthcare.

Literatures suggest that a modelling framework specific to healthcare virtual collaboration is yet to be developed and there are unaddressed challenges relating to the organisation and management aspects of healthcare virtual collaboration. In this thesis, Virtual Breeding Environment (VBE) and Virtual Organisation (VO) concepts are used as theoretical bases to answer research questions relating to modelling, organising and managing virtual collaboration for healthcare. To contribute to the modelling aspect of virtual collaboration in healthcare and address organisational and managerial challenges of healthcare virtual collaboration the first objective of the thesis is to develop a modelling framework to enable system developers model healthcare virtual collaboration in terms of participants and services classification, representation and descriptions. The second objective is to develop a framework based on concepts developed in the modelling framework, to be used as guide to develop systems for organising and managing healthcare virtual collaborations.

To achieve the objectives, a deductive research approach is used to develop theoretical frameworks first, and later implement and evaluate the frameworks. For evaluation purpose, Technology Acceptance Model (TAM) is modified three times by adding new constructs. The extended TAMs are used as theoretical evaluation frameworks to test the acceptability of the technologies developed in this thesis. For each extension, a set of hypotheses are defined to be tested by prospective target users. Survey questionnaire is used as a data collection method, and Structural Equation Modelling (SEM) technique is used to analyse the collected data statistically in AMOS software.

The first contribution of this thesis is a Healthcare Virtual Breeding Environment Modelling Framework (HC-VBE-M-F) which consists of a service and participant classification mechanism, a domain specific modelling language and a service orchestration description language. The frame work is implemented as a Java application and has been tested for acceptance by system developers.

The second contribution is a Healthcare Virtual Breeding Environment Framework (HC-VBE-F) which is based on the first framework. The framework consists of a conceptual description, a member selection mechanism, a service level agreement creation and management mechanism and a provider verification and validation mechanism. The Framework is implemented as a mobile application and it is evaluated by healthcare requesters and providers for acceptance.

The evaluation results show that the frameworks are both acceptable by prospective users and their intensions to use systems developed based on the two developed frameworks are positive and validated empirically. The achievements of this thesis are the developed mechanisms and frameworks that together facilitate the modelling and development of healthcare virtual collaboration systems based on VBE and VO concepts.

Table of Contents

Copyright Statement	1
Abstract	2
Table of Contents	3
List of Figures	8
List of Tables	
Acknowledgements and Dedications	
List of Abbreviations	15
Chapter 1 : Introduction	
1.1 Research Introduction	
1.2 Research Motivations	
1.3 Research Challenges and Questions	
1.4 Research Aims and Objectives	22
1.4.1 Research Objectives	
1.5 Target Groups	24
1.6 Thesis Contributions	25
1.7 Thesis Structure	28
1.8 Chapter 1 Summary	
Chapter 2 : Background	
2.1 Collaboration	
2.2 Virtual Collaboration	
2.3 Virtual Community	
2.4 Healthcare Virtual Community	
2.5 Virtual Breeding Environment and Virtual Organization	
2.6 Modelling Virtual Collaboration	40
2.7 Service Level Agreement	40
2.8 Blockchain technology	
2.9 Technology Acceptance Model	
2.10 Structural Equation Modelling	45

3

2.11 Chapter 2 Summary	47
Chapter 3 : Literature Review	49
3.1 Virtual Collaboration Models	49
3.2 Virtual Collaboration Frameworks for Healthcare	50
3.3 SLA Creation and Management	51
3.4 Partner Selection Mechanism	53
3.5 Service Orchestration	54
3.6 TAM for Healthcare Applications	54
3.7 Chapter 3 Summary	55
Chapter 4 : Methodology	57
4.1 Research Methodology	57
4.2 Research Design	58
4.2.1 Research Design Steps Description	59
4.3 Evaluation Strategy and Methods	61
4.3.1 Frameworks Implementations (Step 1)	62
4.3.1.1 Modelling Framework Implementation	62
4.3.1.2 HC-VBE Framework Implementation	65
4.3.2 TAM Extension and Hypotheses Formulation (Step 2)	70
4.3.2.1 TAM Extension for HC-VBE Modelling Framework	71
4.3.2.2 TAM Extension for HC-VBE-F Providers	73
4.3.2.3 TAM extension for HC-VBE-F Requester	75
4.3.3 Research Instruments (Step 3)	
4.3.4 Research Measurements (Step 4)	
4.3.5 Sampling Methods and Outcomes (Step 5)	79
4.3.6 Applications Testing and Data Collection (Step 6)	81
4.3.6.1 HC-VBE-M-F Application Testing Method	82
4.3.6.2 HC-VBE-F Mobile Application Testing Method	
4.3.7 Data Analysis (Step 7)	85
4.3.7.1 HC-VBE-Modelling Framework AMOS Model	
4.3.7.2 HC-VBE-F-Provider AMOS Model	
4.3.7.3 HC-VBE-F-Requester AMOS Model	

4.3.8 Statistical Results Selection (Step 8)	89
4.3.8.1 Mean	
4.3.8.2 Standard Deviation	
4.3.8.3 Factor Loading	
4.3.8.4 Significant <i>p</i> value	
4.3.8.5 Correlation Coefficient Matrix	
4.3.8.6 AVE (Average Variance Extracted)	
4.3.8.7 CR (Construct Reliability)	
4.3.8.8 Cronbach's Alpha	
4.3.8.9 Kaiser-Meyer-Olkin	
4.3.8.10 Model Fit Indices	
4.4 Chapter 4 Summary	
Chapter 5 : Modelling Framework	
5.1 Healthcare Virtual Collaboration Modelling Framework	
5.2 Service and Participant Classification Mechanism	100
5.2.1 SPCM Description	103
5.3 HC-VBE Domain Specific Modelling Language	106
5.3.1 UML Use Case Notation Extension	109
5.3.2 A Modelling Example	113
5.3.2.1 Using HC-VBE-Leicester to provide care	114
5.4 Service Orchestration Description Language	116
5.4.1 HC-VBE-OM Introduction	118
5.4.2 Orchestration Description Language	122
5.4.3 Orchestration Description Example	125
5.5 HC-VBE-M-F Evaluation Results	130
5.5.1 Overall Analysis Results	130
5.5.2 Intermeasurement Correlation	134
5.5.3 Characteristics of Constructs	136
5.5.4 Data Validity and Reliability	137
5.5.5 Model Fit Indices	138
5.6 HC-VBE-M-F Discussions	140
5.6.1 The HC-VBE-M-F Limitations	144

5.7 Chapter 5 Summary	144
Chapter 6 : Collaboration Framework	146
6.1 Healthcare Virtual Collaboration Framework	146
6.2 The HC-VBE Framework Description	147
6.2.1 The HC-VBE Framework Concept	149
6.2.2 Framework Component Description	152
6.2.2.1 VBE Initiation Manager	152
6.2.2.2 Negotiations and Creation	155
6.2.2.3 Monitoring Manager	158
6.2.2.4 Repository Manager	159
6.2.2.5 VO Monitoring Manager	160
6.2.3 The HC-VBE Framework Working Steps	160
6.3 HC-VBE Member Selection Mechanism	165
6.3.1 Member Selection Mechanism Steps	167
6.4 HC-VBE Provider Verification and Validation Mechanism	169
6.4.1 HC-VBE-PVVM Description	170
6.4.1.1 PVVM Working Steps	172
6.4.1.2 HC-VBE-PVVM Application Example	174
6.5 HC-VBE Service Level Agreement Templates	175
6.5.1 HC-VBE Master SLA Template	179
6.5.2 HC-VBE General SLA Template	182
6.6 HC-VBE-F Evaluation Results	192
6.6.1 HC-VBE-TAM-Provider Evaluation Result	192
6.6.1.1 Overall Data Analysis Results	192
6.6.1.2 Intermeasurement Correlation	196
6.6.1.3 Characteristics of Construct	198
6.6.1.4 Data Validity and Reliability	199
6.6.1.5 HC-VBE-TAM-Provider Model Fit	200
6.6.2 HC-VBE-Requester Evaluation Results	202
6.6.2.1 Overall Data Analysis Results	203
6.6.2.2 Intermeasurement Correlation	206
6.6.2.3 Characteristics of Construct	208

-

6.6.2.4 Data Validity and Reliability	209
6.6.2.5 HC-VBE-TAM-Requester Model Fit	210
6.7 HC-VBE-F Discussions	212
6.7.1 HC-VBE-TAM-Provider Discussion	215
6.7.2 HC-VBE-TAM-Requester Discussion	218
6.7.3 HC-VBE-TAM Providers and Requester Comparison	220
6.7.4 HC-VBE Framework Limitations	222
6.8 Chapter 6 Summary	223
Chapter 7 : Conclusion and Future Works	224
7.1 Conclusion	224
7.2 Research Achievements	225
7.3 Published and Planned Publications	227
7.4 Future Works	229
Bibliography	232
Bibliography	232 257
Bibliography Appendices Appendix A: Research survey consent form	232 257 259
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire	232 257 259 261
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire	232 257 259 261 267
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-M-F System Developer Questionnaire	232 257 259 261 267 273
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-M-F System Developer Questionnaire Appendix E: Official permissions obtained to conduct research	232 257 259 261 267 273 279
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-M-F System Developer Questionnaire Appendix E: Official permissions obtained to conduct research Appendix F: All data collected for the HC-VBE-M-F	232 257 259 261 267 273 279 281
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-M-F System Developer Questionnaire Appendix E: Official permissions obtained to conduct research Appendix F: All data collected for the HC-VBE-M-F Appendix G: All data collected for the HC-VBE-F from healthcare professionals	232 257 259 261 267 273 279 281
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire. Appendix C: HC-VBE-F healthcare requesters survey questionnaire. Appendix D: HC-VBE-F healthcare requesters survey questionnaire. Appendix D: HC-VBE-M-F System Developer Questionnaire Appendix E: Official permissions obtained to conduct research Appendix F: All data collected for the HC-VBE-M-F Appendix G: All data collected for the HC-VBE-F from healthcare professionals Appendix H: All data collected for the HC-VBE-F from healthcare requesters.	232 257 259 261 267 273 279 281 281 282
Bibliography Appendices Appendix A: Research survey consent form Appendix B: HC-VBE-F healthcare professionals survey questionnaire Appendix C: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-F healthcare requesters survey questionnaire Appendix D: HC-VBE-M-F System Developer Questionnaire Appendix E: Official permissions obtained to conduct research Appendix F: All data collected for the HC-VBE-M-F Appendix G: All data collected for the HC-VBE-F from healthcare professionals Appendix H: All data collected for the HC-VBE-F from healthcare requesters Appendix I: HC-VBE-F mobile application screenshots and explanations	232 257 259 261 267 273 273 281 281 282 284

List of Figures

Figure 1.1: The healthcare virtual collaboration vision of the thesis which shows a
number of rectangles representing VBEs with a possible content of a given VBE
shown in the middle
Figure 1.2: Research objectives, objective 2 is developed based on objective 1
Figure 1.3:The thesis structure, showing the sections of each chapter
Figure 2.1: Virtual collaboration settings between human and machine participants 34
Figure 2.2: Virtual communities in a virtual environment
Figure 2.3: VBE and VO life-cycle showing the three stages (initiation, operation and
dissolution)
Figure 2.4: Blocks linked in a chain showing the concept of Blockchain technology 42
Figure 2.5: Original Technology Acceptance Model developed by Davis in 1989
Figure 4.1: General search design used to carry out the research in this thesis
Figure 4.2: Evaluation strategy steps used to evaluate both the HC-VBE-M-F and the HC-
VBE-F frameworks62
Figure 4.3: Implemented HC-VBE-M-F Java application screenshots showing framework
notations and different stages of model development
Figure 4.4: Implemented Orchestration Description Language screenshots and stages of
the description
Figure 4.5: HC-VBE mobile application implementation technologies used to develop the
view layer, the application layer and the database
Figure 4.6: HC-VBE-F mobile application architecture showing the three layers
Figure 4.7: Mobile application screenshots for the prototype working steps developed
based on the HC-VBE-F 70
Figure 4.8: Extended TAM for HC-VBE-M-F showing the added PAM construct
Figure 4.9: Extended TAM for HC-VBE-F for healthcare professionals showing the two
newly added constructed (PCE and PHG)74
Figure 4.10: Extended TAM for HC-VBE Framework for prospective patients showing the
two newly added constructs PHQ and PHA76
Figure 4.11: A screenshot of sample data collected based on the surveyed

8

questionnaires developed for testing both frameworks
Figure 4.12: HC-VBE-M-F Application Evaluation and data collection Steps
Figure 4.13: HC-VBE mobile application evaluation and data collection steps
Figure 4.14: A screenshot of AMOS model created based on one of the extended TAM
and some computed statistical results for the model
Figure 4.15: HC-VBE-M-F-TAM Structural Equation Model developed in AMOS
Figure 4.16: HC-VBE-F-TAM-Provider Structural Equation Model developed in AMOS 88
Figure 4.17: HC-VBE-M-TAM-Requester Structural Equation Model developed in AMOS
Figure 4.18: Structural Equation Model components
Figure 4.19: Structure of correlation coefficient results table
Figure 5.1: HC-VBE Modelling framework components (SPCM, DSML and SODL)
Figure 5.2: Summary of healthcare roles and services identified as a result of studying
resources in the form of articles and websites101
Figure 5.3: Refined healthcare roles, services and environment classifications 102
Figure 5.4: Healthcare service and participants classification mechanism mapped to the
six classes of stakeholders and services104
Figure 5.5: UML use case diagram actor and use case graphical notations
Figure 5.6: UML profile for organiser actor extended from the UML actor notation in this
thesis 109
Figure 5.7: The service and participants classification mechanism mapping to the
developed DSML for healthcare virtual collaboration
Figure 5.8: Use case Diagram for the HC-VBE-Leicester collaboration scenario modlled
uing the developed DSML114
Figure 5.9: Use case diagram for the HC-VO collaboration (SSI-Prevention) modelled
using the developed DSML 116
Figure 5.10: BPEL orchestration process. showing how requests are orchestrated 118
Figure 5.11: the proposed HC-VBE Orchestration mechanism components 119
Figure 5.12: HC-VBE Orchestration mechanism mapping to BPEL 121
Figure 5.13:Sentence structure of the textual orchestration description language
developed 123

Figure 5.14: HC-VBE-TAM-M-F accepted hypotheses 144
Figure 6.1: Components of the healthcare virtual collaboration framework which consist
of a conceptual framework, a member selection mechanism, a provider validation
and verification mechanism and SLA templates147
Figure 6.2: Illustrating the HC-VBE-F description on conceptual, component and process
levels
Figure 6.3: The HC-VBE framework general structure that shows the classification of
collaborating participants and HC-VBE and HC-VO created based on SLA
Figure 6.4: HC-VBE components grouped in 5 groups (Initiation Manger, Negotiation and
Creation Manager, VBE Monitoring Manager, Repository Manager and VO
Monitoring Manager) 152
Figure 6.5: OCL to OWL mapping between OCL expression and OWL tags taken from
[250]
Figure 6.6: A view of the convertor use case working mechanism
Figure 6.7: Master SLA concept where a number of SLA can be linked together under one
Master SLA 160
Figure 6.8. HC-VBE-E processing steps showing the steps numbered and their directions
righte 0.0. He vie i processing steps showing the steps numbered and their directions
specified
specified
specified
specified
 specified
specified
 specified
 specified
 Figure 6.9: Request report example with possible contents
righter 6.8: HC-VBE-F processing steps showing the steps numbered and their directions specified 161 Figure 6.9: Request report example with possible contents 162 Figure 6.10: The framework UML activity diagram that show the flow of activity stage by stage 164 Figure 6.11: A seven steps HC-VBE Member Selection Mechanism, developed to work within the HC-VBE-F 167 Figure 6.12: HC-VBE -MSM activity diagram showing the flow of activities to select a participant 169 Figure 6.13:HC-VBE provider verification and validation mechanism steps and components 172
rigure 0.3. HC-VBE-F processing steps showing the steps numbered and their directions specified 161 Figure 6.9: Request report example with possible contents 162 Figure 6.10: The framework UML activity diagram that show the flow of activity stage by 162 stage 164 Figure 6.11: A seven steps HC-VBE Member Selection Mechanism, developed to work 164 Figure 6.12: HC-VBE-F 167 Figure 6.12: HC-VBE -MSM activity diagram showing the flow of activities to select a 167 Figure 6.13: HC-VBE provider verification and validation mechanism steps and 169 Figure 6.13: HC-VBE provider verification and validation mechanism steps and 172 Figure 6.14: A section of the OWL version of the Master SLA
Figure 6.10: The Framework UML activity diagram that show the flow of activity stage by stage 161 Figure 6.10: The framework UML activity diagram that show the flow of activity stage by stage 164 Figure 6.11: A seven steps HC-VBE Member Selection Mechanism, developed to work within the HC-VBE-F. 167 Figure 6.12: HC-VBE -MSM activity diagram showing the flow of activities to select a participant. 169 Figure 6.13:HC-VBE provider verification and validation mechanism steps and components 172 Figure 6.14: A section of the OWL version of the Master SLA. 177 Figure 6.15: HC-VBE-SLA template package components 177
Figure 6.18: HC-VBE-F processing steps showing the steps numbered and their directions specified 161 Figure 6.9: Request report example with possible contents 162 Figure 6.10: The framework UML activity diagram that show the flow of activity stage by stage 164 Figure 6.11: A seven steps HC-VBE Member Selection Mechanism, developed to work within the HC-VBE-F. 167 Figure 6.12: HC-VBE -MSM activity diagram showing the flow of activities to select a participant. 169 Figure 6.13: HC-VBE provider verification and validation mechanism steps and components 172 Figure 6.14: A section of the OWL version of the Master SLA. 177 Figure 6.15: HC-VBE-SLA template package components 177 Figure 6.16: A screenshot of the Mater SLA template implementation in Protégé 178
rigure 0.3. He-VBL-F processing steps showing the steps numbered and their directions specified 161 Figure 6.9: Request report example with possible contents 162 Figure 6.10: The framework UML activity diagram that show the flow of activity stage by stage 164 Figure 6.11: A seven steps HC-VBE Member Selection Mechanism, developed to work within the HC-VBE-F. 167 Figure 6.12: HC-VBE -MSM activity diagram showing the flow of activities to select a participant 169 Figure 6.13:HC-VBE provider verification and validation mechanism steps and components 172 Figure 6.14: A section of the OWL version of the Master SLA 177 Figure 6.15: HC-VBE-SLA template package components 177 Figure 6.16: A screenshot of the Mater SLA template implementation in Protégé 178 Figure 6.17: Text based Master SLA template developed for the HC-VBE-F 179

Figure 6.18: The main six clauses of the developed general SLA template for HC-VBE. 182
Figure 6.19: HC-VBE-SLA template Clause 1 structure and content 183
Figure 6.20: HC-VBE-SLA template Clause 2 structure and content
Figure 6.21: HC-VBE SLA template Clause 3 structure and content
Figure 6.22: HC-VBE-SLA template Clause 4 structure and content
Figure 6.23: HC-VBE-SLA Template Clause 5 structure and content
Figure 6.24: HC-VBE-SLA template Clause 6 structure and content
Figure 6.25: HC-VBE-TAM-provider accepted hypotheses results with bold lines
indicating the accepted hypotheses 218
Figure 6.26: HC-VBE-TAM-requester accepted hypotheses with bold lines indicating the
accepted hypotheses 220

List of Tables

Table 2.1: Structural Equation Model Notations 45
Table 2.2: A comparison between Structural Equation Model and Traditional Statistical
Method summarised from [146] 46
Table 3.1: e-Contract Mechanism Examples 52
Table 4.1: System developers' demographic data who participated in the survey after
using the Java application developed based on the HC-VBE-M-F
Table 4.2: Healthcare providers and requesters demographic data that tested the
mobile application developed based on the HC-VBE-F
Table 4.3: Recommended model fit indices and recommended values 96
Table 5.1: The UML use case notation extension description that make up the developed
DSML 112
Table 5.2: Orchestration mechanism component description 119
Table 5.3:BPEL4People tag descriptions used in the mapping process
Table 5.4: The developed Textual orchestration description language reserved words 124
Table 5.5:Orchestration description language logical operators 124
Table 5.6:Healthcare provider requirement specification requested by Mr. David 126
Table 5.7:(TreatWound) orchestration described using the Orchestration description
language
Table 5.8: TreatWound Orchestration Explanation 127
Table 5.9: Overall data analysis results for HC-VBE-TAM-Modelling framework which
includes the Mean, Std., Factor loading and Significance value
Table 5.10: HC-VBE-TAM-Modelling framework intermeasurement correlations 135
Table 5.11: Statistical results for HC-VBE-M-F-TAM construct characteristics
Table 5.12: HC-VBE-M-F-TAM data reliability and validity results
Table 5.13: HC-VBE-M-F-TAM fit results140
Table 5.14: HC-VBE-M-F-TAM hypotheses testing results
Table 6.1: Master SLA clause 1 description 180
Table 6.2: Master SLA clause 2 description
Table 6.3 : The general SLA template first clause description 183

Table 6.4: The general SLA template second and third clause description
Table 6.5: The general SLA template fourth clause description 187
Table 6.6: The HC-VBE-SLA template fifth clause description
Table 6.7: The HC-VBE-SLA template fifth clause description
Table 6.8: HC-VBE-TAM-provider overall data analysis results which includes the Mean,
Standard deviation, Factor loading and Significance value
Table 6.9:HC-VBE-TAM-provider intermeasurement correlation results
Table 6.10: HC-VBE-TAM-provider construct characteristics results which includes Mean
and Standard Deviation198
Table 6.11: HC-VBE-TAM-provider Data validity and reliability results
Table 6.12: HC-VBE-TAM-provider model fit results 202
Table 6.13: HC-VBE-TAM-Requester overall analysis results which includes Mean,
Standard deviation, Factor loading and Significance value
Table 6.14: HC-VBE-TAM-Requester intermeasurement correlations
Table 6.15: HC-VBE-TAM-Requester construct characteristics results 208
Table 6.16: HC-VBE-TAM-requester data reliability and validity results
Table 6.17: HC-VBE-TAM-Requester model fit results
Table 6.18: HC-VBE-TAM-provider hypotheses testing results
Table 6.19: HC-VBE-TAM-requester hypotheses testing results
Table 6.20: Comparison between HC-VBE-TAM-provider and requester hypotheses
testing results

Acknowledgements and Dedications

My PhD journey was lonely and difficult at times; however, it would have been a lot more difficult without the kind support and help I received from many individuals who I would like to take this opportunity to thank.

First of all, I would like to thank my supervisor Professor Zhongyu (Joan) Lu whole heartedly for her encouragement and support throughout my journey. Her belief in me was empowering and her guidance and feedbacks were tremendously helpful in the process of completing this thesis. I would also like to thank my co-supervisor Dr. Qiang Xu for his help and feedbacks at different stages of my journey.

I would like to thank my wife and three children (Laily, Nela and Elam) for their unconditional love and support which kept me going for the last six years.

I would like to dedicate this thesis to my deceased mother who will always be in my thoughts and prayers. To my father, your love and advices have always been the source of energy for me, especially in completing this work.

Finally, I am grateful to all the people who have assisted me, especially the Huddersfield Computing and Engineering School staffs, my friends, colleagues and students from the University of Human Development, Kurdistan.

List of Abbreviations

AMOS: Analysis of Moment Structures
AU: Atitude towards Using
BPEL: Business Process Execution Language
DSML: Domain Specific Modelling Language
HC-VBE: Healthcare Virtual Breeding Environment
HC-VBE-A: Healthcare Virtual Breeding Environment Actors
HC-VBE-E: Healthcare Virtual Breeding Environment Environement
HC-VBE-F: Healthcare Virtual Breeding Enviroment framewrok
HC-VBE-M-F: Healthcare Virtual Breeding Environment Modelling Framewrok
HC-VBE-MSM: Healthcare Virtual Breeding Environment Member Selection Mechanism
HC-VBE-ODL: Healthcare Virtual Breeding Environment Service Orchestration
Description Lnaguge
HC-VBE-OM: Healthcare Virtual Breeding environemnt Orchestration Mechanism
HC-VBE-R: Healthcare Virtual Breeding Environment Relationships
HC-VBE-U: Healthcare Virtual Breeding Environement Use cases
HC-VO: Healthcare Virtual Organisation
HVCom: Healthcare Virtual Community
ICT: Information and Communication Technology
IU: Intension to Use
OCL: Object Constaint Language
OWL: Web Ontology Language
PAM: Percieved Ability to Model
PCE: Perceived Clinical Effectiveness
PEU: Percieved Ease of Use
PHA: Percived Healthcare Availability
PHG: Perceived Healthcare Globalisation
PHQ: Perceived Healthcare Quality
PU: Perceived Usefulness
SEM: Structural Equation Modelling

SLA: Service Level Agreement

SODL: Service Orchistration Description Language

SPCM: Service and Participant Classification Mechanism

TAM: Technology Acceptance Model

UML: Unified Modelling Language

VBE: Virtual Breeding Envirnment

VC: Virtual Collaboration

VCom: Virtual Community

VO: Virtual Organisation

This chapter introduces the research topic and put it into context through providing research motivations and challenges in the Sections 1.2 and 1.3. The research aims and objectives are outlined in Section 1.4 and the target groups of the research are defined in Section 1.5. The contributions of the research are listed and described in Section 1.6 and the thesis structure is presented in Section 1.7. Finally, a summary of the chapter is provided in Section 1.8.

1.1 Research Introduction

Healthcare is a fundamental human right and the right was reaffirmed in the world leaders meeting in Alma-Ata [1]. To fulfil their obligations, governments around the world are investing a lot of time and resources in their healthcare systems with the aim to improve care availability, increase timely-care provision and make healthcare more affordable to provide and receive [2][3][4]. According to [5], healthcare spending takes up 17% of US Gross Domestic Product (GDP) and it is expected to rise from 17.4% to 19.6% by 2024. Developed countries such as the United Kingdom and France are also investing more than 10% of their GDP in their healthcare systems [6].

A major reason for the increase in spending is down to the fact that throughout the world the number of population is rising and the demand for healthcare services are growing [7]. For instance, in England, patients with long-term health conditions reached 15 million in 2014 and 70% of available beds and half of General Practitioners' visits went to this type of patients [8]. It is argued by the authors of [6] and [9] that the current centralised model of healthcare cannot answer the complex organisational and managerial questions arises during care provision; and cannot deal with today's healthcare challenges where more timely collaboration and resource sharing is required. In support of this claim, researches such as [10] and [11] indicate that despite the continuous efforts by

governments around the world to improve healthcare, challenges such as cost and timely access to healthcare remain at large. These challenges, if left unaddressed, can lead to inefficiency and patient dissatisfaction as [12] states that access, availability and cost are essential patient satisfaction attributes in healthcare.

To address collaboration and resource sharing challenges in healthcare, the authors of [9] and [13] claim that healthcare provision has to be decentralised in order to increase opportunities to recruit providers and resources for a given care, when and where needed. To decentralise healthcare, cross stakeholder collaborations that can take place regardless of geographical boundaries is seen as the way to provide care in future [14][15][16]. To facilitate the decentralised collaborations, virtual healthcare is emerging as a strong option [17] [18] [19] [20] [21][22]. Virtual care is becoming a reality and it is *"no longer a futuristic idea"* claimed in a report published by the centre for health solution on virtual care programs (Deloitte) in 2018 [23]. The report also states that *"virtual care is a must-have"* for modern healthcare systems as it has the potential to diversify care delivery channels to patients, provide *"convenience and access"* and support *"patient-centricity"*.

Virtual collaboration concept is important to healthcare because collaborating within virtual settings provide the platform for patients to seek support and information beyond the scope of healthcare institutions [24]. Virtual care is made possible by new information and communication technologies which has encouraged team-based collaborations to provide care and has transferred the traditional single care provider model to multi care provider model [25]. In virtual care, technologies such as online messaging, chat rooms and videoconferencing are used as a medium to connect healthcare requesters and providers. During a typical healthcare provision many stakeholders (e.g. healthcare professionals and government agencies) will have to collaborate and share resources which make healthcare provision a complex process. For example, in England, privately run GP practices, community trust, mental health trust and acute care trust are some of the organisations involved in providing care [8]. To cope with the complexity of bringing all required parties together to provide a given care, the authors in [26] suggest that

efficient collaboration mechanisms that are capable of aligning stakeholders and resources are required.

Healthcare virtual collaboration presents many organisational and managerial challenges that requires an environment capable of regulating and formalising the collaboration process. To manage and organise virtual collaboration, researchers have developed the concepts of Virtual Breeding Environment (VBE) [27] and Virtual Organisation (VO) [28]. Researchers are becoming interested in the concepts as virtual collaboration among people are becoming the norm and communication technology advances [29][30] [31][32]. VBE is a long-term alliance between a number of key stakeholders, supported by information and communication technologies that provide the environment for the formation and operation of short-term and goal-based collaborations known as VO [33]. To put the concepts into perspective for healthcare, VBE can serve as a virtual hospital where the need of a patient is addressed by recruiting healthcare professionals and resources on the fly and dedicate a virtual space (VO) for collaboration and service provision for a specific duration and to achieve a specific goal.

In this thesis, VBE and VO concepts have been used as theoretical bases to develop a Healthcare Virtual Collaboration Modelling Framework (HC-VBE-M-F) to enable healthcare system developers understand and model healthcare virtual collaboration scenarios. The concepts are also used to develop a Healthcare Virtual Collaboration Framework (HC-VBE-F) to address organisational and managerial challenges of healthcare virtual collaboration. The frameworks are described and empirically evaluated using Technology Acceptance Model (TAM) [34] in Chapters 5 and 6 respectively.

1.2 Research Motivations

The motivations behind modelling and developing virtual collaboration for healthcare are summarised in the following points:

1- In many fields of science such as engineering, models and visualisation techniques are used to simplify complex scenarios with the aim to facilitate better understandings and manage implementation risks. [35] States that a modelling framework to model the divergent aspect of virtual collaboration in a given community would be interesting. However, "Visualisation of aspects of virtual collaboration has received relatively little research attention in the past" [36], and a similar claim is also made by the authors of [37]. Researchers in the field of virtual collaboration are yet to offer a generic design modelling framework as claimed by authors of [38]. The prospect of providing a modelling framework capable of modelling virtual collaboration for healthcare in a simple, understandable and comprehensive manner is one of the motivations driving this research.

2- In healthcare, timely service provision is crucial for patients and could mean life if the service is provided on time or death if the service is not provided on time. The idea that virtual collaboration could provide services in a timely fashion and across borders, makes the concept very attractive and provide a motivation to research it for possible contributions that it can make to healthcare.

3- Cost plays a great role in the provision of every service, normally high cost of a service means it cannot be provided as often as necessary to those who are in need of the service. To reduce cost, healthcare virtual collaboration as an alternative to the current centralised model of healthcare has the potential to make a great contribution by providing a platform for healthcare services to be provided without the need for patients to travel and use local healthcare resources. The possibility of reducing cost of care and increasing healthcare availability though the utilisation of virtual collaboration, provides a motivation to research it.

4- Service quality attributes such as staff professionalism and service relevance are some of the service quality measuring factors that healthcare providers shoulder most of the responsibilities. In many occasions, time and energy is wasted and service quality is compromised by incompetent service providers which they claim to have a good track record. In virtual collaboration, during member recruitment, the profile of members could be screened and validated electronically before job assignments which can raise the quality of care.

20

5- For an effective healthcare provision, promptness is required, in a typical situation professional, family members, administrators, local authorities and many more have to work together to deal with the situation. Brining all these parties together is a highly complex and time-consuming venture with undesirable consequences if not managed properly. Virtual collaboration, through a described and guided framework, can cater for the management and promptness required especially for the services that can be provided virtually such as consultations and first aid guidance.

1.3 Research Challenges and Questions

The following points summarises the research challenges and questions.

1- Due to the complexity and broadness of the theoretical and practical use of virtual collaboration concept, it is yet to be fully researched. The majority of research literatures available such as [39] focus on the functional aspect of virtual collaboration. The authors of [38] claim that researches carried out so far on virtual collaboration " *do not yet provide a comprehensive view on its design*". Therefore, developing a model for virtual collaboration is a challenge as claimed by [20] and [40] which is also the case for healthcare virtual collaboration.

2- The authors of [41] state that, within a VO, it should be possible to determine roles and services that are going to be provided and [40] suggests for a virtual collaboration to fulfil its purpose the roles of its members should be coordinated. These researches point towards the need for a role and service coordination mechanism. This challenge is mentioned in [42] in a form of service orchestration. Therefore, developing a mechanism capable of classifying roles and services in healthcare virtual collaboration and a mechanism to describe the service provision coordination present challenges.

3- In virtual collaboration many parties have to collaborate to complete a task, these parties need to be organised and managed to have a productive result. Since virtual collaboration takes place in cyberspace the organisation and management of collaborators and resources offer real challenges in terms of member selections, member verification and validation and Service Level agreement (SLA). [43] Identifies a number of

requirements that a typical VO should fulfil, for example a VO can have members that are participants of other VOs, resource could be shared between different VOs and in a given VO the same member can have different roles. Within this context, developing a member selection, a member verification and validation mechanism and an SLA mechanism for healthcare virtual collaboration present real challenges.

4- Developing concepts and frameworks for healthcare virtual collaboration requires proper evaluations to ensure that the developed concepts are sound and usable which is a research challenge.

From the challenges described, a number of research questions (RQ) are deduced to be answered in this thesis as follows:

RQ1: How to classify the main stakeholders and services in virtual collaboration for healthcare?

RQ2: How to model and describe service provision in virtual collaboration for healthcare?

RQ3: How to manage and organise virtual collaboration for healthcare?

RQ4: How to select, verify and validate participants for healthcare virtual collaboration?

RQ5: How to regulate virtual collaboration for healthcare?

1.4 Research Aims and Objectives

The General vision of this thesis is to initiate a new approach in healthcare provision based on VBE and VO concepts where there are many VBEs on global level competing to form VOs to provide real-time care. Figure 1.1 provides a diagrammatic view of the research vision in which the oval shape represents the world and the Healthcare Virtual Breeding Environment (HC-VBE) depicted as squares represent healthcare VBEs around the world. The square with more details in the middle represent the possible content of each HC-VBE which can create many Healthcare Virtual Organisation (HC-VO). A HC-VO is created for a specific purpose with the aim to achieve four goals (reduce cost, raise quality, increase availability and provide timely service) which are placed on the four corners of the diagram. To realise the vision, the aim is to develop a modelling framework specific to healthcare, to help system developers model healthcare virtual collaboration scenarios more effectively. And based on the modelling framework, develop a virtual collaboration framework to guide the understanding and development of virtual healthcare systems.



Figure 1.1: The healthcare virtual collaboration vision of the thesis which shows a number of rectangles representing VBEs with a possible content of a given VBE shown in the middle

1.4.1 Research Objectives

To achieve the aims and answer the research questions, Figure 2 shows the objectives and sub-objects of the thesis which are:

1- Develop and evaluate a generic modelling framework for healthcare virtual collaborations based on VBE and VO concepts that includes:

- A- A service and participant classification mechanism to enable the classification of roles and services of healthcare virtual collaboration scenarios.
- B- A domain specific modelling language to model participants and services in healthcare virtual collaboration scenarios.
- C- An orchestration description language to describe service orchestration in healthcare virtual collaboration scenarios.

2- Develop and evaluate a virtual collaboration framework for healthcare, based on the concepts and models developed in objective one that includes:

- A- A conceptual architecture that outlines the main components and working steps for healthcare virtual collaborations.
- B- A member selection mechanism for healthcare virtual collaborations.
- C- A member verification and validation framework for healthcare virtual collaborations.
- D- A set of SLA templates to frame contracts between participants of healthcare virtual collaborations.



Figure 1.2: Research objectives, objective 2 is developed based on objective 1

1.5 Target Groups

The target groups of this research are healthcare stakeholders and system developers. Healthcare stakeholders can use the modelling framework (HC-VBE-M-F) described in Chapter 5 to visualise and plan ahead for possible virtual collaboration without technical expertise. The modelling framework is simple and text based which requires minimal effort to learn and use. Developers can use the modelling framework to model VBE and VO based healthcare virtual collaboration systems in terms of roles and use cases, and use the service orchestration mechanism to describe service provision coordination. System Developers can also use the HC-VBE-F described in Chapter 6 as an architectural base and a guide for healthcare virtual collaboration system developments.

1.6 Thesis Contributions

The contributions of this thesis are to frameworks for healthcare virtual collaboration which are both evaluated empirically. The first framework is a modelling framework which is called Healthcare Virtual Collaboration Modelling Framework (HC-VBE-M-F). Three specific contributions make up the overall contributions of the modelling framework which are listed below. Technology Acceptance Model (TAM) is extended with one new construct (Perceived Ability to Model) to evaluate the acceptance of system developers of the HC-VBE-M-F which is implemented as a Java application. The new construct has been validated by system developers as an acceptance factor for modelling languages. This is also a contribution that extends the applicability of TAM in examining users' acceptance of new modelling technologies. The extension is explained in Section 5.5.

- A. A unique conceptual Service and Participant Classification Mechanism (SPCM) for VBE and VO based healthcare virtual collaboration is developed as a result of addressing RQ1. The SPCM is a high-level classification of the main roles and services developed with the aim to simplify and help understanding the healthcare sector. On VBE level, the SPCM classifies roles (Human and non-human) into *Organisers* who participant in organising a VBE to provide healthcare virtually and *Support* which represents roles of information and communication support systems that facilitate virtual collaboration in healthcare VBEs. As for the service, all healthcare VBE management tasks are classed as *Task* which represent the use cases required for managing a healthcare VBE. On VO level, the SPCM classifies roles into *Provider* who represent all healthcare providers that participants in healthcare virtual collaboration services such as patients. As for service, on VO level, the SPCM represents all types of healthcare virtual collaboration services as *Service*. The SPCM is described in Section 5.2.
- B. A domain specific modelling language (DSML) to model the structural aspect of VBE and VO based healthcare virtual collaborations is developed to address the modelling part of RQ2. The DSML provides modelling notations to model the participants and

services classified based on the SPCM. The DSML is extended from UML use case diagram notations using the UML profiling notation extension mechanism. The DSML provides five new graphical notations to model HC-VBE participants (Organiser, Requester, Provider and Support) which are all extensions of the UML actor notation modelled as a stickman. To model HC-VBE services (Task and Service), the DSML provides two new graphical notations which are extensions of UML use case notation modelled as an oval shape. The DSML provide a new relationship notation to model automatic relationship which is an extension of the UML relationship modelled as a straight line. The DSML is described in Section 5.3.

C. A unique Service Orchestration Description Language (SODL) for VBE and VO based healthcare virtual collaborations is developed as a result of addressing the service description part of RQ2. The SODL is comprised of two components; the first component is a mechanism that captures scope, attributes and process aspects of an orchestration. The second component is a textual description language developed for describing the sequence of actions during a service orchestration. The language has a specific and defined structure as well as a sentence grammar to ensure consistencies in all service orchestration descriptions. The SODL is described in Section 5.4.

The second contribution is a healthcare Virtual Collaboration framework (HC-VBE-F), developed based on the concepts and models outlined in the modelling framework described in Chapter 5, to address RQ3. The framework is described on conceptual, component and process levels which can be used by system developers as a guide to develop healthcare virtual collaboration systems. The framework contributes towards realising the implementation of VBE and VO concepts for healthcare. On conceptual level, the framework describes how HC-VBEs and HC-VOs can be formed and managed for healthcare virtual collaboration. The conceptual description is provided in Section 6.2.1. On component level, the HC-VBE-F specifies all the main components required for creation, operation and dissolution of HC-VBEs and HC-VOs. Each component is described and their main functions are outlined in Section 6.2.2. On process level, a detail step by step process is provided to show how the framework works, more details can be found in Section 6.2.3.

TAM is extended with four new constructs (Perceived Healthcare Globalisation, Perceived Healthcare Availability, Perceived Healthcare Quality and Perceived Clinical Effectiveness) to create two new TAMs (HC-VBE-TAM-Requester and HC-VBE-TAM-Provider). The new TAMs are used to evaluate healthcare providers and requesters acceptance of the HC-VBE-F which is implemented as a mobile application. The extensions contribute towards further validation of TAM in predicting the acceptance of new healthcare technologies. The extensions are detailed in Section 4.3.2. The collaboration framework is enhanced with three new mechanisms that are developed to address member selection, service level agreement and provider verification and validation challenges in healthcare virtual collaboration. The mechanisms are:

- A. A conceptual Member Selection Mechanism (HC-VBE-MSM) for HC-VBE and HC-VO collaboration is developed as a result of addressing RQ4. The mechanism proposes to select members based on tasks which is different from reputation-based and predefined criterion-based mechanisms proposed by researchers in the field. The HC-VBE-MSM provides a seven steps guide to select the right member for HC-VOs. The mechanism contributes towards addressing the challenge of member selection for healthcare virtual collaboration. It is described in Section 6.3 in detail.
- **B.** A Healthcare Provider Verification and Validation Mechanism (HC-VBE-PVVM) is developed for HC-VBE and HC-VO collaboration as a result of addressing the verification and validation part of RQ4. The HC-VBE-PVVM is a conceptual mechanism which utilises blockchain technology to verify and validate healthcare providers who offer to provide a virtual healthcare service. Unlike similar blockchain-based mechanisms proposed in literature, the mechanism provides a seven steps process which can be used as an implementation guide for the mechanism. The mechanism contributes towards addressing the challenge of user verification and validation in virtual healthcare. It is described in Section 6.4.
- C. Two SLA templates have been developed to regulate and formalise contracts between HC-VBE and HC-VO participants as a result of addressing RQ5. The first template is a Master SLA template which works as an umbrella for all the SLAs approved between participants of a particular HC-VO. The second template is developed to regulate the

27

terms and conditions of a healthcare service between two virtually collaborating participants. The templates are similar to the ones suggested in literature in terms of purpose and structure, but different in details and content presentation. They are text-based, simple to understand and contains the necessary clauses that depict a formal contract. The templates contribute towards the formalisation, standardisation and regulation of healthcare services provided in a given HC-VO. The templates are described in Section 6.5.

1.7 Thesis Structure

The thesis consists of seven chapters as follows:

Chapter 1: Introduces the research topic, motivations, challenges, aims and objectives and the main contributions of the thesis. At the end, a summary of the chapter is provided. The chapter consists of eight sections which are 1.1 Research Introduction, 1.2 Research Motivations, 1.3 Research Challenges and Questions, 1.4 Research Aims and Objectives, 1.5 Target Audience, 1.6 Thesis Contributions, 1.6 Thesis Structure and 1.8 Chapter 1 Summary.

Chapter 2: Provides the scientific background for the VBE and VO theories as well as technologies (e.g. blockchain) used in developing the modelling and collaboration frameworks in Chapters 5 and 6. A brief background of Technology Acceptance Model (TAM) is also provided which is used as a theoretical evaluation framework to evaluate the two frameworks developed. At the end, a summary of the chapter is provided. The chapter consist of eleven sections which are 2.1 Collaboration, 2.2 Virtual Collaboration, 2.3 Virtual Community, 2.4 Healthcare Virtual Community, 2.5 Virtual Breeding Environments and Virtual Organisation, 2.6 Modelling Virtual Collaboration, 2.7 Service Level Agreement, 2.8 Blockchain Technology, 2.9 Technology Acceptance Model, 2.10 Structural Equation Modelling and 2.11 Chapter 2 Summary.

Chapter 3: This chapter provides a brief review of related literatures and identify the gaps that this research is aiming to address. The chapter touches upon the state of the art in healthcare virtual collaboration modelling, healthcare virtual collaboration frameworks

and healthcare SLA creation and management mechanisms. As a justification for choosing to use TAM as theoretical base for evaluation, a review of a number of related researches are provided. Finally, a summary of the chapter is provided at the end. The chapter consists of seven sections which are 3.1 Virtual Collaboration Models, 3.2 Virtual Collaboration Frameworks for Healthcare, 3.3 SLA Creation and Management, 3.4 Partner Selection Mechanism, 3.5 Service Orchestration, 3.6 TAM for Healthcare Applications and 3.7 Chapter 3 Summary.

Chapter 4: Describes the research design and methodology which are used to achieve the aims and objectives of the research. The chapter starts by explaining the main steps taken to conduct the research and end up with zooming into each step to provide fine details about how the step is executed. The chapter also provide implementation details of the frameworks. At the end, a summary of the chapter is provided. The chapter consists of four Sections which are 4.1 Research Methodology, 4.2 Research Design, 4.3 Evaluation Strategy and Methods and 4.4 Chapter 4 Summary.

Chapter 5: The chapter presents the HC-VBE-M-F as well as the framework evaluation results and discussions. The components of the modelling framework which consist of a service and participant classification mechanism, a domain specific modelling language and a service orchestration description language are described. The framework is demonstrated through developing simple healthcare virtual collaboration scenarios and modelling them using the framework components. At the end, a summary of the chapter is provided. The chapter consist of seven sections which are 5.1 Healthcare Virtual Collaboration Modelling Framework, 5.2 Service and Participant Classification Mechanism, 5.3 HC-VBE Domain Specific Modelling Language, 5.4 Service Orchestration Description Language, 5.5 HC-VBE-M-F Evaluation Results, 5.6 HC-VBE-M-F Discussions and 5.7 Chapter 5 Summary.

Chapter 6. The chapter presents the HC-VBE-F. The framework is described on conceptual, component and process levels. As part of the framework, a member selection mechanism for healthcare virtual collaboration is presented which provides a step by step guide as to how potential virtual collaboration members can be selected. The chapter also, provides a detail description of a healthcare virtual collaboration provider

29

verification and validation mechanism. The final part of the framework is the description and explanation of two SLA templates that can be used to regulate and formalise contracts between healthcare virtual collaborators. The evaluation results of a mobile application prototype developed based on the HC-VBE-F are presented and discussed and a summary of the chapter is provided at the end. The chapter consists of eight sections which are 6.1 Healthcare Virtual Collaboration Framework, 6.2 The HC-VBE Framework Description, 6.3 HC-VBE Member Selection Mechanism, 6.4 HC-VBE Provider Verification and Validation Mechanism, 6.5 HC-VBE Service Level Agreement Templates, 6.6 HC-VBE-F Evaluation Results, 6.7 HC-VBE-F Discussions and 6.8 Chapter 6 Summary.

Chapter 7. The research is concluded in this chapter and a list of key achievements are presented. A number of future research ideas are introduced and a list of publications and future publication plans are provided at the end. The chapter consists of four sections which are 7.1 Conclusion, 7.2 Research Achievements, 7.3 Published and Planned Publications and Future Works. Figure 1.3 shows the structure of the thesis in detail.



Figure 1.3:The thesis structure, showing the sections of each chapter

1.8 Chapter 1 Summary

This chapter introduced the research topic, the motivation and challenges for the research. The content of the chapter is summarised in the following points:

- 1- Healthcare is a complex multi-dimensional service where many stakeholders such as medical professionals, governmental agencies and organizations have to collaborate to organise, manage and provide. The demand for healthcare services is rising due to the rise in population numbers and population aging, the rise in demand has led to increase in cost and waiting time for patients.
- 2- The current centralised model of healthcare provision cannot cope with the rising demand for healthcare; therefore, a new decentralised model is required to facilitate collaboration and resource sharing across time and space in order to make healthcare more efficient.
- 3- Virtual collaboration where participants can come together regardless of whereabout they are, to collaborate and share resources to achieve a common goal, has the potential to facilitate a new model of healthcare provision.
- 4- The organisation and management aspect of virtual collaboration present challenges such as how to model healthcare virtual collaboration, how to recruit members for a given virtual collaboration and how to regulate the rights and responsibilities of each participant.
- 5- The aim of this thesis is to develop two frameworks for modelling, organising and managing healthcare virtual collaboration.
- 6- To develop the frameworks, the concepts of Virtual Breeding Environment (VBE) and Virtual Organisation (VO) are used as theoretical bases.
- 7- To evaluate the developed frameworks, Technology Acceptance Model (TAM) is used as a theoretical evaluation framework, survey questionnaire is used as a data collection instrument and Structural Equation Modelling (SEM) is used as a statistical calculation technique in AMOS software.
- 8- The contribution of this thesis is the development of two frameworks; the first framework is a modelling framework that consists of a service and participant classification mechanism, a domain specific modelling language and a service

orchestration language. The second framework is developed based on the first framework to address organisational and managerial challenges of healthcare virtual collaboration.

In Chapter 2, a brief background of the concepts and technologies used in this thesis is provided.

Chapter 2 : Background

This chapter provides background information about the concepts and technologies used in the research to formulate frameworks for modelling and facilitating virtual collaboration for healthcare. The chapter contains nine sections, Section 2.1 provides background on collaboration, and Section 2.2 introduces virtual collaboration. In Section 2.3, the concept of virtual community is introduced. In Section 2.4, background about healthcare virtual community is provided. Section 2.5 explains VBE and VO concepts. Section 2.6 explains the importance of modelling in virtual collaboration. Section 2.7 introduces SLA and Section 2.8 introduces blockchain technology. Technology Acceptance Model as a theoretical base for evaluation is introduced in Section 2.9 and Structural Equation Modelling as an empirical evaluation technique is introduced in Section 2.10. Finally, a summary of the chapter is provided in Section 2.11.

2.1 Collaboration

The term "collaboration" is frequently used to convey the joint actions between two or more parties towards achieving a common goal [44]. In a typical collaboration, all collaborating parties share responsibilities in accomplishing a task and get a joint credit for the outcome [45]. Working together and helping each other have always been part of human life from the days when joint efforts were needed to hunt to today's data driven world. Today, human with human, human with machine and machine with machine collaborations are the norm.

The human demand for a better life has led to a busier life style where the amount of time that people have for a face to face and direct collaboration has been reduced. In the absence of direct collaboration, a virtual one can serve as the alternative. It is for this reason that the concept is widely researched for possible facilitation in providing important services such as education and health [46] [47]. Virtual collaboration is a complex and multi-dimensional process that requires integration of various approaches,

techniques and technologies to realise it. Hereafter, a brief background of the concepts and technologies that serve the base for the healthcare virtual collaboration frameworks developed in the thesis is provided.

2.2 Virtual Collaboration

Advances in the internet and telecommunications have transformed the traditional face to face collaboration; and have introduced a new type known as Virtual Collaboration (VC) where collaborating parties share resources and work together [48][49]. In VC, collaborating parties use modern technologies to connect with each other and share resources. According to [50], telecommunication, databases and multimedia are the three main technologies used in VC, where collectively, they provide means to create collaboration platforms such as websites, blogs, and discussion boards. Figure 2.1 shows a wholistic view of virtual collaboration where information and communication technology (ICT) brings Humans and machines together to create three VC settings (Human-Human, Human-Machine and Machine-Machine) regardless of geographical borders.



Figure 2.1: Virtual collaboration settings between human and machine participants

In Human-Human VC, ICT provides means such as telecommunications to connect two humans together, through which, they can collaborate and exchange information. In Human-Machine VC, humans and machines can communicate and collaborate with each other using specially designed interfaces that works as a translator between them. In
Machine-Machine VC, machines exchange electronic signals that can be interpreted to meaningful commands on both sides. The aim in all VC settings is to complete a common task.

VC is flexible and allow collaboration to happen when and where needed [51]. The flexible nature of VC has helped in reducing cost and improving quality in specialised fields such as manufacturing since collaborators can share information in real-time without the need for them to be at the same location [52]. Researches such as [53] claim that "*collaboration in virtual settings are occurring on a regular and increasing basis*" and according to [54], healthcare collaboration in virtual settings is going to increase due to globalisation and the rising demand for healthcare. This is because, in healthcare, complex considerations are required in diagnosing and treating a given case which imposes the need for collaboration between different healthcare specialists.

VC has also been considered for improving service quality by companies and organisations; for example, customers of Dell computers collaborate virtually with the company to request a specific computer by selecting the components that make up the computer [38]. In order for the services provided through VC to become reliable it requires a persistent environment with advanced ICT at its heart [55]. Technology dependency is one of disadvantages of virtual collaboration, which means, if the technology needed is not available collaboration cannot take place.

2.3 Virtual Community

Virtual Community (VCom) is the product of virtual collaboration, in which, a group of geographically spread individuals get together similar to the real world community to share ideas and exchange services facilitated by ICT [56][57]. The authors of [58] describe VCom as a community of agents (agents can be individual, organisation or machines) collaborating using ICT with the aim to provide services to each other. Howard Rheingold define VCom as "social aggregations that emerge from the Net when enough people carry on those public discussions long enough, with sufficient human feeling, to form webs of personal relationships in cyberspace" [59].

According to [60], shared goals such as solving a particular problem, collaborative participation, shared resource and shared social values such as being supportive are some of the attributes that virtual communities possess. Amazon, eBay and Gumtree are examples of such community. The communities are business oriented where members of the community buy, sell and advertise goods virtually. Health care virtual community is considered as one of target users of the virtual collaboration frameworks developed in this thesis. Figure 2.2 visualises three virtual communities C1, C2 and C3 in a virtual environment. Each community is created by a number of entities modelled as network of different sizes of circles that represents individuals and organisations. Each community is created for a specific purpose based on specific shared goals and values. The communities can also collaborate and share resources between each other (resource sharing is indicated by the blue three-way headed arrow) based on shared goals and values.



Figure 2.2: Virtual communities in a virtual environment

2.4 Healthcare Virtual Community

Healthcare virtual community (HVCom) is composed of patients, doctors, healthcare educators, providers and organisations collaborating and sharing health related information online [61] [62]. HVCom is becoming increasingly important as the number of people searching online for healthcare information is on the rise. Rentrop and Straton Marketing Research Company carried out an investigation in Romania and found 81.9% of 1300 survey responses received, went online in search of health information [35]. In a

similar research, which was carried out in the early years of virtual community emergence, the authors of [63] found that, up to the time of their research, 33 million Americans have searched for health related information online. Individuals who search online for health information are referred to as e-patients. HVCom is known to " empower patients with knowledge, facilitate health information dissemination, and provide social and psychological support" [40].

HVCom can facilitate the provision of some forms of healthcare such as consultation to patients even when the patients and care providers are at two different locations [64] [65]. Different types of HVComs exist and they can be formed between patients to share stories and obtain psychological support from each other which is known as Patient-Centred community, or, they can be formed between healthcare professionals to share ideas which is known as Professional-Centred community [20]. Communication in HVCom communities is normally asynchronous which is not suitable for real-time communication and collaboration. VO is a form of virtual community formed on a short-term base which is considered more suitable for the dynamic nature of healthcare as claimed by [66]. The reason is, members commit for a short period to provide a service in a VO and after the set goal is achieved, they are free to join other VOs. The flexible and temporary nature of VO is one of the determining factors considered in choosing it as a theoretical base for the purpose of concept development in this thesis.

2.5 Virtual Breeding Environment and Virtual Organization

Virtual collaboration comes with challenges such as regulation and organization to ensure that the collaborating parties meet their obligations. Collaboration is taking place between a number of collaborators to achieve a common goal; each collaborator is required to contribute to the process according to their level of obligations, however, this offer challenges in terms of defining the level of obligations and enforcing it. Researchers have been developing VBE and VO concepts to address the challenges [55][56][67][68][31][32]. VBEs are created by individuals, organisations and other relevant parties using ICT to provide a long-term environment for virtual collaboration. The main job of the environment is to recruit and connect collaborators and facilitate the process of resource sharing between them under some enforceable rules and regulations [69]. A possible collaborator recruitment method from a universal pool is suggested by authors in [70]. Theses environments can serve different collaboration purposes, for example, Virtual Research Environment (VRE) developed to help researchers connect and Open Science Grid (OSG) developed to facilitate scientific resource sharing [29][30]. Initiation, operation and dissolutions are the three main life-cycle stages of a VBE as described by authors in [27]. VBE is capable of overseeing the process of creation, operation and dissolutions of a number of parallel and temporary collaboration ensembles known as VOs.

VO is an independent container for a number of collaborating parties that wish to collaborate and share resources to achieve a goal, after which, it is dissolved and the collaborators are free to form or join a new VO. VO is defined by [33] as "A loosely bound consortium of organisations that together address a specific demand that none of them can (at the given time) address alone and once the demand has been satisfied the VO might disband". For the world of business, the authors of [71] define VO as a "temporary network of independent companies, suppliers, customers, and even rivals - linked by information technology to share skills, costs, and access to one another's markets".

Virtual collaboration presents many organisational and management challenges which researchers aim to tackle using the facilities offered by VBE and VO concepts. The concepts are believed to create a common organised virtual space where collaborators can share resources and work together [72]. To understand the management aspect of VBE and VO, the authors of [28] have developed a framework that models the structural and behavioural aspects of the concepts. The concepts are general and requires a different organisational and managerial approach in different sectors. For instance, a framework to manage VBE and VO in the business sectors is proposed by [27] which aims to describe roles and services and the authors clearly sate that the framework is not suitable for the healthcare sector.

In this thesis, a virtual collaboration framework specific to healthcare is developed with the aim to fill in the gap. Healthcare management and provision requires a dynamic mechanism that can adjust to different needs and circumstances, the idea that VO can be formed on goal bases with no long-term commitments from its collaborating members makes it appealing to healthcare.

The life cycle of a VO is similar to that of VBE according to [69] and [73], but with much shorter life-time for each stage. Figure 2.3 is a lifecycle diagram for VBE and VO. In the initiation stage of VBE, parties who are willing to commit to long-term base collaborations and resource sharing are recruited along with required resources (indicated by the arrows pointing to the VBE initiation stage) to provide the operational support needed for the life spam of the VBE. In the operation stage, VBEs accept requests based on which VOs are created (oval shapes inside the VBE operation indicate VO life-cycle stages). The main task of a VBE is to recruit collaborators and provide required resources to initiate the requested VO and oversee the operation and dissolution of the VO. Finally, the VBE is dissolved when its service is no longer needed in which case collaborators and resources are freed to be reused.



Figure 2.3: VBE and VO life-cycle showing the three stages (initiation, operation and dissolution)

2.6 Modelling Virtual Collaboration

To convey information, diagrams play an important role [74], they are created by human, for human, to offer conceptual solutions to problems using geometric and symbolic representation [75]. Diagrams are seen much like a natural language that use words and grammar of the language to form a meaningful sentence [76], since a diagrammatic or visual sentence is also made up of visual notations and grammars [77]. The structure of VBE and VO is important to be described as stated by [78] and to do so, there is a need for a formal modelling language. According to [79] modelling languages can be classified into two groups, general purpose and domain specific.

Due to advancement in technological solutions and the attempt to tackle ever more complex scenarios, domain specific modelling has gained importance recently [80][81][82]. Because of the diversity and complexity of virtual organisation structures, developing an inclusive model usable for VBEs and VOs in all sectors requires an intensive research. Therefore, developing a domain specific modelling language (DSML) can be more achievable for the obvious reason that every sector is different and requires different configurations in their design.

According to [83], DSML consists of a set of abstraction and notations that can be understood by experts in the field to which the DSML is created for. This means that the notations of DSML is more specific compared to general modelling languages such as UML. DSML provide solutions by *"raising the level of abstraction"* and the final products are generated from them. [84] Argues that diagrams created using DSML are formal descriptions of the domain and they are suitable for formal documentation. With this in mind, a DSML for VBE and VO based healthcare virtual collaboration is developed later in section 5.3.

2.7 Service Level Agreement

In virtual environments, SLA is a widely accepted mechanism to regulate collaborations and resource sharing. SLA is simply a contract that regulates service providers, receivers and expected service attributes. It also outlines the responsibilities of each party, expected rewards in case of full service delivery and expected penalties in case of contract violations [85]. SLA is defined by [86] as "a contract that defines the level of services a service provider promises to a service consumer". Formation, deployment, enforcement and termination are the life cycle stages of SLA [87]. [86] Identifies the main components of SLA as (1) expected service description, (2) the reliability level of the service, the monitoring and service reporting, (3) the service conditions. SLA is normally created based on a predefined template which is believed to speed up the process of SLA agreement between service requesters and providers [88][89][90]. SLA is important in a collaborative environment especially when collaborating parties expect services from each other. It provides a framework for working where the interest of all parties is protected and formalised. SLA in conceptual models can increase the level of understanding and service development according to [91]. In this thesis two SLA templates have been developed for healthcare virtual collaboration and their structure and content are presented in the Section.

2.8 Blockchain technology

In this research, blockchain technology is considered as a base to develop a mechanism for healthcare provider validation and verification in Section 6.4. Nakamoto in 2008 proposed the concept of Bitcoin [92], which removes the third-party authentication mechanism during transactions made online. Instead, authentication happens in a peerto-peer network where all peers in the network hold a copy of the same information (Hash value in the Bitcoin context). In blockchain, transactions are recorded between peers in a network using cryptography and store the records in a digital ledger which is called a block. To verify and validate the content of blocks, each block is linked to the other and, during the verification and validation process the content of all blocks should match. Figure 2.4 shows the composition of blocks as block header, hash of previous block header and Merkle root.



Figure 2.4: Blocks linked in a chain showing the concept of Blockchain technology

A unique hash value of the previous block is stored in each block in a chain, in case of requests to change a block all the hash values are compared, for the change to be recorded successfully, they all have to match. Any unconfirmed alteration to a block will render the block invalid as changes to the content of a block will alter the unique hash value which will not match with the hash values stored in other blocks. This mechanism safeguards the validity of the information contained in a block. Data security is an important requirement in healthcare systems, in blockchain, to ensure data security within a block, public key cryptography is used. This is where the information in a block is encrypted using a publicly available key, but to decrypted back to its original information, a private key which is held by the block owner is required [93]. This process verifies the identity of the block owner.

Permissionless and permissioned are the two main types of blockchain, the former allows free requests to change a block by the owner and in the later, requests to change a block is controlled by a central authority that requires the requester to provide identity verification credentials such as user name and password [94]. For the purpose of this research permissioned blockchain is chosen to be used later in the development of the healthcare provider verification and validation mechanism.

2.9 Technology Acceptance Model

The authors of [95] claim that most researches in the field of healthcare informatics have focused on technical aspects of developing and implementing healthcare systems and not enough attentions have been paid to the acceptability of such systems among prospective users. The acceptability level has been used as indicators by system developers to decide whether to continue developing a system or not [96][97][98]. Theory of reasoned action (TRA) and theory of planned behaviour (TPB) [99] are both used in the development of Technology Acceptance Model (TAM) [34].

The TAM pictured in Figure 2.5 was published by Davis in 1989 after a small refinement from the original one he developed in his PhD thesis in 1985 [100]. The TAM proposed by Davis consists of four constructs which are perceived ease of use, perceived usefulness, attitude towards using, and intention to use. Each construct tests an aspect of technology acceptance by prospective users of the technology in question.



Figure 2.5: Original Technology Acceptance Model developed by Davis in 1989

Davis describes Perceived Usefulness (PU) as "the degree to which an individual believes that using a particular system would enhance his or her job performance" and perceived ease of use (PEU) is described as "the degree to which an individual believes that using a particular system would be free of physical and mental effort" [34]. Attitude towards Using (AU) is described as" individuals positive or negative feeling about performing the target behaviour (e.g. using a system) " and Intention to Use (IU) is described as " the degree to which an individual has formulated a conscious plan to perform or not perform some specific future behaviour" [101] [102].

The lines connecting the four constructs represent causal hypothesis used test the causal effect of the constructs on each other. Hypothesis is described as "A proposition to be put to test to determine its validity" [103]. Formulating and testing hypothesis has served as one of the fundamental pillars of research and it is popular among researchers. To evaluate the frameworks developed for healthcare virtual collaboration in this thesis, TAM is used as a theoretical base to develop evaluation models which includes extending the original TAM model with new constructs and formulating new hypotheses. TAM extensions are described in Section 4.3.2. [104] States that TAM has been successful in providing "general explanation of the determinants of computer acceptance and explains computer usage behaviour in a wide range of computer user domains".

According to [105], hypothesis "can be formulated based on previous research and observation", previously tested TAM hypotheses found in literatures such as [106],[107],[108] and [109] and the original TAM hypotheses were used as a base to formulate hypotheses for this research. The aim was to find out users' intention to use the developed healthcare virtual collaboration modelling framework implemented as Java Application and the virtual collaboration framework implemented as a mobile application. The hypotheses adopted and later modified from the original TAM published by Davis are the followings:

 H_1a : "Perceived ease of use will have a significant effect on perceived usefulness".

 H_1b : "Perceived ease of use will have a significant effect on attitude towards using".

H₂a: "Perceived usefulness will have a significant effect on attitude towards using".

H₂b: "Perceived usefulness will have a significant effect on intention to use".

H₃: "Attitude towards using will have a significant effect on intention to use".

2.10 Structural Equation Modelling

To statistically analyse data obtained empirically during this research, Structural Equation Modelling (SEM) is used [110][111]. According to [112], SEM is the most frequently deployed method in analysing data collected from survey questionnaires. The technique utilises path analysis and regression testing to test the extend of relationships between surveyed items [110]. One of the attractive aspects of SEM is the graphical representation of measured and unmeasured constructs and the relationships between them, which has made it popular among researchers who prefer visual models over mathematical formulas. Table 2.1 shows the main graphical notations of SEM.

[110] States that SEM is more flexible than traditional statistical approach since it works based on a defined model, takes into account errors in variables and it is a multivariate estimation technique. Table 2.2 is a brief comparison between SEM and traditional statistical approach. In this research, TAM is modelled using the modelling symbols of SEM in AMOS to compute statistical values that can be interpreted for user acceptance testing purpose. Each construct in TAM is modelled as an oval (latent variable). To measure the latent variables, a number of questions are developed in a questionnaire, each question is represented as a rectangle (measured variable) in SEM. The symbols are linked using the causal effect (single headed arrow) and the circles of SEM are used to represent possible error values in the collected data.

No.	Name	Symbol	Explanation
1	Measured		A rectangle represents a variable that
	Variable		can be measured directly.
2	Latent Variable	\bigcirc	An Oval represents unmeasurable latent variables which requires a number of measured variables to measure it.

No.	Name	Symbol	Explanation
3	Error variable	\bigcirc	Represents possible errors in a dataset of a variable.
4	Causal effect (Direct effect)		Represents the causal effect between variables in a path diagram.
5	Covariance (none- directional path)		Represents covariance relationship between variables.

Table 2.2: A comparison between Structural Equation Model and Traditional Statistical Method summarised from [146].

Structural Equation Modelling	Traditional Statistical Method
 Multivariate (measured and latent variables). 	 Single-variate (measured variable only).
 Appropriate for testing a number of interlinked hypothesis in a wide range of areas such as economy, health, family, self-efficacy. 	 Appropriate for testing a single hypothesis.
• A formal model is required to be estimated and tested.	 The default model is estimated and tested.
• A number of statistical estimation equations are solved in parallel.	 One estimation equation is solved at a time.
• Takes errors into account.	• Does not take errors into account.

Structural Equation Modelling	Traditional Statistical Method
Model fit test is based on multiple	Model fit is based on a single
tests.	straightforward test.
• To estimate a latent variable	Latent variables cannot be
multiple measure are required.	estimated.
Relationship between variables	 No graphical modelling is
can be represented in a graphical	available.
model.	

2.11 Chapter 2 Summary

This chapter provided a brief background of the concepts and technologies used to develop the healthcare virtual collaboration modelling and collaboration frameworks. The content of the chapter is summarised in the following points:

- 1- Virtual collaboration is where a number of parties come together to share resources and work on a common goal without meeting face to face. Virtual collaboration is facilitated by modern information and communication technologies such as the internet and telecommunication.
- 2- Virtual Breeding Environment is described as a permanent setting that is created by a number of collaborating parties with the aim to facilitate the management and creation of short-lived and goal-based virtual collaboration known as Virtual Organisation. The two concepts have been used as a theoretical base two develop the modelling and collaboration framework in this thesis.
- 3- Modelling is a method to simplify complex settings, the technique is used in many fields of science such as engineering and manufacturing. To facilitate the design of a healthcare virtual collaboration system, a domain specific modelling framework is considered necessary by researchers.

- 4- Service Level Agreement is a widely accepted mechanism to regulate the terms and conditions of a collaboration where the rights and responsibilities of each party is stipulated. The creation, deployment, enforcement and termination are considered the life cycle stages of SLA. In this thesis, two SLA templates are developed to be used for regulating the rights and responsibilities of healthcare virtual collaboration participants.
- 5- Blockchain technology is a peer to peer network that records transactions in a block and distribute a copy of the block to all peers in the network. The technology removes the need for a third party to verify and validate the content of a transaction by comparing all copies of a block held by participating peers in the network. This technology is considered to develop a healthcare virtual collaboration provider versification and validation mechanism in this thesis.
- 6- Technology Acceptance Model (TAM) is a widely used theoretical evaluation framework to examine the acceptance of a new technology by users. TAM examines the extend of the effect of *usefulness* and *ease of use* of a technology on the intention of a user to use the technology. The original TAM model is modified by adding new variables in order to adapt for the purpose of this thesis.
- 7- Structural Equation Modelling (SEM) is statistical data analysis technique which is frequently used to analysed data collected from survey questionnaires. The technique makes use of path analysis and regression testing and it is popular among researchers who prefer to use visual models instead of mathematical formulas. In this thesis the technique is used to analyse data collected to evaluate the two developed frameworks.

In Chapter 3, a brief literature review is provided to present the current state of the art in addressing virtual collaboration modelling, organising and managing for healthcare and identify the gaps that puts the need of this research into context.

48

Chapter 3 : Literature Review

This chapter reviews a number of the literatures available on developing models and mechanisms for virtual collaboration. The chapter also presents literature reviews on the use of TAM in evaluating healthcare systems. Section 3.1 presents some proposed modelling approaches for virtual collaboration and Section 3.2 presents reviews of virtual collaboration frameworks developed by researchers in the field. Section 3.3 presents a number of available SLA creation and management mechanisms and Section 3.4 provides some partner selection mechanisms proposed in literatures. A brief review of a number of service orchestration mechanisms are outlined in in Section 3.5. A review of literature on the use of TAM to test healthcare applications is provided in Section 3.6 and finally, a summary of the chapter is provided in Section 3.7. To put frameworks developed in this thesis into the context of current on-going research, more related work reviews will be provided in relevant sections in Chapters 5 and 6.

3.1 Virtual Collaboration Models

Enterprise related modelling approaches have been researched for some times and some of them are currently in use, for instances, Unified Modelling Language (UML) [75] [113], business Processing Modelling Notation [114], Integrated Definition Methods (IDEF) [115], Business Process Modelling Language (BPML) [116]. As for virtual collaboration, the authors of [117] propose the Vienna Development Method and the authors of [38] present a virtual organisation model that combines both "structure and process components". The federation-agent-workflow (FAW) model is suggested by [70] to model collaborating federated agents that come together to accomplish a task. As for the elements of virtual organisation, the authors of [38] present a model as agents, behaviour and autonomous federation. [118] Offers a formal framework for modelling agent-based VOs. The purpose of the framework is to describe services provided by agents in grid

applications and they focus on creating VOs automatically using intelligent agents. Providing explicit constructs of VOs is a feasible approach to define its structure, for this [119] suggests APPEL language for defining the policies of the construct.

The structure of VO is difficult to define due to its dynamic nature, therefore, finding a model capable of capturing important aspects of VO structure is a challenge [120]. To address this challenge the authors of [33] present a Virtual Organisation Modelling language that aims to model the structural and orchestration aspect of VO supported by a persistent breeding environment. The language consists of two sets of notations which are VO-S that define the structure of supported VOs and VO-R that defines the internal reconfiguration of VOs. This modelling language is developed for business environment and the language lacks graphical syntax; however, in developing the HC-VBE-M-F in this thesis, their direction is adapted for healthcare virtual collaboration with added graphical notations.

3.2 Virtual Collaboration Frameworks for Healthcare

Researchers have proposed a number of frameworks for virtual collaboration in healthcare for different purposes. A number of such researches are outlined here to provide a basic review of the different proposals. An image interpretation framework is outlined in [121] to allow healthcare professionals to share images provided by a patient with other healthcare professionals for diagnostic and treatment purpose. The authors of [122] suggest that team-based care is becoming the norm and to support the process, they propose a multi-agent framework where they describe a number of components which they call agents to manage different aspects of team work collaborations in healthcare. The framework is specific to teamwork management within a local healthcare system and dese not outline team creation mechanism.

The authors of [123] propose a Fog technology-based framework for healthcare collaboration which is specific to machines within a healthcare system. They call the collaborating entities "nodes" and the communication between them is powered by Internet of Things (IoT). Another similar framework to the Fog-based framework is

proposed by [124] based on Cloud to connect healthcare Virtual Labs together. The framework is composed of seven components which they call agents, each managing different aspects of the collaboration. The frameworks mentioned are rather technical and do not consider human participants in the collaboration process.

A Smart Healthcare System Framework is proposed by [125] that uses cloud and mobile technologies to provide care. The framework aims to utilize a mix of technologies to enable healthcare providers to create, provide and reuse healthcare services in a virtual independent module. The framework is yet to be implemented and tested for acceptability which is an important stage to be passed by any new technology before its application. The review suggests that a comprehensive healthcare virtual collaboration framework is yet to be developed as the frameworks presented are developed to address specific aspects of virtual collaboration. It is one the objectives of this research to develop a general and comprehensive virtual collaboration framework for healthcare.

3.3 SLA Creation and Management

The importance of SLA as a mechanism to define and manage electronic resources in a virtual world is increasing and it is becoming more detail intensive. For example, the authors of [126] propose a SLA for defining Quality of Service (QoS) constrains and others propose it for service management [127]. In a virtual collaboration setting, automatic management of resource sharing, participant selection and goal achievement measures are required, for this, authors in [127] propose that SLA is the way to automate. A detail definition of roles and obligations for participants of a given VO is vital for achieving their set goals [128]. The importance of SLA management for virtual service provision is realised by many researchers, for instance, the authors of [136] propose SLAC management framework aimed at managing cloud services. They focus on the formal description of SLAs and manage the SLA once it has been agreed by all parties. A SLA framework for providing web services is proposed by [137] for regulating the agreements between clients and service providers. The authors of [89] propose a SLA management framework which deals with electronic contracts for distributed multimedia contents. The framework enables consumers to define the quality of services they require before they

51

are delivered by service providers. TrustCoM proposed in [138] is an example of SLA management framework aimed at ensuring a trustworthy SLA creation. Others propose a logic based SLA management framework where they make use of knowledge representation to manage SLA [139]. As for creating SLA, there are a number of available mechanisms, Table 3.1 is a summary of a number of them.

Years/decades	Reference	e- Contracting Creation Mechanisms
2000	[129]	WSDL, ebXML, UBL
2003	[130]	SLAng
2003	[131]	WSLA
2007	[132]	WS-Agreement
2007	[133]	Сс-рі
2010	[134]	C-Semiring
2010	[135]	SLA*
2014	[136]	SLAC

Table 3.1: e-Contract Mechanism Examples

Not all services provided in healthcare can be provided electronically, human involvement therefore is inevitable. [135] Claims that popular SLA mechanisms such as WSLA [140], CC-Pi [133] and WS-Agreement suffer from flexibility and support for services that have human involvement. All available electronic SLAs mentioned are developed for machines to understand which can be difficult for human user to read and understand. Human and machine readability is an attribute that a SLA for healthcare virtual collaboration should possess to be deemed suitable. With this in mind, it is one of the objectives of this thesis to develop SLA templates that are text-based and support human readability.

On machine level, developing a language to describe SLA is another challenge. Due to the unique form of SLA in different domains, a generic SLA language capable of describing

service provision in all domains is yet to be defined. It is for this reason that researchers in the field have defined a number of domain specific languages to express SLA. For instance researchers in [141] and [142] have developed SLAng specific to network services and the authors of [131] propose languages for web services. Object Constraint Language (OCL) is used in [143] to deal with service violation. WSaAgreement [132] and SLA* [135], are some other researches aimed at developing languages to write machine readable SLA and each is developed for a specific domain. It is not the intension of this research to develop a language for describing a machine-readable SLA, therefore any of the languages mentioned can be adapted if deemed suitable in future researches.

3.4 Partner Selection Mechanism

To achieve the dynamicity required in the formation of a healthcare VO, it is important to specify a mechanism for collaborators selection [69]. Recruiting the right partner into a VO setting is crucial for a successful collaboration. The authors in [144] claim that there is a gap in research on partner selection mechanism. To contribute to the body of literature, they propose a partner selection mechanism based on formal concept analysis. A criteria-based partner selection framework for virtual enterprise (which is similar to VO) is proposed by [145], in which, they survey literature for proposed partner selection criteria and filter the most important ones based on asset procedures. The framework is specific to business enterprises and may not be suitable for healthcare VO, however the method can be considered for possible adaption.

In a similar research, the authors of [146] propose a multi-agent model for partner selection, in which, they use price, quality, delivery time and past performance as selection criteria. The model uses an auction environment where potential partners can bid to offer a service and the criteria are used to select the best offer. Auction model in business could be a workable choice since the aim is to maximise profit, but in the context of healthcare, quality is more important. Moreover, the researchers have not outlined a clear mechanism for selecting partners within their model. Researchers in [70] suggest selecting partners from a universal pool of organisations based on short term objectives and long term objectives. The mechanism is rather ambiguous, it does not answer the

partner selection question fully instead raises many more questions such as how do we classify members into short term and long term based? Literatures reviewed in this section show that there is a need for a clear partner selection mechanism that can be followed to select the right virtual collaborators for healthcare as none of the proposed mechanisms is healthcare specific. It is one of the sub-objectives of this thesis to develop a clear step by step healthcare virtual collaboration partner selection mechanism.

3.5 Service Orchestration

The authors in [42] claim that coordinating resource sharing is a real problem in dynamic and multi-participant grid collaboration; they stress the need for clear resource sharing rules between participants. The term that is most recognised with coordination process in virtual collaboration is "orchestration". [147] Describes orchestration as the activity sequences that takes place in a specific organisation, and the concept as a global coordinator for all elements involved in a virtual collaboration, is explained in [148]. For collaboration and orchestration mechanism, [70] proposes the use of communication languages such as KQML [149][150], as for the member selection and collaboration negotiation, they recommend the language outlined in [151]. The concept of orchestration is not widely researched for healthcare virtual collaboration, there are a few available researches such as [15] that aims to support the "dynamic creation, management and coordination of virtual medical teams for the continuous treatment". One of the sub-objectives of this research is to develop an orchestration mechanism for healthcare virtual collaboration.

3.6 TAM for Healthcare Applications

The use of Technology Acceptance Model (TAM) is extremely popular among researchers for testing the acceptability of newly developed ICT-based applications in healthcare [95][152] [153]. In this section, the uses of TAM in a number of available researches are presented. A home telehealth service for people aged 50 and above was tested for acceptability by authors in [154] using TAM in Slovenia. [155] Uses and extends TAM to examine factors affecting users' intention to use mobile healthcare information system in Jordan. A telemonitoring system was developed in Spain by the authors of [156] and to examine the factors that influence the intention to use the system by users, they used TAM. The acceptability of tablet computers by paediatricians was tested in [157] using an extended TAM and users attitude towards using electronic medical record systems was tested in a similar way in [158]. TAM has been used to study acceptability and implementation barriers of electronic health record systems by physicians in [159]. TAM was also used by the authors of [160] to examine nurses expected usefulness and ease of use with regard to healthcare information systems in Jordanian hospitals.

A mHealth application for elderly patients was developed by researchers in [161] and to research the factors affecting the intention to use the technology by target users, they used a modified version of TAM. The authors in [95] carried out a systematic review of the use of TAM in healthcare informatics up to the year 2017 and they found that mobile application for healthcare is one of the main technologies evaluated for acceptability using TAM. This survey supports the approach of this research to evaluate the developed virtual collaboration concepts through implementing it as a mobile application and use TAM to evaluate it.

3.7 Chapter 3 Summary

This chapter provided a brief literature review on frameworks, mechanisms and techniques developed for different purposes of virtual collaboration in healthcare. The following points summarise the content of the chapter:

- 1- Modelling virtual collaboration has been researched for business purposes and several modelling frameworks such as Unified Modelling Language (UML) and Business Process Modelling Language (BPML) are currently used. There is a gap in literature on modelling virtual collaboration for healthcare. To fill in the gap, a modelling framework for healthcare virtual collaboration is developed and evaluated in Chapter 5.
- Several healthcare virtual collaborations have been presented in literature such as an Image Integration Framework for healthcare to facilitate image interpretation in a

collaborative setting between healthcare providers. A Fog-based collaboration framework is developed to connect medical laboratories. The frameworks are task specific which reveals a gap in literature about a comprehensive collaboration framework for healthcare. To fill in this gap, a healthcare virtual collaboration framework is developed in Chapter 6.

- 3- There are a number of mechanisms presented in literatures for virtual collaboration. For the creation and management of SLA mechanisms such as SLAng and SLAC are developed, for partner selections, mechanisms such as criteria- based partner selection mechanism is developed and for service orchestration, mechanisms such as KQML is developed. None of the mechanisms is healthcare specific. in Chapters 5 and 6, similar mechanisms are developed specifically for healthcare virtual collaboration as part of the modelling and collaboration frameworks.
- 4- Technology acceptance Model (TAM) is widely used to test the acceptance of new healthcare technologies by prospective users. This chapter provides a brief literature review about a number of researches that have extend and used TAM as a theoretical evaluation framework. To evaluate the two developed frameworks in this thesis TAM is used.

The next chapter presents the methodology and evaluation strategy of the thesis.

This chapter describes the research design and methodologies that are practised to achieve the aims and objectives of the study. It starts by providing a general idea about the research method used in this thesis in Section 4.1 and provide a comprehensive research design in Section 4.2. In order to evaluate the frameworks developed for healthcare virtual collaboration effectively, an evaluation strategy is outlined and described in Section 4.3. The strategy contains the concept implementations, testing, data collection and data analysis processes. Finally, a summary of the chapter is provided in Section 4.4.

4.1 Research Methodology

Research is described by [105] as a "A logical and systematic search for new and useful information on a particular topic". Basic research leading to theoretical discoveries as well as conceptual construction of knowledge regarding the use of virtual collaboration in healthcare sector was the focus of this study.

The overall aim of this research is to model and develop virtual collaboration for healthcare, and answer research questions that have been defined in Section 1.3. Modelling healthcare virtual collaboration requires an intense understanding of the sector in terms of healthcare stakeholders and services. To bring about the understanding required, in-depth literature review was conducted. To investigate how virtual collaboration can facilitate healthcare service provision, current virtual collaboration concepts needed to be studied and adapted. To ensure that the developed collaboration concepts are acceptable by healthcare providers and receivers, they had to be tested empirically. In light of these research needs, a mixed methods research approach consisting of literature reviews and survey questionnaire was performed.

4.2 Research Design

Success of a research partly depends on the research design used to carry out the research. Research design is defined as the overall plan to study and collect data [162]. In designing the research for this thesis a deductive research approach was adopted, which aims to develop theoretical concepts before testing it [163]. In this research, theoretical healthcare virtual collaboration frameworks have been developed first before carrying out empirical evaluations. Figure 4.1 shows the overall research design used in this study. The design consists of five major steps. In step one, the healthcare sector is researched from literatures found on the internet, books and published papers. In step 2, the concept of virtual collaboration is researched from literatures found on the internet, books and published papers. In step three, a HC-VBE modelling framework is developed which includes a service and participant classification framework, a domain specific modelling language and a service orchestration description language. In step four, HC-VBE collaboration framework is developed which includes a theoretical architecture, a member selection mechanism, a member validation mechanism and SLA templates. In step five, the developed frameworks are evaluated. The steps are described in Section 4.2.1.



Figure 4.1: General search design used to carry out the research in this thesis

58

4.2.1 Research Design Steps Description

This section provides a brief description of each of the five steps in the research design.

Step 1: Research healthcare Sector

To understand the structure and elements that makes up the healthcare sector in terms of stakeholders and services available, literatures found on the internet is studied and analysed. It was decided that a deductive method is appropriate to identify healthcare services and stakeholders from the literatures found. To begin with, search words such as "healthcare, health care, and health-care" were used in different databases such as Google Scholar, Scopes and Science Direct to obtain references that could provide an overview of the sector.

As a result, more than a 100 books and articles such as the ones cited in this thesis were found and studied. From the result of the study, a summary chart for stakeholders and services of healthcare was drawn which can be found in Figure 5.2. The research outcome of this step was further processed and adapted to develop the service and participant classification mechanism described in Section 5.2.

Step 2: Research virtual collaboration

To understand the concept, purpose, technology and use of virtual collaboration, similar to the previous step, available literatures found in major research database on the internet were reviewed. As a result, relevant literatures found were studied and analysed to paint an overall picture of the current state of virtual collaboration. The main aim of this step was to understand the feasibility, mechanism, scope and applicability of virtual collaboration in healthcare sector.

Step 3: Develop healthcare virtual collaboration modelling framework

From the results of steps one and two, frameworks relating to modelling and applying the concept of virtual healthcare in healthcare were developed. The first framework developed is a generic modelling framework which can be used to model HC-VBE and HC-VO collaboration scenarios. The framework is developed for healthcare system

developers as target audience with the aim to provide them with a tool to classify and model stakeholders and services for healthcare virtual collaboration. The modelling framework is described in Chapter 5. The framework consists of a service and participant classification mechanism described in Section 5.2, a set of modelling graphical notations developed by extending UML use case diagram graphical notations which are described in Section 5.3 and a service orchestration description language described in Section 5.4 that can be used to describe healthcare virtual collaboration service processes.

Step 4: Develop healthcare virtual collaboration frameworks

This is the second framework which provides a step by step VBE and VO based collaboration initiation, management and dissolution. It is developed based on the concepts developed in the modelling framework with the aim to facilitate healthcare provision remotely to those in need. The collaboration framework is described in Chapter 6. The framework consists of a general conceptual collaboration framework presented in Sections 6.1 and 6.2 which describes how a heartcare service is requested, processed and provided in a clear step by step process, a member selection mechanism that formalises the process of member selection for a particular healthcare service requested described in Section 6.3, a healthcare provider validation mechanism described in Section 6.4 and a number of general service level agreement templates that formalises contracts between participants of a healthcare virtual collaboration described in Section 6.5.

Step 5: Framework Evaluation

There are many organizations around the world working to develop healthcare information systems to improve the quality of care and reduce cost [164]. One of the most common ways to ensure that a system achieves its intended aims is to perform user acceptance evaluation. The authors of [165] claim that evaluation can aid advancing healthcare systems through identifying the strength and weakness of a system in terms of acceptability and intention to use by users. Healthcare system evaluation has many dimensions such as human, organizational, clinical and technological which makes it a challenging task that requires different approaches to address it [165].

Healthcare information systems have an unpredictable nature, evaluation is performed to ensure that an information system meets its intended purpose. [166] States that *"health information systems evaluation finds its roots in technology acceptance models"*. To evaluate user' acceptance of the technologies developed in this research, TAM [108] was chosen as a theoretical evaluation framework. The authors of [160] state that TAM is widely used and its validity, reliability and robustness has been proven. Structural Equation Modelling (SEM) [111] was then used as a statistical data analysis tool to empirically show the acceptability of the virtual collaboration frameworks developed in this study.

4.3 Evaluation Strategy and Methods

According to [167], there are three types of strategies that can be considered when it comes to information system evaluation which are "*Goal-based evaluation, Goal-free evaluation and Criteria-based evaluation*". Goal-based evaluation is performed against a set of goals identified prior to the system development, Goal-free evaluation is performed without any prior goals, it is rather performed to discover the potential of a developed system, and finally Criteria-based evaluation is performed against a number of identified criterion that the system must adhere to in order to be deemed successful.

As for implementation strategies, *"IT-system as such"* and *"IT-system in use"* have been identified as two different strategies to perform information system evaluation. The first is concerned with having only evaluators of a system to collect data and carryout the evaluation. The second is concerned with involving users of a system to collect data and let evaluators perform the evaluation [167]. A mixture of Goal-based and IT-System in use evaluation was performed in this thesis. Other information system evaluation approaches such as summative, which evaluates a system at the end of development and formative, which evaluates a system during development are suggested by [166] and [168] with the aim to provide more guidance for evaluation. Evaluation is at the heart of research and to ensure that the evaluation for this research is carried out in a systematic way, an evaluation strategy was developed. The strategy consists of 8 steps that was used as a

roadmap to ensure the quality and reliability of the end results. Figure 4.2 shows all the 8 steps and each step is described in detail in the following sections.



Figure 4.2: Evaluation strategy steps used to evaluate both the HC-VBE-M-F and the HC-VBE-F frameworks

4.3.1 Frameworks Implementations (Step 1)

To evaluate the acceptability of both frameworks developed, they were implemented as a prototype in order to make them available as tools to be used by the target audiences. The HC-VBE-M-F was implemented as a Java stand-alone application that could be downloaded onto a computer and be used directly to model healthcare virtual collaboration scenarios. The HC-VBE-F was implemented as a mobile application that healthcare providers and requesters could install on their mobile phones and use it to provide and request healthcare services whenever and wherever, provided they have an internet connection. The two implementations are described in Sections 4.3.1.1 and 4.3.1.2.

4.3.1.1 Modelling Framework Implementation

UML is widely used by developers and system designers to model different aspects of a system, which means there are a wide range of tools available for UML modelling such as

UMLet, ArgoUML, and StarUML. Since the notations of the modelling language developed for healthcare virtual collaboration is an extension of UML use case diagram, it was natural to follow a similar deign of tools available to model UML. The layout of most UML modelling tools consists of three parts:

1- Graphical notation

Majority of UML tools, including the ones mentioned, have graphical notations ready for the user to drag and drop, after notation selection, editing capabilities such as resizing and moving are provided to edit the notation.

2- Canvas

Majority of UML tools provide a canvas for the users to draw their model on. Multi canvases can be opened each for a separate project.

3- Documentation

The common way to document models is to save them as images and print them. As for model modifications, the tools normally provide the capability to save projects and be reopened for later modifications.

To design and implement the modelling tool, the same design and layout of available tools was followed, but with less functionalities. The aim was to develop a prototype which can be used by healthcare virtual collaboration system developers to model scenarios and provide technology acceptance feedback later. Having this in mind, with the help of one of the author's students, a simple Java-based application prototype was implemented with the aim to support the evaluation process. The prototype is simple and includes all the notations in the modelling framework that can be used to build a model without much efforts.

The tool is composed of two main parts, the first allow users to model healthcare virtual collaboration scenarios through using relevant notations that represent types of roles and services as described in Section 5.2.1. Figure 4.3 shows a few screenshots of the modelling part of the prototype and a few stages of model development. The second part of the prototype implemented the HC-VBE-SODL described in Section 5.4 that can be used to

describe the service orchestration of a particular scenario.. Figure 4.4 shows a few screenshots of the second part with different stages of orchestration description process.



Figure 4.3: Implemented HC-VBE-M-F Java application screenshots showing framework notations and different stages of model development



Figure 4.4: Implemented Orchestration Description Language screenshots and stages of the description

4.3.1.2 HC-VBE Framework Implementation

The virtual collaboration framework developed in Chapter 6 is complex with many features that require a lot of time and resources to implement fully, however, to evaluate the basic concepts of the framework, a mobile application prototype was developed with helps from two of the author's students. The use of mobile technology for providing access to healthcare has been promoted by researchers as an easy and affordable method that has the potential to save time and cost [169][161]. The application encompasses all the main stages of the framework such as virtual organisation initiation, creation and management and implements the main components of the framework. The prototype was developed to be tested by healthcare providers and requesters for acceptability using TAM as a theoretical evaluation framework.

The application is available on the author's personal website for users to download and use for feedback and data collection purposes. The mobile application serves as a virtual breeding environment where healthcare requesters and providers can register, negotiate and provide or receive a healthcare service. The application formalises the agreed service terms and conditions in a form of electronic SLA, based on which, the environment creates a dedicated VO for the agreed participants to collaborate and share resources for the duration they have agreed on. The Created VO will be dissolved as soon as the terms of the contract are fulfilled and the duration has expired. To elaborate more on the protype implementation the following steps were executed.

1- Identify prototype functionalities for implementation

To develop the prototype, its main functions had to be identified that covers the most basic but essential aspects of the developed framework. As a result, the following functionalities were identified before the protype development, as minimum requirements:

- Healthcare requesters should be able to:
 - a) Register as a requester: The prototype shall allow healthcare requesters to register their details in a registration form and submit it to the HC-VBE mobile application successfully.

- **b) Request a healthcare service**: The prototype shall allow healthcare requesters to fill in a healthcare request form and submit it successfully.
- **c)** Negotiate a service: The prototype shall allow healthcare requesters to negotiate a healthcare service with a healthcare provider.
- **d)** View SLA: The prototype shall allow healthcare requesters to view a final agreed SLA.
- e) Approve SLA: The prototype shall allow healthcare requesters to approve a SLA after agreement on the terms and conditions of the SLA with a healthcare provider.
- **f) Virtual contact**: The prototype shall allow healthcare requesters to have a realtime virtual communication via audio or video or both with providers.
- g) Provide feedback: The prototype shall allow healthcare requesters to provide feedback on services they received via the prototype and rate healthcare providers in terms of service satisfaction.
- Healthcare providers should be able to:
 - a) Register as a provider: The prototype shall allow healthcare providers to register their details in a registration form and submit it to the HC-VBE mobile application successfully.
 - b) Offer a healthcare service: The prototype shall allow healthcare providers to view all healthcare requests made by healthcare requesters and make an offer for a service successfully.
 - **c)** Negotiate a service: The prototype shall allow healthcare providers to negotiate a healthcare service with a healthcare requester.
 - **d)** View SLA: The prototype shall allow healthcare providers to view a final agreed SLA.

- e) Approve SLA: The prototype shall allow healthcare providers to approve a SLA after agreement on the terms and conditions of the SLA with a healthcare requester.
- **f) Virtual contact**: The prototype shall allow healthcare providers to have a realtime virtual communication via audio or video or both with a requester.
- Healthcare organisers should be able to:
 - a) Register as an organiser: The prototype shall allow healthcare organisers to register their details in a registration form and submit it to the HC-VBE system successfully.
 - b) View all running VO: The prototype shall allow organisers to view all running VOs for monitoring purposes.
 - c) View all Master SLA: The prototype shall allow organisers to view all master SLA created for a running VO for enforcement purposes.
 - **d)** View SLA: The prototype shall allow organisers to view all SLA within a VO for monitoring and enforcement purposes.

2- Implementation technologies

To develop the prototype, implementation technologies had to be selected and it was decided to use *React Native* technology for developing the mobile application. In developing the mobile application view layer, React Native [170], Android [171] and IOS [172] technologies were used. For the server side, Node.js [173], Express.js [174] and Socket.io [175] were used, and finally MySQL [176] was used to develop the database. To handle the video and audio calls between healthcare requesters and providers the multiplatform opensource video conferencing technology provided by (jitsi.org) was integrated into the mobile application. Figure 4.5 shows the main technology groups, their purpose and relationships used in developing the prototype.



Figure 4.5: HC-VBE mobile application implementation technologies used to develop the view layer, the application layer and the database.

3- Application Architecture

A three-tier architecture was chosen for the mobile application, here a brief description of each layer is provided. The first the layer is the environment that organises and handles requests for registration and healthcare services. The environment creates a master SLA for each HC-VO which records some HC-VO-specific information such as creation date, expiry date, the participants of SLAs and a unique ID that will be used for future referencing. The master SLA acts as an umbrella for all the SLAs in the HC-VO and individual SLAs can be navigated to for modification purposes via the master SLA. Section 6.4.1 provides more detail about the concept of master SLA.

The second layer is a negotiation layer that service as the middle engine between the HC-VBE and the HC-VO management layer. It is responsible for handling negotiations between healthcare service providers and requesters and it will trigger the creation of HC-VOs once negotiations have ended. The third layer is where all HC-VOs are in operation and monitored for SLA enforcements and modifications. Each HC-VO is separate from the others and each is created under specific SLAs for specific purposes and they all run in parallel. Figure 4.6 shows the mobile application architecture described.



Figure 4.6: HC-VBE-F mobile application architecture showing the three layers

4. Prototype working steps

The framework functions in an eight-step process illustrated by the prototype screenshots in Figure 4.7. Initially, the HC-VBE mobile application accepts healthcare service requests from requesters, and then process the requests and announce them to registered providers for them to view. In step three, providers can either accept a request offer made by a requester or make a new offer based on some changes to the initial request. The HC-VBE mobile application will announce the offer to the requester in the same way, the requester can accept the offer or make a new offer to the healthcare provider in contact.

This process is handled by the negotiation engine and is aided by a decision support system in a complete system to speed up the negotiation process. The decision support system is not developed within the scope of this thesis and can be developed in the future. When negotiation ended in step five, the HC-VBE mobile application generates a SLA based on the negotiation result for both parties to approve. In the subsequent steps, the HC-VBE mobile application creates a VO for the requester and provider to collaborate virtually for the duration they have agreed on in the approved SLA. Finally, in step 8, the HC-VBE end the particular HC-VO process with feedbacks and rewards for both requesters and providers after successful service provision or service termination for any reason.





4.3.2 TAM Extension and Hypotheses Formulation (Step 2)

The value of any healthcare system is in its effective use, [152] claims that a great number of information systems developed for healthcare are not in use due to user acceptability concerns. [177] States that there is a lack of information about the acceptability of collaboration systems in healthcare. The intention to use any new healthcare technology is important to be gaged before its full development and installation. To test user acceptance, the original TAM was extended through adding new constructs to develop mew TAMs specific for this research. Hypotheses for each extended TAM model were defined based on the original TAM hypotheses presented in Section 2.9.
It is recognised by Davis [34] that sometimes it is necessary to modify the TAM to accommodate new factors which might be influential in users' acceptance of a new technology. In their review paper, the authors of [178] claim that "added variable approach" is the most common approach to use TAM. This is where researchers add new constructs to the original TAM to adapt the model for their own study. The approach is used by many researchers such as [107], [109], [179] and [155]; following their directions, the same approach was used in this research. Depending on what technology is tested for acceptance, relevant factors have to be identified. Factor identification is a challenge and there are studies that concentrate on acceptance factor identification, for instance, [180] identifies factors for e-Government service acceptance test using the theory of planned behaviour.

To identify new acceptance factors to be added to the original TAM as new constructs, available literatures were studied and as a result five new constructs were defined and added to the original TAM to form three new TAMs and eight new hypotheses were formulated for the three sperate TAMs. In the case of the HC-VBE-M-F, one new construct (Perceived Ability to Model) was added to the TAM and two new hypotheses were formulated to be tested by healthcare system developers. For the HC-VBE-F, two new constructs (Perceived Healthcare Globalisation and Perceived Clinical effectiveness) were added to the TAM and three new hypotheses were defined to be tested by healthcare globalisation and Perceived Clinical effectiveness) were added to the TAM and three new hypotheses were defined to be tested by healthcare for the trans. Two new constructs (Perceived Healthcare Availability and Perceived Healthcare Quality) were added to the TAM and three new hypotheses were defined to be tested by healthcare requesters. The newly extended TAMs and hypotheses are described in Sections 4.3.2.1, 4.3.2.2 and 4.3.2.3.

4.3.2.1 TAM Extension for HC-VBE Modelling Framework

Researchers have used TAM to evaluate the acceptability of modelling languages previously, and their direction is followed to evaluate the developed HC-VBE-M-F implemented as a java application prototype. For examples, TAM has been used by [181] to evaluate System Modelling Language (SysML) and the authors in [182] have used TAM to evaluate the acceptance of UML diagrams by undergraduate students. A basic yet

curtail feature of a modelling language is its ability to model the domain correctly. This feature has been mentioned in DSML related literatures. In [183] the authors suggest that DSML should demonstrate concepts that users in the domain are familiar with, represented by graphical notations that are suitable to model the domain. The concept of relevance is suggested by [184] as a criteria for developing DSML, which refers to how closely models developed using the DSML represent the domain. The ability of a DSML in providing a model solution that is fit for the purpose of the domain is indicated by [185] as a quality criteria, and the authors of [186] state that "language-domain" appropriateness" is a quality measure which measures the extent to which a model represents a domain. Therefore, to evaluate the modelling ability of the HC-VBE-M-F, TAM was extended with one new construct which is Perceived Ability to Model (PAM) and it can be defined as the degree to which an individual believes that models constructed using the modelling framework captures a healthcare virtual collaboration scenario correctly. Figure 4.8 shows the added construct to the original TAM and the extended model has been used as a base for hypothesis development, questionnaire design and data collection.



Figure 4.8: Extended TAM for HC-VBE-M-F showing the added PAM construct

Based on the extended TAM shown in Figure 4.8 and the original TAM hypotheses described in Section 2.9, the following hypotheses were defined to be tested in order to evaluate system developer's acceptance of the HC-VBE-M-F developed in Chapter 5.

- H₁: Perceived ability to model has a significant effect on healthcare system developers' perceived usefulness of the HC-VBE modelling package.
- ii. H₂: Perceived ability to model has a significant effect on healthcare system developers' perceived ease of use of the HC-VBE modelling package.
- iii. H₃a: Perceived ease of use has a significant effect on healthcare system developers' perceived usefulness of the HC-VBE modelling package.
- iv. H₃b: Perceived ease of use has a significant effect on healthcare system developers' attitude towards using the HC-VBE modelling package.
- v. **H₄a:** Perceived usefulness has a significant effect on healthcare system developers' attitude towards using the HC-VBE modelling package.
- vi. **H**₄**b**: Perceived usefulness has a significant effect on healthcare system developers' intention to use the HC-VBE modelling package.
- vii. **H**₅: Attitude towards using has a significant effect on healthcare system developers' intention to use the HC-VBE modelling package

4.3.2.2 TAM Extension for HC-VBE-F Providers

Healthcare professionals look for functionalities in a healthcare system that improves their effectiveness and delivers value for them and their patients [187]. The authors of [166] state that beside technological concern, healthcare information system has "clinical effectiveness" concern that should be addressed. Clinical effectiveness has been defined as " the application of the best knowledge, derived from research, clinical experience, and patient preferences to achieve optimum processes and outcomes of care for patients" [188]. In this context, Perceived Clinical Effectiveness (PCE) was added to the original TAM as an important determinate of healthcare providers attitudes towards using a healthcare system developed based on the framework. PCE can be defined for the purpose of this research as the degree to which an individual believes that the healthcare provided through systems developed based on the HC-VBE-F is clinically effective.

Everyone is concerned about healthcare and due to globalization, options to receive healthcare has widen. The concept of medical tourism which refers to people traveling to a foreign country to receive healthcare services is researched in [189] and [190]. People who travel abroad will have to invest time, money and effort to receive the care they need. To address this issue, the HC-VBE-F in Chapter 6 is developed with healthcare globalisation in mind, which means, making virtual healthcare available and accessible globally to provide care over time and space to those in need. Through this, healthcare providers can reach as many healthcare requesters as possible without the need for them to travel.

Healthcare globalisation is also important for filling in healthcare professional shortages and increase collaboration between healthcare stakeholders. The HC-VBE-F developed can bring together healthcare professionals from different locations to provide care and provide the integrated platform needed for medical research. To evaluate this aspect of the framework and find out whether healthcare professionals who test the mobile application prototype believe that the HC-VBE-F has the potential to globalise healthcare, the second construct which is called Perceived Healthcare Globalisation (PHG) was added to the original TAM. PHG can be defined as *the degree to which an individual believes that using systems developed based on the HC-VBE-F would make healthcare service provision global*. The two new constructs are coloured yellow in the extended TAM pictured in Figure 4.9.



Figure 4.9: Extended TAM for HC-VBE-F for healthcare professionals showing the two newly added constructed (PCE and PHG)

74

Based on the extended TAM shown in Figure 4.9 and the original TAM hypotheses described in Section 2.9, the following hypotheses were defined to be tested in order to evaluate healthcare providers acceptance of the HC-VBE-F implemented as a mobile application prototype.

- i. **H**₁**a:** Perceived healthcare globalization has a significant effect on healthcare providers perceived clinical effectiveness of the HC-VBE application.
- ii. H₁b: Perceived healthcare globalization has a significant effect on healthcare providers perceived ease of use of the HC-VBE application.
- iii. H₁c: Perceived healthcare globalization has a significant effect on healthcare providers perceived usefulness of the HC-VBE application.
- iv. H₂: Perceived clinical effectiveness has a significant effect on healthcare providers perceived usefulness of the HC-VBE application.
- v. **H**₃**a:** Perceived ease of use has a significant effect on healthcare providers perceived usefulness of the HC-VBE application.
- vi. **H**₃**b**: Perceived ease of use has a significant effect on healthcare providers attitude towards using the HC-VBE application.
- vii. **H₄a:** Perceived usefulness has a significant effect on healthcare providers attitude towards using the HC-VBE application.
- viii. **H**₄**b:** Perceived usefulness has a significant effect on healthcare providers intention to use the HC-VBE application.
- ix. H₅: Attitude towards using has a significant effect on healthcare providers intention to use the HC-VBE application.

4.3.2.3 TAM extension for HC-VBE-F Requester

To evaluate the HC-VBE-F acceptance from healthcare requesters perspective the original TAM was extended with two new constructs. The first is Perceived Healthcare Availability (PHA) and the second is Perceived Healthcare Quality (PHQ).

The availability of healthcare services where and when needed is an important attribute of an effective healthcare service. Service convenience, which refers to the availability of healthcare service at requesters' time and place of choice in the context of this research, is a determinant adopted form [191] and [109]. In the later study, convenience is described as the extent to which healthcare requesters believe that an IT-based healthcare system minimises "the amount of physical and mental energy spent when in need of medical help and/or advice". Therefore, PHA can be defined for the purpose of this study as the degree to which a healthcare requester believes systems developed based on the HC-VBE framework increases healthcare availability.

The quality of healthcare service provided is another important factor in convincing healthcare requesters to use a system. Different scholars have interpreted service quality differently, for instance [192] sees service quality as *"on-time professional and personalised service"* and others have referred to quality in terms of response time and accessibility [193]. To assess the extent of users' belief with regard to the quality of healthcare services provided though systems developed based on the HC-VBE-F, the second construct which is called Perceived Healthcare Quality (PHQ) was added to the original TAM described in Section 2.9. For the purpose of this research PHQ can be defined as *the degree to which a healthcare requester believes that the quality of healthcare provided through systems developed based on the HC-VBE framework is acceptable"*. Figure 4.10 shows the extended TAM that is used to examine healthcare requesters acceptance of the HC-VBE-F implemented as a mobile application prototype.



Figure 4.10: Extended TAM for HC-VBE Framework for prospective patients showing the two newly added constructs PHQ and PHA

76

Based on the extended TAM shown in figure 4.10 and the original TAM hypotheses described in Section 2.9, the following hypotheses were defined to be tested in order to evaluate healthcare requesters acceptance of the HC-VBE-F implemented as a mobile application prototype.

- i. **H**₁**a:** Perceived healthcare availability has a significant effect on healthcare requesters perceived healthcare quality of the HC-VBE application.
- ii. **H**₁**b:** Perceived healthcare availability has a significant effect on healthcare requesters perceived ease of use of the HC-VBE application.
- iii. H₁c: Perceived healthcare availability has a significant effect on healthcare requesters perceived usefulness of the HC-VBE application.
- iv. H₂: Perceived healthcare quality has a significant effect on healthcare requesters perceived usefulness of the HC-VBE application.
- v. **H**₃**a:** Perceived ease of use has a significant effect on healthcare requesters perceived usefulness of the HC-VOBE application.
- vi. **H**₃**b**: Perceived ease of use has a significant effect on healthcare requesters attitude towards using the HC-VBE application.
- vii. **H**₄**a:** Perceived usefulness has a significant effect on healthcare requesters attitude towards using the HC-VBE application.
- viii. **H**₄**b**: Perceived usefulness has a significant effect on healthcare requesters intention to use the HC-VBE application.
- ix. H₅: Attitude towards using has a significant effect on healthcare requesters intention to use the HC-VBE application.

4.3.3 Research Instruments (Step 3)

Researchers have used various methods to examine new technology acceptance by users, for example authors in [194] have used open question and answer sessions to gage users thoughts and feedbacks on users satisfaction of a clinical system they have proposed. [193] Used one-to-one and one-to-many interviews to test nurses acceptance of their nursing care plan system and [195] used observations to collect data for their critical care information system acceptance evaluation. However, according to [107], survey

questionnaire is the most widely used data collection method for user acceptance evaluation of a new healthcare information system. Therefore, survey questionnaire method was selected to be used in this research as a data collection instrument. Three sets of questionnaires based on measurement items for each construct in the given extended TAMs in Sections 4.3.2.1, 4.3.2.2 and 4.3.2.3 were designed and printed on A4 sheets. The questionnaires were then given to prospective users of the two developed frameworks to fill in.

The filled in questionnaires were collected straight after they were filled in by survey participants. The data were then transferred into SPSS datasheets to be processed further. A questionnaire set was composed of two separate forms, the first form was to obtain the participants consent to take part in the research and the second form was the questionnaire to be filled for the acceptance testing. The questionnaire sets were translated to the Kurdish language by a professional legal translator which is the local language were the data is collected. The translated version was given out alongside the English version to help those whose level of English was not adequate to understand the questions. The participants were asked to read the Kurdish version of the questionnaire items but fill in the English version as no scales were provided on the Kurdish version. Translating questionnaire items to the local language was similarly performed by authors in [159].

The filled in questionnaires were screened for completeness to ensure that there is no missing data. If any unfilled section spotted the relevant participant was asked to revise it. All filled in questionnaires are numbered and kept for future references and verifications along with the participant details. At the beginning of data collection sessions, survey participants were informed of their rights to withdraw from the process at any time they wished as participation was voluntary.

4.3.4 Research Measurements (Step 4)

The questionnaires were designed based on the TAMs extended in Section 4.3.2, each model consisted of 5-6 unobserved variables (latent variable). For each latent variable,

4-5 observed items were defined based on previously defined and validated questionnaire items from studies such as [196], [177], [197], [157] and [108]. Unobserved variables cannot be measured directly by survey participants as their scope is broad and flexible to answer, however, observed items are measurable directly by survey participants as they are formulated to capture a specific aspect of users' intention to use a system. Unobserved variables are measured as a result of analysing the collective answers for all its observed variables.

To measure each item and gage participants feeling, Linkert Scale measurement was used as it is one of the most popular measurement scales used by researchers [107]. Questionnaire items were measured on a seven-point Linkert-type scale and the measures were (1) strongly disagree (2) quite disagree, (3) disagree, (4) neutral, (5) slightly agree, (6) quite agree and (7) strongly agree. To ensure the appropriateness of the questionnaire items and make necessary changes to its structure and meanings, they were pilot tested by three small groups per the target users before their actual use. As a result, the wording and the structure of several questionnaire items were changed before their actual use.

4.3.5 Sampling Methods and Outcomes (Step 5)

The surveyed participants of this research were mainly from the Kurdistan Region of Iraq where the author currently works as a lecturer and some were from Finland where the author stayed for a month as a pedagogy student. The target audience for the first part of this study was system developers including software engineers, system designers, programmers, project managers and university lecturers who teach software developments. They were contacted though paying them a visit in their working places or acquiring their personal contact numbers. They were then given locations, dates and times for conducting the research. The original target number was 100 participants however, there aren't many system developers in the region and therefore only 28 participants turned up for this part of the research. In total 82% of the surveyed participants were male and 18% were female. Table 4.1 shows the demography of the surveyed participants.

79

Table 4.1: System developers' dem	nographic data who participated	in the survey after	using the
Java application developed b	pased on the HC-VBE-M-F		

Variable	Item	System Developer Frequency (N=28)	%
	Male	23	82
Sex	Female	5	18
	Age 15-20	0	0
Age	Age 20-30	9	32
	Age 30-40	15	54
	Age 40-50	4	14
	Age 50-60	0	0
	Age 60-above	0	0
Occupation	System developers	28	100

The target audiences for the second part of the study were healthcare providers and requesters. To recruit healthcare providers, hospitals and healthcare directorates were contacted in the region. Permissions such as the ones in Appendix E were requested to conduct the research inside hospitals during working hours or outside working hours in cultural centres. Incentives such as offering free dinners were one of the techniques used to recruit research participants. The concept of healthcare provider is a general one, therefore, all healthcare related professionals were invited such as doctors, nurses, pharmacist, first aiders etc. As for the healthcare requesters, lecturers and students from the University of Human development in the Kurdistan Region were recruited. The target number for both healthcare providers and requesters were 100 each, i.e. 200 in total. For this part of the research, the targets were met.

For healthcare providers, in total 56% of the surveyed participants were male and 44% were female. The highest age rang was 30-40 years with 32% and the highest healthcare provider participant were nurses with 53%. As for the requesters, in total 61% were male and 39% were female. The highest age range was 20-30 with 44% and the highest prospective healthcare requester participants were University lecturers with 57%. Table 4.2 shows the demography of the surveyed participants.

Table 4.2: Healthcare providers and requesters demographic data that tested the mobile	ļ
application developed based on the HC-VBE-F	

Variable	Item	Healthcare Provider	%	Healthcare Requester	%
		Frequency (N=100)		Frequency (N=100)	
	Male	56	56	61	61
Sex	Female	44	44	39	39
	Age 15-20	0	0	3	3
	Age 20-30	27	27	44	44
Age	Age 30-40	32	32	36	36
	Age 40-50	30	30	9	9
	Age 50-60	11	11	7	7
	Age 60-above	0	0	1	1
	Nurse	53	53	0	0
	Doctor	15	15	0	0
	Other medical	32	32	0	0
Occupation	staff				
	Student	0	0	43	43
	Lecturer	0	0	57	57

4.3.6 Applications Testing and Data Collection (Step 6)

There were two applications developed for both the modelling and the virtual collaboration frameworks described in Chapters 5 and 6. The first application was implemented as a Java stand-alone application and the second application was a mobile application. They were both developed to be used by prospective users as part of user acceptance evaluation process which is designed based on TAM.

The application testing method for each of the applications were slightly different as described in the next two sections, however a direct face to face application testing method was chosen over an online one following the direction of others such as [161]. They noticed that the method has the advantage of addressing concerns and questions of participants which they may have during testing and filling in the questionnaire; the

same advantage was noticed during the data collection process for this thesis. The data collected was documented in SPSS, Figure 4.11 shows some of the data documented after following the applications testing methods described in Sections 4.3.6.1 and 4.3.6.1. The first left column represents participants and the numbers on the rest of the columns represent answers given by the participants for each questionnaire item specified by each column header. The Column headers are abbreviations for questionnaire items which are described in Sections 2.9 and 4.3.2.

👍 *vb	e-profe	sional da	ta-modif	ied.sav [D	ataSet	:1] - IBM SI	PSS Stat	istics Da	ata Edito	r																_	٥	×
<u>F</u> ile	<u>E</u> dit	<u>V</u> iew	<u>D</u> ata	Transform	m <u>/</u>	Analyze	<u>G</u> raph	s <u>U</u> ti	lities	Extensio	ns <u>W</u> ir	ndow	<u>H</u> elp															
				r	2		Ł		Ч	h			A 14)														
5:																									Vi	isible: 2	4 of 24 Va	riable
		🐥 PHG 1	PH 2	G 💊 PH	IG	PHG 4	PU 1	8 PU 2	💰 PU 3	💰 PU 4	뤚 PEU 1	💰 PEU 2	PEU	💰 PEU 4	RCE	💰 PCE 2	PCE 3	RCE	뤚 AU 1	💰 AU 2	💰 AU 3	💰 🕺	뤙 <mark>IU</mark> 1	🚴 <mark>IU</mark> 🤹	8 IU 3	₿ <mark>IU</mark> 4	var	
1		(6	7	7	5	5	7	6	6	5		6 6	6	7	7	6	6	6	6	7	6	6	5	4	5		1
2	2	ł	i	6	6	7	4	6	6	7	4		5 5	5	6	6	5	6	7	7	5	6	6	6	6	6		
3	3	5	5	3	5	5	7	7	6	7	4		4 5	3	4	4	4	5	7	6	7	5	6	5	5	6		
4		1	1	6	6	7	7	7	6	7	5		65	7	6	7	7	6	6	7	4	5	6	7	7	6		
5	;	2	2	1	1	2	2	1	1	1	7		6 6	6	3	2	2	2	3	3	2	4	5	5	4	5		
6	6	5	5	6	7	6	3	4	3	5	3		4 3	5	3	4	3	5	7	6	5	4	5	6	4	6		
7	7	6	6	5	5	5	4	6	5	6	5		55	4	4	5	6	5	5	6	4	4	5	5	6	5		
8	3	(6	5	7	7	6	6	5	7	5		6 4	3	5	6	4	6	7	6	5	7	7	6	7	6		
9)	6	6	7	6	6	5	6	6	6	6		55	7	5	7	6	5	7	6	7	7	7	7	7	7		
1	0	(6	7	7	6	7	6	6	7	5		56	6	7	7	6	7	7	6	6	7	6	7	7	7		
1	1	6	6	5	4	6	5	7	6	5	5		65	6	6	7	4	6	7	6	7	7	6	7	6	7		
13	2	(6	6	4	6	6	6	7	7	5		7 6	7	7	7	6	7	7	6	7	7	7	6	6	7		
1	3	6	6	7	6	7	7	7	6	7	7		67	7	7	7	7	7	7	7	6	7	5	6	6	6		
14	4	1	7	7	7	7	7	7	7	7	6		7 7	7	7	7	7	6	7	7	7	5	6	7	7	6		
- 1	5	6	6	6	6	7	4	6	5	6	4		4 5	5	6	5	5	4	6	5	6	6	7	5	5	5		
10	6	7	7	7	7	7	7	7	7	7	5		57	6	7	7	7	7	6	6	6	6	6	7	6	7		
1	7	1	7	7	7	7	7	7	7	7	5		66	7	6	7	7	6	7	6	7	7	6	6	7	6		
- 1	8	1	1	7	7	7	7	7	7	7	7		7 7	7	7	7	7	7	7	7	7	7	7	7	7	7		
- 19	9	(6	5	6	7	5	6	6	7	7		67	7	5	6	7	6	7	6	7	7	7	7	7	7		
2	0	4	L I	5	6	7	6	5	7	7	7		67	7	6	7	6	7	7	7	6	7	7	7	7	7		

Figure 4.11: A screenshot of sample data collected based on the surveyed questionnaires developed for testing both frameworks

4.3.6.1 HC-VBE-M-F Application Testing Method

The developed Java application was made available on a number of laptops for use, also perspective users (healthcare system developers) were given the option to put the application on their laptops and desktops if they wished. There were occasions that research participants thought a demonstration of the application was good enough for them to fill in the questionnaires. The testing process was given about an hour and it was performed in four steps explained below and pictured in Figure 4.12.

1- In short seminar style presentation, the modelling framework concepts and components were explained to participants. The aim was to familiarise participants

with the concept behind the modelling framework as well as the components and modelling steps required to model a healthcare virtual collaboration scenario. During the presentation, concerns and questions of participants were answered which helped them to progress quickly during the testing stage.

- 2- As a second step, a healthcare virtual collaboration scenario was provided to the participants to model using the Java application.
- 3- In this step, the relevant questionnaire was handed out to the participants to fill in. They were asked to select a number that best represent their answer with regard to each questionnaire item on a 7-point Linkert Scale.
- 4- This step was the final step where completed questionnaires were collected and checked for completeness. The data was then transferred into SPSS data sheets to be processed later during data analysis.



Figure 4.12: HC-VBE-M-F Application Evaluation and data collection Steps

4.3.6.2 HC-VBE-F Mobile Application Testing Method

To test the virtual collaboration framework developed in this study, the functionalities and working steps of the mobile application which is developed based on the framework were demonstrated to research participants. Demonstrating the working steps of the mobile application allowed users to experience the full working process of the prototype without the need for them to spend too much time on getting to know the different functionalities of the application before actual use.

Although the option to use the protype was given to participants but in most cases, users felt that the demonstration is good enough for them to understand the concept behind the application. Demonstrating the mobile application saved time for both the researcher and the users during the process of data collection. This method has also been used by other researchers, for example the authors of [198] showed patients the way that their developed online posting for Emergency Department waiting time worked rather than letting users use the website; and the authors of [199] used a similar method to test a home telemedicine system.

To demonstrate the working process of the prototype the following five steps were taken which are pictured in Figure 4.13.

- 1- Two or three prospective users (requester and providers) from the recruited research participants were invited to download the application onto their smart phones if they wanted, if not, readily set up mobile devices were provided.
- 2- They were instructed to register as either a healthcare provider or healthcare requester.
- 3- Once they registered, they were guided step by step regarding how to make a healthcare service request in the case of requesters, and how to make a service offer in the case of providers.
- 4- The HC-VO creation and operation by the HC-VBE mobile application was demonstrated from start to finish, each step of the process was announced to the attended audience and the outcome was shown around. Internet was needed for

each smart phone in the demonstration process as the mobile application had to connect to a database hosted externally which was developed as a part of the application.

5- After allowing some times for questions and answers from the participants, the printed questionnaires were distributed to be filled in by the participants. The filled in questionnaires were then collected and the data from them were documented using SPSS software.



Figure 4.13: HC-VBE mobile application evaluation and data collection steps

4.3.7 Data Analysis (Step 7)

Structural Equation Modelling (SEM) was utilised to analyse the data obtained from the survey questionnaires filled in as a result of testing the applications. SEM is one of the most widely used techniques to statistically analyse collected data for the constructs of a TAM [112]. SEM uses path analysis, regression testing and other methods to statistically estimate and examine the extent of relationships between unobserved variables and their observed items in a specific model [110]. To draw the extended TAMs for this study and

perform the SEM analysis, AMOS version 23, SPSS version 25 and Microsoft excel 2019 were used. AMOS is a tool that utilizes structural model to compute several different statistical results for acceptance testing such as standardised estimates for each test path [180]. Figure 4.14 is screenshots of the AMOS program and outcomes of some of the analysed data, such as factor loading between observed items and their related unobserved variable.



Figure 4.14: A screenshot of AMOS model created based on one of the extended TAM and some computed statistical results for the model

To understand how to use AMOS, IBM provides and updates AMOS user manual regularly [200]. The AMOS models developed for the three extended TAMs are similar to the one in Figure 4.14 and they are presented and explained in Section 4.3.7.1, 4.3.7.2 and 4.3.7.3. The arrows between the latent variables indicate the causal effects between them. The arrows between the latent variables and the observed items indicate the item loading on each latent variable. The strength of the loading is indicated by a number on each loading

which is between 0 and 1. The closer the value to 1 the stronger the item loading on the latent variable. The circles pointing to the latent variables and observed items represent error values in the item loading. The generation of such values on a SEM diagram indicates that the model has been run by AMOS successfully without any error.

4.3.7.1 HC-VBE-Modelling Framework AMOS Model

Figure 4.15 is the SEM model developed in AMOS based on the extended TAM explained in Section 4.3.2.1. The figure shows all the latent (unobserved variable) with their related measured items (observed variable). Each latent variable is a construct in the corresponding TAM. In this case, there are five latent variables represented by the oval shapes, namely Perceived Ability to Model (PAM), Perceived Ease of Use (PEU), Perceived Usefulness (PU), Attitude towards Using (AU) and Intention to Use (IU). In total, there are 20 observed items, 4 for each latent variable represented by the squares.



Figure 4.15: HC-VBE-M-F-TAM Structural Equation Model developed in AMOS

87

4.3.7.2 HC-VBE-F-Provider AMOS Model

Figure 4.16 is the SEM model developed in AMOS based on the extended TAM explained in Section4.3.2.2. The model consists of six latent (unobserved) variables represented by the oval shapes, namely Perceived Clinical Effectiveness (PCE), Perceived Healthcare Globalisation (PHG), Perceived Ease of Use (PEU), Perceived Usefulness (EU), Attitude towards Using (AU) and Intention to Use (IU). In total there are 24 observed items, 4 items for each latent variable denoted by the squares.



Figure 4.16: HC-VBE-F-TAM-Provider Structural Equation Model developed in AMOS

4.3.7.3 HC-VBE-F-Requester AMOS Model

Figure 4.17 is the SEM model developed in AMOS based on the extended TAM explained in Section 4.3.2.3. It shows all the latent (unobserved constructs) with their related measured items (observed variable). The model consists of five latent variables represented by the oval shapes, namely Perceived Healthcare Quality (PHQ), Perceived Healthcare Availability (PHA), Perceived Ease of Use (PEU), Perceived Usefulness (PU), Attitude towards Using (AU) and Intention to Use (IU). In total there are 25 observed items, 4 for each latent variable except PHQ which has 5 items.



Figure 4.17: HC-VBE-M-TAM-Requester Structural Equation Model developed in AMOS

4.3.8 Statistical Results Selection (Step 8)

Many different statistical results can be computed for data analysed in AMOS and SPSS, in this section, a brief introduction is provided for each of the statistical values selected to be computed to support the evaluation process and show users acceptance of the developed healthcare virtual collaboration modelling and provision frameworks. The selected statistical results are Mean, Standard Deviation, Factor Loading, Significant *p* Value, Correlations Coefficient Matrix, Average Variance Extracted, Construct Reliability, Cronbach's Alfa, Kaiser-Meyer-Olkin and Model Fit Indices.

4.3.8.1 Mean

Mean is defined as the sum of all observations in a distribution divided by the total number of all observations [201]. It is calculated to show the average of all values in a specific distribution; in the context of this research two different mean (average) values are computed. The first one is the average of all values given to a particular observed item, which means, if we have 24 observed items in a TAM then we have calculated 24 different means. The second one is the average value of all the observed item for a particular unobserved variable, which means, if we have 6 unobserved variables in a TAM then we have 6 different average values. Mean present researchers with a single value that is representative of all the answers given to a particular questionnaire item and it is a powerful tool that enable researchers to easily draw conclusions. Equation 4.1 is used to calculate Mean (average).

$$A = \frac{1}{n} * \sum_{i=1}^{n} X_i$$

Equation 4.1: The equation for calculating "Mean"

where:

- *A* = average (or arithmetic mean)
- *n* = the number of participants in a research
- *x* = the values given to a particular questionnaire item.

4.3.8.2 Standard Deviation

Standard deviation measures how far the values in a particular distribution are from the average value [202]. Big standard deviation value indicates widely spread values from the mean; the bigger the standard deviation value the more widely spread are the values in a given set of values. In the context of this research small standard deviation value is preferred since it indicates that participants' perception towards a particular item in a questionnaire are close to each other. It provides support to conclude whether answers

given to a specific item by perspective users are consistent or not. Equation 4.2 is the mathematical formula for standard deviation.:

$$s = \sqrt{\frac{\sum_{i=1}^{n} (X_i - A)^2}{n - 1}}$$

Equation 4.2: The equation for calculating standard deviation

Where:

- S = standard deviation
- *x* = the value of each data item
- *A* = average values
- *n* = number of data items

4.3.8.3 Factor Loading

Factor loading is the indication of relationship strength between an unobserved variable and it's observed variable the higher the loading value the stronger the relationship [203]. In SEM the latent variable is indicated by an oval shape, the observed item is indicated by a square and the error variance for each observed item is indicated by a circle [204]. Figure 4.18 shows a SEM diagram and Equation 4.3 is the general formula for factor loading calculation.

$$X_m = \lambda_m F_m + e_m$$

Equation 4.3: The equation for calculating factor loading

Where:

X is the observed item, F is the unobserved variable, λ is the factor loading of X on F, e is the measurement error of X and m is the observed item number. For instance, if we have a latent variable (*F*) that has four observed items ($X_1...X_4$) then the formula for each factor loading (λ) is:



Figure 4.18: Structural Equation Model components

4.3.8.4 Significant p value

To accept or reject a hypothesis, researchers use "significant p value" which is a number between 0 and 1 and the acceptable level of significance is set as 0.05 [205]. A hypothesis is accepted if the significance value calculated is less than the set value. p value is calculated using Equation 4.4.

$$z = \frac{\hat{p} - p0}{\sqrt{\frac{p0(1 - p0)}{n}}}$$

Equation 4.4: The equation for calculating significant *p* value

where:

- *z* is a value to be looked up in significance value table
- \hat{p} is the sample proportion
- *p*0 is the assumed population proportion in the null hypothesis
- *n* is the sample size

4.3.8.5 Correlation Coefficient Matrix

Correlation coefficient is also known as Pearson Product-Moment Correlation, measures the covariance of two variables [206]. Usually a collection of variables together provides an overview of a goal, for example in this research more than 20 questionnaire items (observed variables) are used together with the aim to show whether users accept the proposed technologies for healthcare virtual collaboration or not. Correlation coefficient matrix is a method to summarise data where values above 0.3 is considered acceptable according to [207]. Figure 4.19 shows the structure of a correlation coefficient results table.



Variables for which correlation coefficients are calculated

Figure 4.19: Structure of correlation coefficient results table

4.3.8.6 AVE (Average Variance Extracted)

AVE measures the convergent validity that is used to examine measurements, and the calculated value indicates the internal consistency of answers for a set of related questionnaire items [208]. Values above 0.5 is considered acceptable and Equation 4.5 is used to calculate AVE.

$$AVE = \frac{\sum \lambda^2}{n}$$

Equation 4.5: The equation for calculating Average Variance Extracted

Where:

- AVE is the average variance extracted
- λ^2 is the square of factor loading
- *n* is the number of indicators

4.3.8.7 CR (Construct Reliability)

CR measures internal consistency of measurements similar to Cronbach Alpha [209]. *CR* values above 0.7 is considered to represent acceptable reliability. Equation 4.6 is used to calculate *CR*.

$$CR = \frac{\left(\sum \lambda\right)^2}{\left(\sum \lambda\right)^2 + \left(\sum \delta\right)}$$

Equation 4.6: The equation for calculating composite reliability

Where:

- CR is the composite reliability
- λ is factor loading
- δ is measurement error

4.3.8.8 Cronbach's Alpha

Lee Cronbach developed the statistical model which is known as "Cronbach alpha" to help researchers calculate the reliability of measures of a test [210]. Cronbach alpha is the statistical measure used for internal research variable reliability and validity which is a value between 0 and 1 and measures ranging above 0.6 is considered a good reliability and validity indicator [211].

In the context of this research Cronbach alpha helps to decide whether the observed items (questions in a questionnaire defined for a particular unobserved variable) correctly

measures the unobserved variables such as Perceived Usefulness. Equation 4.7 is used to calculate Cronbach alpha.

$$\alpha = \frac{n^* \overline{A}}{\overline{V} + (n-1)^* \overline{C}}$$

Equation 4.7: The equation for calculating Cronbach Alpha

Where:

- *n* = the number of items.
- \overline{A} = average covariance between two observed items.
- \bar{v} = average variance.
- α = calculated Cronbach's Alpha value

4.3.8.9 Kaiser-Meyer-Olkin

Kaiser-Meyer-Olkin test is carried out on data sets to assess the adequacy of the data sample to carry out factor analysis [212]. A sample values above 0.5 is considered adequate for factor analysis. Equation 4.8 is used to calculate the value.

$$KMO_{j} = \frac{\sum_{i \neq j} r_{ij}^{2}}{\sum_{i \neq j} r_{ij}^{2} + \sum_{i \neq j} u_{ij}^{2}}$$

Equation 4.8: The equation for calculating KMO

Where:

- KMO is the sample adequacy measurement value
- *r* is the correlation matrix
- *u* is the partial covariance matrix
- the calculation will be performed under the assumption that *j≠i*

4.3.8.10 Model Fit Indices

Model fit statistical values are calculated to show how best a model represents theories presented in a research and there are many different fit indices which have made it difficult to bring about consensus among researchers regarding their use [213]. According to [214], the most used fit indices are chi-square (χ 2), GFI, TLI, CFI, RMSEA and SRMR for which they provide a table where the definition and recommended values of the indices are outlined. Kline in his famous book [215] (p.221) suggests that as a minimum, Chi-Square, RMSEA, CFI and SRMR should be reported for model fit analysis. According to [213] absolute fit indices such as Chi-Squared test, RMSEA, GFI and SRMR "provide the most fundamental indication of how well the proposed theory fits the data". Based on their suggestions fit indices listed in Table 4.3 were used to evaluate the fitness of the extended TAMs in this study.

Model fit index	Description	Recommended Value
CMIN/DF	Chi Square in AMOS	Between 1 and 3
CFI	Comparative fit index (baseline comparison)	CFI ≥ 0.90
TLI	Tucker Lewis Index (baseline comparison)	TLI ≥ 0.90
IFI	Incremental fit index (base line comparison)	IFI ≥ 0.90
RMSEA	Root Mean Square Error of Approximation (Absolute fit indices)	RMSEA ≤ 0.08
SRMR	(Standardized) Root Mean Square Residual	SRMR ≤ 0.08

4.4 Chapter 4 Summary

This chapter provided a detail account of the methodology and evaluation strategy used to achieve the aims and objectives of this thesis. The following points summaries the main content of the chapter:

1- A deductive research approach is followed to complete this research. This is where theoretical frameworks are developed first and then evaluated empirically. The

research design consists of 5 stages which care research the healthcare sector, research virtual collaboration, develop virtual collaboration modelling framework, develop virtual collaboration framework and finally implement and evaluate the frameworks.

- 2- An eight steps evaluation strategy is followed to implement and evaluate the two developed frameworks. The first step is to implement the frameworks; the modelling framework is implemented as a Java stand-alone application and the collaboration framework is implemented as a mobile application. In step two, TAM is extended three times to adapt it for this research. The first TAM is extended with one new construct (Perceived Ability to Model) for evaluating the healthcare virtual collaboration model. The second TAM is extended with two new constructs (Perceived Healthcare Globalisation and Perceived Clinical Effectiveness) to examine the acceptance of the collaboration framework from healthcare providers perspective. The third TAM is extended with two new constructs (Perceived Healthcare availability) to examine the collaboration framework from healthcare Quality and Perceived Healthcare availability) to examine the collaboration framework from healthcare TAMs to be tested by relevant target audience.
- 3- Survey questionnaire is used as a data collection instrument and research measurements are developed based on 7 points Likert-Scale measurements.
- 4- Survey participants from the Kurdistan Region of Iraq, where the author currently works as a lecturer, are invited to participant in the research. In total, 28 system developers participated in evaluating the healthcare virtual collaboration modelling framework presented in Chapter 5, 100 healthcare requesters and 100 healthcare providers participated in evaluating the healthcare virtual collaboration framework presented in Chapter 6.
- 5- Structural Equation Modelling (SEM) is used to model the three extended TAM using AMOS software. The SEM models are used to analyse collected data and compute statistical results such as mean, standard deviation, factor loading and model fit indices.

The next chapter presents the Healthcare Virtual Collaboration Modelling Framework (HC-VBE-M-F).

Chapter 5 : Modelling Framework

This chapter presents a modelling framework specific to healthcare virtual collaboration, developed based on VBE and VO concepts. Section 5.1 provides a brief opening about the characteristics and components of the modelling framework. The framework consists of three parts which are: A Service and Participant Classification Mechanism (SPCM) described in Section 5.2, a Domain Specific Modelling Language (DSML) described in Section 5.3 and a Service Orchestration Description Language (SODL) described in Section 5.4. The framework is implemented as a Java application and evaluated by users' (System designers and software developers) for acceptance using TAM. Section 5.5 presents the evaluation results and in Section 5.6 the results are discussed. Finally, a summary of the chapter is provided in Section 5.7

5.1 Healthcare Virtual Collaboration Modelling Framework

Developing a modelling framework that is understandable by stakeholders is a research challenge [216]. Modelling VO for healthcare allow healthcare stakeholders to understand virtual collaboration at a level of abstraction that doesn't require technological expertise. Modelling also allow picturing a VO structure independent from its operational aspect [216]. In this study, a modelling framework for healthcare virtual collaboration is developed with following attributes in mind:

- 1- Be able to model VBE and VO based collaboration scenarios for the healthcare domain.
- 2- Be simple, understandable and usable by healthcare stakeholders.
- 3- Be platform and technology independent.

Developing a modelling framework for a complex sector as healthcare, and be able to capture an aspect of collaboration (virtual collaboration in this case) between its

stakeholders to provide a healthcare service, is a challenging task. Some backgrounds and literature reviews in this regard are provided in Sections 2.6 and 3.1. The aim here is to answer the two specific research questions relating to the modelling aspect of this research which are:

- 1- How to classify the main stakeholders and services in virtual collaboration for healthcare?
- 2- How to model and describe service provision in virtual collaboration for healthcare?

To answer the research questions relating to the modelling aspect of this research, the direction of [28] is followed to model healthcare virtual collaboration structure and behaviour. The modelling framework developed for VBE and VO based healthcare virtual collaboration consists of three parts which are Service and Member Classification Mechanism, Domain Specific Modelling Language and Service Orchestration Description Language. The framework components are pictured in Figure 5.1.



Figure 5.1: HC-VBE Modelling framework components (SPCM, DSML and SODL)

5.2 Service and Participant Classification Mechanism

Developing a classification mechanism that is representative of all stakeholders and services in a sector is a research challenge [216]. To classify services and participants in healthcare on an abstract level, one must identify the main components in terms of environments, stakeholders and services first, for this purpose, research articles such as [217] and [218] as well as online sources such as websites about healthcare is studied . Figure 5.2 presents the findings of the study in terms of different types of roles and services that exists in healthcare sector.

In general, 6 classes of roles and services are identified, which are, care providers, executives, consumers, care environments, electronic devices and healthcare services. Care providers can be broken down to a number of sub-categories such as healthcare equipment manufacturer, healthcare communities and agencies, academic institutions, financial agencies, governmental agencies, pharmaceutical companies and medical professionals. Each of the sub-categories is composed of a number of sub-sub-categories, for instance medical professionals include doctors, nurses and pharmacists.

The executive class represents all individuals and organisations who oversee and manage healthcare environments. Consumers represent patients and people around patients who are affected during a healthcare service provision. Care environments represent places where healthcare is provided such as hospitals and patient homes. Electronic devices represent all electronic equipment that support the process of healthcare provision used by both patients and doctors. Finally, the service can be divided over healthcare related administrative services and care related services.

In a typical virtual collaboration scenario, many of the identified stakeholders will have to collaborate to provide a service. This clearly presents a complex collaboration setting that requires a simplification mechanism to understand. Abstract models can simplify the complexity, this where a modelling language that is simple and understand by stakeholders play a vital role.

-100



Figure 5.2: Summary of healthcare roles and services identified as a result of studying resources in the form of articles and websites

The findings presented in Figure 5.2 have served as the base for developing the service and participant classification framework which is outlined in Section 5.2.1. In order to develop an abstract and generic classification that is simple and inclusive, the categories are further refined into three main classes which are stakeholders, environment and services as shown in Figure 5.3. Here, a brief description of each refined category is provided:

- 1- Stakeholder: The class represents all individuals, organisations, governmental and none governmental agencies who participant in providing, consuming and managing healthcare. They can be classified into three sperate sub-classes:
 - a- *Care providers*: These are healthcare professionals and organisations that provide care.

- b- *Executives:* These are healthcare organisers and managers who ensure the smooth running of healthcare related operations.
- c- *Care consumers*: These are patients and family members who receive healthcare services.
- 2- **Environment**: This is the environment where healthcare is provided. Generally, they consist of two types:
 - a- *Physical environment:* This type of environment is the traditional healthcare environments where healthcare is provided such as hospitals, clinics and institutions.
 - b- *Electronic environment*: This type includes all the electronic systems where care is provided using information and communication technologies.
- 3- **Services**: This class is concerned with all the services required to provide, run and maintain healthcare services; they can be classified into two types:
 - a- *Care related service*: This type includes all the services that are concerned with providing care to consumers.
 - b- *Management related service*: This type of service is concerned with managing the healthcare environments to ensure the orderly operation of care provision.



Figure 5.3: Refined healthcare roles, services and environment classifications

-102

5.2.1 SPCM Description

To describe virtual collaboration scenarios in healthcare in a manner that both healthcare system developers and stakeholders understand, a simple service and participant classification mechanism would contribute to the process positively. In developing the SPCM, the author has followed the direction of other researchers who have suggested classification frameworks similar to what is outline in this section. For instance the authors in [28] have proposed a general business-oriented actor and resource classification framework to model VBEs and VOs at abstract levels. Grid technology has been used by the authors in [219] to define resources and assign roles and services in managing healthcare virtual organisation.

The authors of [27] propose a framework by which general roles and service relating to VBE and VO in business can be defined, however, they state that "Clearly neither Competencies nor resources of the organizations in the VBE for Healthcare (e.g. Doctors practice office, insurance company, ambulance services, etc.) can be defined by the same ontology". A framework that uses the concept of service-based agents to classify virtual organisation participants is outlined by the authors of [118]. To define the SPCM for VBE and VO based healthcare virtual collaborations, the direction of [15] is followed. In this thesis, a role-based classification method is used to develop the mechanism using the roles and services summarised in Section 5.2. The mechanism is aimed to achieve two objectives, first to be general and wholistic in classifying services and participants in healthcare virtual collaboration and second, be simple and understandable by healthcare stake holders.

Figure 5.4 shows the SPCM classes and their mapping with the healthcare classes identified in Section 5.2. The mechanism is developed to classify participants and services for VBE and VO-based healthcare virtual collaboration. It is comprised of six classes which is can be applied to all healthcare virtual collaboration scenarios. The mechanism addresses the structural organisation of VBE and VO-based healthcare collaboration on an abstract level. It is designed with healthcare system developers in mind to provide

them with a mechanism that they can use to breakdown complex scenarios for system design purposes.



Figure 5.4: Healthcare service and participants classification mechanism mapped to the six classes of stakeholders and services

As illustrated in Figure 5.4, the mechanism classifies roles and services on VBE and VO levels as below:

1. HC-VBE level: Participants are classified into two distinct classes of participants, the first is **Organiser** and the second is **Support**. Service is classified as one class which is **Task**.

- a) Organiser: Roles that participant in the establishment of a HC-VBE are called Organiser which can be individuals and organisations. This class is denoted by <<Organiser>> Name and has the following responsibilities:
 - Create HC-VBE to provide an environment for healthcare service provision virtually.

- Oversee and manage a HC-VBE environment and operations.
- Recruit other participants who wish to contribute to the creation and management aspect of HC-VBE.
- Ensure resource availability and accessibility as required for operating a HC-VBE and HC-VO.
- Ensure all necessary facilities are available for timely HC-VO creation.
- b) Support: Represent electronic resources such as decision support systems and databases which provide operational support to HC-VBE or HC-VO. It is denoted by <<Support>> Name and has following responsibilities:
 - Data storage.
 - Organise communications between healthcare service provider and requester.
 - Collaboration and sharing facilitation.
 - SLA creation and management.
 - Provide electronic Input mechanisms for resources as required from HC-VBE internal and external sources.
 - Provide output mechanisms for managers, care providers and receivers.
- c) **Task:** All HC-VBE management activities that are carried out by *Organisers* are classed as *Task,* which is denoted by *<<Task>> Name* and has the following responsibilities:
 - Provide a common interface for *Organiser* collaborations.
 - Administer resource allocations.
 - Define and manage resource access privileges.
 - Define Organiser rights and responsibilities within a HC-VBE.

2. HC-VO level: Participants are classified into two distinct classes of participants, the first is **Requester** and the second is **Provider**. Service is classified as one class which is **Service**.

- a) Requester: This class represents virtual collaboration participants who request a healthcare service from a HC-VBE; it is documented as <<Requestor>> Name and has the following responsibilities:
 - Request virtual healthcare services.

- Respond to queries that are sent by HC-VBE participants (Organiser and provider).
- Negotiate a healthcare service to be received.
- Approve SLA after negotiation.
- Pay for healthcare services provided.
- Rate Providers and services provided.
- b) Provider: Healthcare professionals who are recruited by a HC-VBE to provide services in HC-VOs are classed as *Provider*. For example, a psychologist can be recruited to provide a consultation service to a patient. It is documented as <<*Provider>> name* and has the following responsibilities:
 - Provide a healthcare service.
 - Negotiate a healthcare service to be provided.
 - Approve SLA after negotiation.
 - Expect to be rewarded after complete service provision.
- c) Service: The class represents short-lived virtual collaborations known as HC-VOs which are created by HC-VBEs to provide a specific service for a specific period to a *Requester*. The responsibilities of *Service* are similar to those of *Task* and it is documented as <<*Service*>> Name.

5.3 HC-VBE Domain Specific Modelling Language

Theoretically, the dynamic nature of VO has the potential to offer solutions to many different problems that require timely solutions, however, when it comes to the application of the concept, there are a number of structural, functional and behavioural barriers that are yet to be finalised. Describing a process in detail helps in simplifying its complexity; models can provide a form of description that *"increases the readability of a process and its evaluation"* [220]. The concept of VO supports collaboration over time and space, but before such a concept is realised it has to be modelled as models facilitate implementation process [221].

Models and visualisation techniques are used in many fields of science such as engineering and manufacturing to facilitate a better understanding of a particular
scenario, however "Visualization of aspects of virtual collaboration has received relatively little research attention in the past" [36]. A structural model that could visualise how different stakeholders in a typical HC-VO is structured is a significant contribution to the realisation and implementation of the concept in healthcare [78]. The authors of [15] have implemented a virtual medical team creation and management system in Cyprus which is an example of the kind of systems that the DSML developed in this thesis can contribute to its design and development.

Following the directions of [222] and [118], conceptual models independent from implementation details is explicitly chosen to provide a generic view of virtual collaboration for healthcare. Authors in [223] claim that "Using non-formal representation often helps to gain a common understanding of the problems at hand". The authors in [224] state that to facilitate the description of challenges facing the healthcare domain due to its complexity, conceptual models can be used. The ability of Service Oriented Architecture (SOA) to show a number of service providers working together, makes it a good candidate to be used for modelling in healthcare. But, SOA has mainly been used to model the technical aspects of interoperability in healthcare systems "not modelling the actual health care processes" [14].

Unified Modelling Language (UML) is a popular modelling language which is used widely by system developers to model the behavioural, structural, and conceptual aspect of a process [225]. There are researchers who have attempted to use UML diagrams to model some aspects of healthcare. For instance, UML has been used to model a number of healthcare related processes such as optimising hospital processes [226], organising hospital-based cancer registration process [227]. UML models are capable of modelling different system aspects, one of which is a use case diagram. UML use case is the model that is chosen as the base to develop the DSML for healthcare virtual collaboration in this thesis.

UML use case diagram has become the tool to draw simple and static views of a system where actors and functions of a system are modelled and documented; it is easy to understand and require limited introduction to the model notation [228]. The graphical notations offered by UML to draw a use case diagram are generic not specific, for example

the stickman and the oval shape notations pictured in Figure 5.5 are used to represent all types of actors and functionalities of a system, regardless of the type and behaviour of the actors and function. The authors of [185] claim that adopting general purpose modelling languages such as UML are not always suitable to model problems in specific domains, hence the development of a DSML could be considered as the alternative.



Figure 5.5: UML use case diagram actor and use case graphical notations

The complexity of virtual healthcare provision suggests that there is always the opportunity for a wider range of stakeholders to participate in providing a care as healthcare virtual collaboration can take place on a global level. During a typical healthcare service provision, many stakeholders come together to provide the service, some healthcare professionals and others outside the healthcare sector. Researches such as [13] has used UML to model the execution procedure of treatment provision to patients and they have drawn the conclusion that UML diagrams "cannot be used directly in the hospital domain". The author of this thesis has researched the suitability of using UML to model healthcare in [229] and the outcome points towards a similar conclusion that UML diagrams cannot model healthcare scenarios effectively.

Therefore, a healthcare virtual collaboration DSML becomes an important tool for modelling in the sector. In developing the DSML, UML use case diagram is extended using UML profile extension mechanism. The extension is necessary to ensure that the model notations represent specific actors and use cases that are visually separable from each other.

5.3.1 UML Use Case Notation Extension

Domain specific graphical languages can simplify complex problems by allowing for abstractions [223]; building these languages are often complex and require collective effort and time, hence finding the right abstraction for a specific domain is a challenge. For this reason, customising and extending an available general modelling language such as UML can be more affordable as well as more acceptable since it is widely used.

Profiling as a lightweight extension mechanism is being promoted by UML which has played a great role in the widespread use of the language [230]. The mechanism facilitates the introduction of new modelling elements without the need to define a totally new modelling language [230]. [231] Explains how the UML profiling mechanism can be used to develop a new DSML. The mechanism is used in a number of other studies, for examples, [223] proposes the use of UML profile for developing a DSML for embedded systems, and the authors in [129] use the mechanism to define a Service Modelling Language. Authors in [232] propose a domain ontology that can aid the evaluation process of DSML developed using UML profile. Figure 5.6 shows the organiser actor which is part of the HC-VBE-A package extended using UML profiling mechanism form the use case actor profile.



Figure 5.6: UML profile for organiser actor extended from the UML actor notation in this thesis

The UML use case diagram extensions performed to create the healthcare virtual collaboration DSML in this thesis consist of four packages. The packages are specifically developed to model the components of the proposed service and participant classification mechanism presented in Section 5.2. The packages are called healthcare virtual collaboration actor (HC-VBE-A), healthcare virtual collaboration use case (HC-VBE-U), healthcare virtual collaboration relationship (HC-VBE-R) and Healthcare virtual collaboration environment (HC-VBE-E). In HC-VBE-A package, the traditional actor notation is customised using the profile mechanism and four new notations are introduced representing virtual collaboration actors in healthcare as specified in the SPCM which are *Requester, Provider, Organiser and Support.*

Figure 5.7 shows the mapping between the extended use case notations to the service and participant classification mechanism outlined in Section 5.2 and the description of these extensions are presented in Table 5.1.



Figure 5.7: The service and participants classification mechanism mapping to the developed DSML for healthcare virtual collaboration

In extending the UML actor notation, different shapes have been used to modify the default UML stickman notation's head which is circle by default, with the aim to sperate the four new actors visually and aid visual recognition by modellers. The inspiration for selecting the-shapes has come from the common traffic signs that are recognisable easily.

For the healthcare "provider" actor, the stickman notation's head is a circle shape with a cross in the middle which is the sign for healthcare related ensembles such as hospitals. For the healthcare "requester" actor, the stickman notation's head is a triangle shape, indicating (give ways to) which can be interpreted as (need help). For the healthcare "organiser" actor, the stickman notation's head is a Dimond shape, the shape is used in traffic concept for warning and directing drivers i.e. organising their driving. For the "support" actor, the stickman notation's head is a trapezoid shape, representing electronic devices.

In HC-VBE-U package, the use case oval shape notation is customised and two extensions are introduced representing *Tasks* and *Services* of HC-VBE and HC-VO which are specified in the SPCM. Both the "task" and "service" shaped are visually separated by drawing horizontal and vertical lines across the original oval shape. In HC-VBE-R package, the association line is customised and two new association types have been introduced representing electronic and semi-electronic communications. The two new association line representing electronic and an added square bracket representing semi-electronic communication.

Semi-electronic means a communication that involves both direct face to face and virtual at the same time. For example, a doctor could be providing a service to a patient in his office directly while collaborating with another doctor virtually about the case of the patient. The final package is the system boundary notation customised to represent the HC-VBE-E where collaboration take place.

Packages	No.	Graphical Notation	Description
	2	< <organiser>> Name</organiser>	The notation depicts <i>Organiser</i> who participate in the creation and management of HC-VBE. This notation always appears on the top of a HC-VBE model. The notation depicts the healthcare providers who join a HC-VBE to provide a specific virtual healthcare service for a specific duration. This notation always appears on the right of a HC-
HC-VBE-A	3	< <provider>>Name</provider>	VBE model. The notation depicts virtual healthcare service
		< <requester>>Name</requester>	to ask HC-VBE for a service which subsequently be provided by creating a HC-VO for the request. This notation always appears on the left of a HC-VBE model.
	4	<support>>Name</support>	The notation depicts HC-VBE and HC-VO electronic resources such as databases that provide operational and management supports. This notation always appears on the bottom of a HC-VBE model.
HC-VBE-U	1	<service>> name</service>	The notation depicts HC-VOs that come to existence based on a request made by a requester. The service exists for a specific duration and it is created to achieve a specific goal. It gets dissolved once the goal is achieved. This notation always appears inside a HC-VBE model.

Table 5.1: The UML	use case	notation	extension	description	that m	nake up	the o	developed	DSML

Packages	No.	Graphical Notation	Description
	2	< <task>> name</task>	The notation depicts HC-VBE tasks where the task is created for management of a HC-VBE. Tasks are long- term based and may continue for the duration of the HC-VBE life time. This notation always appears inside a HC-VBE model.
HC-VBE-R	1		This notation is used to indicate electronic collaboration using communication technologies between two entities of a HC-VBE. This notation always appears crossing a HC-VBE model between Service or Task and Requester, or Provider or Support or Organiser.
	2		This notation is used to indicate a mix of direct and electronic collaborations between two entities of a HC-VBE. This notation always appears crossing a HC-VBE model between Service or Task and Requester, or Provider or Support or Organiser.
HC-VBE-E	1	< <environment>> name</environment>	This notation represents the HC-VBE where HC- VO is created and live to provide a service.

5.3.2 A Modelling Example

To demonstrate how the DSML developed facilitate the process of modelling a healthcare virtual collaboration, the following scenario is considered.

To reduce cost and provide on demand consultation for patients who require after treatment monitoring, the local healthcare authority in Leicester decides to setup a HC-VBE-Leicester and appoints following participants:

- 1. A clinical operation manager as Organiser
- 2. An accountant as Organiser
- 3. The local authority central patient record system as Support

The participants collaborate in a *Task* to manage the operation of the HC-VBE-Leicester to ensure the smooth running of the virtual environment and the availability of resources for HC-VO creations. Figure 5.8 shows the structure of the HC-VBE-Leicester modelled using the DSML described in Section 5.3.



Figure 5.8: Use case Diagram for the HC-VBE-Leicester collaboration scenario modlled uing the developed DSML

5.3.2.1 Using HC-VBE-Leicester to provide care

To model a healthcare service provision by the HC-VBE-Leicester, a case study has been extracted and simplified form [218] as follows:

A decision has been taken to discharge a male patient named John who had left hip total joint arthroplasty in the past 15 days. The progress of the patient will be monitored at his

home. The patient's movement is limited because of his operation and he has to be assisted in his demotics needs as he lives alone.

Surgical Site Infection is one of the most common possibilities after surgery, if the patient is not cared for properly. the infection is caused by "contamination of an incision with microorganisms from the patient's own body during surgery" [233]. To prevent the patient from being infected collaborations between the patient and healthcare professionals in the hospital is crucial. For such a case the HC-VBE-Leicester is asked by Mr. Johan as directed by his surgeon to provide a real-time consultation service which they call *SSI-Prevention* that facilitates:

- 1. The collaboration between the patient and healthcare providers.
- 2. Allow his surgeon to access real-time information recorded by a monitoring health device installed at his home.
- 3. Select and recruit a pharmacy to provide him with the right medication when needed.
- 4. Select and recruit a carer to look after him in terms of cooking and cleaning.
- 5. Find and recruit a nurse to treat his surgical wounds.

To model the scenario, the participants of the "SSI-Prevention" can be classified based on the SPCM developed as follows:

- 1. **Organiser**: The Surgeon who performed the operation.
- 2. **Providers**: A nurse to treat Mr. John's wounds, a pharmacy to provide medication when needed and a carer to cook and clean.
- 3. **Support**: The local authority's central database and a health monitoring device installed at the patient's home.
- 4. **Requester**: Mr. John is the service requester
- 5. **Service**: *SSI-Prevention*.

Figure 5.9 shows the virtual collaboration model for the scenario modelled using the DSML developed.



Figure 5.9: Use case diagram for the HC-VO collaboration (SSI-Prevention) modelled using the developed DSML

5.4 Service Orchestration Description Language

To show the processing steps of online services, the term "orchestration" is a familiar word among researchers. The aim of orchestration is simply to show the workflow details of an online activity [234]. The other term that is also used to describe processing steps of online services is "choreography" and the difference is, orchestration is known to represent centrally controlled collaboration whereas in choreography the service providers are loosely controlled [235]. It was decided that orchestration is more suitable for healthcare virtual collaboration and the reason is, in order to ensure all collaborating

parties, deliver the right services following enforced terms and conditions, a centrally controlled mechanism is more appropriate.

To model service orchestration there are a number of approaches suggested, for instance, the authors in [236] are proposing to use UML activity diagram. Their approach has great usability potentials as the diagram is frequently used among business service modellers and software developers. However, it suffers from lack of details as the diagram only shows activities without the required data that is needed for each step of the orchestration. The Authors of [234] propose the use of "Reo" which is a graphical modelling language to model orchestration. Their proposed mechanism only shows the connection and data flows between web services with the aim to generate technical code. Their approach lacks details such as the condition under which the orchestration takes place.

Business Process Execution Language (BPEL) [237], is one of the most popular orchestration workflow languages that is becoming industry standard and supported by tech companies such as Microsoft and IBM. BPEL provides the mechanism to implement orchestration of web services. In BPEL, the orchestration between service providers over the web is encoded in XML, and provides a mechanism to send and receive XML messages sequentially and in parallel [237]. Initially, BPEL didn't have the support for human activities, but later, the language was extended and BPEL4People was introduced [238]. The new extension is essential for the purpose of this research since in healthcare virtual collaboration human is involved as service providers alongside machines.

In BPEL, first a service requester sends an input message to be processed and once the request message is received service attributes will be identified as variables and necessary service provision modules are assigned to be invoked. The result of service provision will be sent back to the service requester in the final stage. Figure 5.10 shows the steps BPEL uses to orchestrate services. To address the gaps identified in this section, a mechanism specific to healthcare virtual collaboration service orchestrations is developed and presented in the next section which consists of an orchestration mechanism (HC-VBE-OM) and a service orchestration description language (HC-VBE-SODL).



Figure 5.10: BPEL orchestration process. showing how requests are orchestrated

5.4.1 HC-VBE-OM Introduction

Here, an orchestration mechanism is developed that captures three aspects of healthcare virtual collaboration service orchestration which are, the scope of the orchestration, the required information for the orchestration to take place and the actual orchestration process. This is different from BPEL as BPEL only shows the orchestration process without showing the scope and required information. Each aspect of the mechanism captures a number of details of an orchestration. Figure 5.11 shows the three aspects of the mechanism and Table 5.2 provides a short description of each aspect of the HC-VBE-OM.



Figure 5.11: the proposed HC-VBE Orchestration mechanism components

No.	Orchestration	Reserved word	Explanation
	Aspect		
1	Scope	1. Master SLA	Represent the scope of the orchestration as
			each HC-VO will have a Master SLA under which
			all SLAs are linked. More details regarding
			Master SLA is provided in Chapter 6.
		2. SLA	In a typical HC-VO there are collaboration
			between participants, the terms and conditions
			of the collaboration is assumed to be formalised
			in an SLA. More details are provided regarding
			SLA in Chapter 6.
2	Orchestration	1. ExpectedService	Represent the expected service that is going to
	Attributes		be orchestrated. This service is identified in the
			SLA in the Scope section. There might be several

Table 5.2: Orchestration mechanism component description

No.	Orchestration	Reserved word	Explanation
	Aspect		
			expected services in an SLA; if that is the case
			then there is an orchestration for each.
		2. InitiatedBy	Indicates the participant who has requested the
			service as specified in the SLA in the Scope.
		3. ImplementedBy	Indicates the participant who provides the
			service as specified by the SLA identified in the
			Scope.
		4. UnderCondition	This section contains all the constraints under
			which the orchestration is going to be carried
			out. If any of the conditions are not true the
			orchestration does not take place.
3	Orchestration	1. Action	This indicates what activity is the orchestration
	Process		is for and starts the orchestration process once
			the request for the activity is received.
		2. Call	This step is triggered by (Action) and it will
			invoke the service provider to start providing
			the service.
		3. Answer	Represents the outcome of the orchestration.
			Penalty will be issued if the service provision
			has not met the conditions specified in the SLA
			otherwise reward will be issued.
		4. Influence	This part identifies the next orchestration that
			is going to be triggered after the completion of
			the current orchestration.

To facilitate the implantation of the developed orchestration mechanism in the future, it has been mapped to BPEL4People. Figure 5.12 shows the mapping between the HC-VBE-OM and BPEL4People.



Figure 5.12: HC-VBE Orchestration mechanism mapping to BPEL

To perform the mapping, a similar direction as in [239] was followed to map the developed orchestration framework to BPEL4People. In the mapping, the three main aspects of the orchestration mechanism and their sub sections are stated and mapped to their equivalent BPEL4People XML tags which are described in [237] and summarised in Table 5.3. The mapping is as follows:

- 1- The orchestration scope which is decided by a *Master SLA* and the *SLA* which the orchestration is for, can be implemented as <scope> tag in BPEL4People.
- 2- The process component of the orchestration mechanism consists of *ExpectedSerivce* which can be implemented as <variable> or <task> tags, *InitiatedBy* can be implemented as <partnerLinK> or <processInitiator>, *ImplementedBy* can be implemented as <partnerLinK> or <processOwner> and *underCondition* can be implemented as <received <condition>.
- 3- The third component of the orchestration mechanism consists of *Action* which can be implemented as <receive>, *Call* which can be implemented as <invoke>, *Answer* which

can be implemented as <reply> and *Influence* which can be implemented as <variable> or <task>.

Now the question is, if we can map the orchestration mechanism to BPEL4People why not use BPEL4People instead? The answer is, BPEL4People is an implementation language based on XML that require technical skills to be created and understood. The orchestration mechanism outlined here has a human understandable textual structure that does not require technical skills to create and understand and it is developed for modelling purposes.

No.	Тад	Description
1	<variable></variable>	Describes the attributes of a service or a requester or
		provider.
2	<task></task>	Describes a human task.
3	<partnerlink></partnerlink>	Describes the service initiator or owner.
4	<processinitiator></processinitiator>	Describes service initiator.
5	<potentialowner></potentialowner>	Describes service owner.
6	<receive <condition="">></receive>	It is used to specify the conditions of the service that is
		expected to be provided.
7	<receive></receive>	The tag is used to specify a service to be provided.
8	<invoke></invoke>	The tag is used to initiate a service to be provided by a
		partner.
9	<reply></reply>	The tag contains a response of an accepted service
		request described in <receive>.</receive>

Table 5.3:BPEL4People tag descriptions used in the mapping process

5.4.2 Orchestration Description Language

As a result of the research carried out on orchestration for this thesis, it was concluded that a graphical orchestration model cannot frame the required information in a given orchestration alone. Therefore, a textual description language to describe an orchestration, following the direction of others such as [240] was developed. Healthcare is complex and there are many details that have to be accounted for in order to start and end an orchestration. The textual orchestration description language presented in this section is designed to describe the three aspects of the orchestration mechanism as presented in the previous section (scope, attributes and process). To standardise the orchestration textual description language, here the sentence structure and reserved words of the language are outlined. Figure 5.13 shows the structure that each sentence will have. It consists of two parts, the first is called Section Identifier (SI) and the second is called Orchestration Line Statement (OLS). The second part consists of two sub-parts which are processing line requirements and the outcome of the line.

In the first section (1. Section Identifier), there is always one of the 9 reserved words defined in Table 5.4 in the order presented in the table. This part of the sentence is separated from the second part using a colon (:). The next part contains one or more statements, each separated by a dot (.) and then followed by relevant properties of the statement. The "2.1 requirement" sub-section is linked to the "outcome" sub-section using either an equal (=) sign to imply that the two sides of the statement should be equal in order for the next line of the orchestration to be triggered or a right arrow (\rightarrow) sign to imply that sub section new is the outcome of the sub section one. The "2.2 Outcome" sub-section has the same structure as sub Section 2.1. Each statement in section two (2. Orchestration line statement) ends with a semi-colon (;). In addition to the sentence structure, Tables 5.4 and 5.5 provide a number of logical operators that can be used in the textual orchestration description language.



Figure 5.13:Sentence structure of the textual orchestration description language developed

No.	Reserved Word	Purpose
1	Invoke	Request to perform an action
2	Send	Transfer content of an interaction
3	Receive	Arrival of a request from requester
4	Reply	Transfer the content of a completed interaction
5	Wait	Wait while required conditions are not met for an action
6	Complete	An interaction is completed
7	Cancelled	An interaction is cancelled before reaching the planed end
8	Trigger	Initiate an interaction
9	True	A condition is met
10	False	A condition is not met
11	Empty	A request has no content
12	Expire	A request is out of date
13	If	Conditional statement

Table 5.4: The developed Textual orchestration description language reserved words

Table 5.5:Orchestration description language logical operators

No.	Logical operator	Purpose
1	(OR)	Optional action
2	& (AND)	Parallel action
3	()	Statement container
4	→ (Causal effect)	Results in
5	All mathematical	
	symbols as it may be	
	needed can be used.	

5.4.3 Orchestration Description Example

To illustrate the applicability of the orchestration language for healthcare virtual collaboration the following simple case study is described using the language.

A local software development company, a private healthcare company and a finance company in the city of Leeds-UK come together to form a HC-VBE called (Leeds-VBE) to provide the following services:

- 1- Facilitate HC-VO formation based on service requests.
- Recruit participants (healthcare providers) for HC-VOs to provide the requested service.
- **3-** Create SLA between participants for the services they agree to provide and receive.
- 4- Monitor the implementation of the SLA in terms of rating services, fines and rewards.
- 5- Disband HC-VOs and terminate SLAs after accomplishing set goals.
- 6- Archive required data and information for future use.

To show how Leeds-VBE can serve healthcare service requesters we consider the following simple scenario:

Mr. David is a 40 years old male who recently had a car accident and had an operation on his back after being injured in the accident. He spent two weeks in hospital after his operation and then dismissed and sent home. Due to the operation his movements are restricted and require care at home. He has asked Leeds-VBE to find a carer to take care of his domestic needs and a nurse to treat his operation wounds.

Mr. David has requested that the individuals recruited should fulfil the requirements specified in Table 5.6. It is now the job of Leeds-VBE to process the request and find the right participants based on Mr. David's requirement specification. Once all participants are recruited Leeds-VBE will from a HC-VO called *TreatWound* to implement and monitor the collaboration.

Participant	Service	Working hours	Availability (time)	Qualification	Distance (miles) from his home	Cost (per hour)
Carer	Treat wound	7am – 3 pm	7am – 5pm	High school	15 miles	£12
Nurse	Domestic house work	3pm – 6pm	24 hours	Qualified Nurse	10 miles	£30

Table 5.6:Healthcare provider requirement specification requested by Mr. David

To demonstrate the application of the HC-VBE-SODL, the *TreatWound* orchestration which is provided by a Nurse is described using the orchestration description language in Table 5.7 and each line of the orchestration is explained in Table 5.8.7

Sections	Section	HC-VBE-Orchestration (TreatWound)
	Components	
Orchestration	MSLA, SLA	Master.ID = 2, SLA.ID = 5
Scope		
Orchestration	ExpectedService:	TreatWound.Service = Master.SLA.Service;
Attributes		
	InitiatedBy:	Patient.Requester = Master.SLA.Requestor;
	ImplementedBy:	Nurse.Provider = Master.SLA.Provider;
	UnderCondition:	TreatWound.Duration = (StartDate= 01/2019 &
		EndDate = 03/2019);
		& TreatWound.Payment = £30/hr;
		& TreatWound.OCL(Frequency)= 1 per Day;
	Action:	TreatWound.Service = TreatWound.Receive;

Table 5.7: (TreatWound) orchestration described using the Orchestration description language

Sections	Section	HC-VBE-Orchestration (TreatWound)
	Components	
Orchestration		& TreatWound.StartDate = True;
Process	Call:	$(TreatWound.Service = Ture) \rightarrow invoke.Nurse:$
	Answer:	(Nurse.Reply = Completed) \rightarrow TreatWound.Payment =
		True;
		If TreatWound.Duration = ((StartDate= 01/2019 &
		EndDate = 03/2019)= True); &
		((TreatWound.OCL(Frequency)= 1 per Day)=True);
		(Nurse.Reply = Cancelled) \rightarrow TreatWound.Payment
		= False;
		If TreatWound.Duration = ((StartDate= 01/2019 &
		EndDate = 03/2019)= False); &
		((TreatWound.OCL(Frequency)= 1 per Day)=False);
	Influence:	(TreatWound.Payment = True) \rightarrow SLA.Archive;
		(TreatWound.Payment = False) \rightarrow SLA.Archive;

Table 5.8: TreatWound Orchestration Explanation

	HC-VBE Orchestration (TreatWound)			
1. Scope:	Master.ID = 2, SLA.ID = 5			
1. Explanation:				
(Scope) represent the bou	indary in which the orchestration takes place. In this example the			
content of scope navigate	s to where exactly the orchestration information is obtained. In the			

content of scope navigates to where exactly the orchestration information is obtained. In the statement, the content is obtained from service level agreement with identification number 5 which is administered by a master service level agreement with identification number 2.

HC-VBE Orchestration (TreatWound)

2. ExpectedService:	TreatWound.Service = Master.SLA.Service;
---------------------	--

2. Explanation:

(ExpectedService) represent the service that is agreed on to be provided. The statement indicates that the expected service is (TreatWound) of collaboration type (Service). This service is equal to the (Service) agreed on in the identified SLA which is navigated to by the statement (Master.SLA.Service).

3. InitiatedBy:	Patient.Requester = Master.SLA.Requester;
	· · · · · · · · · · · · · · · · · · ·

3. Explanation:

(InitiatedBy) represent the requester (human or device) that has requested the service agreed on in the SLA. The statement indicates that a (Patient) of participant type (Requester) has requested the service. For this orchestration to take place the requester is the same as the one declared in the SLA.

3. ImplementedBy:	Nurse.Provider = Master.SLA.Provider;
5. Implemented by.	Nulse. Tovider – Master.sLA. Tovider,

3.Explanation:

(ImplementedBy) represent the provider (human or device) that is going to take the responsibility to provide the service. In this example a (Nurse) of participant type (Provider) is going to provide the (TreatWound) service.

4. UnderCondition:	TreatWound.Duration = (StartDate= 01/2017 & EndDate = 03/2017);
	& TreatWound.Payment = £30/hr;
	& TreatWound.OCL(Frequency)= 1 per Day;

4. Explanation:

(UnderCondition) represent the conditions under which the service is implemented. The success of the orchestration process will be validated against the stated conditions. These are the conditions under which the two participants (Patient) and (Nurse) have agreed to collaborate. In

HC-VBE Orchestration (TreatWound)

this example the conditions are that the duration of the service is 2 months during which a visit is paid daily for which the nurse will be paid £30 per hour.

5. Action:	TreatWound.Service = TreatWound.Receive;
	& TreatWound.StartDate = True;

5. Explanation:

(Action) represent the activity that is going to take place in the orchestration. In this example the action is the (TreatWound) service. The action will start when it is received by the (process) part of the orchestration and the start date is valid.

6. Call:	(TreatWound.Service = Ture) \rightarrow invoke.Nurse;

6. Explanation:

(Call) represent the start of the process. In this example when the action condition is true the participant of type (Nurse) is requested to start the action (TreatWound service in this case).

7. Answer:	(Nurse.Reply = Completed) \rightarrow TreatWound.Payment = True;
	If TreatWound.Duration = ((StartDate= 01/2017 & EndDate =
	03/2017)= True); & ((TreatWound.OCL(Frequency)= 1 per
	Day)=True);
	$ (Nurse.Reply = Cancelled) \rightarrow$ TreatWound.Payment = False;
	If TreatWound.Duration = ((StartDate= 01/2017 & EndDate =
	03/2017)= False); & ((TreatWound.OCL(Frequency)= 1 per
	Day)=False);

7. Explanation:

(Answer) represent the result of the action. In this case if the Nurse returned the (Completed) message with all conditions observed, payment will be made otherwise payment will not be made.

HC-VBE Orchestration (TreatWound)				
8. Influence:	(TreatWound.Payment = True) \rightarrow SLA.Archive;			
	(TreatWound.Payment = False) \rightarrow SLA.Archive;			
8. Explanation:				
(Influence) represent the	next orchestration that will be triggered to start. In this case the			
orchestration for the (Arch	ive) service will be triggered.			

5.5 HC-VBE-M-F Evaluation Results

Framework implementation, testing methods, data collection and analysis process are already described in Section 4.3. In this section, system developers' acceptance evaluation results for the HC-VBE-M-F are presented and discussed. The results are produced using data collected during the survey, modelled and analysed using Structural Equation Modelling technique in AMOS software. The SEM model developed based on the HC-VBE-TAM-Provider is pictured in Figure 4.15 and described in Section 4.3.7.1.

5.5.1 Overall Analysis Results

This section provides the results for Mean, Standard Deviation, Factor loading, and Significance *p* value obtained as a result of analysing the collected data using the questionnaire designed based on the HC-VBE-M-F-TAM. The questionnaire used can be found in Appendix D. "Mean" is the average of all values given to an observed item by surveyed participants. The Mean results for the observed items are between 4.71 and 5.25 which indicate agreeability of participants with the measuring statements in the questionnaire. The result is significant and conveys that the concepts developed in the HC-VBE-M-F and demonstrated through the developed Java application prototype is acceptable by system developers.

Standard deviation measures the deviation of answers from the mean and the smaller the better. The standard deviation values are between 1.101 and 1.693 which indicates that

the answers given by research participants were close to each other since the standard deviation values are not too large. This result shows that research participants view about the modelling tool were fairly consistent. This finding increases the credibility of the results and provides a good acceptability support for the modelling framework. The result of individual item factor loading indicates strong link between observed items and their unobserved variable with values between 0.768 and 0.942 which are all above the acceptable cut-off value 0.5 [241]. The factor loading results suggests the effectiveness of individual measurement items in measuring the unobserved acceptance factors in the extended TAM which subsequently increases measurement reliabilities.

To measure the significance of relations between the unobserved variables and their observed items, significance p value is computed for each construct. Significance test is necessary in SEM to ensure that the result is not produced by chance and it is widely accepted that a value is considered significant when (p<0.05) [205]. Significance computed values are indicated as *** in AMOS analysis results output which indicates that p<0.001. The result of all items is significant which suggests a high probability that the results are not produced by chance. Table 5.9 summarises the overall data analysis results for the modelling framework acceptability evaluation. The table shows statistical results for each questionnaire item within the context of their unobserved variables.

Unobserved	Observed Variable and its		Std.	Factor	<i>p</i> < 0.05
Variable	related questionnaire	Mean	Deviation	Loading	(Significance)
	item				
Perceived	PAM1	4.79	1.500	0.925	<i>p</i> <0.001
Ability to	Models produced using the HC-VBE modelling package				
Model (PAM)	represents virtual collaboration in healthcare.				
	PAM2	4.82	1.541	0.837	<i>p</i> <0.001
	Models produced using the HC-VBE modelling package				
	simplifies complex heal	thcare	virtual co	llaboration	
	scenarios.				

Table 5.9: Overall data analysis results for HC-VBE-TAM-Modelling framework which includes the Mean, Std., Factor loading and Significance value

Unobserved	Observed Variable and its		Std.	Factor	<i>p</i> < 0.05
Variable	related questionnaire	Mean	Deviation	Loading	(Significance)
	item				
	PAM3	4.96	1.598	0.917	<i>p</i> <0.001
	Models produced using the	ne HC-V	BE modellin	g package	
	represent roles of virtual healthcare stakeholders.				
	PAM4	4.71	1.357	0.867	<i>p</i> <0.001
	Models produced using the	ne HC-V	BE modellin	g package	
	describes the collaborati	on pro	cess betwee	en virtual	
	healthcare stakeholders.				
Perceived	PU1	4.96	1.374	0.879	<i>p</i> <0.001
Usefulness	Using the HC-VBE mode	elling p	ackage imp	roves my	
(PU)	healthcare virtual system de	evelopme	ent process.		
	PU2	5.25	1.404	0.942	<i>p</i> <0.001
	Using the HC-VBE modellin	g packa	ge saves me	effort and	
	time to understand a he	ealthcare	e virtual col	llaboration	
	scenario.				
	PU3	4.86	1.297	0.812	<i>p</i> <0.001
	Using the HC-VBE model	ling pac	kage would	be more	
	effective to use for	modellin	ig healthcar	e virtual	
	collaboration systems than	other mo	odelling langu	ages.	
	2114	4.00	4.070	0.046	
		4.82	1.278	0.846	<i>p</i> <0.001
	Overall, I find the HC-VBE modelling package useful for				
	modelling healthcare virtual collaboration systems.				
Dorcoived	DEL11	4.06	1 /79	0.002	n<0.001
	Loarning to use HC MPE med	4.90		0.902	μ<0.001
(DEII)		ening pa	ukage would	ue easy for	
	IIIE.				

Unobserved	Observed Variable and its		Std.	Factor	<i>p</i> < 0.05
Variable	related questionnaire	Mean	Deviation	Loading	(Significance)
	item				
	PEU2	4.79	1.101	0.858	<i>p</i> <0.001
	I would find it easy to model	healthc	are virtual co	llaboration	
	systems using the HC-VBE modelling package.				
	PEU3	5.07	1.386	0.909	<i>p</i> <0.001
	It would be easy for me to	become	skilful at usi	ng the HC-	
	VBE modelling package.				
	PEU4	5.07	1.585	0.942	<i>p</i> <0.001
	I would find the HC-VBE mo	delling p	ackage easy t	o use.	
Attitude	AU1	5.14	1.693	0.950	<i>p</i> <0.001
towards using	Using the HC-VBE modelling	package	e would be a g	good idea.	
(AU)					
	AU2	4.89	1.257	0.895	<i>p</i> <0.001
	Using the HC-VBE modell	ing pac	kage makes	modelling	-
	healthcare systems more in	teresting	ξ.		
	AU3	4.93	1.609	0.935	<i>p</i> <0.001
	Using the HC-VBE modellin	g packa	ge would be	a pleasant	
	experience.				
	AU4	4.82	1.219	0.871	<i>p</i> <0.001
	I would like to use the HC-VBE modelling package.				
Intention to	IU1	4.86	0.970	0.848	<i>p</i> <0.001
use (IU)	I intend to use the HC-VBE modelling package.				
	IU2	4.82	1.090	0.768	<i>p</i> <0.001
	It is likely that I will use the HC-VBE modelling package.				-

Unobserved	Observed Variable and its		Std.	Factor	<i>p</i> < 0.05
Variable	related questionnaire	Mean	Deviation	Loading	(Significance)
	item				
	IU3	4.82	0.983	0.888	<i>p</i> <0.001
	I expect to use the HC-VBE r				
	IU4	5.00	1.361	0.866	<i>p</i> <0.001
	I am willing to recommend other people to use the HC-VBE modelling package.				

5.5.2 Intermeasurement Correlation

Intermeasurement correlation shows the link between all observed items, Table 5.10 shows the analysis result of intermeasurement correlations for the questionnaire items examining the HC-VBE-M-F-TAM. There are 20 measurement items in the questionnaire. Each questionnaire item is correlated to all other questionnaire items in a questionnaire including itself. A perfect correlation value which is 1, is produced when a questionnaire item is correlated with itself perfectly, but the values would be less than 1 in all other cases.

The intermeasurement results for all questionnaire item in this case are between 0.4 and 0.9 which are above the 0.3 acceptable cut off point. The results prove that the scores given to all items as answers are closely related. This means that research participants shared similar opinions on measurement items and they were all positive. Intermeasurement values provide a detail map of the scores given to individual items in a questionnaire. If the majority of correlation values are above 0.3 that indicates that the questionnaire items have adequately captured users view of the technology being tested in this context.

	PAM1	PAM2	PAM3	PAM4	PU1	PU2	PU3	PU4	PEU1	PEU2	PEU3	PEU4	AU1	AU2	AU3	AU4	IU1	IU2	IU3	IU4
PAM1	1.00																			
PAM2	0.78	1.00																		
PAM3	0.85	0.75	1.00																	
PAM4	0.84	0.68	0.78	1.00																
PU1	0.79	0.80	0.74	0.73	1.00															
PU2	0.84	0.72	0.90	0.78	0.83	1.00														
PU3	0.69	0.71	0.64	0.69	0.81	0.75	1.00													
PU4	0.77	0.77	0.74	0.78	0.76	0.77	0.72	1.00												
PEU1	0.72	0.78	0.80	0.64	0.71	0.83	0.75	0.76	1.00											
PEU2	0.82	0.68	0.82	0.73	0.68	0.83	0.63	0.74	0.79	1.00										
PEU3	0.77	0.73	0.79	0.74	0.64	0.71	0.67	0.68	0.82	0.69	1.00									
PEU4	0.75	0.73	0.76	0.72	0.70	0.79	0.71	0.70	0.84	0.79	0.91	1.00								
AU1	0.87	0.78	0.85	0.81	0.78	0.86	0.72	0.75	0.77	0.83	0.78	0.81	1.00							
AU2	0.79	0.66	0.74	0.74	0.77	0.88	0.72	0.77	0.76	0.81	0.64	0.78	0.86	1.00						
AU3	0.76	0.71	0.82	0.80	0.74	0.84	0.76	0.77	0.81	0.77	0.78	0.83	0.87	0.84	1.00					
AU4	0.79	0.59	0.80	0.68	0.70	0.83	0.62	0.60	0.67	0.83	0.62	0.70	0.87	0.78	0.79	1.00				
IU1	0.74	0.68	0.78	0.70	0.61	0.71	0.57	0.73	0.75	0.77	0.67	0.68	0.80	0.72	0.80	0.73	1.00			
IU2	0.63	0.60	0.65	0.59	0.56	0.61	0.40	0.53	0.59	0.65	0.52	0.56	0.68	0.61	0.71	0.70	0.67	1.00		
IU3	0.75	0.71	0.80	0.63	0.71	0.78	0.62	0.65	0.79	0.75	0.69	0.70	0.82	0.73	0.83	0.81	0.75	0.73	1.00	
IU4	0.76	0.78	0.80	0.70	0.75	0.78	0.69	0.75	0.85	0.67	0.79	0.77	0.79	0.76	0.85	0.65	0.70	0.70	0.75	1.00

Table 5.10: HC-VBE-TAM-Modelling framework intermeasurement correlations

5.5.3 Characteristics of Constructs

At construct level, the result supports strong acceptability for the concepts developed in the HC-VBE-M-F. The mean value for each construct variable computed as an average for all observed item answers for a particular construct is close to 5 with standard deviations between 0.978 and 1.371. Table 5.11 shows the result for all constructs tested, the number of observed items per construct, means and standard deviation computed for the 28 samples. The significance of means and standard deviations results on the acceptability evaluation has already been explained in Section 4.3.8. The result suggests that on construct level the survey participants' acceptance attitude towards the modelling framework were positive since the values are close to 5, which equals to "Agree" in Likert-Style measures. The result also suggests that on construct level the answers were similar since the standard deviation values are not too large.

Construct statistics								
Constructs	Observed	Mean	Std.	N				
	Items		Deviation					
Perceived Ability to Model (PAM)	4	4.821	1.371	28				
Perceived Usefulness (PU)	4	4.973	1.220	28				
Perceived Ease of Use (PEU)	4	4.973	1.288	28				
Attitude Towards Using (AU)	4	4.946	1.356	28				
Intention to Use (IU)	4	4.875	0.978	28				

Table 5.11: Statistical results for HC-VBE-M-F-TAM co	onstruct characteristics
---	--------------------------

5.5.4 Data Validity and Reliability

In survey research, one indicator of measurement validity and reliability is internal consistency reliability of items in a questionnaire; valid and reliable measurement should produce high consistency of answers between items measuring the same construct. To ensure measurement validity and reliability, internal consistency reliability is computed for the collected data based on three most used statistical validity and reliability calculation methods which are Cronbach's Alpha [210], Construct Reliability (*CR*) and Average Variance Extracted (*AVE*) [208].

The acceptable value for *AVE* is 0.5 or above and the value computed for all constructs in this part of the study are above the cut off value ranging from 0.712 to 0.834. These results suggest that the amount of variance produced as a result of correct data collected for each construct is much higher than the amount of variance which may have been produced as a result of errors in the data collected. The suggestion supports strong data reliability which is vital for validating the technology acceptance claims made with regard to the modelling framework developed in this thesis.

CR acceptable value is 0.7 or above, the result for all constructs are above the value starting from 0.908. The acceptable value for Cronbach's Alpha is 0.7 again the result for all constructs are above the value starting at 0.902 and the overall test value is 0.982. These results suggest strong data validity and reliability which supports the conclusions reached for the *AVE* results. Kaiser-Meyer-Olkin Measure has been computed to ensure sampling adequacy and the result is 0.885 which indicates that the sample used in the study is adequate as values above 8 is considered acceptable [242]. The adequacy of the sample data is important to be proven statistically which in turn increases the findings made as result of analysing the sample data.

The Individual construct factor loading indicate strong connections between the construct and its measured questionnaire item as all values are above 0.9 which is much higher than the 0.5 acceptable value. The correlations between the constructs in the model as values start with the correlation between IU and PU being the lowest (0.814) which is much higher than the 0.3 acceptable value. The strong correlation between the constructs supports the validity of the model which in turn support the empirical conclusions made with regard to the acceptability of the modelling framework. Table 5.12 shows the summary of results obtained for collected data reliability and validity.

Constructs	AVE >0.5	CR>0.7	Cronbach's Alpha> <i>0.7</i>	Factor Loading	PAM	Da	PEU	AU	D	Whole Cronbach's Alpha> <i>0.7</i>	Kaiser-Meyer-Olkin Measure of Sampling Adequacy
Perceived Ability to Model (PAM)	0.787	0.937	0.933	0.949	1						
Perceived Usefulness (PU)	0.759	0.926	0.931	0.945	0.908	1					
Perceived Ease of Use (PEU)	0.816	0.947	0.940	0.931	0.882	0.853	1			0.982	0.885
Attitude Towards Using (AU)	0.834	0.953	0.947	0.956	0.895	0.889	0.879	1			
Intention to Use (IU)	0.712	0.908	0.902	0.945	0.876	0.814	0.852	0.902	1		

Table 5.12: HC-VBE-M-F-TAM data reliability and validity results

5.5.5 Model Fit Indices

Model fit indices are calculated to find out whether the model used fits the data and it is acceptable. As explained in Section 4.3.8.10 that many fit indices have been suggested by researchers but the most recommend ones are Chi Square (CMIN in AMOS), CFI, TLI, IFI, RMSEA and SRMR [214][215][213]. Based on the recommendations, Table 5.13 summarises the model fit indices computed in AMOS for the HC-VBE-M-F-TAM.

The computed Chi Square value for the model is 1.318 which is within the recommend value of 1-3. Chi Square is a measure of badness of fit, a value that is not significant

(*p*>0.05) indicates that the model is acceptable. This means that the covariance matrix calculated based on the collected data is similar to the predicted covariance matrix by the model. In this case, the Chi Square value suggest that the model designed to test the acceptability of the framework is acceptable and is fit for the intended purpose. CFI compares the fit value of the model with a fit value of a base model (alternative model) with value closer to 1 indicates an acceptable fit. The CFI value computed is 0.925 which has exceeded the recommended 0.90 cut of value and hence indicates a good fit. These results suggest that the model used to test the acceptability of the framework is closer to ideal, hence indicate the fitness of the model.

TLI and IFI are both comparative fit indices similar to CFI with computed values of 0.913 and 0.928 respectively. They both support the claim that the model fit is acceptable since the values are above the 0.9 cut off value. RMSEA calculates the difference between covariance matrix value for the same observed item in both the tested and predicted model. The recommended value is 0.08 or smaller, the value computed for the HC-VBE-M-F-TAM is 0.108 which is slightly bigger (0.0208 difference), however since all other fit measures are within the acceptable cut off values, this can be overlooked and be considered as acceptable.

SRMR is the final fit value computed which measures the standardised difference between predicted and observed covariance matrix with a recommended value of 0.08 or smaller. The computed SRMR value for the HC-VBE-M-F-TAM is 0.048 which is smaller than 0.08 and suggests a good model fit. The model fit values collectively support the fitness of the model to test the acceptability of the framework by prospective users and the results produced from it can be relied on.

All the results so for far presented were computed to show whether the results produced in testing the hypothesises are statistically sound or not. Collectively the results suggest the hypothesis results are statistically sound. The results of the hypotheses testing are presented in the next section. Table 5.13: HC-VBE-M-F-TAM fit results

Overall fit	Description	Computed	Recommended Value		
index		Value			
CMIN/DF	Chi Square/ Degree of Freedom	1.318	Between 1 and 3		
CFI	Comparative fit index (baseline comparison)	0.925	CFI ≥ 0.90		
TLI	Tucker Lewis Index (baseline comparison)	0.913	TLI ≥ 0.90		
IFI	Incremental fit index (base line compression)	0.928	IFI ≥ 0.90		
RMSEA	Root Mean Square Error of Approximation (Absolute fit indices)	0.108	RMSEA ≤ 0.08		
SRMR	(Standardized) Root Mean Square Residual	0.048	SRMR ≤ 0.08		

5.6 HC-VBE-M-F Discussions

This section discusses the results obtained for this part of the evaluation with regard to the two research questions stated in Chapter 1. The questions focus on the modelling aspect of healthcare virtual collaboration and the extent to which the research questions have been answered is examined.

RQ1: How to classify the main stakeholders in virtual collaboration for healthcare?

To be able to model healthcare virtual collaboration the first step is to develop a generic classification model for the roles and services in the sector. With this in mind and to

answer the research question, an in-depth literature search was conducted to understand healthcare in terms of stakeholders and services. The challenge was to deduce a generic classification mechanism that is applicable to all possible scenarios of healthcare virtual collaboration. Healthcare is a broad and complex sector which is quite difficult to carry out a holistic analysis for in terms of roles and services. To guide the research, VBE and VO concepts are used as a theoretical base and as a result, a generic virtual collaboration role and service classification mechanism which is called Service and Participant Classification Mechanism (SPCM) is developed and described in Section 5.2. The mechanism classifies services and participants for a given virtual collaboration scenario at an abstract level. The mechanism is simple and can be applied to any healthcare virtual collaboration scenario to classify the roles and service. The applicability of the mechanism is demonstrated with a simple healthcare virtual collaboration example in Section 5.2.1. The evaluation results show that the mechanism is seen by system developers as acceptable for classifying roles and service in healthcare. This result is significant and important since the rest of the concepts developed in this thesis are based on the SPCM.

RQ2: How to model and describe service provision for healthcare virtual collaboration?

The research outcome obtained in the process of answering RQ1 served the based to answer RQ2, which led to developing a DSML for healthcare virtual collaboration. In Section 5.3 a modelling language which is called HC-VBE-DSML is developed and described. The language enables healthcare virtual collaboration system developers to model a scenario in terms of roles and services. The modelling language serves as a generic tool that provides graphical notations specific to certain roles and services as well as a collaboration orchestration mechanism for healthcare, based on VBE and VO concepts. The graphical notations are packaged in a DSML which is developed based on UML use case diagram. The DSML is developed with the aim to be simple, easy to learn and comprehensive in representation. With regard to the service description part of RQ2 a Service Orchestration Description language is developed. The language is text-based which can be used by system developers to describe service provision in a given healthcare virtual collaboration scenario. The orchestration is mapped to BPEL4People for implementation purposes in the future. To evaluate the modelling approach for healthcare virtual collaboration the modelling language was implemented as a Java application and the application was evaluated for user acceptance using TAM as a theoretical evaluation framework. TAM is adapted through extending it for this part of the study, a detail description for the extension process is provided in Section 4.3.2. Based on the HC-VBE-M-F-TAM seven hypotheses were defined in Section 4.3.2.1. The result of testing those hypotheses based on the data analysis carried out in AMOS show that H₁ with a significance value of less than 0.05 is accepted; which means PAM has a significant effect on PU. The result provides empirical proof that research participants believe that the modelling framework is capable of modelling virtual collaboration in healthcare domain (indicated by *p*-value for PAM construct in Table 5.14) and this will influence its usefulness. The *p*-value for the causal effect of PAM on PEU hypothesised in H₂ is 0.001 which is less than 0.05. This means that H₂ is accepted and PAM has a positive effect on PEU. The interpretation of this result is that research participants believe PAM of the modelling framework will contribute to its PEU.

 H_{3a} which hypothesises the causal effect of PEU on PU is rejected with a *p*-value of 0.927. This leads to the conclusion that the research participants didn't believe that PEU will have a causal effect on PU. A possible reason for this finding could be the fact that the modelling framework is based on UML use case which is considered easy to use and useful, not one causing the other. This explanation could also be applied to H_{3b} which is rejected with a *p*-value of 0.116 which hypothesises the causal effect of PEU on AU. The result indicates that research participants didn't believe that PEU of the modelling framework changes their AU as they have already indicated their intention to use, shown by the result of IU construct with a mean value close to 5 (Table 5.3 in Section 5.5.3).

The causal effect of PU on AU hypothesised by H₄a is accepted with a *p*-value of 0.001 which indicates that since the research participants found the modelling framework useful, they also think of using it and the usefulness factors will influence their AU. This finding contradicts the result of H₄b which hypothesises the causal effect of PU on IU and it is rejected with a *p*-value of 0.646. According to this result PU will not lead to IU. Having said that, the evaluation results show that research participants intend to use the
modelling framework as the mean value of IU is close to 5 (Table 5.3 in Section 5.5.3) and H_5 which hypothesises the causal effect of AU on IU is accepted with a *p*-value of 0.015. Overall, these results show that the surveyed participants were positive about the contribution that the HC-VBE-M-F makes towards modelling the structural and aspect of VBE and VO based virtual collaboration in healthcare. Table 5.14 summarises the hypotheses test results and the bold lines in Figure 5.14 illustrates the accepted paths (hypotheses) in the HC-VBE-M-F-TAM.

				Sig.Level	
Hypos.	Evaluated Causal Effect			at <i>p</i> <0.05	Acceptance
H1	Perceived Usefulness (PU)	<	Perceived Ability to Model (PAM)	p<0.001	Accepted
H ₂	Perceived Ease of Use (PEU)	<	Perceived Ability to Model (PAM)	0.001	Accepted
H₃a	Perceived Usefulness (PU)	<	Perceived Ease of Use (PEU)	0.927	Not Accepted
H₃b	Attitude Towards Using (AU)	<	Perceived Ease of Use (PEU)	0.116	Not Accepted
H₄a	Attitude Towards Using (AU)	<	Perceived Usefulness (PU)	0.001	Accepted
H₄b	Intention to Use (IU)	<	Perceived Usefulness (PU)	0.646	Not Accepted
H5	Intention to Use (IU)	<	Attitude Towards Using (AU)	0.015	Accepted

Table 5.14: HC-VBE-M-F-TAM hypotheses testing results



Figure 5.14: HC-VBE-TAM-M-F accepted hypotheses

5.6.1 The HC-VBE-M-F Limitations

The HC-VBE-M-F is developed with the aim to be simple, generic and capable to model all VBE and VO based healthcare virtual collaboration scenarios. The structural aspect of the model which is based on UML use case diagram provides a set of simple notations to model roles and functions in healthcare virtual collaboration, however there are several important system feature that requires to be accommodated in a model such as system performance, availability, security which cannot be modelled using the developed modelling language. The process aspect of the model which is developed as an orchestration description language can describe possible steps in a given healthcare virtual collaboration service statically, however, it lacks a formal simulation or validation procedure to evaluate the correctness of processes described.

5.7 Chapter 5 Summary

This chapter presented the Healthcare Virtual Collaboration Modelling Framework (HC-VBE-M-F) and the concepts and mechanisms that make up the framework are explained in detail. The content of the chapter is summarised in the following points:

1- In order to understand the healthcare sector, a detail map of the roles and services of the sector is deduced from literatures available on major research data-bases such as

Springer and Elsevier. In total 6 classes of roles and services are identified which are care providers, executives, consumers, care environments, electronic devices and healthcare services. The identified roles and services are used as a base to develop a participant and service classification mechanism.

- 2- A Service and Participant Classification Mechanism (SPCM) is developed that classifies healthcare virtual collaboration participants and services. The mechanism uses the VBE and VO concepts as a guide for the classification. On VBE level, healthcare virtual collaboration participants and services are classified into *Organiser*, *Support* and *Task*. On VO level the participants and service are classified into *Provider*, *Requester* and *Service*. The mechanism is generic and can be used to classify participants and service of all VBE-based healthcare virtual collaboration.
- 3- A domain specific modelling language based on UML use case diagram graphical notations is developed to model the roles and services identified by the SPCM. To develop the modelling language, UML profiling technique is used to extend the notations. The actor notation (stickman notation) is extended four times with different head shapes to represent the four identified participants (*Organiser, Support, Provider* and *Requester*). The use case oval shape of UML is extended twice to represent *Task* and *Service*.
- 4- A Service Orchestration Mechanism (SOM) is developed to describe the processing flow of a healthcare virtual collaboration service. The mechanism includes an orchestration description language that uses structed sentences to describe the steps in a given service.
- 5- The evaluation results show that the framework is acceptable by system developers participated in the survey and their intension to use systems developed based on the modelling framework is positive.

In the next chapter, a healthcare virtual collaboration framework is presented which is developed based on the modelling framework presented in this chapter. This chapter presents a healthcare virtual collaboration framework that has been developed based on the modelling framework in Chapter 5. The framework is comprised of a conceptual description presented in Sections 6.1 and 6.2 with aim to manage and organise virtual collaborations for healthcare. A member selection mechanism is described in Section 6.3, a healthcare provider validation mechanism is described in Section 6.4 and a set of service level agreements (SLA) templates that formalises the agreements between collaborating participants are described in Section 6.5. The results and findings of the user acceptance evaluation for the framework which is implemented as a mobile application are presented and discussed in Section 6.7. Finally, a summary of the chapter is provided in Section 6.8.

6.1 Healthcare Virtual Collaboration Framework

The current centralised model of healthcare provision cannot deal with today's healthcare challenges such as the rise in demand and increased cost of care [9]. Moving the healthcare sector towards a multi-provider model where a number of individuals and organisations collaborate together to provide services are seen as the future of healthcare [14][15][16]. In this section, the author presents and describes a virtual collaboration framework for healthcare which is called Healthcare Virtual Breeding Environment Framework (HC-VBE-F) developed based on the role and service concepts developed in the modelling framework in Chapter 5. The aim of the framework is to facilitate and organise virtual collaboration and resource sharing for healthcare. Figure 6.1 shows the framework components which are a conceptual framework, a member selection mechanism, a healthcare provider validation and verification framework and a set of SLA templates.

This chapter aims to answer the remaining research questions as listed below:

- 1- How to manage and organise virtual collaboration for healthcare?
- 2- How to select, verify and validate participants for healthcare virtual collaboration.
- 3- How to regulate virtual collaboration for healthcare?



Figure 6.1: Components of the healthcare virtual collaboration framework which consist of a conceptual framework, a member selection mechanism, a provider validation and verification mechanism and SLA templates.

6.2 The HC-VBE Framework Description

The theoretical base of the framework is drawn from two well-researched virtual collaboration concepts which are VBE and VO [27] [31] [32]. The concepts originally developed for the world of business to address collaboration management and regulation challenges in virtual settings and recently they have been considered for application in

education, e-commerce and teleworking [46] [47]. In healthcare, the concepts have been researched for specific healthcare services, for example, VO concept is considered in a TeleCare pilot-study for collaboration and patient record sharing between stroke specialists in The Netherlands [243]. The researchers describe VO but they fall short in providing a working framework and they have not explained how the concept of VO is utilised in their approach. The authors of [244] explain a virtual hospital project in Finland in which VO is used as theoretical base for virtual collaboration in a hospital without presenting the workflow and management process for the collaboration.

As a solution to provide a type of mental healthcare which is not bound to physical structures in hospitals, the authors in [245] suggest the use of VO but similarly they don't provide any clarity as to how the concept is going to be used to provide the virtual care. To create a platform for radiologist to collaborate and share medical images and other resources a MammoGrid Virtual Organisation is suggested by [246]. They describe a number of functionalities of the VO on technical level without explaining the processing steps that the VO implements in organising the collaboration and resource sharing. These studies focus on a specific aspect of healthcare and none provide a comprehensive framework for healthcare collaboration and resource sharing in a virtual organisation setting.

There is clearly a gap in researches carried out on the use of VBE and VO concepts in healthcare as the studies cited are the only ones managed to be found and they are more than 13 years old. To contribute to the body of literature and fill in the gaps identified in the use of VBE and VO in healthcare, a comprehensive framework is presented in the next few sections. The framework is described on conceptual level, component level and process level as shown in Figure 6.2. On conceptual level, the figure shows the general architecture in terms of roles and services. On component level, the figure shows the main components that make up the framework which will be explained in detail in terms of component functionalities and groupings in later sections. Finally, on process level, the figure shows a flowchart like diagram which describes how the framework works to organise and facilitate virtual collaboration for healthcare.



Figure 6.2: Illustrating the HC-VBE-F description on conceptual, component and process levels

6.2.1 The HC-VBE Framework Concept

Healthcare has a dynamic nature in which the duration of care and the required resources are varied from one case to another [247]. The dynamicity of care provision is even more prominent in virtual healthcare since collaboration and resource sharing requires realtime synchronisations and managements. Therefore, a framework to facilitate the management and coordination aspects of healthcare virtual collaboration becomes essential. The framework developed here aims to facilitate the creation of virtual environments where healthcare professionals, health institutions and other organisations, supported by ICT infrastructures, can come together to provide healthcare services. The framework also facilitates the management of healthcare collaboration on both HC-VBE and HC-VO levels. A single HC-VBE can create many HC-VOs based on requests for healthcare services send to the HC-VBE by requesters. The HC-VBE facilitates parallel but sperate running of created HC-VOs. Electronic Service Level Agreement (SLA) serves the base for the terms and conditions under which each HC-VO is created. SLAs are agreed between requesters and providers prior to creation of the HC-VO for a particular request. In virtual environments, SLA is a widely accepted mechanism to regulate collaborations and resource sharing.

One of the advantages of employing SLA is guaranteeing a trustful relationship between collaborating parties. In virtual organization, members are recruited and discharged as per task requirements, therefore, it is possible that many of the members collaborate with each other for the first time. A major part of healthcare services depend on trust as it provides psychological comfort to service providers and consumers [138]. It is in human nature to be cautious about trusting someone you meet or work with for the first time; this is even more difficult in virtual settings as collaborating members don't meet each other face to face. The concerns can be addressed in an understandable and enforceable SLA [248]. SLA can also formalise service expectations as claimed by [249].

In a virtual setting where unidentified number of participants are online to provide a service, it is always a challenge to select the right one for the services requested. With this challenge in mind, in Section 6.3, a member selection mechanism is presented which is an important part of the HC-VBE framework. The framework makes use of SLA templates generated based on healthcare service requests send to the HC-VBE, and selects the right participant based on a number of criteria such as qualification and reputation. Once the right participant is identified and selected from a pool of providers, verifying and validating the credentials of the selected participant presents another challenge that has to be addressed. To address the challenge, in Section 6.4 a virtual healthcare provider verification and validation framework that uses blockchain technology is presented.

Figure 6.3 shows the overall infrastructure of the framework. In the diagram, *Requesters* appear on the left of the system boundary, *Providers* appear on the right, *Organisers* who manage the VBE appear on top and *Support* appears at the bottom of the system

boundary. Participants and resources join force under independently managed entities which is called *Service* in the case of short goal-based collaboration and *Task* in the case of long-term management-oriented collaboration. All participants are bound by SLAs approved by participants in a given collaboration settings. Before explaining the components, which make up the framework, it has to be noted that the framework is designed under the assumptions that:

- 1. Human participants have the necessary competence to use computers and smart devices.
- 2. The internet and local networks are the mediums facilitating the communications between participants.
- 3. Expected healthcare services are the ones which are virtually achievable such as consultation services.





6.2.2 Framework Component Description

HC-VBE and HC-VO participants in the framework can be human or electronic devices collaborating and sharing resources through communication technologies [27]. In designing the framework, the SPCM described in Section 5.2 and published in [250] is used. In this section, each of the components of the framework which is shown in Figure 6.4 are described.



Figure 6.4: HC-VBE components grouped in 5 groups (Initiation Manger, Negotiation and Creation Manager, VBE Monitoring Manager, Repository Manager and VO Monitoring Manager)

6.2.2.1 VBE Initiation Manager

The component manages healthcare requests sent by participants (e.g. patients) to the HC-VBE for a service (e.g. regular heart rate check) and has five use cases as defined below. A service request could be a simple function triggered by a registered *Requester*

or *Organiser* or a detailed request for a service [250]. In both cases, the HC-VBE generates a request report. The report contains the necessary information for creating an SLA and makes it available to the *Request Checker* use case for processing.

Request checker

It analyses the request report sent by the HC-VBE and carries out the following checks:

- 1- Validity check: Checks the validity of the following attributes:
- a- Date of the request.
- b- Time of the request.
- c- Requester details such as ID and attributes.
- d- Requested service description, by checking that the necessary information is provided in specified fields of the request report.
- 2- Scope check: It is carried out to ensure that the service requested is within the scope of the HC-VBE. The check mechanism was not intended to be specified here but it can be for example by searching for keywords in the service description attributes of the request report.
- 3- Completeness check: Checks the availability of the required data to trigger an SLA creation process.

• Request parser

The function of this use case is to parse the content of the *request report* into internal conceptual representation which is the classification of roles and services as per the SPCM [250].

• Requester identifier

Once the parsed request report is made available, this use case identifies the requesters of the service by searching for the followings:

- a- Name and ID of participants
- b- Role types of the requester (Requester or Organiser)

• Type of service identifier

The function of this use case is to identify the requested service type in the request report. To do so, it searches through the parsed report to identify if the request report is for a *Service* or *Task*. If it is not specified in the report then the use case searches for indications to decide what is the request for, and it uses a number of indications to make the decision:

- a- To decide the request is for *Service*, information generated by the *Requester identifier* use case is used. If the request is made by a participant with role type *Requester* then the use case decides that the requested service is of type *Service*, this is because *Requester* can only request a *Service* formation [250].
- b- To decide the request is for *Task*, it uses the information generated by the *Requester identifier* use case. If the request is made by participant with role type *Organiser* then the use case decides that the request service is of type *Task*, this is because *Organiser* can only request a *Task* formation [250].

• Constraint identifier

Constraints play a vital role in specifying meticulous details of an SLA such as duration of service, quality of service, rewards, and penalties. The use case checks for constraints in the request report and make it available to the *Constraint negotiator* use case described later in this section. Since SLA contains terms to be fulfilled by each participant, a constraint-based mechanism is necessary to regulate the terms; for this, Object Constraint Language (OCL) can be used following the footsteps of others such as [85] and [141] and also for the following reasons:

- It has become the default language for expressing specification requirements [251].
 It is flexible and facilitates future extensions as separate OCL expressions can be written for each parameter based on the purpose of the parameter [85].
- 2- It is a general purpose and text-based specification language which requires a low learning curve. A text-based human readable SLA can be an effective way to speed up the negotiation process required for VO formation. Such mechanism can be

implemented and provided as a negotiation tool to human participants; such tool has already been considered and implemented by other researchers [252].

3- OCL-based constraint can be converted to Web Ontology Language (OWL) which is machine readable and helpful during framework implementation, as claimed by authors of [253]. The authors provide a direct mapping between OCL and OWL which is pictured in Figure 6.5. The mapping is done between specific OCL expression and OWL tags, for instance, OCL *context* is mapped to <sch:pattern id> tag.



Figure 6.5: OCL to OWL mapping between OCL expression and OWL tags taken from [250]

6.2.2.2 Negotiations and Creation

This component facilitates the negotiation and creation of SLA, based on the healthcare service request processed by the *Initiation Manager* component. It supports the following use cases:

• Participant catalogue

This use case has a number of important functions as follows:

a- Keeps record of previous participants of a HC-VBE in terms of member identification attributes, ratings, availability, specialty, performance history and any other required information.

- b- Search for suitable participants within the record to identify a suitable participant for the requested service. This set uses the participant selection mechanism described in Section 6.3.
- c- Trigger an advertisement for specific participants such as *Provider* and *Support* when such participants are not available within the internal catalogues.
- d- Approve participant applications in collaboration with the registration use case of the HC-VBE.

Task catalogue

The functions of this use case are as follows:

- a- Keep record of previous *Task* implemented by a HC-VBE in terms of task descriptions, task identification attributes and task durations.
- b- Search for reusable *Task* based on the requested service requirements. This is to speed up the *Task* formation and implementation process within the VBE to deliver the fastest service possible to healthcare service requesters.
- c- Process the details of a new *Task* as it is formed in order to stipulate them in the SLA that is going to be created for the participants of the *Task*.

• Service catalogue

The functionalities here are similar to the one of *Task catalogue* but instead of *Task* the functions are performed for *Service*.

• Constraints negotiator

Constraint negotiation is one of the essential tasks in bringing all participants together to agree on an SLA. The function of this use case is crucial in the framework. It is responsible for presenting all available options to participants so that they can make a decision on the conditions under which they want to participant or receive the healthcare service. The use case has the following main functions:

- a- Fetch the constraints in the service request report
- b- Make the constraints available to all prospective participants
- c- Present all participants who are willing to fulfil the specified constraints in the request report. The mechanism of the presentation is not specified but could be based on percentage of constraint acceptance by each participant.

d- Stipulate in an SLA, the confirmed and agreed on constraints.

• SLA creator

The use case puts together all gathered information and generate SLAs. The task here is closely linked with the result of the other use cases detailed so far. An SLA is generated for each HC-VBE participant that participate in service provision in a given HC-VBE collaboration settings so that each participant knows their exact roles, obligations and rewards.

• Convertor

Not all who participate in healthcare virtual collaborations are computer or system experts, normally ordinary people are the main customers of such services. Therefore, it is important to have an SLA which is written in human readable and understandable format. The functions of this use case are:

- a- Convert machine readable SLA into human readable text.
- b- Convert SLA created by a HC-VBE into machine readable OWL.

If any changes made to the clauses of an SLA during service provision the convertor will convert the changes from machine readable format into human readable format and vies-versa. Figure 6.6 illustrates the conversion process where the converter is the bridge between text-based SLA and OWL-based SLA and the double headed arrows indicate bi-directional conversion capability. The conversion process and mechanism are not outlined in this thesis since it is an implementation matter that may differ from one virtual collaboration system to the next.



Figure 6.6: A view of the convertor use case working mechanism

6.2.2.3 Monitoring Manager

Once an SLA is created and sent for implementation, it has to be monitored to ensure that the terms and conditions are observed and respected. The Monitoring Manager component performs the monitoring function through the following use cases:

• Enforcer

The function of this use case is to ensure that the terms and conditions stipulated in an SLA are observed and implemented by performing the followings:

- a- Match continuously the participants attributes stipulated in the implemented SLA to the ones provide the service in the *Task*. The matching can be performed by comparing participant IDs and Names for example.
- b- Monitor the implementation of the attributes of services stipulated in an SLA through checking service satisfaction indicators such as satisfaction ratings by service requesters to ensure that the full agreed services are provided.
- c- Trigger the process of rewarding participants in terms of ratings and payments once the services requested are fully delivered and confirmed by service requesters.
- d- Trigger the process of fining participants who have violated the terms and conditions of an SLA.

• Change manager

The use case manages and detects changes that may be made to an SLA during implementation. Changes can be requested by participants of an SLA before and during SLA execution. For instance, changes can be requested to enhance SLA terms and conditions to better the services provided or the other way around. In case of changes, the use case performs the followings:

- a- Gather information such as the change requester IDs and the change requested.
- b- Specify the clause numbers in the SLA that the change affects.
- c- Notify all affected participants of the change.

- d- Trigger the *Convertor* use case to convert the changes to text and OWL version.
- e- Replace an old SLA with a new one and make it available for the *Monitoring Manager* use case for implementation.

• Dispute manager

This use case processes all complaints sent to a HC-VBE regarding an SLA content adherence, for example, a service requester could rate a service provider poorly for the service provided. In such a case, the service provider can complain about the rating and the use case has to process the complaint, for this, it takes the following actions:

- a- Identify the SLA which the complaint is about by searching for the SLA attributes provided in the complaint such as ID of the SLA.
- b- Identify the parties involved in the complaint from the identified SLA.
- c- Compare results gathered during SLA implementation with the information provided in the complaint.
- d- Inform involved parties of the complaint outcome.
- e- Make changes to records held about the SLA and affected participants.

6.2.2.4 Repository Manager

This is the last component which includes two use cases as follows:

• Maintainer

To keep all records about a specific SLA up to date, regular checks have to be made in collaboration with other use cases especially the *Change manager*. Therefore, regular checks and keeping records up to date is the main function of this use case.

• Archiver

This use case has the following functions:

a- Create a Master SLA for each *Task* and *Service* where SLAs for all participants are placed under one umbrella for the purpose of implementations and linkage. Figure

6.7 illustrates the Master SLA where (p) represents a participant within a specific virtual collaboration to which an SLA is created. All changes and referral requests for SLA clauses are made through the Master SLA to which the SLA is belong.

b- Archive SLAs created and provide access to each one for future references and reuses. This part of the use case plays an essential role in dispute management since it holds records about all SLAs and make them available for access at any time.



Figure 6.7:Master SLA concept where a number of SLA can be linked together under one Master SLA

6.2.2.5 VO Monitoring Manager

This component performs the same functions as *VBE Monitoring Manager* component except here the monitoring is performed for HC-VOs which have different participants and the duration of service provision is shorter than the one of HC-VBE.

6.2.3 The HC-VBE Framework Working Steps

This section provides a step by step description of the HC-VBE-F working process. The descriptions provided are none-technical as technical implementations are left for framework implementers' preference in the future. The framework organises and manages virtual collaboration for healthcare in 16 steps pictured in Figure 6.8.



Figure 6.8: HC-VBE-F processing steps showing the steps numbered and their directions specified

Before any participant can initiate a healthcare virtual collaboration, they will have to register in a HC-VBE system. Participants (requesters, providers and organisers) will have to fill in a registration form to ensure that all required participant's information is available for SLA creation and enforcement purposes. Once participants filled in the registration form and their details are approved by a HC-VBE, they can start making service requests in the case of requesters, provide a service in the case of providers and provide management services in the case of organisers.

The main processing flows for a HC-VBE system are:

1- After a service request form is submitted by a service requester, the HC-VBE creates a request report with similar content to the one shown in Figure 6.9 and sends it to the *checker* use case to be checked. The request report will be created based on the SLA templates described in Section 6.5.

Request report					
Requester details:					
1-	Id: (e.g. 10)				
2-	Name:				
3-	Address : (e.g. Huddersfield, UK)				
4-	Contract information:(e.g. <u>hoger@gmail.com</u>)				
Service requested					
1-	Service title: (e.g. head ache consultation)				
2-	Type of service:(e.g. service)				
3-	Service description: (e.g. continuing headache at front part of head)				
4-	Service constraints (duration, payment rate):(e.g. consultation for 1 hours				
	on 24-12-2018 between 14:00 till 15:00 will pay 50 pounds)				
Providers required					
1-	Title :(e.g. Neurologist doctor)				
2-	Qualification :				
3-	Constraints (availability):.(e.g. be available in the afternoon)				

Figure 6.9: Request report example with possible contents

- 2- The report will be checked by the *checker* function of the system which carries out the following checks to make sure that the request report is complete. If any information is missing the requester will be notified to provide the missing information.
 - a. Requester details
 - b. Service requested
 - c. Providers required
 - d. Service conditions
 - e. Offered payment amount

- 3- After the check is completed, the system sends the request report to be parsed (separate the information contained in it) for providers and service required. The parsing process uses the SPCM described in Section 5.2.
- 4- The parsed information from the request report will then be sent to the *recruiter and creator* components to search the catalogues for the right participants. It also initiates a *Service* or a *Task,* based on the request made.
- 5- To recruit the requested participants, first, the internal database is searched, in the event that the right participant could not be found the system advertises the request to external HC-VBEs to find the right participants.
- 6- Once the right participants and services are identified, the information will be collected and sent to the *negotiation* component to start negotiating the terms and conditions of the service.
- 7- The *negotiation* component then generates an SLA (OWL-based version) based on the identified information.
- 8- The HC-VBE sends the OWL-based SLA to the *converter* use case.
- 9- The *converter* use case converts the OWL-based SLA to a text-based human readable SLA.
- 10- The converted template will be sent to both requesters and providers for review and approval.
- 11- Providers and requesters read the SLA and either approve it or request changes to it.In the case of change requests, the agreement will be sent back for renegotiation.This process cycles through until the final agreement is reached.
- 12- Once the SLAs are approved by all parties they will be converted back to OWL-SLA and sent to the *Monitoring and enforcement* component for implementation. At this stage, a *Service* (HC-VO) will be created if the original request was for a *Service* and a *Task* (VBE management) will be created if the original request was by organisers for *Task*.

- 13- After a *Service* is created and running, if any change was requested to be made to the approved SLAs during the run, the request will be sent back to the *negotiation* use case to be approved by all parties before implementation.
- 14- In the event of change requests approval, they will be accommodated in the old SLAs and converted to OWL-based SLA and send it back to the *enforcement and monitoring* component.
- 15- After the service provision is complete, then the *monitoring* use case either reward the provider and charge the requester or it will fine the provider for not observing the terms and conditions of the SLA.
- 16- Finally, all SLAs will be archived in a repository for referral and reuse in the future. Figure 6.10 is the overall UML activity diagram for the HC-VBE-F which shows the activities performed by each component and the flow of information between the components.



Figure 6.10: The framework UML activity diagram that show the flow of activity stage by stage

Finally, as described, the developed framework is a generic one that supports the identification, recruitment and management of healthcare virtual collaboration participants which may come together to provide a healthcare service. The framework has advantages over other proposed frameworks. For example a virtual telehealth framework is suggested in [254]. Similar to the proposed framework in this thesis, the authors have developed the framework with the aim to facilitate healthcare provision virtually. The framework is designed specifically to manage stroke cases collaboratively and facilitate doctor to doctor consultation.

One of the main differences is, the authors have stated that patients using their framework can only request a service in the time allocated by healthcare professionals, which clearly is a limitation on the use of the framework. The framework described in this thesis enable requesters to request a healthcare service on demand and be serviced in real-time. This is because patients can request services at any time they may wish and there will always be a healthcare provider to provide the service requested. The second difference is that the authors state that healthcare providers can only be recruited from the local repository but the developed HC-VBE framework is designed for global level collaboration where providers are recruited from a pool of providers around the world. The third difference is that the framework is designed to provide specific services whereas the developed HC-VBE framework is generic and can be used to provide all types of healthcare virtual services.

6.3 HC-VBE Member Selection Mechanism

Collaborating and working in a virtual world lacks direct performance supervision, which means, enforcing timely performance requirements on collaborating parties may not be achieved easily. One way to deal with this issue is to make sure that the right collaborating participant is selected. Researches such as [255] emphasises the importance of effective mechanism for member selection in dynamic domains and [256] states that, in dynamic and complex environments where many possible partners and technologies exist, member selection process is a challenge. The Authors of [257] state that having the right team member in a team is an essential success factor that should be taken seriously.

How to select the right member? Under what criterion? Are questions researched by researchers in different fields and the issue is yet to be resolved. The authors of [258] claim that there is a lack of unified framework for team member selection. The challenge in virtual collaboration is even greater, because besides having the right professional skills, collaborating virtually, requires participants to be technically competent in the technology that is going to be used for the collaboration [259].

According to [256], there are two member selection processes, the first considers historical member performance and the second uses predetermined criterions to assess the suitability of new members. Historical member performance can be gaged through reputation. The concept of reputation has been considered in some researches for the purpose of member selection. For instance, reputation accumulation processes where members can rate other members based on personal experience and collaboration results, are considered by authors in [260] and [261]. Defining the right criterion is also a challenge that is being researched, for instance, [262] identifies partner characteristics, knowledge and capability, and degree of fitness as three criterions to be considered for partner selection. Over recruitment and under recruitment is another issue realised by researchers in the field [257]. In this section a task-based member selection mechanism is outlined by which problems such as over recruitment is avoided.

To tackle the issue in virtual enterprises, a partner selection mechanism is developed by [255] that considers both time and cost as the two main selection determinate criterion. In a similar attempt, a temporal team member selection framework is suggested by [258]. The authors of [263] use combinational auction to deal with partner selection issue which may work in a domain where time is not an important factor as the selection takes time to be finalised based on who offers more for a product. However, this kind of mechanism is not suitable for healthcare domain as service quality and fitness has to be considered alongside cost and time.

To select the right member for healthcare virtual collaboration, a 7-step member selection mechanism (HC-VBE-MSM) pictured in Figure 6.11 is developed. The mechanism makes use of the SPCM described in Section 5.2 and the HC-VBE-F described in

Section 6.2 to select a member for virtual collaboration. The mechanism's working steps are described in the next section.



Figure 6.11: A seven steps HC-VBE Member Selection Mechanism, developed to work within the $\rm HC\text{-}VBE\text{-}F$

6.3.1 Member Selection Mechanism Steps

In this section each step in the mechanism is described.

Step 1: Read parsed request report

It is assumed per the HC-VBE-F, as a result of processing the initial request from a healthcare requester a parsed report has been generated in which participants in the report are separated based on the SPCM. The member selection mechanism makes use of the parsed report, reads and analyses its content and start initiating the member selection process based on the information provided in the record.

Step 2: Identify participant

Having analysed the parsed report in the previous step, now the required participant is identified. For instance, if a healthcare member of type *provider* and title "psychologist"

is requested, this step in the mechanism will note this information and pass It on to the next stage of the member selection process.

Step 3: Identify participant attributes

The information in the previous step now has to be analysed further to identify the attributes of the participant. The process in this step is vital to ensure all the necessary details of the identified participant is noted and processed before passing it on to the next stage of the selection process.

Step 4: Internal or External search

In this step, participants who possess the required attributes and roles are searched for in order to be contacted for possible service provision. First, the search will be in the HC-VBE repository; this is to speed up an SLA finalisation process as details of previous participants are already saved and the search list is shorter. The search will be moved to external HC-VBEs if the right participant could not be found or negotiation fails. As per the HC-VBE-F, in this case, the participant requirements are advertised to external HC-VBEs for possible offers.

Step 5: Propose participant

As the result of the search, if the required participant is found, then the member credentials are going to be verified and validated using the provider verification and validation mechanism described in Section 6.4. After verification and validation, the HC-VBE-MSM proposes the details of the provider to the requester for possible negotiation. At this stage, the status of the proposed member is undecided in terms of recruitment and it will be decided in the next step.

Step 6: Approve participants

If the negotiation in the previous step is successful then the mechanism notifies both requesters and providers to approve each other. The approval will serve as a base for the SLA that is going to be created for both. This is a preliminary approval just to enable the process of SLA creation to be triggered, the final approval is going to be made when the created SLA is approved by both requesters and providers.

Step 7: Recruit participant

After both collaborators (requester and provider) approve each other, at this stage, the SLA between them are finalised and the provider will be formally recruited. This will lead to the start of HC-VO formation process. Figure 6.12 is the UML activity diagram for the HC-VBE-MSM process. The diagram shows the activities that are performed throughout the seven-step mechanism.



Figure 6.12: HC-VBE -MSM activity diagram showing the flow of activities to select a participant

6.4 HC-VBE Provider Verification and Validation Mechanism

Technology-based healthcare systems have altered the way care is provided to those in need with emphasis on raising the quality of care and reducing the cost of care [264]. Virtual healthcare is one of the ways that technology is playing a decisive role in its delivery, however, there are challenges that are yet to be addressed. In virtual healthcare settings, it is highly probable that healthcare providers and receivers do not know each other, which formulates the user validation and verification challenge. Patients are entitled to be served by qualified healthcare professionals with their credentials verified and validated. Moreover, to prevent information falsification and identity assumption it is vital that virtual collaboration systems in healthcare offer the assurance needed regarding the identity and attributes of the collaborators.

This section describes a healthcare provider verification and validation mechanism (HC-VBE-PVVM) which is developed based on blockchain technology. A brief background on blockchain technology is provided in Section 2.8 which serves the foundation for the development of the mechanism. Blockchain technology is sighted for huge potentials in many areas of services, and reputable companies such as the Tierion/Philips partnership (Netherlands), GEM (U.S.), IBM, Guardtime (Europe) and Brontech (Australia), are all considering the technology in addressing challenges related to their needs [265]. Healthcare providers are taking a keen interest in applying the technology to solve healthcare provision related challenges as claimed in a study by Deloitte [266].

There are studies such as [267] that claims current user authentication systems such as password and usernames have not been very successful and suggest blockchain is the technology that can provide the full solution. The thesis author does not intend to comment on the technical implementation of blockchain for healthcare provider verification and validation but rather provide a conceptual description of the mechanism. The implementation details and challenges may require a separate research to be carried out in the future. In this section, the main participants, the authentication system conceptual structure and the processing steps that serve the bases of the mechanism are provided. To shed some lights on the applicability of the mechanism it is applied to a simple healthcare scenario. The HC-VBE-PVVM described in the next section has been published in a peer reviewed journal in [268].

6.4.1 HC-VBE-PVVM Description

Blockchain has been considered in a diverse fields of research ever since it was reported in 2008 [92], and its famous crypto currency application (Bitcoin) in 2009 [269]. Healthcare is one of the fields that researchers are considering the technology as a possible solution to some healthcare provision related challenges. The technology is already used for patient identification [270], patient record control [271] and healthcare record sharing between stakeholders securely [264]. A swiss company (Healthbank (<u>www.healthbank.coop</u>)) is developing a blockchain-based system to control healthcare data transaction verification and validation. The technology is considered for healthcare provider qualification validation and verification in [272] in a similar way as the proposed mechanism without specifying the verification and validation process.

Care provider reputation plays an important role in raising quality as healthcare requesters seek providers with high reputation; to secure the authenticity of provider reputation, blockchain has been used by the authors of [273]. To prevent drug counterfeiting, Hyperledger is using blockchain to develop a system in collaboration with a number of well-known companies such as IBM, Cisco and Intel [274]. To provide a secure platform to enable healthcare specialists to share information, a network based on blockchain has been launched by Gem Health in the US [275]. Blockchain technology is believed to play a vital role in healthcare which has created a buzz to an extend that some healthcare researchers call it a revolution [287]. For long, the concern around to be sure that the person you talk to online is who they claim to be, has been present. Despite the invention of many authentication techniques the concern is yet to be addressed fully.

The mechanism presented here aims to aid healthcare virtual collaboration system developers with a step by step roadmap that enables them to use blockchain technology for healthcare service provider validation and verification purpose. The mechanism works under the following assumptions:

1- Healthcare systems are developed based on the HC-VBE-F concepts developed in this thesis where any request for healthcare service is processed and provided virtually.

2- A number of virtual collaborators including HC-VBEs, healthcare institutes, governmental agencies and academic institutes participate in creating a blockchain to share and validate healthcare providers credentials. HC-VBEs seek the verification and validation of healthcare providers that they intend to recruit for service provision. Healthcare institutions validate reputational attributes such as length of experience and

professionalism of the healthcare provider to whom the verification and validation are required. Academic institutes are responsible to verify and validate the academic qualifications that the healthcare providers claim to hold.

Under the above assumptions, a HC-VBE-PVVM that performs its function in a seven-step process. The mechanism components and working steps are shown in Figure 6.13. Each working step of the mechanism is described in Section 6.4.1.1.



Figure 6.13:HC-VBE provider verification and validation mechanism steps and components

6.4.1.1 PVVM Working Steps

The following steps describe how the HC-VBE-PVVM works:

1- Service request initiation: In a HC-VBE, requests will be received for a service from a requester in a form of a request report as explained in Section 6.2.3. Once this took place, the HC-VBE takes the necessary steps as in Section 6.3.1 to find and contact a service provider.

- 2- Provider identification: If the provider requested is found within the local database it means the provider has already gone through the verification and validation process, in that case, the provider will be recruited to provide the service after mutual acceptance by both healthcare providers and requesters. Otherwise, a newly unverified and unvalidated provider can be found and contacted from a pool of external providers.
- 3- Provider validation and verification request: After the newly found provider declares its intention to provide a particular healthcare service, the HC-VBE sends a verification and validation request to the formed blockchain. The mechanism broadcasts the information provided by the healthcare provider in the blockchain to be verified and validated.
- 4- Blockchain verification and validation initiation: As result of the previous step, now all nodes in the chain has received the request and the process of verification and validation can start. In blockchain, verifying and validating a transaction which is known as mining, comes with an incentive (e.g. financial payment) and the node with most computational power wins the incentive. Having made that clear, now the nodes in the chain compete to verify and validate the providers information requested by the HC-VBE. The competition speeds up the verification and validation process.
- 5- Provider information comparison: In this step, the information received by the nodes are processed and compared with blocks held by relevant authorities (e.g. academic institute or hospital where the provider has claimed to have worked). In blockchain technology, there are several methods that is used for the comparison process such as Proof of Work (PoW).
- 6- Verification and validation result: When the process of information comparison is completed, the requester HC-VBE will be notified of the outcome and the result of the process will be broadcasted to other nodes to update their records.
- 7- VBE result processing: If the provider's information and credentials are verified and validated successfully then the HC-VBE will take the necessary steps to recruit the provider, and subsequently form a HC-VO for the healthcare requester and the

approved provider. Otherwise, the HC-VBE will take steps to find new healthcare providers for the requested service and repeat the steps 2-7.

Finally, the initiation of using blockchain technology to address a well-recognised challenge in healthcare virtual collaboration, sums up the original contribution of this part of the research. However, the mechanism is conceptual and it is yet to be implemented. The mechanism has the potential to be a possible future PhD research as further research and implementation is required to fully realise the mechanism and prove its contribution in the field of healthcare, empirically.

Researchers mention that despite the acclaimed contribution of blockchain technology to the healthcare sector conceptually, the implementation and adaption of the technology is very slow due to the fact that it requires significant computational resources and the cost of its use is challenging to estimate [278]. Having said that, the technology deemed full of potential and there are studies that suggest the opportunities for its use in healthcare. For example, the authors of [282] suggest several potential services that blockchain can provide in healthcare such as bringing stakeholders of healthcare closer by removing third parties during collaboration and sharing resources, hence reducing transaction charges and keeping records secure and trusted.

6.4.1.2 HC-VBE-PVVM Application Example

The mechanism is designed to provide verification and validation for virtual healthcare service providers that offer to provide a service in HC-VBEs. One of the most common virtual healthcare services is consultation. The service's purpose is to hold conversation with a patient in order to guide the patient on healthcare concerns. To show the applicability of the mechanism the scenario below is considered:

Mr. David has recently developed a pain in his back after a fall while playing football. After a number of hospital visits for close check-ups, he has been advised by a specialist that he has to perform a number of specific exercises in the next few months to manage the pain. His local doctor has introduced him to a HC-VBE where he can request consultations with a physiotherapist on a weekly basis. Now Mr. David can use the virtual services provided by the HC-VBE to receive guidance regarding his exercise without the need to visit his local healthcare institute. In order to receive the care, he takes the following steps:

- 1- He fills in a request form after registering himself in the HC-VBE system for a physiotherapist consultation. In the form, he explains his health issue and specifies the attributes of the physiotherapist as (having good reputation, minimum 6 years of experience, qualified by an EU university and should speak good French as he is a French national). He also specifies the duration of the service and the amount he is willing to pay for the service. He sends the request to the HC-VBE for processing.
- 2- The HC-VBE takes the steps outlined in Section 6.3.1 to find a physiotherapist that matches the request made by Mr. David. The HC-VBE search in the service providers local repository does not yield a close match. Now the HC-VBE advertises the request to external HC-VBEs and a physiotherapist who claims that he is a French national with 9 years of experience and an EU graduate with close matches to the other attributes specified in the request, offers to provide the service.
- 3- The HC-VBE now uses the outlined verification and validation mechanism to ensure the claims made by the provider are correct or false. To perform the verification and validation checks the HC-VBE takes the steps 3-7 described in section 6.4.1.1.

6.5 HC-VBE Service Level Agreement Templates

Managing SLA in a complex, multiparty and dynamic environment such as healthcare virtual collaboration is a challenge. The challenge is recognized by researchers in the area and they all point to the need of a comprehensive SLA framework [276][277]. To manage SLA in virtual and distributed environments, various approaches such a modelling SLA based on business objectives have been suggested by authors in [278][279][280]. The main management tasks in SLA lifecycle have been researched intensely and there is a common agreement that negotiation, creation, monitoring and closure are the four major tasks in the cycle [281][282] [283][284] [87]. On a practical level, for SLA negotiation and creation, efficient protocols are required, for such requirements some significant works

are on-going such as the extension to WS-Agreement by [285]. The approaches are proposed for domains other than healthcare and they do not present a clear and comprehensive SLA management process. Similar to the approach taken in this thesis, the authors of [286] propose a SLA framework for federated virtual organisation without providing details as to how the SLAs are created and managed. In Section 6.2 within the HC-VBE-F a clear SLA life-cycle management process is provided where the SLA negotiation, creation and monitoring process is described.

As for the structure and content of SLA, the concept of SLA template is commonly used to provide a consistent structure and content for SLAs created for a specific domain or purpose [88][89][90]. A template, provides the structure and the main sections that should be included in an SLA, such as involved parties, service attributes and constraints. Templates also speed up the agreement process during negotiation since participants learn what is required from them and what they will get in return. The majority of virtual service providers such as those in could computing businesses, offer SLAs based on specific templates [287]. In a fast-changing environment such as healthcare having an electronically enforceable SLA to guarantee the delivery of the right service at the right time and at the right cost is essential. Another important point is the fact that not all services provided in healthcare can be provided electronically, human involvement therefore is inevitable. [135] Claims popular SLA mechanisms such as WSLA [140], CC-Pi [133] and WS-Agreement suffer from flexibility and support for services that have human involvement.

All available electronic SLAs are machine readable and could be difficult for ordinary human user to read and understand. The SLA templates presented in this section have a text-based version which supports human readability and involves human decision making in the process which makes it different from others. This work is a step forward in formalising the structure of SLAs for HC-VBE. As for the HC-VBE-SLA template implementation they can be implemented in OWL, and OCL can be used to define service constraints. Figure 6.14 is a section of the Master SLA template implemented in OWL. In a separate approach, the authors in [288] propose an RDF-based language to represent contracts, however the language is semantic web specific which limits its use as not all virtual organisations necessarily base their collaboration on semantic web architecture.



Figure 6.14: A section of the OWL version of the Master SLA

To enforce, monitor and manage SLAs, the authors of [289] suggests GSMA architectures and the authors of [131] Identify the main components of a typical SLA which is used as a guide to develop the clauses of the developed SLA templates in this section. Here, two templates are going to be presented; the first template is a Master SLA described in Section 6.5.1 and the second template is a general SLA described in Section 6.5.2. Figure 6.15 shows the template package developed for the HC-VBE-F which consists of two templates namely HC-VBE-SLA and HC-VBE-Master-SLA.



Figure 6.15: HC-VBE-SLA template package components

Each template consists of a number of clauses and the naming convention of each clause is as follows:

 Clause Number: <<HC-VBE-SLA>> Name, where Number and Name are variables i.e., they are different for each clause. To ensure that the templates are designed objectively, the following attributes are identified to be considered in the templates:

- 1- Be generic, this is to ensure that it can be used as a base for all possible HC-VBE collaborations SLA creation.
- 2- Be simple and clear in terms of structure and content.
- 3- Be readable by both human and machine.
- 4- Be modified and adapted with ease.

The templates are text-based, human readable, scalable and are the base for all SLAs created in VBE and VO based healthcare virtual collaborations. For better readability, referrals, navigations and amendments, each clause and its sections are numbered. The same numbering system shall appear in the OWL-based version of the SLA templates. Figure 6.16 shows a section of the OWL version of the Master template detailed in the next section which is implemented using Protégé software.



Figure 6.16: A screenshot of the Mater SLA template implementation in Protégé
6.5.1 HC-VBE Master SLA Template

The purpose of having a Master SLA in the repository manger is already explained in Section 6.2.2.4; here the basic structure and content of a master template is described. As it is shown in Figure 6.17, the master consists of two clauses:

- 1- Master details: The function of this clause is to hold basic required data about the Master SLA for identification purposes such as id, creation date, creation time, status of the master, the number of SLAs it contains and finally the id of each SLA contained in the master. The information contained in this clause is described in Table 6.1.
- 2- Changes: The clause holds a record of all the changes that are going to take place in the SLAs contained in a given Master SLA. The information contained in this clause is described in Table 6.2.

	Master SLA				
Clause 1: < <hc-vbe-sla>> 1</hc-vbe-sla>	Master				
1.1: MasterID: in 1.2: CreationDate	t : date				
1.3: CreationTime: time					
1.4: Status: active waiting completed cancelled 1.5: CollaborationType: task service					
1.6: SlaNo: number					
Clause 2: < <hc-vbe-sla>></hc-vbe-sla>	Changes				
2.1: ChangeInSlal	D: number				
2.2: ChangeInClau	use: number				
2.2.1: Propert	ies:				
2.2.1.1:	NewDetails: text	, OCL, Number, Currency, Time,			
	Date, percentage,	, active, waiting, completed,			
	cancelled				
2.2.1.2:	Affects: clause	numbers, sub-clause number,			
	property number				

Figure 6.17: Text based Master SLA template developed for the HC-VBE-F

Table 6.1: Master SLA clause 1 description

Clause	Data	Туре	Description
Section			
1	< <hc-vbe-sla>> Master</hc-vbe-sla>		This indicates that this is a master-SLA template.
1.1	MasterID	integer	A unique Identification number by which the master is identified and referred to during implementation and monitoring.
1.2	CreationDate	date	The date that the Master SLA is created and the type of data that it holds is <i>date</i> .
1.3	CreationTime	time	The time that the Master SLA is created and the type of data that it holds is <i>time</i> .
1.4	Status	active waiting completed cancelled	 This is the status of the Master SLA which can only be at one of the four status below: 1- Active: Indicates that the master is currently active and being implemented. 2- Waiting: Indicates that the master is yet to be approved or activated. 3- Completed: Indicates that the master was successfully implemented. 4- Cancelled: Indicates that the master was cancelled before it was activated or completed. The statuses are separated by () which is an <i>OR</i> logical operator.
1.5	CollaboratationType	task service	The type of the collaboration for which the master is created which is either <i>Task</i> or <i>Service</i> . The two different types of HC-VBE services are separated by () which is an <i>OR</i> logical operator.
1.6	SlaNo	number	Indicates the number of SLAs contained within the master. The type of data that is held in the line is

Clause Section	Data	Туре	Description
			number, starting with 1 as no Master SLA can be created without containing at least 1 SLA.
1.7	AllSlaIDs	enum	Each SLA that is linked to the Master SLA has a unique ID. Here all SLAs' ID contained in the Master SLA are listed.

Table 6.2: Master SLA clause 2 description

Clause	Data	Туре	Explanations
section			
2	< <hc-vbe-< th=""><th></th><th>This is where the changes with regard to SLAs</th></hc-vbe-<>		This is where the changes with regard to SLAs
	SLA>> Changes		contained within the Master SLA are recorded.
2.1	ChangeInSlaID	integer	The ID of the SLAs in which the changes occur and
			the type of data it holds is <i>integer</i> .
2.2	ChangeInClause	number	The clause number of the SLA in which the change
			occurs. The type of data it holds is a number since
			each clause is numbered.
2.2.1	Properties		The attributes of the change are under this section.
2.2.1.1	NewDetails	Text (,&)	The new details that will be replaced in the affected
		OCL (,&)	SLA is recorded here. The data it holds can be a
		Number (<i>,</i> &)	mixture of the data types specified. The data types
		Currency (,&)	change per the line number of the SLA that the
		Time (,&)	change takes place. As each line numbered holds a
		Date (,&)	different data type, therefore the reason this
		Percentage	section has so many different data types is because
		(,&)	it has to accommodate changes requested to any
		active (,&)	section of the SLA. The (,&) are the OR and AND
		waiting (,&)	logical operator which indicates that the data type
		completed	can be alone or with other data types.
		(,&)	

Clause	Data	Туре	Explanations
section			
		cancelled (,&)	
2.2.1.2	Affects	ID(,&)	This indicates the SLAs that are going to be affected
		clause numbers	as the result of the changes described in the
		(,&)	previous section. which can be any of the data
		sub-clause	types specified. The (,&) are the OR and AND
		number (<i>,</i> &)	logical operator which indicates that the data type
		property	can be alone or with other data types.
		number (<i>,</i> &)	

6.5.2 HC-VBE General SLA Template

This section describes a general SLA template to be used as a base for all the SLAs that will be created by a HC-VBE. The template consists of six clauses which are General, Collaborator 1, Collaborator 2, Expected Service, Duration and Enforcement. Each clause is designed for a specific purpose and a diagrammatical view of the template shown in Figure 6.18. In this section the content of each clause is described.



Figure 6.18: The main six clauses of the developed general SLA template for HC-VBE

1- <<HC-VBE-SLA>> General: The purpose of this clause is simply to present a basic information about an SLA such as the SLA ID, creation date and time. Figure 6.19 shows the structure and content of the clause as will be seen by healthcare requesters and providers and Table 6.3 describe each line in the clause.

```
Clause 1: <<HC-VBE-SLA>> General

1.1: ID: int

1.2: CreationDate: date

1.3: CreationTime: time

1.4: Status: active | waiting | completed | cancelled

1.5: MasterID: inter
```

Figure 6.19: HC-VBE-SLA template Clause 1 structure and content

Table 6.3 : The general SLA template first clause description

Clause	Data	Туре	Explanations
Sections			
1	< <hc-vc-sla>></hc-vc-sla>	text	This is the header name of the clause and the
	General		information contained in this clause is "general"
			and relevant to the whole SLA.
1.1	ID	integer	A unique identification number by which the SLA
			is recognised and referred to. The data type of
			the line in the SLA is <i>integer.</i>
1.2	CreationDate	date	This is the creation date of the SLA and it is
			represented by a <i>date</i> data type.
1.3	CreationTime	time	This is the time of SLA creation and it is
			represented by a <i>time</i> data type.
1.4	Status	active	This is where the status of the SLA is shown and
		waiting	there are four states:

Clause	Data	Туре	Explanations
Sections			
		completed	Active: Means the SLA is currently being
		cancelled	implemented.
			Waiting: Means the SLA is awaiting activation or
			approval.
			Completed: Means the SLA has been
			successfully implemented.
			Cancelled: Means the SLA is cancelled before it
			was activated or completed.
			The statuses are separated by () which is an OR
			logical operator.
1.5	MasterID	integer	This is the Master SLA ID to which the SLA is
			belong. It holds an integer data type.

2- <<HC-VBE-SLA>> Collaborator 1: The clause contains required information about the participant (requester) who the SLA is created for. The clause's structure and content are shown in Figure 6.20 and described in Table 6.4 which is also the same for clause 3. The assumption here is that every participant has pre-registered and their details and credentials are approved.

```
Clause 2: <<HC-VBE-SLA>> Collaborator 1

2.1: Participant: Requestor | provider | Organiser |

Support

2.1.1: Properties:

2.1.1.1: ID: int

2.1.1.2: Name: text

2.1.1.3: ContactDetails: text and number

2.1.1.4: Credentials: textand number

2.1.1.5: Constraints: OCL
```

Figure 6.20: HC-VBE-SLA template Clause 2 structure and content

Table 6.4: The general SLA template second and third clause description

Clause	Data	Туре	Explanations
sections			
2&3	< <hc-vc-sla>></hc-vc-sla>	text	The word "Collaborator" indicates that this
	Collaborator		section holds data about a HC-VBE participant.
2&3.1	Participant	Requestor	The participant type for whom the SLA is created
		Provider	and must be one of the four types. The role of
		Organizer	each type is explained in section 4.1. The
		Support	statuses are separated by () which is an OR
			logical operator.
2&3.1.1	Properties		The properties of the participants are recorded
			under this section.
2&3.1.1.1	ID	integer	When participants register for the first time,
			they are assigned a unique ID by which they are
			recognised and referred to. Here the ID of type
			<i>integer</i> is displayed.
2&3.1.1.2	Name	text	The name of the participant that was recorded
			during registration, it is represented as <i>text</i> data
			type.
2&3.1.1.3	ContactDetails	text and	The contact details of the participant recorded
		number	during registration, could be both <i>text</i> and
			number, text is for address and number is for
			telephone for examples.
2&3.1.1.4	Credentials	Text and	This is a brief description of the approved
		number	credentials of the participant such as
			qualifications and years of experience.
2&3.1.1.5	Constraints	OCL	This part displays all the constraints that the
			participants has put in place in order to
			participate in the collaboration. Constraints are
			written in OCL to make it short and concise for
			human and aid the conversion process to OWL

Clause sections	Data	Туре	Explanations
			version later. An example of constraint can be the duration that the participant is available to collaborate.

3- <<HC-VBE-SLA>> Collaborator 2: This clause contains required information about the participant (Provider) who provides a healthcare service; Figure 6.21 shows the structure of the clause and Table 6.4 presents the information recorded in this section.

```
Clause 3: <<HC-VBE-SLA>> Collaborator 2

3.1: Participant: Requestor | provider | Organiser |

Support

3.1.1: Properties:

3.1.1.1: ID: int

3.1.1.2: Name: text

3.1.1.3: ContactDetails: text and number

3.1.1.4: Credentials: textand number

3.1.1.5: Constraints: OCL
```

Figure 6.21: HC-VBE SLA template Clause 3 structure and content

4- <<HC-VBE-SLA>>Expected Service: The clause contains details of the expected service to be provided by the provider named in the SLA, Figure 6.22 shows its structure and content and Table 6.5 describes the content of the clause.

```
Clause 4: <<HC-VBE-SLA>> ExpectedService
4.1: HasRollIn: Service | Task
4.1.1: Properties:
4.1.1.1: Description: text
4.1.1.2: Constraints: OCL
```



Table 6.5: The general SLA template fourth clause description

Clause	Data	Туре	Explanations
sections			
4	< <hc-vc-sla>></hc-vc-sla>		This is the name of the section which
	ExpectedService		indicates that purpose of the clause.
4.1	HasRollIn	service	Here the type of the service that is going to
		task	be provided is indicated to be Service or
			Task as per the SPCM in section 4.1.
4.1.1	Properties		The attributes of the expected service will
			be listed under this section.
4.1.1.1	Description	Text	A text-based description of the expected
			service will be provided as part of this line.
4.1.1.2	Constraints	OCL	The constraints that shapes the expected
			service such as the method of service
			provision, the rate of the service and the
			type of the service are provided under this
			section.

5- <<HC-VBE-SLA>> Duration: This clause contains the details of the duration in terms of start and end date and time of the expected service. Figure 6.23 shows the structure and content of the clause and Table 6.6 describes the content of the clause.

Clause 5: < <hc-vbe-sla>> Duration</hc-vbe-sla>	
5.1: StartDate: date	
5.2: StartTime: time	
5.3: Constraints: OCL	
5.4: EndDate: date	
5.5: EndTime: time	
5.6: Constraints: OCL	

Figure 6.23: HC-VBE-SLA Template Clause 5 structure and content

Table 6.6: The HC-VBE-SLA template fifth clause description

Clause	Data	Туре	Explanations
sections			
5	< <hc-vc-sla>></hc-vc-sla>		The name of the section which indicates that
	Duration		it contains duration related data.
5.1	StartDate	date	The start date of the expected service is noted here with a data of type <i>date</i> .
5.2	StartTime	time	The start time of the expected service is noted here with data of type <i>time</i> .
5.3	Constraints	OCL	All the constraints relevant to the start time and date are recorded here using OCL such as the time is am or pm.
5.4	EndDate	date	The end date of the expected service is noted here with a data of type <i>date</i> .
5.5	EndTime	time	The end time of the expected service is noted here with data of type <i>time</i> .
5.6	Constraints	OCL	All the constraints relevant to the end time and date of the expected service are recorded here using OCL such as am or pm.

6- <<HC-VBE-SLA>> Enforcement: The clause contains the SLA enforcement attributes that must be monitored during implementation. Figure 6.24 shows the structure and content of the clause and Table 6.7 describes the content of the clause.



Figure 6.24: HC-VBE-SLA template Clause 6 structure and content

Table 6.7: The	HC-VBE-SLA	template fifth	clause	description
	110 102 02/1		010.010.0	

Clause	Data	Туре	Explanations
sections			
6	< <hc-vc-sla>></hc-vc-sla>		The name of the section which indicates the
	enforcement		SLA enforcement related data will be held here.
6.1	QoS (Quality of	criterions	This sub-section is concerned with the quality
	Service)		of the expected service as described in the next
			sub-sections which holds data of type QoS.
6.1.1	Properties		These are the quality of service properties that
			are going to be monitored during service
			implementation.

Clause	Data	Туре	Explanations
sections			
6.1.1.1	Availability	percentage	The percentage that the service is expected to
			be available. Measurement mechanism is
			beyond the scope of this research.
			Measurement type should be in percentage for
			transparency reasons.
6.1.1.2	Accessibility	percentage	The percentage that the expected service is
			expected to be accessible during collaboration.
			Measurement mechanism is beyond the scope
			of this research. Measurement type should be
			in percentage for transparency reasons.
6.1.1.3	Satisfaction	percentage	This is the satisfaction rate expected to be
			achieved after service provision from
			healthcare requesters point of view.
			Measurement mechanism is beyond the scope
			of this research. Measurement type should be
			in percentage for transparency reasons.
6.1.1.4	ResponseTime	time	The speed of the required response during
			expected service provision. Measurement
			mechanism is beyond the scope of this
			research. Measurement type should be in time
			for transparency reasons.
6.1.1.5	Constraints	OCL	All constraints related to the quality of the
			expected service are recorded here using OCL
			such as expected allowed waiting period for a
			response to be provided.
6.2	Reward	Payment	The expected reward if the criterion of the
		and	quality of expected service is fulfilled. It can be
		promotion	

Clause	Data	Туре	Explanations
sections			
			in a form of payment as well as promotions
			done through rating the service provider.
6.2.1	Properties		The properties of the reward are described
			under this section.
6.2.1.1	Payment	currency	The amount of payment in currency that the
			participant should receive after successful
			service provision.
6.2.1.2	Rating	number	The rate of promotion that the participant will
			receive after successful service provision.
			Rating are provided as numbers. The rating
			mechanism is technical and beyond the scope
			of this research.
6.2.1.3	Constraints	OCL	All the constraints related to the reward are
			recorded here using OCL.
6.3	Penalties	payment	This part records the details of penalties the
		and	participant incurs in terms of payment and
		promotion	promotion.
6.3.1	Properties		The properties of the expected penalties
6.3.1.1	Payment	currency	The amount of payment in currency that the
			participant will be fined after unsuccessful
			service provision.
6.3.1.2	Rating	number	The rate of demotion the participant will
			receive after unsuccessful service provision.

Clause sections	Data	Туре	Explanations
6.3.1.3	Constraints	OCL	All the constraints related to the penalties are recorded here using OCL such as the precise conditions that the healthcare service provider will incurs the penalty.

6.6 HC-VBE-F Evaluation Results

This section provides two separate sets of evaluation results computed to examine users' acceptance of the HC-VBE-F implemented as a mobile application prototype. The first set of results are the evaluation results from healthcare providers perspective and the second set of results are the evaluation results from healthcare requesters perspective. The framework was evaluated by 200 participants, 100 of which were healthcare providers and the other 100 were healthcare requesters using the steps outlined in Sections 4.3. In Section 6.7 the results of the hypotheses testing defined for both extended TAMs (HC-VBE-TAM-Provider and HC-VBE-TAM-Requester) are discussed compared.

6.6.1 HC-VBE-TAM-Provider Evaluation Result

Framework implementation and data collection process is already described in Sections 4.31 and 4.3.6. Here, the healthcare providers acceptance evaluation results for the HC-VBE-F are presented and discussed. The results are produced using the data collected during the survey and analysed using Structural Equation Modelling technique in AMOS software. The SEM model developed based on the HC-VBE-TAM-Provider is pictured in Figure 4.16 and described in Section 4.3.7.2.

6.6.1.1 Overall Data Analysis Results

This section reports on Mean, Standard deviation, Factor loading, Significance value and Intermeasurement Correlation values which were computed for the data collected using the questionnaire designed for healthcare providers and can be found in appendix B. Mean is the average of all values given to an observed item by surveyed participants. The Mean results for the observed items are between 6.08 and 6.54 which indicate strong agreeability of participants with the measuring statements in the questionnaire. The result is significant in the context of this research which conveys that the concepts developed in the HC-VBE-F and demonstrated through the developed mobile application prototype is acceptable by healthcare providers. This is because scale 6 represent high agreeability in the Likert Scale questionnaire as stated in Section 4.3.4.

Standard deviation measures the deviation of answers from the mean and the smaller the better as small values indicate that participants answers to individual questionnaire items were close. The standard deviation values are between 0.769 and 1.203 which indicates that the answers given by research participants were close to each other. This result shows that research participants views about the concepts behind the mobile application protype were fairly consistent. This finding increases the credibility of the results and provides a good acceptability support for the HC-VBE-F. The result of individual item factor loading indicates a strong link between observed items and their unobserved variables with values between 0.710 and 0.912 which are all above the acceptable cut-off value 0.5 [241]. The factor loading results suggest the effectiveness of individual measurement items in measuring the unobserved acceptance factors in the HC-VBE-TAM-Provider model which subsequently increases measurement reliabilities which is important for the validity of the hypotheses testing results.

To measure the significance of relations between the unobserved variables and their observed items, the significance value which is symbolised as p value is computed. Significance test is necessary in SEM to ensure that the result is not produced by chance and it is widely accepted that the significance value is when (p<0.05) [205]. Significance computed values are indicated as *** in AMOS analysis results output, which indicates that p<0.001. The result of all items is significant which suggests a high probability that the results are not produced by chance. Table 6.8 summarises the overall data analysis results for HC-VBE-F acceptability evaluation from healthcare providers perspective. The table shows statistical results for each questionnaire item within the context of their unobserved variables.

Table 6.8: HC-VBE-TAM-provider overall data analysis results which includes the Mean, Standard deviation, Factor loading and Significance value

Unobserved	Observed Variable and	Mean	Std.	Factor	<i>p</i> < 0.05
Variable	Related Questionnaire		Deviation	Loading	(Significance)
	Item				
Perceived	PHG1	6.38	0.982	0.764	<i>p</i> <0.001
healthcare	Using the HC-VBE application				
Globalisation	collaboration.				
(PHG)	PHG2	<i>p</i> <0.001			
	Using the HC-VBE application	-			
	widely available.				
	PHG3	6.48	0.990	0.792	<i>p</i> <0.001
	Using the HC-VBE application				
	serve healthcare seekers aro				
	PHG4	6.54	0.834	0.750	<i>p</i> <0.001
	Using the HC-VBE applicat	ion wo	uld be instru	umental in	-
	globalising healthcare.				
Perceived	PCE1	6.49	0.893	0.840	<i>p</i> <0.001
Clinical	Using the HC-VBE applicatio	n would	facilitate pro	oviding the	
Effectiveness	right care through making pa	itient his	story available	2.	
(PCE)	PCE2	6.51	0.916	0.808	<i>p</i> <0.001
	Using the HC-VBE applicatio	n would	facilitate pro	oviding the	
	right care through recruiting	specialis	st care provid	er.	
	PCE3	6.22	1.031	0.762	<i>p</i> <0.001
	Using the HC-VBE applicat	ion wo	uld facilitate	providing	
	timely care through anytime	collabor	ration capabil	ity.	
	PCE4	6.47	0.881	0.845	<i>p</i> <0.001
	Using the HC-VBE application	n would	facilitate pro	viding care	
	in the right place through an	ywhere	collaboration	capability.	

Unobserved	Observed Variable and	Mean Std.	Factor	<i>p</i> < 0.05			
Variable	Related Questionnaire	Deviation	Loading	(Significance)			
	Item						
Perceived	PU1	6.23 1.162	0.815	<i>p</i> <0.001			
Usefulness	Using the HC-VBE application	ormance in					
(PU)	providing care.						
	PU2	6.38 1.003	0.834	<i>p</i> <0.001			
	Using the HC-VBE applicati						
	provide care.						
	PU3	6.36 1.010	0.912	<i>p</i> <0.001			
	Using the HC-VBE appli	y patient					
	management tasks.						
	PU4	<i>p</i> <0.001					
	Overall, I find the HC-VBE	application useful in	n providing				
	healthcare.						
Perceived	PEU1	6.08 1.203	0.855	<i>p</i> <0.001			
ease of use	Learning to use the HC-VBE a	pplication would be e	easy for me.				
(PEU)	PEU2	6.16 0.982	0.784	<i>p</i> <0.001			
	I would find it easy to pro-	ovide care using the	e HC-VOBE				
	application.						
	PEU3	6.33 0.975	0.846	<i>p</i> <0.001			
	It would be easy for me to be	come skilful at using	the HC-VBE				
	application						
	PEU4	6.22 1.088	0.710	<i>p</i> <0.001			
	I would find the HC-VBE appl	ication easy to use.					
Attitude	AU1	6.61 0.777	0.766	<i>p</i> <0.001			
towards using	Using the HC-VBE application	n would be a good ide	a.				
(AU)	AU2	6.47 0.810	0.793	<i>p</i> <0.001			
	Using the HC-VBE application	on makes providing	healthcare				
	more interesting.						
	AU3	6.39 0.920	0.777	<i>p</i> <0.001			
	Using the HC-VBE application	would be a pleasant	experience.	•			

Unobserved	Observed Variable and	Mean	Std.	Factor	<i>p</i> < 0.05							
Variable	Related Questionnaire		Deviation	Loading	(Significance)							
	Item											
	AU4	6.36	0.938	0.752	<i>p</i> <0.001							
	I would like to use the HC-VE	-										
Intention to	IU1	<i>p</i> <0.001										
use (IU)	I intend to use the HC-VBE a											
	IU2	6.33	0.922	0.722	<i>p</i> <0.001							
	It is likely that I will use the H	•										
	IU3	6.35	0.903	0.863	<i>p</i> <0.001							
	I expect to use the HC-VBE a	•										
	IU4	6.43	0.769	0.829	<i>p</i> <0.001							
	I am willing to recommend other people to use the HC-V											
	application.											

6.6.1.2 Intermeasurement Correlation

Intermeasurement correlation shows the link between all observed items, Table 6.9 shows the analysis results of intermeasurement correlations for the questionnaire items examining the HC-VBE-TAM-Provider. There are 24 measurement items in the questionnaire. Each questionnaire item is correlated to all other questionnaire items in a questionnaire including itself. A perfect correlation value which is 1 is produced when a questionnaire item is correlated to itself, but the values would be less than 1 in all other cases. The intermeasurement results are between 0.264 and 0.756 which indicates that some of the intermeasurement are weak since the acceptable cut off point is 0.3. Having said that the vast majority of the intermeasurement results prove that the scores given to most of the items are closely related. This means that the research participants shared similar opinions on measurement items. Intermeasurement values provide a detail map of the scores given to individual items in a questionnaire. If the majority of correlation values are above 0.3 that indicates the questionnaire items have adequately captured users view of the technology being tested in this context.

	PHG1	PHG2	PHG3	PHG4	PU1	PU2	PU3	PU4	PEU1	PEU2	PEU3	PEU4	ECE1	ECE2	ECE3	ECE4	AU1	AU2	AU3	AU4	IU1	IU2	IU3
PHG1	1.00																						
PHG2	0.68	1.00																					
PHG3	0.62	0.73	1.00																				
PHG4	0.51	0.66	0.60	1.00																			
PU1	0.61	0.48	0.58	0.49	1.00																		
PU2	0.64	0.62	0.60	0.61	0.76	1.00																	
PU3	0.59	0.65	0.60	0.63	0.75	0.73	1.00																
PU4	0.53	0.55	0.55	0.62	0.64	0.61	0.76	1.00															
PEU1	0.44	0.40	0.26	0.36	0.31	0.31	0.35	0.28	1.00														
PEU2	0.42	0.43	0.40	0.44	0.49	0.45	0.53	0.41	0.69	1.00													
PEU3	0.51	0.47	0.33	0.38	0.38	0.33	0.43	0.34	0.76	0.60	1.00												
PEU4	0.34	0.44	0.29	0.36	0.34	0.37	0.37	0.27	0.55	0.60	0.62	1.00											
ECE1	0.58	0.75	0.50	0.54	0.62	0.62	0.72	0.51	0.45	0.46	0.56	0.44	1.00										
ECE2	0.43	0.65	0.55	0.51	0.60	0.65	0.70	0.53	0.27	0.46	0.39	0.47	0.73	1.00									
ECE3	0.51	0.62	0.54	0.53	0.54	0.61	0.67	0.58	0.42	0.50	0.49	0.49	0.63	0.64	1.00								
ECE4	0.56	0.64	0.60	0.61	0.64	0.66	0.74	0.62	0.39	0.45	0.45	0.41	0.71	0.66	0.61	1.00							
AU1	0.49	0.54	0.57	0.58	0.50	0.57	0.62	0.58	0.30	0.47	0.33	0.37	0.48	0.48	0.42	0.62	1.00						
AU2	0.60	0.63	0.62	0.59	0.59	0.59	0.66	0.59	0.36	0.51	0.38	0.39	0.60	0.57	0.57	0.71	0.66	1.00					
AU3	0.51	0.54	0.53	0.53	0.58	0.60	0.65	0.55	0.33	0.40	0.42	0.33	0.60	0.57	0.56	0.63	0.65	0.61	1.00				
AU4	0.39	0.48	0.49	0.59	0.53	0.58	0.59	0.50	0.48	0.58	0.37	0.42	0.55	0.57	0.46	0.61	0.58	0.51	0.62	1.00			
IU1	0.32	0.38	0.32	0.42	0.35	0.37	0.47	0.43	0.43	0.53	0.41	0.41	0.39	0.36	0.40	0.49	0.44	0.44	0.44	0.61	1.00		
IU2	0.24	0.34	0.31	0.40	0.30	0.32	0.38	0.30	0.24	0.43	0.29	0.26	0.35	0.35	0.39	0.49	0.41	0.45	0.35	0.42	0.53	1.00	
IU3	0.42	0.43	0.40	0.46	0.51	0.40	0.60	0.56	0.40	0.55	0.42	0.35	0.46	0.48	0.46	0.60	0.50	0.56	0.41	0.55	0.67	0.66	1.00
																							-

Table 6.9:HC-VBE-TAM-provider intermeasurement correlation results

6.6.1.3 Characteristics of Construct

At construct level, the result supports strong acceptability by the healthcare providers for the concepts developed in the HC-VBE-F. The mean value for each construct variable computed as an average for all observed items for a particular construct is above 6 with standard deviation of less than 1. Table 6.10 shows the constructs, the number of items used to calculate the means, their mean and standard deviation values. The result suggests that on construct level the survey participants' acceptance attitude towards the HC-VBE-F were positive since the values are above 6 which are equals to "quite Agree" in Likert-Scale measures. The result also suggests that on construct level the answers were similar since the standard deviation values are less than 1.

	Item Statistics			
Constructs	Observed	Mean	Std. Deviation	N
	Items			
Perceived Clinical Effectiveness	4	6.423	0.803	100
(PCE)				
Perceived Healthcare Globalisation	4	6.475	0.808	100
(PHG)				
Perceived Usefulness (PU)	4	6.370	0.922	100
Perceived Ease of Use (PEU)	4	6.198	0.906	100
Attitude Towards Using (AU)	4	6.458	0.722	100
Intention to Use (IU)	4	6.373	0.747	100
1	1		1	

Table 6.10: HC-VBE-TAM-provider construct characteristics results which includes Mean and Standard Deviation

6.6.1.4 Data Validity and Reliability

Sections 4.3.8.(6,7,8 and 9) have already explained the importance of measurement validity and reliability as well as the most common validity and reliability statistical methods. Here the statistical data validity and reliability values computed for the data collected from healthcare providers through the survey questionnaire designed for HC-VBE-TAM-Provider are presented. The acceptable value for *AVE* is 0.5 or above and the values computed for all constructs in this part of the study are above the cut-off value ranging from 0.596 to 0.702. These results suggest that the amount of variance produced as a result of correct data collected for each construct is much higher than the amount of variance which may have been produced as a result of errors in the data collected. The suggestion supports strong data reliability which is vital for validating the technology acceptance claims made with regard to the HC-VBE framework developed in this thesis.

CR acceptable value is 0.7 or above, the result for all constructs are above the value starting from 0.855. The acceptable value for Cronbach's Alpha is 0.7 again the result for all constructs are above the value starting at 0.856 and the overall Cronbach's Alpha value is 0.959. These results suggest strong data validity and reliability. Kaiser-Meyer-Olkin Measure is used to measure sampling adequacy and the computed value is 0.919 which indicates that the sample used in the study is adequate as values above 8 is considered excellent [242]. The adequacy of the sample data is important to be proven statistically which in turn increases the findings made as result of analysing the sample data.

The Individual construct factor loading indicates strong connections between the construct and its measured questionnaire item as all values are above 0.7 which is much higher than the 0.5 acceptable value. The correlations between the constructs in the model as values start with the correlation between PEU and PU being the lowest (0.488) which is higher than the 0.3 acceptable value. The strong correlation between the constructs supports the validity of the model which in turn support the empirical conclusions made with regard to the acceptability of the HC-VBE framework. Table 6.11 shows the summary of results obtained for collected data reliability and validity.

Table 6.11: HC-VBE-TAM-provider	Data validity and reliability results
---------------------------------	---------------------------------------

Latent Variables	AVE >0.5	CR>0.7	Cronbach's Alpha> 0.7	Factor Loading	PCE	DHG	PU	PEU	AU	Ð	Whole Cronbach's Alpha> 0.7	Kaiser-Meyer-Olkin Measure
Perceived Clinical	0.6	0.8	0.8	0.9	1						0.9	0.9
Effectiveness (PCE)	64	87	84	10							59	19
Perceived	0.6	0.8	0.8	0.8	0.7	1						
Healthcare	36	75	73	66	72							
Globalisation (PHG)												
Perceived	0.7	0.9	0.9	0.8	0.8	0.7	1					
Usefulness (PU)	02	04	04	88	18	73						
Perceived Ease of	0.6	0.8	0.8	0.7	0.5	0.5	0.4	1				
Use (PEU)	41	77	72	16	99	33	88					
Attitude Towards	0.5	0.8	0.8	0.9	0.7	0.7	0.7	0.5	1			
Using (AU)	96	55	56	04	75	52	80	59				
Intention to Use (IU)	0.6	0.8	0.8	0.7	0.5	0.4	0.5	0.5	0.6	1		
	43	77	74	50	92	99	67	19	62			

6.6.1.5 HC-VBE-TAM-Provider Model Fit

Model fit values indicate that the results generated from the model can be reproduced and the model is statistically acceptable. In Section 4.3.8.10 the most recommended fit indices are described which are Chi Square (CMIN in AMOS), CFI, TLI, IFI, RMSEA and SRMR [214][215][213]. The model fit indices computed for the HC-VBE-TAM-Provider are summarised in Table 6.12. The computed Chi Square value for the model is 1.460 which is within the recommend value of 1-3. Chi Square is a measure of badness of fit, a value that is not significant (p>0.05) indicates that the model is acceptable. This means that the covariance matrix calculated based on the collected data is similar to the predicted covariance matrix by the model. In this case the Chi Square value suggest that the model designed to test the acceptability of the framework is acceptable and is fit for the intended purpose

CFI compares the fit value of the model with a fit value of a base model (alternative model) with value closer to 1 indicates an acceptable fit. The CFI value computed is 0.914 which has exceeded the recommended 0.90 cut-of value and hence indicates a good fit. These results suggest that the model used to test the acceptability of the framework is closer to ideal, hence indicate the fitness of the model. TLI and IFI are both comparative fit indices similar to CFI with computed values of 0.903 and 0.916 respectively which are above the 0.9 cut off value. The model fit results indicate a good fit with the data and therefore, the model is acceptable.

RMSEA calculates the difference between covariance matrix value for the same observed item in both the tested and predicted model. The recommended value is 0.08 or smaller, the value computed for the HC-VBE-TAM-Provider is 0.08 exactly. The final fit measure is SRMR which measures the standardised difference between observed and predicted covariance matrix with a recommended value of 0.08 or smaller. The computed SRMR value for the model specified is 0.063 which is smaller than 0.08 and this result again indicates a good model fit.

The model fit values collectively support the fitness of the model to test the acceptability of the framework by prospective users and the results produced from it can be relied on. All the results so far presented were computed to show whether the results produced in testing the hypothesises are statistically sound or not. Collectively the results suggest the hypothesis results are statistically sound. The results of the hypotheses testing are presented in Section 6.7. Table 6.12: HC-VBE-TAM-provider model fit results

Overall Fit		Computed	Recommended Value
Index		Value	
CMIN/DF		1.460	Between 1 and 3
CFI	Comparative fit index (baseline comparison)	0.914	CFI ≥ 0.90
ТЦ	Tucker Lewis Index (baseline comparison)	0.903	TLI ≥ 0.90
IFI	Incremental fit index (base line comparison)	0.916	IFI ≥ 0.90
RMSEA	Root Mean Square Error of Approximation (Absolute fit indices)	0.080	RMSEA ≤ 0.08
SRMR	(Standardized) Root Mean Square Residual	0.063	SRMR ≤ 0.08

6.6.2 HC-VBE-Requester Evaluation Results

This section presents healthcare requesters acceptance evaluation results and discussions for the HC-VBE-F. The results produced using the data collected during the survey and analysed using Structural Equation Modelling in AMOS software. The SEM model developed based on the HC-VBE-TAM-Requester is pictured in Figure 4.17 and described in Section 4.3.7.3.

6.6.2.1 Overall Data Analysis Results

This section reports on Mean, Standard deviation, Factor loading, Significance value and Intermeasurement Correlation values which are computed for the data collected using the questionnaire designed for healthcare requesters and can be found in appendix C. Mean is the average of all values given to an observed item by surveyed participants. The Mean results for the observed items are between 5.02 and 5.63 which indicate agreeability of participants with the measuring statements in the questionnaire. The results indicate that the concepts developed in the HC-VBE-F and demonstrated through the developed mobile application prototype are acceptable by healthcare requesters.

Standard deviation measures the deviation of answers from the mean and the smaller the better. The standard deviation values computed are between 1.059 and 1.478 which indicates that the answers given by research participants were close to each other since the standard deviation values are not too large. This result shows that research participants views about the concepts behind the mobile application protype were fairly consistent. The result of individual item factor loading indicates strong link between observed items and their unobserved variable with values between 0.751 and 0.907 which are all above the acceptable cut-off value 0.5 [241]. Significance p value is computed to measure relations between the unobserved variables and their observed items.

Significance test is necessary in SEM to ensure that the result is not produced by chance and it is widely accepted that the acceptable significance value is when (p<0.05) [205]. Significance computed values are indicated as *** in AMOS analysis results output which indicates that p<0.001. The result of all items is significant which suggests a high probability that the results are not produced by chance. Table 6.13 summarises the overall data analysis results for the HC-VBE-F acceptability evaluation from healthcare requesters perspective. The table shows statistical results for each questionnaire item within the context of their unobserved variables. Table 6.13: HC-VBE-TAM-Requester overall analysis results which includes Mean, Standard deviation, Factor loading and Significance value

Unobserved	Observed Variable	Mean	Std.	Factor	<i>p</i> < 0.05
Variable			Deviation	Loading	(Significance)
Perceived	PHA1	5.12	1.335	0.818	<i>p</i> <0.001
healthcare	Using the HC-VBE a	applicatio	on facilitate fa	ir access to	
availability	healthcare.				
(PHA)	PHA2	5.28	1.478	0.751	<i>p</i> <0.001
	Using the HC-VBE ap	plication	provides anytir	ne anywhere	
	access to healthcare.				
	РНАЗ	5.13	1.338	0.858	<i>p</i> <0.001
	Using the HC-VBE a	pplicatio	n makes more	varieties of	
	healthcare available.				
	PHA4	5.29	1.409	0.863	<i>p</i> <0.001
	Using the HC-VBE	applicati	on makes mo	ore specialist	
	healthcare provider av	ailable			
Perceived	PHQ1	5.12	1.437	0.889	<i>p</i> <0.001
Healthcare	Using the HC-VBE helps	s me find	the right health	care provider.	
Quality (PHQ)	PHQ2	5.21	1.336	0.884	<i>p</i> <0.001
	Using the HC-VBE a	oplicatior	helps me fin	d healthcare	
	respectful of and respo	onsive to	my preferences	and needs.	
	PHQ3	5.24	1.357	0.802	<i>p</i> <0.001
	Using the HC-VBE a	pplicatio	n helps me fi	nd the right	
	healthcare for reasona	ble price			
	PHQ4	5.33	1.393	0.875	<i>p</i> <0.001
	Using the HC-VBE a	pplicatio	n helps me fi	nd the right	
	healthcare in a reasona	able time			
	PHQ5	5.42	1.319	0.818	<i>p</i> <0.001
	Using the HC-VBE a	pplicatio	n helps me fi	nd the right	
	healthcare from a loca	tion of m	y choice.		

Unobserved	Observed Variable	Mean	Std.	Factor	<i>p</i> < 0.05
Variable			Deviation	Loading	(Significance)
Perceived	PU1	5.21	1.274	0.777	<i>p</i> <0.001
Usefulness	Using the HC-VBE app	nce to get on			
(PU)	demand healthcare.				
	PU2	5.51	1.367	0.832	<i>p</i> <0.001
	Using the HC-VBE appli	ication sa	ves me effort an	d time to find	
	needed healthcare.				
	PU3	5.02	1.378	0.803	<i>p</i> <0.001
	Using the HC-VBE appli	ication we	ould be more co	nvenient than	
	visiting healthcare inst	itutions.			
	PU4	5.23	1.325	0.822	<i>p</i> <0.001
	Overall, I find the H	C-VBE ap	plication useful	in receiving	
	healthcare.				
Perceived	PEU1	5.18	1.466	0.867	<i>p</i> <0.001
ease of use	Learning to use the HC	-VBE app	lication would b	e easy for me.	
(PEU)	PEU2	5.22	1.323	0.807	<i>p</i> <0.001
	I would find it easy	to rece	eive care using	the HC-VBE	
	application.				
	PEU3	5.47	1.359	0.846	<i>p</i> <0.001
	It would be easy for me	e to beco	me skilful at usir	ng the HC-VBE	
	application.				
	PEU4	5.30	1.382	0.893	<i>p</i> <0.001
	I would find the HC-VB	E applica	tion easy to use.		
Attitude	AU1	5.63	1.338	0.907	<i>p</i> <0.001
towards using	Using the HC-VBE appl	ication w	ould be a good i	dea.	
(AU)	AU2	5.31	1.361	0.836	<i>p</i> <0.001
	Using the HC-VBE ap				
	more interesting.				
	AU3	5.34	1.157	0.812	<i>p</i> <0.001
	Using the HC-VBE appli	cation wo	ould be a pleasai	nt experience.	

Unobserved	Observed Variable	Mean	Std.	Factor	<i>p</i> < 0.05		
Variable			Deviation	Loading	(Significance)		
	AU4	5.59	1.164	0.794	<i>p</i> <0.001		
	I would like to use the	HC-VBE a	pplication	1			
Intention to	IU1	5.49	1.040	0.833	<i>p</i> <0.001		
use (IU)	l intend to use the HC-	VBE appli	cation.	1			
	IU2	5.36	1.142	0.864	<i>p</i> <0.001		
	It is likely that I will use	the HC-\	VBE application.	1			
	IU3	<i>p</i> <0.001					
	IU4	5.63	1.143	0.839	<i>p</i> <0.001		
	1						

6.6.2.2 Intermeasurement Correlation

Intermeasurement correlation shows the relationship between all observed items, Table 6.14 shows the analysis result of intermeasurement correlations for the questionnaire items examining the HC-VBE-TAM-Requester. There are 25 measurement items in the questionnaire. Each questionnaire item is correlated to all other questionnaire items in a questionnaire including itself. A perfect correlation value which is 1 is produced when a questionnaire item is correlated to itself, but the values would be less than 1 in all other cases. The intermeasurement results are between 0.30 and 0.82 which are within or above the acceptable cut-off point of 0.3. The results indicate strong correlation between them, which means, research participants shared similar opinions on measurement items in a questionnaire. If the majority of correlation values are above 0.3 that indicates that the questionnaire items have adequately captured users view of the technology being tested in this context.

	PHA1	PHA2	PHA3	PHA4	PU1	PU2	PU3	PU4	PEU1	PEU2	PEU3	PEU4	PHQ1	PHQ2	PHQ3	PHQ4	PHQ5	AU1	AU2	AU3	AU4	IU1	IU2	IU3	1U4
PHA1	1.00																								
PHA2	0.63	1.00																							
PHA3	0.72	0.66	1.00																						
PHA4	0.68	0.70	0.76	1.00																					
PU1	0.62	0.50	0.58	0.61	1.00																				
PU2	0.65	0.66	0.64	0.71	0.76	1.00																			
PU3	0.60	0.48	0.68	0.64	0.62	0.67	1.00																		
PU4	0.61	0.49	0.64	0.71	0.63	0.63	0.72	1.00																	
PEU1	0.52	0.48	0.52	0.51	0.53	0.55	0.53	0.45	1.00																
PEU2	0.63	0.64	0.68	0.59	0.61	0.73	0.56	0.57	0.69	1.00															
PEU3	0.54	0.55	0.56	0.60	0.54	0.58	0.48	0.62	0.72	0.68	1.00														
PEU4	0.54	0.43	0.53	0.55	0.52	0.57	0.52	0.53	0.82	0.68	0.75	1.00													
PHQ1	0.65	0.56	0.67	0.62	0.65	0.70	0.63	0.66	0.51	0.75	0.59	0.54	1.00												
PHQ2	0.65	0.50	0.60	0.61	0.70	0.70	0.62	0.67	0.47	0.68	0.55	0.53	0.82	1.00											
PHQ3	0.58	0.57	0.64	0.59	0.49	0.58	0.50	0.62	0.50	0.65	0.56	0.51	0.68	0.66	1.00	1									
PHQ4	0.67	0.53	0.65	0.62	0.61	0.65	0.54	0.59	0.58	0.67	0.65	0.66	0.75	0.79	0.74	1.00									
PHQ5	0.61	0.57	0.64	0.62	0.54	0.63	0.52	0.57	0.59	0.70	0.63	0.57	0.71	0.69	0.74	0.72	1.00								
AU1	0.70	0.59	0.72	0.72	0.56	0.62	0.71	0.70	0.64	0.70	0.70	0.69	0.72	0.65	0.64	0.74	0.66	1.00							
AU2	0.62	0.50	0.64	0.63	0.55	0.59	0.62	0.73	0.50	0.59	0.64	0.58	0.72	0.62	0.60	0.69	0.59	0.76	1.00						
AU3	0.61	0.45	0.67	0.68	0.58	0.62	0.66	0.63	0.55	0.58	0.57	0.57	0.66	0.61	0.54	0.66	0.62	0.74	0.73	1.00					
AU4	0.45	0.47	0.53	0.56	0.48	0.53	0.57	0.61	0.55	0.57	0.63	0.64	0.60	0.58	0.56	0.63	0.51	0.73	0.69	0.58	1.00				
IU1	0.51	0.40	0.46	0.53	0.44	0.44	0.51	0.50	0.55	0.54	0.49	0.54	0.57	0.49	0.45	0.48	0.45	0.62	0.51	0.56	0.63	1.00			
1U2	0.30	0.32	0.29	0.44	0.46	0.46	0.46	0.47	0.46	0.44	0.46	0.46	0.56	0.51	0.41	0.43	0.40	0.52	0.50	0.50	0.64	0.72	1.00		
IU3	0.36	0.39	0.37	0.48	0.44	0.44	0.41	0.42	0.49	0.50	0.40	0.48	0.51	0.52	0.41	0.44	0.41	0.58	0.44	0.48	0.66	0.69	0.79	1.00	
1U4	0.39	0.41	0.41	0.53	0.48	0.60	0.52	0.54	0.53	0.57	0.46	0.60	0.62	0.60	0.47	0.51	0.52	0.60	0.50	0.56	0.67	0.71	0.71	0.71	1.00

Table 6.14: HC-VBE-TAM-Requester intermeasurement correlations

6.6.2.3 Characteristics of Construct

At construct level, the results suggest strong acceptability by healthcare requesters for the concepts developed in the HC-VBE-F. The mean value for each construct variable computed as an average value for all observed item of a particular construct is above 5 with standard deviation between 0.97 and 1.23. Table 6.15 shows the result for all constructs tested. The result suggests that on construct level (each construct tests an aspect of the technology being tested) the survey participants' acceptance attitude towards the HC-VBE-F were positive since the values are above 5 which are equals to "Agree" in Likert-Style measures. The result also suggests that on construct level the answers were similar since the standard deviation values are not too large.

Item Statistics										
Constructs	Observed	Mean	Std. Deviation	N						
	Items									
Perceived	4	5.21	1.22	100						
Healthcare Availability (PHA)										
Perceived	5	5.26	1.21	100						
Healthcare Quality (PHQ)										
Perceived	4	5.24	1.16	100						
Usefulness (PU)										
Perceived	4	5.29	1.23	100						
Ease of Use (PEU)										
Attitude Towards Using (AU)	4	5.47	1.11	100						
Intention to Use (IU)	4	5.50	0.97	100						

Table 6.15: HC-VBE-TAM-Requester construct characteristics results

6.6.2.4 Data Validity and Reliability

In Sections 4.3.8.(6,7,8 and 9) the importance of measurement validity and reliability as well as the most common statistical methods are already explained. Here, the statistical data validity and reliability values computed for the data collected from healthcare requesters through the survey questionnaire designed for HC-VBE-TAM-Requester are presented. The acceptable value for *AVE* is 0.5 or above and the value computed for all constructs in this part of the study are above the cut off value starting from 0.679. These results suggest that the amount of variance produced as a result of correct data collected for each construct is much higher than the amount of variance which may have been produced as a result of errors in the data collected. The suggestion supports strong data reliability which is vital for validating the technology acceptance claims made with regard to the modelling framework developed in this thesis.

CR acceptable value is 0.7 or above, the result for all constructs are above the value starting from 0.894. The acceptable value for Cronbach's Alpha is 0.7 again the result for all constructs are above the value starting at 0.890 and the overall Cronbach's Alpha value is 0.972. These results suggest strong data validity and reliability. To ensure sampling adequacy Kaiser-Meyer-Olkin Measure of sampling adequacy was computed and the result is 0.939 which indicates that the sample used in the study is adequate as values above 8 is considered excellent [242]. The adequacy of the sample data is important to be proven statistically which in turn increases the findings made as result of analysing the sample data.

The Individual construct factor loading indicate strong connections between the construct and its measured questionnaire item as all values are above 0.7 is much higher than the 0.5 acceptable value. The correlations between the constructs in the model as values start with the correlation between IU and PHA being the lowest (0.526) which is much higher than the 0.3 acceptable value. The strong correlation between the constructs supports the validity of the model which in turn support the empirical conclusions made with regard to the acceptability of the modelling framework. Table 6.16 shows the summary of results obtained for the collected data reliability and validity.

Latent Variables	AVE >0.5	CR>0.7	Cronbach's Alpha> 0.7	Factor Loading	РНА	рнд	PU	PEU	AU	I	Whole Cronbach's Alpha> 0.7	Kaiser-Meyer-Olkin Measure
Perceived Healthcare	0.6	0.8	0.8	0.8	1						0.9	0.9
Availability (PHA)	79	94	98	77							72	39
Perceived Healthcare	0.7	0.9	0.9	0.9	0.7	1						
Quality (PHQ)	17	27	31	07	81							
Perceived Usefulness	0.7	0.9	0.8	0.9	0.8	0.7	1					
(PU)	02	04	90	04	08	94						
Perceived Ease of Use	0.6	0.8	0.9	0.8	0.7	0.7	0.7	1				
(PEU)	87	98	13	70	07	51	18					
Attitude Towards Using	0.7	0.9	0.9	0.9	0.7	0.8	0.8	0.7	1			
(AU)	19	11	04	26	73	07	00	72				
Intention to Use (IU)	0.7	0.9	0.9	0.7	0.5	0.6	0.6	0.6	0.7	1		
	15	09	11	70	26	20	16	28	10			

Table 6.16: HC-VBE-TAM-requester data reliability and validity results

6.6.2.5 HC-VBE-TAM-Requester Model Fit

Model fit values indicate that the results generated from the model can be reproduced and the model is statistically acceptable. As stated already the most recommended fit indices are Chi Square (CMIN in AMOS), CFI, TLI, IFI, RMSEA [214] [215] [213]. The model fit indices computed for the HC-VBE-TAM-Requester are summarised in Table 6.17. Chi Square is a measure of badness of fit, a value that is not significant (*p*>0.05) indicates model acceptability. The recommended Chi Square value is between 1 and 3 for a model to be acceptable; the computed Chi Square value for the model is 1.774 which indicates that the model and the data are a good fit. This means that the covariance matrix calculated based on the collected data is similar to the predicted covariance matrix by the model. CFI compares the fit value of the model with a fit value of a base model (alternative model) with value closer to 1 indicates an acceptable fit. In this case the Chi Square value suggest that the model designed to test the acceptability of the framework is acceptable and is fit for the intended purpose.

The CFI value computed is 0.911 which has exceeded the recommended 0.90 cut-of value and hence indicates a good fit. These results suggest that the model used to test the acceptability of the framework is closer to ideal, hence indicate the fitness of the model. TLI and IFI are both comparative fit indices similar to CFI with computed values of 0.900 and 0.913 respectively which are above the values are within the 0.9 cut-off value. Both results indicate that the model fits wells with the data and it is acceptable. RMSEA calculates the difference between covariance matrix value for the same observed item in both the tested and predicted model. The recommended value is 0.08 or smaller, the value computed for the HC-VBE-TAM-Provider is 0.088.

The final fit measure is SRMR which measures the standardised difference between measured and predicted covariance matrix with a recommended value is 0.08 or smaller. The computed SRMR value for the model specified is 0.064 which is smaller than 0.08 which indicates that the model is a good fit.

The model fit values collectively support the fitness of the model to test the acceptability of the framework by prospective users and the results produced from it can be relied on. All the results for far presented were computed to show whether the results produced in testing the hypothesises are statistically sound or not. Collectively the results suggest the hypothesis results are statistically sound. The results of the hypotheses testing are presented in the next section.

Table 6.17: HC-VBE-TAM-Requester model fit results	
I	

Overall Fit		Computed	Recommended
Index		Value	Value
CMIN/DF		1.774	Between 1 and 3
CFI	Comparative fit index (baseline comparison)	0.911	CFI ≥ 0.90
ти	Tucker Lewis Index (baseline comparison)	0.900	TLI ≥ 0.90
IFI	Incremental fit index (base line comparison)	0.913	IFI ≥ 0.90
RMSEA	Root Mean Square Error of Approximation (Absolute fit indices)	0.088	RMSEA ≤ 0.08
SRMR	(Standardized) Root Mean Square Residual	0.064	SRMR ≤ 0.08

6.7 HC-VBE-F Discussions

Governments and healthcare providers are in continuous search to find new technologybased platforms to provide on time and affordable healthcare [290]. Healthcare is complex and there is always a high demand for up-to-date knowledge and skills in the field. To cope with the demand, healthcare professionals and institutions are encouraged to collaborate and share resources, and for this, virtual care technologies are gaining importance [291]. Forms of virtual care have existed for some times as telemedicine and telehealth. However, the overall concept of virtual care which is defined as *"the* convergence of digital media, health technology, mobile devices, text messaging, digital voice assistants, and decision support tools powered by artificial intelligence and augmented/virtual reality to create a continuous connection between patients, physicians, and other caregivers" is yet to be realised in a manageable and organised settings [23].

In Section 1.3, three research questions were stated to be answered for healthcare virtual collaboration; here the researched answers to each of the questions are discussed.

RQ3: How to manage and organise virtual collaboration for healthcare?

To answer this question a healthcare virtual collaboration framework (HC-VBE-F) is developed based on the modelling framework described in Chapter 5. The framework facilitates both the organisation and management of virtual collaboration for healthcare. The HC-VBE-F consists of a number of components each having a specific function in organising and managing virtual collaboration in healthcare. The framework working steps are clearly outlined and explained. To demonstrate the feasibility of the concepts developed in the framework, a mobile application prototype is developed that implements the main functionalities and working steps in the framework. The results of the acceptance evaluation show that the framework concepts and mechanisms have passed the user acceptance test which is a vital first step in developing a virtual collaboration system for healthcare.

RQ4: How to select, verify and validate virtual healthcare providers?

To answer the first part of the question, with regard to provider selection, a member selection mechanism is developed and presented in Section 6.3. Ensuring the right member is selected to join a given healthcare virtual collaboration, increases trust in both the service provided and between the members that collaborate. As for the second part of the question, a user validation and verification mechanism which is described in Section 6.4 is developed. The mechanism uses blockchain technology as a theoretical base to verify and validate healthcare providers that wish to provide a service in HC-VBE collaborations. The mechanism provides a detailed step by step guide as to how

healthcare service providers are going to be verified and validated and it is designed to increase trust between participants of a given virtual collaboration. The mechanism is conceptual and the author has not attempted to provide any technical implementation details as it requires a separate research, however the backbone of the mechanism is blockchain technology which has already been implemented for various purposes such as crypto currency.

RQ5: How to regulate virtual collaboration for healthcare?

To answer RQ5, an SLA management mechanism is developed. The mechanism manages the life-cycle of SLA which is formation, deployment, enforcement and termination [87] and it is described in Section 6.2. The mechanism is specific to VBE and VO based healthcare virtual collaboration. Having an electronic contract between collaborating parties in virtual collaboration is one of the methods suggested by researchers to address the issue of trust.

To facilitate the implementation of the SLA mechanism, a set of SLA templates are defined in Section 6.5 that embodies the necessary details for contracts to be approved by parties in a given virtual collaboration. The templates are text-based human readable which are designed to increase transparency and separation of rights and responsibilities of the contract approvers. A mechanism is also suggested to convert the text-based contracts into machine readable OWL. The basic concepts of the templates are implemented as a mobile application prototype and are evaluated by prospective users. The results show that prospective users believe that the templates can contribute to the formalisation and regulation of SLA in healthcare virtual collaboration.

To examine the user acceptance of the concepts developed in the HC-VBE-F, it was implemented in a form of a mobile application prototype and examined using Technology Acceptance Model [34]. One of the mechanisms to investigate the suitability of a technology in a sector is to examine users' acceptance of the technology. The importance of studying the acceptance of users towards a healthcare information technology has been recognised by researchers [101].
TAM is one of the most widely used models used to examine users' acceptance of a new technology. In evaluating the acceptability of the HC-VBE-F, in Section 4.3.2, the original TAM is extended with four new constructs to develop two new TAMs (HC-VBE-TAM-Provider and HC-VBE-TAM-Requester). The results of the acceptance examinations are discussed in the next two sections.

This study also empirically evaluated four new technology acceptance factors specific to healthcare virtual collaboration system which are PCE, PHG, PHQ and PHA. The factors have not been evaluated in previous studies in the context of healthcare information technology systems and statistical results show that they are significant healthcare technology acceptance indicators. This result is an important contribution to the list of latent variables that have been validated previously through extending TAM for healthcare information technology systems.

6.7.1 HC-VBE-TAM-Provider Discussion

To investigate the healthcare providers acceptance of technologies developed in the HC-VBE-F for healthcare virtual collaboration based on the HC-VBE-TAM-Provider, nine hypotheses were defined in Section 4.3.2.2. Here, the result of testing those hypotheses based on the data analysis carried out in AMOS are presented and discussed. The authors of [109] state that significant path coefficient is a good tool to empirically accept or reject a hypothesis in TAMs and the recommend acceptance statistical significance value is equal or less than 0.05.

The evaluation results show that H₁a is accepted with a significance *p* value of less than 0.001 indicated by *** in AMOS; which means PHG has a significant effect on PCE. The result provides empirical proof that research participants believe a globally accessed healthcare virtual collaboration system will affect positively on its clinical effectiveness. A possible explanation for this could be the fact that a system that allows healthcare providers around the world to provide care will have more resources to be effective and provide a timely service. H₁b which hypothesises the causal effect of PHG on PEU is also

accepted indicated by a significance value of less than 0.001. It means that healthcare providers think that a healthcare system which can be accessed globally will be easier to use. This is because "anytime anywhere" attribute of a system that can be used on a timely and on demand base increases system availability which makes it easier to use.

 H_1c which hypothesises the causal effect of PHG on PU is rejected with a significance value of 0.390 which is more than 0.05. The result contradicts with expectation since one would expect that if a healthcare system can be used globally it would be more useful as more people can benefit from using it. H₂ which hypothesises the causal effect of PCE on PU is accepted with a significance value of less than 0.001. This result indicates that healthcare providers believe that clinical effectiveness of a virtual healthcare system will influence its usefulness. The causal effect of PEU on PU hypothesised in H₃a is rejected with a significance value of 0.232; which suggests that healthcare provides don't think that if a system is easy to use it means it is also useful. H_3b hypothesises the causal effect of PU on AU is accepted with a significance value of less than 0.001. The result suggests that healthcare providers believe that the usefulness of a virtual healthcare system will influence their attitude towards using that system. H₄a is accepted as the computed significance value is 0.004. H₄a hypothesises the causal effect of PEU on AU; which indicates that healthcare providers attitude towards using healthcare systems developed based on the HC-VBE framework is affected by how easy they think the system is to use. The causal effect of AU on IU is accepted by healthcare providers which is hypothesised in H₄b. The acceptance of this hypothesis indicates that healthcare providers intention to use a virtual healthcare system developed based on the HC-VBE concept will lead them to heightened intention to use it. The causal effect of PU on IU which is hypothesised in H_5 is rejected with a significance value of 0.379. This result is also one of the unexpected ones since if a system is useful it would be expected to be used.

Overall, these results show that the participants were positive about the contribution that the HC-VBE-F can make to virtual collaboration for healthcare. Table 6.18 summarises the hypotheses test results and Figure 6.25 shows the accepted causal paths indicated by bold lines.

- 216

		Sig.Level	Acceptance		
Hypotheses	Evalua	ted Caus	al Effect	at <i>p</i> <0.05	
H ₁ a	Perceived clinical	<	Perceived	<i>p</i> <0.001	Accepted
	effectiveness		healthcare		
			globalisation		
H ₁ b	Perceived ease of	<	Perceived	<i>p</i> <0.001	Accepted
	use		healthcare		
			globalisation		
H1C	Perceived	<	Perceived	0.390	Not Accepted
	usefulness		healthcare		
			globalisation		
H ₂	Perceived	<	Perceived clinical	<i>p</i> <0.001	Accepted
	usefulness		effectiveness		
Н₃а	Perceived	<	Perceived ease of	0.232	Not Accepted
	usefulness		use		
H₃b	Attitude towards	<	Perceived	<i>p</i> <0.001	Accepted
	using		usefulness		
H₄a	Attitude towards	<	Perceived ease of	0.004	Accepted
	using		use		
H₄b	Intention to use	<	Attitude towards	0.002	Accepted
			using		
H₅	Intention to use	<	Perceived	0.379	Not Accepted
			usefulness		

Table 6.18: HC-VBE-TAM-provider hypotheses testing results



Figure 6.25: HC-VBE-TAM-provider accepted hypotheses results with bold lines indicating the accepted hypotheses.

6.7.2 HC-VBE-TAM-Requester Discussion

To investigate the healthcare requesters acceptance of technologies developed in the HC-VBE-F for healthcare virtual collaboration, based on the HC-VBE-TAM-Requester, nine hypotheses were defined in Section 4.3.2.3. Here, the result of testing those hypotheses are discussed based on the data analysis carried out in AMOS software.

The evaluation results show that H₁a is accepted with a significance value of less than 0.001 indicated by *** in AMOS; which means PHA has a significant effect on PHQ. The result indicates that healthcare requesters believe that a virtual healthcare system availability will influence the quality of services provided through it. H₁b which hypothesises the causal effect of PHA on PEU is accepted which is indicated by a significance value of less than 0.001. This means that healthcare requesters perceive that the availability of a virtual healthcare system will influence its easiness to use. H₁c which hypothesises the causal effect of PHA on PU is accepted with a significance value of less than 0.001. This means that healthcare requesters perceive that the availability of a virtual healthcare system will influence its easiness to use. H₁c which hypothesises the causal effect of PHA on PU is accepted with a significance value of less than 0.001. This is accepted with a significance value of less than 0.001. This with a virtual healthcare system that is available to be used when needed will heighten its usefulness.

H₂ which hypothesises the causal effect of PHQ on PU is accepted with a significance value of 0.014. This result indicates that healthcare requesters believe that the quality of

services provided in applications developed based on the HC-VBE-F will be a contributing factor in its usefulness. The causal effect of PEU on PU hypothesised in H₃a is rejected with a significance value of 0.737; which suggests that healthcare requesters don't believe that if a system is easy to use means it is also useful. H₃b hypothesises the causal effect of PEU on AU is accepted with a significance value of less than 0.001. The result suggests that healthcare requesters believe that the usefulness of a virtual healthcare system will influence their attitude towards using that system.

H₄a is accepted as the computed significance value is less than 0.001. H₄a hypothesises the causal effect of PU on AU; which indicates that healthcare requesters attitude towards using healthcare systems developed based on the HC-VBE framework is affected by how useful they think the system is. The causal effect of PU on IU which is hypothesised in H₄b is rejected with a significance value of 0.444. This result is unexpected since if a system is useful it would be expected to be used. The causal effect of AU on IU is accepted by healthcare requesters which is hypothesised in H₅. The acceptance of this hypothesis indicates that healthcare requesters intention to use a virtual healthcare system developed based on the HC-VBE concept will lead them to heightened intention to use it. Table 6.19 shows the hypotheses testing results and Figure 6.26 shows the accepted paths.

		Sig.Level	Acceptance		
Hypotheses	Evaluate	at <i>p</i> <0.05			
H ₁ a	Perceived <		Perceived healthcare	<i>p</i> <0.001	Accepted
	healthcare quality		availability		
H₁b	Perceived ease of	<	Perceived healthcare	<i>p</i> <0.001	Accepted
	use		availability		
H₁c	Perceived	<	Perceived healthcare	<i>p</i> <0.001	Accepted
	usefulness		availability		
H ₂	Perceived <		Perceived healthcare	0.014	Accepted
	usefulness		quality		

Table 6 10, UC VEE TAM rec	unostar hypothase	s tosting rocults
TADIE 0.19: HC-VBE-TAIVI-LEO	uester hypothese	es lesling results

				Sig.Level	Acceptance
Hypotheses	Evaluate	at <i>p</i> <0.05			
Н₃а	Perceived usefulness	<	Perceived ease of use	0.737	Not Accepted
H₃b	Attitude towards using	<	Perceived ease of use	<i>p</i> <0.001	Accepted
H₄a	Attitude towards using	<	Perceived usefulness	<i>p</i> <0.001	Accepted
H₄b	Intention to use	<	Perceived usefulness	0.444	Not Accepted
H₅	Intention to use	<	Attitude towards using	<i>p</i> <0.001	Accepted



Figure 6.26: HC-VBE-TAM-requester accepted hypotheses with bold lines indicating the accepted hypotheses

6.7.3 HC-VBE-TAM Providers and Requester Comparison

The mobile application prototype developed based on the HC-VBE-F was evaluated by both healthcare providers and requesters. Each was tested with a slightly different TAMs, however the framework they were examining was the same. Therefore, it is important to compare the hypotheses testing results for both groups and discuss similarities and differences between them.

In both cases the extended TAM was comprised of 6 constructs two of which were newly added to examine relevant acceptability factors. Since the newly added constructs are examining different factors from different perspectives (Healthcare providers and healthcare requesters) their results cannot be compared. However, the TAMs have four original constructs (PEU, PU, AU and IU) and five hypotheses (H₃a, H₃b, H₄a, H₄b and H₅) in common, which allows their results and findings to be compared and discussed.

Table 6.20 shows the results of the comparison and perhaps not surprisingly both healthcare providers and requesters are the same in terms of accepting and rejecting hypotheses. Both users have accepted hypotheses H₃b, H₄a, H₄b and have rejected hypotheses H₃a and H₅. From the comparison result one can conclude that the developed virtual collaboration framework for healthcare was received equally by both healthcare providers and receivers and their intension to use systems that are going to be developed based on the HC-VBE-F is positive.

Uunoth	Evoluated Causel Effect			Healthcare Providers		Healthcare		
пуротп	Evaluated Causal Effect			пуротнез	Hypotheses Result		Requesters	
eses				Hypotheses Result			es Result	
				Sig.Level	Acceptance	Sig.Level	Acceptance	
				at		at <i>p</i> <0.05		
				<i>p</i> <0.05				
H₃a	Perceived	<	Perceived	0.232	Not	0.737	Not	
	usefulness		ease of		Accepted		Accepted	
			use					
H₃b	Attitude	<	Perceived	<i>p</i> <0.001	Accepted	<i>p</i> <0.001	Accepted	
	towards		usefulness					
	using							

Table 6.20: Comparison between HC-VBE-TAM-provider and requester hypotheses testing results

	Evaluated Causal Effect			Healthcare Providers		Healthcare		
Hypoth				Hypotheses Result		Requesters		
eses						Hypothes	Hypotheses Result	
				Sig.Level	Acceptance	Sig.Level	Acceptance	
				at		at <i>p</i> <0.05		
				<i>p</i> <0.05				
H₄a	Attitude	<	Perceived	0.004	Accepted	<i>p</i> <0.001	Accepted	
	towards		ease of					
	using		use					
H₄b	Intention to	<	Attitude	0.002	Accepted	<i>p</i> <0.001	Accepted	
	use		towards					
			using					
H₅	Intention to	<	Perceived	0.379	Not	0.444	Not	
	use		usefulness		Accepted		Accepted	

6.7.4 HC-VBE Framework Limitations

This study was carried out on a local level which may not reflect the globalised view of the HC-VBE framework. The research would have carried more credit if participants were recruited from different regions and countries to provide feedback on the developed concept and prototype. However, because the framework needed to be presented and explained to prospective users before the prototype demonstration, it would have been difficult to gather and engage users beyond the local area. Another limitation is that the mobile application protype was developed in English but evaluated in a none-English speaking country, this posed a limitation on researchers to only invite those who had some basics of English. This may have affected the sample study representativeness of the local population negatively. To dilute the effect of this limitation the questionnaires used in the surveys were translated into the local language and the author answered questions and concerns during meetings held for data collection.

6.8 Chapter 6 Summary

This chapter presented a Healthcare Virtual Collaboration Framework (HC-VBE-F) that consists of a conceptual description, a member selection mechanism, a healthcare provider verification and validation mechanism and a set of SLA templates. The content of the chapter is summarised in the following points:

- 1- The HC-VBE-F is described on conceptual level which includes the components required to provide the use cases that facilitate the initiation, negotiation, creation, monitoring and archiving life-cycle of a given healthcare virtual collaboration. The framework working steps are explained in detail which spans over 16 steps.
- 2- A Member Selection Mechanism (HC-VBE-MSM) is developed that facilitates the selection process of suitable healthcare virtual collaboration participants. The mechanism is made up of a seven steps process. The applicability of the mechanism is demonstrated by applying it to a simple healthcare virtual collaboration case study.
- 3- A Healthcare Virtual Collaboration Provider Verification and Validation Mechanism (HC-VBE-PVVM) is developed using blockchain technology as a theoretical base. The mechanism presents a step by step guide as to how to verify and validate a healthcare provider who offers to provide a service to a requester. The applicability of the mechanism is demonstrated by applying it to a simple case study.
- 4- To standardise the terms and conditions of a healthcare virtual collaboration service and to regulate the rights and responsibilities of participants, a set of SLA templates are developed. The templates are text-based and human readable; designed to speed up the negotiation process between service requesters and service providers. The content of each clause of the SLA templates are explained in detail.
- 5- The HC-VBE-F evaluation results show that the framework is acceptable by healthcare service providers and requesters and their intension to use systems developed based on the framework is positive.

The research is concluded in the next chapter which contains research achievements and future works.

This chapter concludes the research in Section 7.1, outline the main achievements in Section 7.2, presents publications in Section 7.3 and presents future works in Section 7.4.

7.1 Conclusion

Literatures in the field of healthcare show that the demand for healthcare is rising and the current model of healthcare provision which relies on centralised healthcare organisations such as hospitals are not coping well. The demand for healthcare is partly because of the rinse in population numbers and partly because of an aging population that is becoming a phenomenon in many countries such as Japan. The rise in demand has increased cost and affected resource availability in care institutions. To reduce cost and increase resource availability, virtual collaboration is seen as an answer for healthcare. However, the modelling, organisation and management of virtual collaboration for healthcare presents many challenges such as, role and service classification, member selection mechanisms and SLAs which are yet to be addressed.

The aim of this research was to model and develop virtual collaboration frameworks for healthcare. At the initial stage of the research, a number of challenges regarding the use of virtual collaboration in healthcare were presented from which 5 research questions were formulated. How to model and describe service provision in virtual collaboration for healthcare and how to organise and manage healthcare virtual collaboration are some of the questions researched in this thesis. System developers use models to simplify complex scenarios, in this thesis, a modelling framework to model healthcare virtual collaboration is developed. Based on the modelling framework, a healthcare virtual collaboration framework is developed to address the organisation and management aspects of virtual care provision.

Both frameworks are implemented and evaluated empirically using Technology Acceptance Modelling (TAM) as a theoretical evaluation framework and Structural Equation Modelling (SEM) is used as a statistical analysis technique to analyse the data collected during the evaluation process. AMOS is used as tool to compute statistical results based on the SEM models and the data collected for each of the developed frameworks.

The results of the evaluations show that both frameworks are received positively by prospective users (system developers in the case of the modelling framework and healthcare requesters and providers in the case of the collaboration framework). The contributions of this thesis are the modelling and collaboration frameworks for healthcare virtual collaboration. The main achievments of the thesis is summarised in the next section.

7.2 Research Achievements

The main achievements of this research can be summarised in the following points:

- A comprehensive literature review has been carried out for healthcare sector in general and as a result the main roles and services in healthcare have been classified. This is considered as a noticeable achievement because it simplifies the structure of healthcare to a point that specialists and non-specialists can understand it.
- 2. A simple and generic service and participant classification mechanism (SPCM) has been developed for VBE and VO based healthcare virtual collaboration. The mechanism provides a conceptual classification of roles and services in healthcare which facilitate in simplifying complex virtual collaboration scenarios for the sector. It also helps healthcare system developers and professionals to plan ahead and identify requirements for a given virtual collaboration.

- 3. To provide a modelling capability specific to healthcare virtual collaboration, UML use case notations are extended to develop a domain specific modelling language (DSML) using the profiling technique provided by Object Management Group. The DSML is simple to understand and use, which can be used to model all HC-VBE and HC-VO collaboration scenarios.
- 4. A healthcare virtual collaboration service orchestration mechanism that is capable of modelling service workflows has been developed. The orchestration mechanism covers the scope and processes needed to show the steps of healthcare service provision virtually. It consists of a textual description language which has been developed to describe the orchestration mechanism. The language has a sentence structure and a set of vocabularies to standardise the description.
- 5. A general conceptual healthcare virtual collaboration framework that describes how collaborations are organised and managed has been developed. The framework is described in detail in terms of the number of components that make up the framework and how it works step by step.
- 6. A healthcare virtual collaboration member selection mechanism has been developed which is simple and effective. The mechanism has been described in detail and a step by step selection process is presented. The mechanism is developed in line with the other part of the research for ease of integration later in one package.
- 7. A healthcare provider verification and validation mechanism based on blockchain technology has been developed. The mechanism is feasible and effective if implemented and has the potential to address the issue of trust between healthcare stakeholders which is one of the most outstanding issue in healthcare and it is yet to be completely addressed by researchers.
- 8. A set of healthcare virtual collaboration SLA templates have been developed to regulate contracts between participants during HC-VBE collaborations. The templates are simple and generic which are capable of documenting all possible virtual collaboration agreements. The templates provide a base for all SLAs that might be created during a

healthcare virtual collaboration setting. The templates facilitate faster agreements between the collaborating participants and they are presented with fine details and possible implementation mechanisms. The templates are both human and machine readable.

9- Technology Acceptance Model (TAM) is extended and validated with 5 new constructs. Perceived Ability to Model (PAM) was added to the original TAM to evaluate the modelling framework by system developers. To evaluate the HC-VBE-F, Perceived Healthcare Araciality (PHA) and Perceived Healthcare Quality (PHQ) were added to the original TAM to evaluate healthcare requesters acceptance. Perceived Clinical Effectiveness (PCE) and Perceived Healthcare Globalisation (PHG) were added to the original TAM to evaluate the HC-VBE-F by healthcare providers. The outcome of this part of the research contributes towards further expanding and validating the applicability of TAM to evaluate new healthcare technologies. The new constructs have not been used in previous studies as user acceptance factors. The new constructs further develop TAM as a theoretical evaluation framework for healthcare technologies which is an achievement.

7.3 Published and Planned Publications

1. Published papers

A. H. Mahmud, M. Mohammadi, N. Ali, T. A. R. Khan, N. K. Al-Salihi, R. M. D. Omer, and J. Lu, "Technologies in medical information processing," Advances in Telemedicine for Health Monitoring: Technologies, Design and Applications, p. 31, 2020.

I was the main author of this book chapter in which we present a number of technologies in use in the field of medical information processing. The chapter in general is relevant to this thesis but in Section 1.6 we present a specific framework that is based on VBE and VO concept for healthcare information interpretation.

B. H. Mahmud, J. Lu and Q. Xu, "A Blockchain-based Service Provider Validation and Verification Framework for Healthcare Virtual Organization," UHD Journal of Science and Technology, vol. 2, no. 2, pp. 24–31, 2018. I was the main author of this paper in which we have presented a healthcare service provider verification and validation framework for healthcare virtual collaboration. The paper is based on the framework described in Section 6.4.

C. H. Mahmud, "Modeling Virtual Organization for Home Healthcare Using UML," Int. J. Comput. Sci. Eng, vol. 4, pp. 22–31, 2016.

I was the sole author of this paper in which I explain the limitation of UML modelling language in modelling healthcare virtual collaboration. The paper evaluates UML, based on a number of modelling criterions and discuss UML limitations. The paper was written as part of justification for developing the DSML for healthcare virtual collaboration.

D. H. Mahmud and J. Lu, "A GENERIC VOBE FRAMEWORK TO MANAGE HOME HEALTHCARE COLLABORATION," *Journal of Theoretical and Applied Information Technology*, vol. 80, no. 2, p. 362, 2015.

I was the main author of this paper in which we present and describe a framework to classify roles and services for home healthcare based on VBE and VO concept. This paper is part of the SPCM described in Section 5.2.

2- Planed papers to be published in the future

- A. A Healthcare Virtual Collaboration Framework: Evaluating Users' Acceptance Using Technology Acceptance Model.
- **B.** A Generic SLA template for VBE- based Healthcare Virtual Collaborations.
- **C.** The Role of Robot in VBE-based Healthcare Virtual Collaborations.
- **D.** A Member Selection Mechanism for VBE-based healthcare Virtual Collaborations.
- E. An Orchestration Mechanism for VBE-based healthcare Virtual Collaborations.
- **F.** A Domain Specific Modelling Language for VBE-based healthcare Virtual collaborations
- G. An OWL-based Model for Healthcare Virtual Collaboration Service Level Agreement

7.4 Future Works

Healthcare virtual collaboration is multi-faceted and requires much more research to address challenges and issues in realising it. This section presents a number of future works that would extend the implications and contributions of this thesis.

1. Cultural differences during HC-VBE-based collaborations

Culture may be an influential factor in the usability and acceptability of virtual healthcare as people are directed to use an electronic medium to explain their health issues rather that a face to face method. In some cultures, due to religious beliefs or cultural boundaries it may not be acceptable for people to use an online virtual method to speak to someone they meet for the first time. The frameworks developed in this research encourage such encounters and healthcare providers can be recruited from anywhere in the world. It would be interesting and useful to study the influence of culture in healthcare virtual collaborations.

Language is another important factor in the success of systems developed based on the HC-VBE framework. Healthcare requesters and providers may not be able to communicate directly with each other but through an interpreter since they may be from different countries. Tackling this challenge would also be an interesting topic to research. Therefore, culture and language barriers are two new electronic healthcare system acceptability variables that can be investigated in future studies.

2. HC-VBEs visibility

The concepts presented in this thesis are global where many HC-VBEs can be created around the world to provide services, collaborate and share resources. The globalised concept of HC-VBE presents a visibility challenge. The question here is how to make HC-VBEs visible in order for them to be discoverable to each other. This is an area of virtual collaboration for healthcare that can be researched in the future.

3. Service orchestration simulation and validation

The service orchestration mechanism presented in this thesis is theoretical and text based with no simulation and validation mechanism. Simulation can provide the facility to visualise healthcare virtual collaboration service provision in action. Service orchestration validation is crucial to check the soundness of the flow of activities to achieve a goal before its implementation. Developing a mechanism to simulate and validate the service orchestration in healthcare is a future work that offers theoretical and technical challenges to future researchers.

4. Healthcare requester data security and privacy

The issue of data security and privacy of users is well recognised and the importance of the topic is well researched. However, the theoretical mechanism and the technical implementation varies depending on the purpose of a system. Healthcare virtual collaborations based on the HC-VBE-F presented in this thesis surely present its own unique security and privacy challenges which can be researched in the future.

5. SLA enforcement and monitoring mechanism

One of the most obvious challenge presented in the HC-VBE-F is to do with SLA monitoring. After an SLA is agreed between healthcare requesters and providers, a robust mechanism is required for the SLA enforcement and monitoring. The mechanism could involve artificial intelligence and smart decision support systems to cope with the dynamic nature of healthcare virtual collaboration. Developing a mechanism to carry out SLA enforcement and monitoring is a research challenge which can be researched in the future.

6. SLA dispute management

Service in HC-VBE based system are provided based on SLA which defines the terms and conditions of the service. The question is, if there is a dispute regarding the extend of observed terms and conditions by one of the signatory parties of the contract how the dispute is managed as communication is virtual. This suggests the need for a simple and effective dispute management mechanism which is a future research opportunity.

7. HC-BE-PVVM development and implementation

The healthcare provider verification and validation mechanism outlined in Section 6.4 is a conceptual one and requires further research and implementation. The mechanism is based on blockchain technology which is a fairly new technology that offers a number of conceptual and implementational challenges that can be researched in the future.

8. Member selection mechanism development and implementation

The HC-VBE-MSM described in Section 6.3 is a conceptual one that is described in an abstract level which may need further research and implementation to make it usable in real systems. Despite the fact that it has been described in a step by step process but it lacks implementation details which offers development challenges that can be addressed by further research in the future.

Bibliography

- B. M. Till, A. W. Peters, S. Afshar, and J. G. Meara, "From blockchain technology to global health equity: can cryptocurrencies finance universal health coverage?," *BMJ global health*, vol. 2, no. 4, p. e000570, 2017.
- [2] A. Renedo, C. A. Marston, D. Spyridonidis, and J. Barlow, "Patient and Public Involvement in Healthcare Quality Improvement: How organizations can help patients and professionals to collaborate," *Public Management Review*, vol. 17, no. 1, pp. 17–34, 2015.
- [3] S. Wiig, G. Robert, J. E. Anderson, E. Pietikainen, T. Reiman, L. Macchi, and K. Aase, "Applying different quality and safety models in healthcare improvement work: Boundary objects and system thinking," *Reliability Engineering & System Safety*, 2014.
- [4] A. O. Association and others, "National Strategy for Quality Improvement in Health Care," *Optometry-Journal of the American Optometric Association*, vol. 82, no. 7, pp. 441–443, 2011.
- [5] "health expenditure in US." [Online]. Available: https://www.cms.gov/Research-Statistics-Data-and-Systems/Statistics-Trends-and-Reports/NationalHealthExpendData/Downloads/proj2014.pdf. [Accessed: 01–09-2015].
- [6] B. K. R. L. Timothy G. Buchman, *Modelling in Healthcare*. Complex Systems Modelling Group (CSMG).
- [7] S. Arbaoui, N. Cislo, and N. Smith-Guerin, "Home healthcare process: Challenges and open issues," *arXiv preprint arXiv:1206.5430*, 2012.
- [8] D. Goldberg, "Can the NHS deliver integration? Lessons from around the world," 2014. [Online]. Available: https://www.good-governance.org.uk/wpcontent/uploads/2017/04/Goldberg-III-Can-the-NHS-deliver-integration-Lessonsfrom-around-the-world.pdf. [Accessed: 24-Apr-2019].
- [9] A. L. Menezes, C. E. Cirilo, J. L. De Moraes, W. L. De Souza, and A. F. do Prado, "Using archetypes and domain specific languages on development of ubiquitous applications to pervasive healthcare," in *Computer-Based Medical Systems* (CBMS), 2010 IEEE 23rd International Symposium on, 2010, pp. 395–400.

- [10] A. K. S. Kumar, L. C. Chen, M. Choudhury, S. Ganju, V. Mahajan, A. Sinha, and A. Sen, "Financing health care for all: challenges and opportunities.," *Lancet*, vol. 377, no. 9766, pp. 668–79, 2011.
- [11] J.-F. Levesque, M. F. Harris, and G. Russell, "Patient-centred access to health care: conceptualising access at the interface of health systems and populations.," Int J Equity Health, vol. 12, p. 18, 2013.
- [12] L. Huycke and A. C. All, "Quality in health care and ethical principles," *Journal of advanced nursing*, vol. 32, no. 3, pp. 562–571, 2000.
- [13] J. Barzdins, J. Barzdins, E. Rencis, and A. Sostaks, "Graphical modeling and query language for hospitals," *Health Information Science and Systems*, vol. 1, no. 1, p. 14, 2013.
- [14] M. Benyoucef, C. Kuziemsky, A. Afrasiabi Rad, and A. Elsabbahi, "Modeling healthcare processes as service orchestrations and choreographies," *Business Process Management Journal*, vol. 17, no. 4, pp. 568–597, 2011.
- [15] A. Pitsillides, B. Pitsillides, G. Samaras, M. Dikaiakos, E. Christodoulou, P. Andreou, and D. Georgiadis, "DITIS: A collaborative virtual medical team for home healthcare of cancer patients," in *M-Health*, Springer, 2006, pp. 247–266.
- [16] A. M. and D. W. J. E. Bardram, "Pervasive Computing in Healthcare," *CRC Press*, 2007.
- [17] G. Maniatopoulos, I. McLoughlin, R. Wilson, and M. Martin, "Developing virtual healthcare systems in complex multi-agency service settings: The OLDES Project," *Electronic Journal of e-Government*, vol. 7, no. 2, pp. 163–170, 2009.
- [18] D. M. Brennan, S. Mawson, and S. Brownsell, "Telerehabilitation: enabling the remote delivery of healthcare, rehabilitation, and self management," *Stud Health Technol Inform*, vol. 145, pp. 231–248, 2009.
- [19] G. Demiris, "The diffusion of virtual communities in health care: concepts and challenges," *Patient education and counseling*, vol. 62, no. 2, pp. 178–188, 2006.
- [20] C. El Morr, "Health care virtual communities: Challenges and opportunities," in Handbook of research on developments in e-health and telemedicine: Technological and social perspectives, IGI Global, 2010, pp. 278–298.
- [21] I. Chorbev, M. Sotirovska, and D. Mihajlov, "Virtual communities for diabetes chronic disease healthcare," *International journal of telemedicine and*

applications, vol. 2011, p. 11, 2011.

- [22] C. Porumb, S. Porumb, B. Orza, and D. Budura, "Computer-supported collaborative work and its application to e-health," in *Advances in Mesh Networks* (*MESH*), 2010 Third International Conference on, 2010, pp. 75–80.
- [23] K. Abrams, S. Burrill, and N. Elsner, "What can health systems do to encourage physicians to embrace virtual care?," *Deloitte Insights*, 2018. [Online]. Available: https://www2.deloitte.com/content/dam/insights/us/articles/4407_Virtual-caresurvey/DI_Virtual-care-survey.pdf. [Accessed: 05-May-2019].
- [24] H. Krcmar, Y. Arnold, M. Daum, and J. M. Leimeister, "Virtual communities in health care: the case of krebsgemeinschaft. de," ACM SIGGROUP Bulletin, vol. 23, no. 3, pp. 18–23, 2002.
- [25] M. Benyoucef, C. Kuziemsky, A. Afrasiabi Rad, and A. Elsabbahi, "Modeling healthcare processes as service orchestrations and choreographies," *Business Process Management Journal*, vol. 17, no. 4, pp. 568–597, 2011.
- [26] A. Gupta and R. Sharda, "Improving the science of healthcare delivery and informatics using modeling approaches," *Decision Support Systems*, vol. 55, no. 2, pp. 423–427, 2013.
- [27] H. Afsarmanesh and L. M. Camarinha-Matos, "A framework for management of virtual organization breeding environments," in *Collaborative networks and their breeding environments*, Springer, 2005, pp. 35–48.
- [28] L. Bocchi, J. Fiadeiro, N. Rajper, and S. Reiff-Marganiec, "Structure and behaviour of virtual organisation breeding environments," *arXiv preprint arXiv:1001.4413*, 2010.
- [29] M. Fraser, "Virtual research environments: overview and activity," *Ariadne*, no. 44, 2005.
- [30] I. Foster, "Service-oriented science: Scaling eScience impact," in *Web Intelligence*, 2006. WI 2006. IEEE/WIC/ACM International Conference on, 2006, pp. 9–10.
- [31] C. Zirpins and W. Emmerich, "Virtual Organisation by Service Virtualisation: Conceptual Model and e-Science Application," *Research Notes RN/07/07, University College London, Dept. of Computer Science*, 2007.
- [32] E. Ermilova and H. Afsarmanesh, "Modeling and management of profiles and competencies in VBEs," *Journal of Intelligent Manufacturing*, vol. 18, no. 5, pp.

561-586, 2007.

- [33] S. Reiff-Marganiec and N. J. Rajper, "Modelling virtual organisations: Structure and reconfigurations," in *Adaptation and Value Creating Collaborative Networks*, Springer, 2011, pp. 297–305.
- [34] F. D. Davis, "A technology acceptance model for empirically testing new end-user information systems: Theory and results," 1985.
- [35] V. L. PURC\uAREA, I. R. GHEORGHE, and C. M. PETRESCU, "Investigating the Impact of Virtual Communities on the Reputation of Romanian Health Care Organizations. A Qualitative Case Study."
- [36] R. P. Biuk-Aghai, "Visualizing structural and behavioural aspects of virtual collaboration," in *Enabling Technologies: Infrastructure for Collaborative Enterprises, 2001. WET ICE 2001. Proceedings. Tenth IEEE International Workshops on,* 2001, pp. 279–284.
- [37] R. P. Biuk-Aghai, "Patterns of virtual collaboration," 2003.
- [38] W. Saabeel, T. Verduijn, L. Hagdorn, and K. Kumar, "A model of virtual organisation: a structure and process perspective," *Electronic Journal of Organizational Virtualness*, vol. 4, no. 1, pp. 1–16, 2002.
- [39] U. Franke and B. Hickmann, "Is the Net-broker an entrepreneur? What role does the net-broker play in virtual Webs and virtual corporations," *Sieber, P.* & *Griese, J.(eds)*, pp. 23–24, 1999.
- [40] M. Gurstein, *Community informatics: Enabling communities with information and communications technologies*. IGI Global, 1999.
- [41] L. J. Winton, "A simple virtual organisation model and practical implementation," in *Proceedings of the 2005 Australasian workshop on Grid computing and eresearch-Volume 44*, 2005, pp. 57–65.
- [42] I. Foster, C. Kesselman, and S. Tuecke, "The anatomy of the grid: Enabling scalable virtual organizations," *International journal of high performance computing applications*, vol. 15, no. 3, pp. 200–222, 2001.
- [43] P. Kunszt, "European DataGrid project: Status and plans," *Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment*, vol. 502, no. 2, pp. 376–381, 2003.

- [44] M. Schrage, *Shared minds: The new technologies of collaboration*. Random House Inc., 1991.
- [45] J. M. Liedtka, "Collaborating across lines of business for competitive advantage," *The Academy of Management Executive*, vol. 10, no. 2, pp. 20–34, 1996.
- [46] P. R. Messinger, E. Stroulia, and K. Lyons, "A typology of virtual worlds: Historical overview and future directions," *Journal For Virtual Worlds Research*, vol. 1, no. 1, 2008.
- [47] J. M. Balkin and B. S. Noveck, State of Play: Law, Games, and Virtual Worlds: Law, Games, and Virtual Worlds (Ex Machina: Law, Technology, and Society). NYU Press, 2006.
- [48] R. P. Biuk-Aghai and S. Simoff, "Patterns of virtual collaboration in online collaboration systems," in *Proceedings of the IASTED International Conference on Knowledge Sharing and Collaborative Engineering, St. Thomas, USVI, November,* 2004, pp. 22–24.
- [49] L. Wainfan and P. K. Davis, *Challenges in virtual collaboration: Videoconferencing, audioconferencing, and computer-mediated communications*. Rand Corporation, 2004.
- [50] Y. P. Shao, S. Liao, and H. Wang, "A model of virtual organisations," *Journal of information science*, vol. 24, no. 5, pp. 305–312, 1998.
- [51] D. Antonacci and N. Modaress, "Second Life: The educational possibilities of a massively multiplayer virtual world (MMVW)," in *EDUCAUSE Western Regional Conference*, 2005, vol. 26.
- [52] D. Te"eni, "A cognitive-affective model of organizational communication for designing IT," *MIS quarterly*, vol. 25, no. 2, pp. 251–312, 2001.
- [53] D. L. Paul, "Collaborative activities in virtual settings: A knowledge management perspective of telemedicine," *Journal of Management Information Systems*, vol. 22, no. 4, pp. 143–176, 2006.
- [54] M. Alavi and A. Tiwana, "Knowledge integration in virtual teams: The potential role of KMS," *Journal of the American Society for Information Science and Technology*, vol. 53, no. 12, pp. 1029–1037, 2002.
- [55] B. Reeves, T. W. Malone, and T. O Driscoll, "Leadership's online labs," *Harvard Business Review*, vol. 86, no. 5, p. 58, 2008.

- [56] H. Rheingold, "A slice of my life in my virtual community," *High noon on the electronic frontier: Conceptual issues in cyberspace*, pp. 413–36, 1996.
- [57] J. Koh, Y.-G. Kim, and Y.-G. Kim, "Sense of virtual community: A conceptual framework and empirical validation," *International Journal of Electronic Commerce*, vol. 8, no. 2, pp. 75–94, 2003.
- [58] U. Lechner and B. F. Schmid, "Communities and media-towards a reconstruction of communities on media," in *System Sciences, 2000. Proceedings of the 33rd Annual Hawaii International Conference on*, 2000, p. 10–pp.
- [59] H. Rheingold, *The virtual community: Homesteading on the electronic frontier*. MIT press, 1993.
- [60] S. Whittaker, E. Isaacs, and V. O'Day, "Widening the net: workshop report on the theory and practice of physical and network communities," ACM SIGCHI Bulletin, vol. 29, no. 3, pp. 27–30, 1997.
- [61] G. Demiris, "Virtual communities in health care," in *Intelligent paradigms for healthcare enterprises*, Springer, 2005, pp. 121–137.
- [62] G. Demiris, "Virtual Communities in Health Care," *Studies in Fuzziness and Soft Computing*, vol. 184, pp. 121–137, 2005.
- [63] T. K. Houston, L. A. Cooper, and D. E. Ford, "Internet support groups for depression: a 1-year prospective cohort study," *American Journal of Psychiatry*, vol. 159, no. 12, pp. 2062–2068, 2002.
- [64] J. L. Haggerty, R. J. Reid, G. K. Freeman, B. H. Starfield, C. E. Adair, and R. McKendry, "Continuity of care: a multidisciplinary review," *BMJ: British Medical Journal*, vol. 327, no. 7425, p. 1219, 2003.
- [65] A. Chryssanthou, I. Varlamis, and C. Latsiou, "A security model for virtual healthcare communities," in *Virtual communities, social networks and collaboration*, Springer, 2012, pp. 75–113.
- [66] I. Varlamis and I. Apostolakis, "Self-supportive virtual communities," *International Journal of Web Based Communities*, vol. 6, no. 1, pp. 43–61, 2009.
- [67] R. Grenier and G. Metes, *Going virtual: Moving your organization into the 21st century*. Prentice Hall PTR Upper Saddle River, 1995.

- [68] M. Turoff, "Virtuality," *Communications of the ACM*, vol. 40, no. 9, pp. 38–43, 1997.
- [69] L. M. Camarinha-Matos and H. Afsarmanesh, "A framework for virtual organization creation in a breeding environment," *Annual Reviews in Control*, vol. 31, no. 1, pp. 119–135, 2007.
- [70] H. Zhuge, J. Chen, Y. Feng, and X. Shi, "A federation-agent-workflow simulation framework for virtual organisation development," *Information* \& *Management*, vol. 39, no. 4, pp. 325–336, 2002.
- [71] C. Scholz, "The virtual corporation: empirical evidences to a three dimensional model," in *Academy of Management.-Conference in Toronto*, 2000.
- [72] H. Walker, "The virtual organisation: a new organisational form?," *International Journal of Networking and Virtual Organisations*, vol. 3, no. 1, pp. 25–41, 2006.
- [73] B. Hardy and R. Affentranger, "Collaborative virtual organisation and infrastructure for drug discovery," *Drug discovery today*, 2013.
- [74] L. Perini, "Visual representations and confirmation," *Philosophy of Science*, vol. 72, no. 5, pp. 913–926, 2005.
- [75] D. Moody and J. van Hillegersberg, "Evaluating the visual syntax of UML: An analysis of the cognitive effectiveness of the UML family of diagrams," in *Software Language Engineering*, Springer, 2009, pp. 16–34.
- [76] A. El Kouhen, S. Gérard, C. Dumoulin, and P. Boulet, "A component-based approach for specifying DSML's concrete syntax," in *Proceedings of the Second Workshop on Graphical Modeling Language Development*, 2013, pp. 3–11.
- [77] D. L. Moody, "The 'physics' of notations: toward a scientific basis for constructing visual notations in software engineering," *Software Engineering, IEEE Transactions on*, vol. 35, no. 6, pp. 756–779, 2009.
- [78] J. W. Bryans, J. S. Fitzgerald, C. B. Jones, and I. Mozolevsky, "Formal modelling of dynamic coalitions, with an application in chemical engineering," in *Leveraging Applications of Formal Methods, Verification and Validation, 2006. ISoLA 2006. Second International Symposium on*, 2006, pp. 91–98.
- [79] A. Van Deursen and P. Klint, "Domain-specific language design requires feature descriptions," *CIT. Journal of computing and information technology*, vol. 10, no. 1, pp. 1–17, 2002.

- [80] C. Atkinson and R. Gerbig, "Harmonizing textual and graphical visualizations of domain specific models," in *Proceedings of the Second Workshop on Graphical Modeling Language Development*, 2013, pp. 32–41.
- [81] D. A. Sadilek, "Prototyping domain-specific language semantics," in *Companion to the 23rd ACM SIGPLAN conference on Object-oriented programming systems languages and applications*, 2008, pp. 895–896.
- [82] R. C. Gronback, *Eclipse modeling project: a domain-specific language (DSL) toolkit*. Pearson Education, 2009.
- [83] A. Van Deursen, P. Klint, and J. Visser, "Domain-Specific Languages: An Annotated Bibliography.," *Sigplan Notices*, vol. 35, no. 6, pp. 26–36, 2000.
- [84] J. Vatjus-Anttila, J. Kreku, and K. Tiensyrjä, "Domain-specific front-end for virtual system modeling," *EIAC-RTESMA*, vol. 12, pp. 2–5, 2012.
- [85] M. Abbasipour, F. Khendek, and M. Toeroe, "A Model-Based Framework for SLA Management and Dynamic Reconfiguration," in *International SDL Forum*, 2015, pp. 19–26.
- [86] R. Ghannoum, O. Cerrato, G. Bostrom, P. Giambiagi, and L. Yngstrom, "A Signing Protocol for Service Level Agreements in Virtual Organizations."
- [87] A. D. Heiko, H. Ludwig, and G. Pacifici, "Web services differentiation with service level agreements," 2003.
- [88] V. Binu and N. Gangadhar, "A Cloud Computing Service Level Agreement Framework with Negotiation and Secure Monitoring," in *Cloud Computing in Emerging Markets (CCEM), 2014 IEEE International Conference on*, 2014, pp. 1–8.
- [89] J. Abawajy, M. F. Fudzee, M. M. Hassan, and M. Alrubaian, "Service level agreement management framework for utility-oriented computing platforms," *The Journal of Supercomputing*, vol. 71, no. 11, pp. 4287–4303, 2015.
- [90] M. Maurer, V. C. Emeakaroha, M. Risch, I. Brandic, and J. Altmann, "Cost and benefit of the SLA mapping approach for defining standardized goods in cloud computing markets," in *International Conference on Utility and Cloud Computing* (UCC 2010) in conjunction with the International Conference on Advanced Computing (ICoAC 2010), 2010.
- [91] W. W. Song, A. Soomro, Y. Li, and Q. Liu, "A Structural Analysis of SLAs and Dependencies Using Conceptual Modelling Approach," in *Computer Software and Applications Conference Workshops (COMPSACW), 2013 IEEE 37th Annual*, 2013,

pp. 439-444.

- [92] S. Nakamoto and others, "Bitcoin: A peer-to-peer electronic cash system," 2008.
- [93] M. Anoop, "Public Key Cryptography Applications Algorithms and Mathematical Explanations," *Tata Elxsi Ltd, India*, 2007.
- [94] A. Collomb and K. Sok, "Blockchain/Distributed Ledger Technology (DLT): What Impact on the Financial Sector?," *Communications* \& *Strategies*, no. 103, p. 93, 2016.
- [95] B. Rahimi, H. Nadri, H. L. Afshar, and T. Timpka, "A systematic review of the technology acceptance model in health informatics," *Applied clinical informatics*, vol. 9, no. 03, pp. 604–634, 2018.
- [96] H.-Y. Yang and M.-J. Wang, "What factors affect physicians' decisions to use an ehealth care system," *Health*, vol. 4, no. 11, pp. 1023–1028, 2012.
- [97] M. Heidari, R. Norouzzadeh, and M. Salari, "Effective factors in Information Technology (IT) acceptance in the view of the nurses working in ICU," 2013.
- [98] G. Strudwick, "Predicting nurses' use of healthcare technology using the technology acceptance model: an integrative review," *CIN: Computers, Informatics, Nursing*, vol. 33, no. 5, pp. 189–198, 2015.
- [99] D. E. Montano and D. Kasprzyk, "Theory of reasoned action, theory of planned behavior, and the integrated behavioral model," *Health behavior: Theory, research and practice*, vol. 70, no. 4, p. 231, 2015.
- [100] F. D. Davis, R. P. Bagozzi, and P. R. Warshaw, "User acceptance of computer technology: a comparison of two theoretical models," *Management science*, vol. 35, no. 8, pp. 982–1003, 1989.
- [101] A. R. Ahlan and B. I. Ahmad, "An overview of patient acceptance of Health Information Technology in developing countries: a review and conceptual model," *International Journal of Information Systems and Project Management*, vol. 3, no. 1, pp. 29–48, 2015.
- [102] D. A. Adams, R. R. Nelson, and P. A. Todd, "Perceived usefulness, ease of use, and usage of information technology: a replication," *MIS quarterly*, pp. 227–247, 1992.

- [103] Y. K. Singh, *Fundamental of research methodology and statistics*. New Age International, 2006.
- [104] M. Rahimpour, N. H. Lovell, B. G. Celler, and J. McCormick, "Patients' perceptions of a home telecare system," *International journal of medical informatics*, vol. 77, no. 7, pp. 486–498, 2008.
- [105] S. Rajasekar, P. Philominathan, and V. Chinnathambi, "Research Methodology, 1-53," 2006.
- [106] M. S. Rahman, M. Ko, J. Warren, and D. Carpenter, "Healthcare Technology Self-Efficacy (HTSE) and its influence on individual attitude: An empirical study," *Computers in Human Behavior*, vol. 58, pp. 12–24, 2016.
- [107] F.-Y. Pai and K.-I. Huang, "Applying the technology acceptance model to the introduction of healthcare information systems," *Technological Forecasting and Social Change*, vol. 78, no. 4, pp. 650–660, 2011.
- [108] F. D. Davis, "Perceived usefulness, perceived ease of use, and user acceptance of information technology," *MIS quarterly*, pp. 319–340, 1989.
- [109] E. J. Lanseng and T. W. Andreassen, "Electronic healthcare: a study of people's readiness and attitude toward performing self-diagnosis," *International Journal of Service Industry Management*, vol. 18, no. 4, pp. 394–417, 2007.
- [110] D. Suhr, "The basics of structural equation modeling," *Presented: Irvine, CA, SAS* User Group of the Western Region of the United States (WUSS), 2006.
- [111] R. G. Lomax and R. E. Schumacker, *A beginner's guide to structural equation modeling*. psychology press, 2004.
- [112] C. D. Melas, L. A. Zampetakis, A. Dimopoulou, and V. Moustakis, "Modeling the acceptance of clinical information systems among hospital medical staff: an extended TAM model.," J Biomed Inform, vol. 44, no. 4, pp. 553–64, 2011.
- [113] G. Booch, J. Rumbaugh, and I. Jacobson, *The unified modeling language user guide*. Pearson Education India, 1999.
- [114] S. A. White, "Introduction to BPMN," *IBM Cooperation*, vol. 2, no. 0, p. 0, 2004.
- [115] A. Cheng-Leong, K. Li Pheng, and G. R. Keng Leng, "IDEF*: a comprehensive modelling methodology for the development of manufacturing enterprise systems," *International Journal of Production Research*, vol. 37, no. 17, pp. 3839–

3858, 1999.

- [116] B. List and B. Korherr, "An evaluation of conceptual business process modelling languages," in *Proceedings of the 2006 ACM symposium on Applied computing*, 2006, pp. 1532–1539.
- [117] H. Gou, B. Huang, W. Liu, and X. Li, "A framework for virtual enterprise operation management," *Computers in Industry*, vol. 50, no. 3, pp. 333–352, 2003.
- [118] J. McGinnis, K. Stathis, and F. Toni, "A formal framework of virtual organisations as agent societies," *arXiv preprint arXiv:1001.4405*, 2010.
- [119] K. J. Turner, S. Reiff-marganiec, L. Blair, G. A. Cambpell, F. Wang, K. J. Turner, S. Reiff-marganiec, L. Blair, G. A. Cambpell, and F. Wang, "APPEL: An Adaptable and Programmable Policy Environment and Language," 2007.
- [120] M. H. Danesh, B. Raahemi, S. A. Kamali, and G. Richards, "A framework for process and performance management in service oriented virtual organisations," *International Journal of Computer Information Systems and Industrial Management Applications*, vol. 5, pp. 203–215, 2013.
- [121] F. Okuyama, T. Hirano, Y. Nakabayasi, H. Minoura, S. Tsuruoka, and Y. Okayama, "Telemedicine imaging collaboration system with virtual common information space," in *The Sixth IEEE International Conference on Computer and Information Technology (CIT'06)*, 2006, pp. 214–214.
- [122] C. E. Kuziemsky and A. Harris, "An Agent Based Framework for Healthcare Teamwork," in *Proceedings of the 2019 2nd International Conference on Information Science and Systems*, 2019, pp. 271–275.
- [123] M. Al-Khafajiy, L. Webster, T. Baker, and A. Waraich, "Towards fog driven IoT healthcare: challenges and framework of fog computing in healthcare," in *Proceedings of the 2nd International Conference on Future Networks and Distributed Systems*, 2018, p. 9.
- [124] M. B. Erdem, A. Kiraz, H. Eski, Ö. Çiftçi, and C. Kubat, "A conceptual framework for cloud-based integration of Virtual laboratories as a multi-agent system approach," *Computers* \& *Industrial Engineering*, vol. 102, pp. 452–457, 2016.
- [125] H. Demirkan, "A smart healthcare systems framework," IT Professional, vol. 15, no. 5, pp. 38–45, 2013.
- [126] H. Ludwig, A. Keller, A. Dan, R. King, and R. Franck, "A service level agreement language for dynamic electronic services," *Electronic Commerce Research*, vol. 3,

no. 1–2, pp. 43–59, 2003.

- [127] I. Brandic, "Towards self-manageable cloud services," in Computer Software and Applications Conference, 2009. COMPSAC'09. 33rd Annual IEEE International, 2009, vol. 2, pp. 128–133.
- [128] L. Camerini, N. Diviani, and S. Tardini, "Health virtual communities: is the Self lost in the Net?," *Social Semiotics*, vol. 20, no. 1, pp. 87–102, 2010.
- [129] J. Abreu, L. Bocchi, J. L. Fiadeiro, and A. Lopes, "Specifying and composing interaction protocols for service-oriented system modelling," in *Formal Techniques for Networked and Distributed Systems-FORTE 2007*, Springer, 2007, pp. 358–373.
- [130] D. D. Lamanna, J. Skene, and W. Emmerich, "SLAng: a language for service level agreements," 2003.
- [131] A. Keller and H. Ludwig, "The WSLA framework: Specifying and monitoring service level agreements for web services," *Journal of Network and Systems Management*, vol. 11, no. 1, pp. 57–81, 2003.
- [132] A. Andrieux, K. Czajkowski, A. Dan, K. Keahey, H. Ludwig, T. Nakata, J. Pruyne, J. Rofrano, S. Tuecke, and M. Xu, "Web services agreement specification (WS-Agreement)," in *Open Grid Forum*, 2007, vol. 128.
- [133] M. G. Buscemi and U. Montanari, "Cc-pi: A constraint-based language for specifying service level agreements," in *European Symposium on Programming*, 2007, pp. 18–32.
- [134] S. Bistarelli, U. Montanari, and F. Rossi, "Semiring-based constraint satisfaction and optimization," *Journal of the ACM (JACM)*, vol. 44, no. 2, pp. 201–236, 1997.
- [135] K. T. Kearney, F. Torelli, and C. Kotsokalis, "SLA★: An abstract syntax for Service Level Agreements," in *Grid Computing (GRID), 2010 11th IEEE/ACM International Conference on*, 2010, pp. 217–224.
- [136] R. B. Uriarte, F. Tiezzi, and R. D. Nicola, "SLAC: A formal service-level-agreement language for cloud computing," in *Proceedings of the 2014 IEEE/ACM 7th International Conference on Utility and Cloud Computing*, 2014, pp. 419–426.
- [137] E. Badidi, "A framework for brokered Service Level agreements in SOA environments," in Next Generation Web Services Practices (NWeSP), 2011 7th International Conference on, 2011, pp. 37–42.

- [138] T. Dimitrakos, D. Golby, and P. Kearney, "Towards a trust and contract management framework for dynamic virtual organisations," *eAdoption and the Knowledge Economy: eChallenges 2004*, pp. 27–29, 2004.
- [139] A. Paschke, J. Dietrich, and K. Kuhla, "A logic based sla management framework," in *Iswc'05: Proceedings of the semantic web and policy workshop*, 2005, pp. 68– 83.
- [140] H. Ludwig, A. Keller, A. Dan, R. P. King, and R. Franck, "Web service level agreement (WSLA) language specification," *IBM Corporation*, pp. 815–824, 2003.
- [141] J. Skene and W. Emmerich, "Generating a contract checker for an SLA language," 2004.
- [142] J. Skene, D. D. Lamanna, and W. Emmerich, "Precise service level agreements," in Proceedings of the 26th International Conference on Software Engineering, 2004, pp. 179–188.
- [143] M. Debusmann, R. Kroger, and K. Geihs, "Unifying service level management using an MDA-based approach," in *Network Operations and Management Symposium, 2004. NOMS 2004. IEEE/IFIP*, 2004, vol. 1, pp. 801–814.
- [144] M. Z. Solesvik and S. Encheva, "Partner selection for interfirm collaboration: the context of ship design," in *International Conference on Cooperative Design*, *Visualization and Engineering*, 2009, pp. 141–144.
- [145] N. A. C. Mat, Y. Cheung, and H. Scheepers, "A framework for partner selection criteria in virtual enterprises for SMEs," in 2008 International Conference on Service Systems and Service Management, 2008, pp. 1–7.
- [146] B. L. Sadigh, F. Arikan, A. Ozbayoglu, H. Unver, and S. E. Kilic, "A multi-agent system model for partner selection process in virtual enterprise," *Procedia Computer Science*, vol. 36, pp. 367–372, 2014.
- [147] R. S. P. Maciel and J. M. N. David, "WGWSOA: A service-oriented middleware architecture to support groupware interoperability," in *Computer Supported Cooperative Work in Design, 2007. CSCWD 2007. 11th International Conference on*, 2007, pp. 556–561.
- [148] A. G. Garc'\ia, I. B. Espert, and V. H. Garc'\ia, "SLA-driven dynamic cloud resource management," *Future Generation Computer Systems*, vol. 31, pp. 1–11, 2014.

- [149] M. A. Covington, "Speech acts, electronic commerce, and KQML," *Decision Support Systems*, vol. 22, no. 3, pp. 203–211, 1998.
- [150] T. Finin, R. Fritzson, D. McKay, and R. McEntire, "KQML as an agent communication language," in *Proceedings of the third international conference on Information and knowledge management*, 1994, pp. 456–463.
- [151] S. Kraus, J. Wilkenfeld, and G. Zlotkin, "Multiagent negotiation under time constraints," *Artificial intelligence*, vol. 75, no. 2, pp. 297–345, 1995.
- [152] A. R. Ahlan and B. I. Ahmad, "User acceptance of health information technology (HIT) in developing countries: A conceptual model," *Procedia Technology*, vol. 16, pp. 1287–1296, 2014.
- [153] P. Ketikidis, T. Dimitrovski, L. Lazuras, and P. A. Bath, "Acceptance of health information technology in health professionals: An application of the revised technology acceptance model," *Health informatics journal*, vol. 18, no. 2, pp. 124–134, 2012.
- [154] M. Cimperman, M. M. Bren\vci\vc, and P. Trkman, "Analyzing older users' home telehealth services acceptance behavior—applying an Extended UTAUT model," *International journal of medical informatics*, vol. 90, pp. 22–31, 2016.
- [155] M.-I. R. M. Jaradat and Z. M. A. Smadi, "Applying the technology acceptance model to the introduction of mobile healthcare information systems," *International Journal of Behavioural and Healthcare Research*, vol. 4, no. 2, pp. 123–143, 2013.
- [156] M. P. Gagnon, E. Orruño, J. Asua, A. B. Abdeljelil, and J. Emparanza, "Using a modified technology acceptance model to evaluate healthcare professionals' adoption of a new telemonitoring system," *Telemedicine and e-Health*, vol. 18, no. 1, pp. 54–59, 2012.
- [157] A. J. Ducey and M. D. Coovert, "Predicting tablet computer use: An extended Technology Acceptance Model for physicians," *Health Policy and Technology*, vol. 5, no. 3, pp. 268–284, 2016.
- [158] M. Noh, H. Jang, and G. Khongorzul, "User Acceptance Model of Electronic Medical Record," *Indian Journal of Science and Technology*, vol. 9, no. 41, 2016.
- [159] M. Beglaryan, V. Petrosyan, and E. Bunker, "Development of a tripolar model of technology acceptance: Hospital-based physicians' perspective on EHR," *International journal of medical informatics*, vol. 102, pp. 50–61, 2017.

- [160] A. Tubaishat, "Perceived usefulness and perceived ease of use of electronic health records among nurses: Application of Technology Acceptance Model," *Informatics for Health and Social Care*, vol. 43, no. 4, pp. 379–389, 2018.
- [161] R. Hoque and G. Sorwar, "Understanding factors influencing the adoption of mHealth by the elderly: An extension of the UTAUT model," *International journal of medical informatics*, vol. 101, pp. 75–84, 2017.
- [162] K. S. Bordens and B. B. Abbott, *Research design and methods: A process approach*. McGraw-Hill, 2002.
- [163] L. K. Soiferman, "Compare and Contrast Inductive and Deductive Research Approaches.," *Online Submission*, 2010.
- [164] P. Yu, "A multi-method approach to evaluate health information systems.," *Stud Health Technol Inform*, vol. 160, no. Pt 2, pp. 1231–5, 2010.
- [165] M. M. Yusof, A. Papazafeiropoulou, R. J. Paul, and L. K. Stergioulas, "Investigating evaluation frameworks for health information systems," *International journal of medical informatics*, vol. 77, no. 6, pp. 377–385, 2008.
- [166] S. Eivazzadeh, "Health Information Systems Evaluation," 2015.
- [167] S. Cronholm and G. Goldkuhl, "Strategies for information systems evaluation-six generic types," *Electronic Journal of Information Systems Evaluation*, vol. 6, no. 2, pp. 65–74, 2003.
- [168] S. Chen, N. M. Osman, J. Nunes, and G. C. Peng, "Information systems evaluation methodologies," in *Proceedings of the IADIS International Workshop on Information Systems Research Trends, Approaches and Methodologies*, 2011.
- [169] K. Källander, J. K. Tibenderana, O. J. Akpogheneta, D. L. Strachan, Z. Hill, A. H. ten Asbroek, L. Conteh, B. R. Kirkwood, and S. R. Meek, "Mobile health (mHealth) approaches and lessons for increased performance and retention of community health workers in low-and middle-income countries: a review," *Journal of medical Internet research*, vol. 15, no. 1, p. e17, 2013.
- [170] C. Gackenheimer, "What is react?," in *Introduction to React*, Springer, 2015, pp. 1–20.
- [171] A. Developers, "What is android," 2011. [Online]. Available: Android Developers, http://developer. android. com/guide/basics/what-is~.... [Accessed: 20-Jul-2019].

- [172] K. W. Tracy, "Mobile Application Development Experiences on Apple?` s iOS and Android OS," *leee Potentials*, vol. 31, no. 4, pp. 30–34, 2012.
- [173] S. Tilkov and S. Vinoski, "Node. js: Using JavaScript to build high-performance network programs," *IEEE Internet Computing*, vol. 14, no. 6, pp. 80–83, 2010.
- [174] A. Mardan, "Domain and Express. js," in *Pro Express. js*, Springer, 2014, pp. 199–204.
- [175] A. Mardan, "Real-Time Apps with WebSocket, Socket. IO, and DerbyJS," in *Practical Node. js*, Springer, 2018, pp. 307–330.
- [176] E. C. Foster and S. Godbole, *Database systems: a pragmatic approach*. Apress, 2016.
- [177] S. Terrizzi, S. Sherer, C. Meyerhoefer, M. Scheinberg, and D. Levick, "Extending the technology acceptance model in healthcare: Identifying the role of trust and shared information," 2012.
- [178] R. J. Holden and B.-T. Karsh, "The technology acceptance model: its past and its future in health care," *Journal of biomedical informatics*, vol. 43, no. 1, pp. 159– 172, 2010.
- [179] E. V. Wilson and N. K. Lankton, "Modeling patients' acceptance of providerdelivered e-health," *Journal of the American Medical Informatics Association*, vol. 11, no. 4, pp. 241–248, 2004.
- [180] S.-Y. Hung, C.-M. Chang, and T.-J. Yu, "Determinants of user acceptance of the e-Government services: The case of online tax filing and payment system," *Government Information Quarterly*, vol. 23, no. 1, pp. 97–122, 2006.
- [181] M. S. Soares and R. P. do Nascimento, "Evaluation of SysML diagrams to document requirements using TAM," in *Proceedings of the 7th Euro American Conference on Telematics and Information Systems*, 2014, p. 9.
- [182] A. Lopes, I. Steinmacher, and T. Conte, "UML Acceptance: Analyzing the Students' Perception of UML Diagrams," in *Proceedings of the XXXIII Brazilian Symposium* on Software Engineering, 2019, pp. 264–272.
- [183] U. Frank, "Domain-specific modeling languages: requirements analysis and design guidelines," in *Domain Engineering*, Springer, 2013, pp. 133–157.
- [184] U. Frank, "Some guidelines for the conception of domain-specific modelling languages," *Enterprise modelling and information systems architectures (EMISA*

2011), 2011.

- [185] P. Mohagheghi and \Oystein Haugen, "Evaluating domain-specific modelling solutions," in International Conference on Conceptual Modeling, 2010, pp. 212– 221.
- [186] O. I. Lindland, G. Sindre, and A. Solvberg, "Understanding quality in conceptual modeling," *IEEE software*, vol. 11, no. 2, pp. 42–49, 1994.
- [187] J. Wu, "Healthcare system-use behavior: a systematic review of ITs determinants," *Journal of International Technology and Information Management*, vol. 25, no. 4, p. 5, 2016.
- [188] "Clinical effectiveness." [Online]. Available: https://www.greaterprestonccg.nhs.uk/quality-and-clinical-effectiveness. [Accessed: 24-Jul-2019].
- [189] E. Lee, T. Hamelin, and J. Daugherty, "Globalization of health service: Sharing of best practices in perianesthesia nursing care, a case study of cross-border institutional collaboration," *Journal of PeriAnesthesia Nursing*, vol. 33, no. 2, pp. 209–219, 2018.
- [190] M. M. Ali and A. Medhekar, "Globalization, medical travel and healthcare management in Bangladesh," *Problems and Perspectives in Management*, vol. 14, no. 2, p. 360, 2016.
- [191] L. L. Berry, K. Seiders, and D. Grewal, "Understanding service convenience," *Journal of marketing*, vol. 66, no. 3, pp. 1–17, 2002.
- [192] A. M. Mosadeghrad, "Healthcare service quality: towards a broad definition.," Int J Health Care Qual Assur, vol. 26, no. 3, pp. 203–19, 2013.
- [193] T. T. Lee, C.-H. Yeh, and L.-H. Ho, "Application of a computerized nursing care plan system in one hospital: experiences of ICU nurses in Taiwan," *Journal of Advanced Nursing*, vol. 39, no. 1, pp. 61–67, 2002.
- [194] P. A. Hortman and C. B. Thompson, "Evaluation of user interface satisfaction of a clinical outcomes database," CIN: Computers, Informatics, Nursing, vol. 23, no. 6, pp. 301–307, 2005.
- [195] M. Lising and C. KENNEDY, "A multimethod approach to evaluating critical care information systems," CIN: Computers, Informatics, Nursing, vol. 23, no. 1, pp. 27–37, 2005.

- [196] J.-H. Wu, W.-S. Shen, L.-M. Lin, R. A. Greenes, and D. W. Bates, "Testing the technology acceptance model for evaluating healthcare professionals' intention to use an adverse event reporting system," *International Journal for Quality in Health Care*, vol. 20, no. 2, pp. 123–129, 2008.
- [197] V. Venkatesh, M. G. Morris, G. B. Davis, and F. D. Davis, "User Acceptance of Information Technology: Toward a Unified View," *MIS Q.*, vol. 27, no. 3, pp. 425– 478, 2003.
- [198] J. Jewer, "Patients' intention to use online postings of ED wait times: A modified UTAUT model," *International journal of medical informatics*, vol. 112, pp. 34–39, 2018.
- [199] M. Cimperman, M. M. Brenčič, P. Trkman, and M. de L. Stanonik, "Older adults' perceptions of home telehealth services.," *Telemed J E Health*, vol. 19, no. 10, pp. 786–90, 2013.
- [200] J. L. Arbuckle and others, "IBM SPSS Amos 21 user's guide," *Crawfordville, FL: Amos Development Corporation*, 2012.
- [201] D. Chalmers, *Cambridge International AS and A Level Mathematics: Probability* \& *Statistics 1 Coursebook*, vol. 1. Cambridge University Press, 2018.
- [202] D. G. Altman and J. M. Bland, "Standard deviations and standard errors," *Bmj*, vol. 331, no. 7521, p. 903, 2005.
- [203] R. Cornish, "Statistics: 3.3 Factor Analysis," *Mathematics Learning Support Centre.* http://www.statstutor.ac.uk/resources/uploaded/factoranalysis.pdf Penn state education https://onlinecourses.science.psu.edu/stat505/node/78, 2007.
- [204] A. G. Yong and S. Pearce, "A beginner's guide to factor analysis: Focusing on exploratory factor analysis," *Tutorials in quantitative methods for psychology*, vol. 9, no. 2, pp. 79–94, 2013.
- [205] D. B. Figueiredo Filho, R. Paranhos, E. C. da Rocha, M. Batista, J. A. da Silva Jr, M. L. W. D. Santos, and J. G. Marino, "When is statistical significance not significant?," *Brazilian Political Science Review*, vol. 7, no. 1, pp. 31–55, 2013.
- [206] P. Sedgwick, "Pearson's correlation coefficient," *Bmj*, vol. 345, p. e4483, 2012.
- [207] M. M. Mukaka, "A guide to appropriate use of correlation coefficient in medical research," *Malawi Medical Journal*, vol. 24, no. 3, pp. 69–71, 2012.

- [208] C. Fornell and D. F. Larcker, "Evaluating structural equation models with unobservable variables and measurement error," *Journal of marketing research*, vol. 18, no. 1, pp. 39–50, 1981.
- [209] R. A. Peterson and Y. Kim, "On the relationship between coefficient alpha and composite reliability.," *Journal of Applied Psychology*, vol. 98, no. 1, p. 194, 2013.
- [210] L. J. Cronbach, "Coefficient alpha and the internal structure of tests," *Psychometrika*, vol. 16, no. 3, pp. 297–334, 1951.
- [211] J. F. Hair Jr, M. Wolfinbarger, A. H. Money, P. Samouel, and M. J. Page, *Essentials* of business research methods. Routledge, 2015.
- [212] B. Williams, A. Onsman, and T. Brown, "Exploratory factor analysis: A five-step guide for novices," *Australasian Journal of Paramedicine*, vol. 8, no. 3, 2010.
- [213] D. Hooper, J. Coughlan, and M. Mullen, "Structural equation modelling: Guidelines for determining model fit," *Articles*, p. 2, 2008.
- [214] S. Parry, "Fit Statistics commonly reported for CFA and SEM," *Cornell Statistical Consulting Unit: Cornell University*, 2017.
- [215] R. B. Kline, *Principles and practice of structural equation modeling*. Guilford publications, 2015.
- [216] N. J. Rajper, "VOML: virtual organization modelling language," 2012.
- [217] M. S. Rahman and M. Ko, "Toward Systematic Identification of Stakeholders for Healthcare Information Systems: A Feature-Based Method," 2013.
- [218] L. Neal-Boylan, *Clinical Case Studies in Home Health Care*, First. The Atrium, Southern Gate, Chichester, PO19 8SQ, UK: John Wiley & Sons Ltd, 2011.
- [219] W. Song and X. Li, "A conceptual modeling approach to virtual organizations in the grid," in *Grid and Cooperative Computing-GCC 2005*, Springer, 2005, pp. 382– 393.
- [220] S. Ferrante, S. Bonacina, and F. Pinciroli, "Modeling stroke rehabilitation processes using the Unified Modeling Language (UML)," *Computers in biology and medicine*, vol. 43, no. 10, pp. 1390–1401, 2013.
- [221] M. Berg and P. Toussaint, "The mantra of modeling and the forgotten powers of paper: a sociotechnical view on the development of process-oriented ICT in health care," *International journal of medical informatics*, vol. 69, no. 2, pp. 223–
234, 2003.

- [222] D. RADOIU, "VIRTUAL ORGANIZATIONS CONCEPTUAL MODELLING," INFORMATICA, vol. Volume LIII, no. 1, pp. 53–64, 2008.
- [223] T. Wegeler, F. Gutzeit, A. Destailleur, and B. Dock, "Evaluating the benefits of using domain-specific modeling languages: an experience report," in *Proceedings* of the 2013 ACM workshop on Domain-specific modeling, 2013, pp. 7–12.
- [224] M. Burwitz, H. Schlieter, and W. Esswein, "Modeling Clinical Pathways-Design and Application of a Domain-Specific Modeling Language.," in Wirtschaftsinformatik, 2013, p. 83.
- [225] P. Kumarapeli, S. De Lusignan, T. Ellis, and B. Jones, "Using Unified Modelling Language (UML) as a process-modelling technique for clinical-research process improvement," *Informatics for Health and Social Care*, vol. 32, no. 1, pp. 51–64, 2007.
- [226] G. T. Jun, J. Ward, Z. Morris, and J. Clarkson, "Health care process modelling: which method when?," *International Journal for Quality in Health Care*, p. mzp016, 2009.
- [227] S. Garde, B. Baumgarten, O. Basu, N. Graf, R. Haux, R. Herold, U. Kutscha, F. Schilling, B. Selle, C. Spiess, and others, "A meta-model of chemotherapy planning in the multi-hospital/multi-trial-center-environment of pediatric oncology," *Methods Inf Med*, vol. 43, no. 2, pp. 171–183, 2004.
- [228] D. G. Rosado, E. Fernández-Medina, J. López, and M. Piattini, "Developing a Secure Mobile Grid System through a UML Extension.," J. UCS, vol. 16, no. 17, pp. 2333–2352, 2010.
- [229] H. Mahmud, "Modeling Virtual Organization for Home Healthcare Using UML," Int. J. Comput. Sci. Eng, vol. 4, pp. 22–31, 2016.
- [230] P. Langer, K. Wieland, M. Wimmer, J. Cabot, and others, "EMF Profiles: A Lightweight Extension Approach for EMF Models.," *Journal of Object Technology*, vol. 11, no. 1, pp. 1–29, 2012.
- [231] B. Selic, "A systematic approach to domain-specific language design using UML," in Object and Component-Oriented Real-Time Distributed Computing, 2007. ISORC'07. 10th IEEE International Symposium on, 2007, pp. 2–9.
- [232] F. Gailly and G. Poels, "Conceptual modeling using domain ontologies: Improving the domain-specific quality of conceptual schemas," in *Proceedings of the 10th*

Workshop on Domain-Specific Modeling, 2010, p. 18.

- [233] N. C. C. for Women's, C. H. (UK, and others, "Surgical Site Infection," 2008.
- [234] S.-S. T. Jongmans, F. Santini, M. Sargolzaei, F. Arbab, and H. Afsarmanesh,
 "Orchestrating web services using Reo: from circuits and behaviors to automatically generated code," *Service Oriented Computing and Applications*, vol. 8, no. 4, pp. 277–297, 2014.
- [235] C. Peltz, "Web services orchestration and choreography," *Computer*, vol. 36, no. 10, pp. 46–52, 2003.
- [236] P. Mayer, A. Schroeder, and N. Koch, "MDD4SOA: Model-driven service orchestration," in *Enterprise Distributed Object Computing Conference, 2008. EDOC'08. 12th International IEEE*, 2008, pp. 203–212.
- [237] C. Ouyang, W. Van der Aalst, and M. Dumas, "Service-oriented processes: an introduction to BPEL," in *Semantic web services: theory, tools and applications*, IGI Global, 2007, pp. 155–190.
- [238] M. Kloppmann, D. Koenig, F. Leymann, G. Pfau, A. Rickayzen, C. von Riegen, P. Schmidt, and I. Trickovic, "Ws-bpel extension for people-bpel4people," *Joint white paper, IBM and SAP*, vol. 183, p. 184, 2005.
- [239] L. Bocchi, Y. Hong, A. Lopes, and J. L. Fiadeiro, "From BPEL to SRML: a formal transformational approach," in *International Workshop on Web Services and Formal Methods*, 2007, pp. 92–107.
- [240] L. Bocchi, S. Gorton, and S. Reiff-Marganiec, "From StPowla processes to SRML models," *Formal aspects of computing*, vol. 22, no. 3–4, pp. 243–268, 2010.
- [241] R. Maskey, J. Fei, and H.-O. Nguyen, "Use of exploratory factor analysis in maritime research," *The Asian journal of shipping and logistics*, vol. 34, no. 2, pp. 91–111, 2018.
- [242] H. Kaiser, "An index of factorial simplicity," *Psychometrika*, vol. 39, no. 1, pp. 31– 36, 1974.
- [243] M. GROOTVELD, J. SWAAK, R. BRUSSEE, R. H. IN'T VELD, and M. MICHEL-VERKERKE, "Virtual Organisations in Healthcare: Organisation, People and Technology."
- [244] E. Syrijanen, "Building a virtual hospital," https://pdfs.semanticscholar.org/a6e5/577ba2ba12886b5d531ab530305bcbdae

859.pdf: Accessed 13-06-2018.

- [245] C. Huffington, "Enabled to care: community mental health care as a virtual organisation.," *Int J Soc Psychiatry*, vol. 45, no. 3, pp. 157–61, 1999.
- [246] R. McClatchey and D. Rogulin, "The mammogrid virtual organisation-federating distributed mammograms," *arXiv preprint cs/0507042*, 2005.
- [247] M. Deng, M. Petkovic, M. Nalin, and I. Baroni, "A Home Healthcare System in the Cloud-Addressing Security and Privacy Challenges," in 2011 IEEE 4th International Conference on Cloud Computing, 2011, pp. 549–556.
- [248] D. J. Mancini, "Building organizational trust in virtual teams," *Journal of Behavioral Studies in Business*, vol. 2, pp. 1–5, 2010.
- [249] J. Goo, R. Kishore, H. R. Rao, and K. Nam, "The role of service level agreements in relational management of information technology outsourcing: an empirical study," *MIS quarterly*, pp. 119–145, 2009.
- [250] H. Mahmud and J. Lu, "A GENERIC VOBE FRAMEWORK TO MANAGE HOME HEALTHCARE COLLABORATION," Journal of Theoretical and Applied Information Technology, vol. 80, no. 2, p. 362, 2015.
- [251] J. Cabot and M. Gogolla, "Object constraint language (OCL): a definitive guide," in *Formal methods for model-driven engineering*, Springer, 2012, pp. 58–90.
- [252] A. I. Oliveira, L. M. Camarinha-Matos, and M. Pouly, "Agreement negotiation support in virtual organisation creation-an illustrative case," *Production Planning and Control*, vol. 21, no. 2, pp. 160–180, 2010.
- [253] J. Mal, "Evaluation of OCL Expressions over XML Data Model," *Journal of Universal Computer Science*, vol. 20, no. 3, pp. 329–365, 2014.
- [254] S. Kareem and I. S. Bajwa, "A virtual telehealth framework: Applications and technical considerations," in 2011 7th International Conference on Emerging Technologies, 2011, pp. 1–6.
- [255] N. Wu and P. Su, "Selection of partners in virtual enterprise paradigm," *Robotics* and computer-integrated manufacturing, vol. 21, no. 2, pp. 119–131, 2005.
- [256] H. Wi, S. Oh, and M. Jung, "Virtual organization for open innovation: semantic web based inter-organizational team formation," *Expert Systems with Applications*, vol. 38, no. 7, pp. 8466–8476, 2011.

- [257] S. Mohammed, J. Cannon-Bowers, and S. Foo, "Selection for team membership: A contingency and multilevel perspective," *Handbook of employee selection*, pp. 801–822, 2010.
- [258] J. Pinto, "Viewing team selection through a temporal lens," *Organizational Psychology Review*, vol. 7, no. 2, pp. 171–194, 2017.
- [259] L. E. Agust'\in-Blas, S. Salcedo-Sanz, E. G. Ortiz-Garc'\ia, A. Portilla-Figueras, Á. M. Pérez-Bellido, and S. Jiménez-Fernández, "Team formation based on group technology: A hybrid grouping genetic algorithm approach," *Computers* \& *Operations Research*, vol. 38, no. 2, pp. 484–495, 2011.
- [260] S. Javaid, A. Majeed, and H. Afzal, "A reputation management system for efficient selection of disaster management team," in Advanced Communication Technology (ICACT), 2013 15th International Conference on, 2013, pp. 829–834.
- [261] A. Arenas, B. Aziz, and G. Silaghi, "Reputation management in grid-based virtual organisations," in *International Conference on Security and Cryptography (SECRYPT 2008)*, 2008.
- [262] I. Polyantchikov and E. Shevtshenko, "Partner Selection Criteria for Virtual Organization Forming," in *Proceedings of 9th International Conference of DAAAM Baltic INDUSTRIAL ENGINEERING*, 2014, pp. 24–26.
- [263] V. D. Dang, "Coalition formation and operation in virtual organisations," 2004.
- [264] Z. Alhadhrami, S. Alghfeli, M. Alghfeli, J. A. Abedlla, and K. Shuaib, "Introducing blockchains for healthcare," in *Electrical and Computing Technologies and Applications (ICECTA), 2017 International Conference on,* 2017, pp. 1–4.
- [265] C. Stagnaro, "White paper: Innovative blockchain uses in health care," *Freed Associates*, 2017.
- [266] C. P. Transaction, "Blockchain: Opportunities for Health Care," 2016.
- [267] G. Zyskind, O. Nathan, and others, "Decentralizing privacy: Using blockchain to protect personal data," in 2015 IEEE Security and Privacy Workshops, 2015, pp. 180–184.
- [268] H. M. Hussen, "A Blockchain-based Service Provider Validation and Verification Framework for Healthcare Virtual Organization," UHD Journal of Science and Technology, vol. 2, no. 2, pp. 24–31, 2018.

- [269] Z. Zheng, S. Xie, H. Dai, X. Chen, and H. Wang, "An overview of blockchain technology: Architecture, consensus, and future trends," in *Big Data (BigData Congress), 2017 IEEE International Congress on*, 2017, pp. 557–564.
- [270] L. A. Linn and M. B. Koo, "Blockchain for health data and its potential use in health it and health care related research," in ONC/NIST Use of Blockchain for Healthcare and Research Workshop. Gaithersburg, Maryland, United States: ONC/NIST, 2016, pp. 1–10.
- [271] M. Swan, Blockchain: Blueprint for a new economy. "O'Reilly Media, Inc.," 2015.
- [272] M. Sharples and J. Domingue, "The blockchain and kudos: A distributed system for educational record, reputation and reward," in *European Conference on Technology Enhanced Learning*, 2016, pp. 490–496.
- [273] D. Carboni, "Feedback based reputation on top of the bitcoin blockchain," *arXiv* preprint arXiv:1502.01504, 2015.
- [274] P. Taylor, "Applying blockchain technology to medicine traceability," *Securing Industry*, 2016.
- [275] G. Prisco, "The blockchain for healthcare: Gem launches gem health network with philips blockchain lab," *Bitcoin Magazine*, vol. 26, 2016.
- [276] W. Shen, Y. Li, H. Ghenniwa, C. Wang, and others, "Adaptive negotiation for agent-based grid computing," *Journal of the American Statistical Association*, vol. 97, no. 457, pp. 210–214, 2002.
- [277] L. Green, "Service level negotiation in a heterogeneous telecommunication environment," in *Proceeding International Conference on Computing, Communications and Control Technologies (CCCT04), Austin, TX, USA*, 2004.
- [278] B. Mitchell and P. Mckee, "SLAs a key commercial tool," *Innovation and the Knowledge Economy: Issues, Applications, Case Studies*, vol. 4, pp. 2–2, 2005.
- [279] M. J. Buco, R. N. Chang, L. Z. Luan, C. Ward, J. L. Wolf, and P. S. Yu, "Utility computing SLA management based upon business objectives," *IBM Systems Journal*, vol. 43, no. 1, pp. 159–178, 2004.
- [280] A. Ching, L. Sacks, and P. McKee, "Sla management and resource modelling for grid computing," in London Communications Symposium (LCS 2003), London, UK, 2003.

- [281] M. Fudzee and M. Farhan, "Broker-based service-oriented content adaptation framework," 2012.
- [282] F. H. Zulkernine and P. Martin, "An adaptive and intelligent SLA negotiation system for web services," *IEEE Transactions on Services Computing*, vol. 4, no. 1, pp. 31–43, 2011.
- [283] H. Wada, J. Suzuki, Y. Yamano, and K. Oba, "E\&\# xb3;: A Multiobjective Optimization Framework for SLA-Aware Service Composition," *IEEE Transactions* on Services Computing, vol. 5, no. 3, pp. 358–372, 2012.
- [284] M. Palacios, J. Garc'\ia-Fanjul, J. Tuya, and G. Spanoudakis, "Coverage-based testing for service level agreements," *IEEE Transactions on Services Computing*, vol. 8, no. 2, pp. 299–313, 2015.
- [285] A. Pichot, O. Wäldrich, W. Ziegler, and P. Wieder, "Towards dynamic service level agreement negotiation: an approach based on ws-agreement," in *International Conference on Web Information Systems and Technologies*, 2008, pp. 107–119.
- [286] T. Nurmela and L. Kutvonen, "Service level agreement management in federated virtual organizations," in *IFIP International Conference on Distributed Applications and Interoperable Systems*, 2007, pp. 62–75.
- [287] P. Patel, A. H. Ranabahu, and A. P. Sheth, "Service level agreement in cloud computing," 2009.
- [288] N. Oren, A. Preece, and T. Norman, "Service level agreements for semantic web agents," in *Proceedings of the AAAI Fall Symposium on Agents and the Semantic Web*, 2005, pp. 47–54.
- [289] P. Balakrishnan, S. T. Selvi, and G. R. Britto, "GSMA based automated negotiation model for grid scheduling," in *Services Computing*, 2008. SCC'08. IEEE International Conference on, 2008, vol. 2, pp. 569–570.
- [290] I. Maglogiannis, D. Constantinos, and L. Kazatzopoulos, "Enabling collaborative medical diagnosis over the Internet via peer-to-peer distribution of electronic health records.," J Med Syst, vol. 30, no. 2, pp. 107–16, 2006.
- [291] B. R. Winthereik and S. Vikkelsø, "ICT and integrated care: some dilemmas of standardising inter-organisational communication," *Computer Supported Cooperative Work (CSCW)*, vol. 14, no. 1, pp. 43–67, 2005.

Appendices

Below a brief introduction to each of the appendices included in this thesis is provided:

Appendix A

During data collection a consent form was given to each survey participants along with the relevant questionnaire to fill in, appendix A is the consent from.

Appendix **B**

The appendix is composed of two parts, B1 which is an image of the questionnaire which was given to healthcare providers to fill in for the HC-VBE-F mobile application acceptance testing and appendix B2 which is the Kurdish translation of the questionnaire.

Appendix C

The appendix is composed of two parts, C1 which is an image of the questionnaire which was given to healthcare requesters to fill in for the HC-VBE-F mobile application acceptance testing and appendix C2 which is the Kurdish translation of the questionnaire.

Appendix D

The appendix is composed of two parts, D1 which is an image of the questionnaire which was given to system developers to fill in for the HC-VBE-M-F java application acceptance testing and appendix D2 which is the Kurdish translation of the questionnaire.

Appendix E

This appendix contains examples of official permission letters obtained to conduct part of the research at certain healthcare authorities and institution.

Appendix F

The appendix provides images of all the data collected from survey participants of the HC-VBE-M-F java application.

Appendix G

The appendix provides images of all the data collected from healthcare providers during the HC-VBE-F mobile application evaluation.

Appendix H

The appendix provides images of all the data collected from healthcare requesters during the HC-VBE-F mobile application evaluation.

Appendix I

The appendix is a number of tables with the HC-VBE-F mobile application screenshots and explanations.

Appendix J

This appendix is an example of AMOS data analysis output from which a selection has been used for evaluation purpose in this thesis. The analysis output is for the HC-VBE-F-Provider acceptance .data collected from healthcare providers.

Appendix A: Research survey consent form

RESEARCH CONSENT FORM

Research Details
Full Name of Researcher: Hoger Hussen Mahmud
Student Number: u1450239
Title of Research: Modelling and Investigating Virtual Collaboration for Healthcare
Research Description:
The aim of this research is to develop modelling and collaboration concepts for healthcare virtual collaboration, based on Virtual Breeding Environment (VBE) and Virtual Organisation (VO) concepts. The developed concepts then will be evaluated using Technology Acceptance Model (TAM) mechanism to establish the level of acceptability of the developed concepts amongst healthcare providers, healthcare receivers and virtual system developers.
Research mode: PhD in Computer Science and informatics
Department: Computer Science
School: School of Computing & Engineering
University: Huddersfield - UK

Participant Details						
Full Name of Participant:						
Age: 15-20 20-30 30-40 40-50 50-60	Above 60 🗖					
Sex: Male Female						
Contact Number:						
Email Address:						
Education: Secondary School 🔲 High School 🔲 Graduate 🔲	Post-Graduate 🔲					
Profession:						
Place of Work:						

Please read and tick the boxes as appropriate and sign and date the declaration at the end. Please ask if you want more clarification on any of the points.

- The research has been explained to me in verbal and / or written form by the researcher.
- I understand that the research will involve filling in a number of questionnaires regarding concepts being developed for this research.
- I understand that the research involves using personal mobile phones to test a software prototype developed for the research.



Page 1 of 2

•	I understand that the research process takes 45 minutes to complete.	Yes 🔲	No 🗖
•	I understand that I may withdraw from this study at any time without having to give an explanation.	Yes 🔲	No 🔲
•	I understand that all information about me will be treated in strict confidence and that I will not be named in any written work arising from this study.	Yes 🗖	No 🗖
•	I understand that any recordings of me (Video, sound and image) will be used for research purposes only and I give my consent to their uses.	Yes 🗖	No 🗖

I freely give my consent to participate in this research study and I have been given a copy of this form for my own information.

Signature:

Date:

Page 2 of 2

Appendix B: HC-VBE-F healthcare professionals survey questionnaire

Appendix B1

Questionnaire 1 Questionnaire 1 (Healthcare professionals) The purpose of this questionnaire is to examine Technology Acceptance Model of developed VOBE-based concepts for healthcare virtual collaboration by healthcare professionals.

Keys: 1- Strongly Disagree 2- Quite Disagree 3- Slightly Disagree 4- Neutral 5- Slightly Agree 6- Quite Agree 7- Strongly Agree

1. Expected healthcare globalization	Strong Disag	ly ree	N	leutral		St	rongly Agree
 Using the HC-VOBE application enhances healthcare providers collaboration. 	1	2	3	4	5	6	7
 Using the HC-VOBE application helps making healthcare more widely available. 	1	2	3	4	5	6	7
 Using the HC-VOBE application provides the opportunity to serve healthcare seekers around the world. 	1	2	3	4	5	6	7
 Using the HC-VOBE application would be instrumental in globalizing healthcare. 	1	2	3	4	5	6	7

2. Expected usefulness (convenience)	Strongly Disagree		Neutral			Strong) Agree		
 Using the HC-VOBE application improves my performance in providing care. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application improves my availability to provide care. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application improves my patient management tasks. 	1	2	3	4	5	6	7	
 Overall, I find the HC-VOBE application useful in providing healthcare. 	1	2	3	4	5	6	7	

Page 1 of 3

3. Expected ease of use	Strong Disag	ly ree	N	eutral		Str	ongly Agree
 Learning to use the HC-VOBE application would be easy for me. 	1	2	3	4	5	6	7
 I would find it easy to provide care using the HC-VOBE application. 	1	2	3	4	5	6	7
 It would be easy for me to become skilful at using the HC-VOBE application. 	1	2	3	4	5	6	7
 I would find the HC-VOBE application easy to use. 	1	2	3	4	5	6	7

4. Expected Clinical effectiveness	Strongly Disagree		Neutral			S	trongly Agree
 Using the HC-VOBE application would facilitate providing the right care through making patient history available. 	1	2	3	4	5	6	7
 Using the HC-VOBE application would facilitate providing the right care through recruiting specialist care provider. 	1	2	3	4	5	6	7
 Using the HC-VOBE application would facilitate providing timely care through anytime collaboration capability. 	1	2	3	4	5	6	7
 Using the HC-VOBE application would facilitate providing care in the right place through anywhere collaboration capability. 	1	2	3	4	5	6	7

Page 2 of 3

5. Attitude towards using	Strong Disag	;ly ree	N	eutral		Sti	rongly Agree
 Using the HC-VOBE application would be a good idea. 	1	2	3	4	5	6	7
 Using the HC-VOBE application make providing healthcare more interesting. 	1	2	3	4	5	6	7
 Using the HC-VOBE application would be a pleasant experience. 	1	2	3	4	5	6	7
 I would like to use the HC-VOBE application. 	1	2	3	4	5	6	7

6. Intention to use	Strongly Disagree		Neutral			S	trongly Agree
1. I intend to use the HC-VOBE application.	1	2	3	4	5	6	7
 It is likely that I will use the HC-VOBE application. 	1	2	3	4	5	6	7
3. I expect to use the HC-VOBE application.	1	2	3	4	5	6	7
 I am willing to recommend other people to use the HC-VOBE application. 	1	2	3	4	5	6	7

Thank you for completing this questionnaire

If you have any questions regarding this questionnaire please email: hoger.mahmud@hud.ac.uk

Page 3 of 3

Appendix B2

Questionnaire 1

Questionnaire 1 (Healthcare professionals)

The purpose of this questionnaire is to examine Technology Acceptance Model of developed VOBE-based concepts for healthcare virtual collaboration by healthcare professionals.

Keys: 1= Strongly Disagree 2= Quite Disagree 3= Slightly Disagree 4= Neutral 5= Slightly Agree 6= Quite Agree 7= Strongly Agree

1. Expected healthcare globalization	وهرگېراو بۇ كوردى
1. Using the HC-VOBE application enhances	1.بەكارھىتانى ئەپلىكەيشنى HC–VOBE ھەساھەنگى لە
healthcare providers collaboration.	نيوان دابينكەرانى خزمەتگوزارى تەندروستى باشتر دەكات.
 Using the HC-VOBE application helps 	2.بەكارھىنانى ئەپلىكەيشىنى HC–VOBE ھاوكارد ئە
making healthcare more widely available.	دابليكردنى خزمەتگوزارى ئەندروسىتى بە شىپوەيەكى
	فراوانتر.
3. Using the HC-VOBE application provides the	3.بەكارھىتانى ئەپلىكەيشنى HC-VOBE دەرفەت
opportunity to serve healthcare seekers	دەرەخسىنىيىت كە خىزمەتى داواكارانى خىزمەتگورارى
around the world.	تەندروسىتى بكريت ئە ھەرشوينېكى بن ئە دونيا.
4. Using the HC-VOBE application would be	4.بەكارھينانى ئەپلىكەيشىنى HC-VOBE ھاوكارە ئە
instrumental in globalizing healthcare.	بەجبھائىكردنى خزمەتگوزارى تەئىروسىتى

2. Expected usefulness (convenience)	ومرکبراو بق کوردی
 Using the HC-VOBE application improves 	1.بەكارھىتانى تەپلىكەيشنى HC-VOBE تواناى
my performance in providing care.	بابينكردنى خزمەتكوزارى تەندروستىم باشتردەكات.
2. Using the HC-VOBE application improves	2.بەكارھىنانى ئەپلىكەيشنى HC-VOBE بەردەست بورىنم
my availability to provide care.	ېق داېينکرينې خزمهتگوزاری تەندروستى زياتر دمکات.
Using the HC-VOBE application improves	3.بەكارھىنانى ئەپلىكەيشنى HC-VOBE تواناي
my patient management tasks.	بەربىردىنى ئەخۆشىەكانم باشىتر دەكات.
4. Overall, I find the HC-VOBE application	4.بەشيوديەكى كشتى پېم وابە بەكارھىتانى تەپلىكەيشنى
useful in providing healthcare.	HC-VOBE به سوده بق دابينکردنی خزمهتگهزاری
	تەندروسىتى.

3. Expected ease of use	وهرظیرای یق کوردی
 Learning to use HC-VOBE application would	1.غۇربوونى بەكارمىتانى ئەپئىكەيشنى HC-VOBE
be easy for me.	ئاسان دەبىت بۇ من.
 I would find it easy to provide care using the HC-VOBE application. 	2.بەلامەرە ئاسان دەيېت كە خزمەتگوزارى تەئدروستى ىابىن بكەم لە رېگەى بەكارەيتانى ئەپلىكەيشىنى HC- VOBE.
 I would be easy for me to become skilful at	3.بۇ من ئاسان دەبېت كە شارەزابم لە بەكارەينانى
using the HC-VOBE application.	ئەپلىكەيشنى HC-VOBE .
 I would find HC-VOBE application easy to	4.بەكارھىتانى تەپلىكەيشنى HC-VOBE بۇ من ئاسان
use.	دەييت

4. Expected Clinical effectiveness	وجرنظيرای بنی عوردی
 Using the HC-VOBE application would	1.بەكارھىتانى ئەپلىكەيشنى HC-VOBE ئاسانكارى
facilitate providing the right care through	دەكات لە ئابىنكردنى خزمەتگىزارى دروست ئە رېڭەى
making patient history available.	بەردەستكردىنى مېزووى ئەخۋشەكە.
 Using the HC-VOBE application would	2.بەكارھىتانى ئەپلىكەيشنى HC-VOBE ئاسانگارى
facilitate providing the right care through	دەكات لە ئابىنكردنى خزمەتگوزارى تەندروستى دروست ئە
recruiting specialist care provider.	رېڭەي دۆزېنەودى كەسى پسېۋې.
 Using the HC-VOBE application would	3.بەكارەيتانى ئەپلىكەيشىنى HC-VOBE ئاسانكارى
facilitate providing timely care through	دەكات ئە بابىنكرىنى خزمەتگوزارى ئانى ئەريگەى
anytime collaboration capability.	بابىنكرىنى تولناى ھەماھەنگى ئانى.
 Using the HC-VOBE application would	4.بەكارھېتانى ئەپلىكەيشنى HC-VOBE ئاسانكارى
facilitate providing care in the right place	دەكات لە ئابىنكردنى خۇمەتگوزارى شوينى لەرپىگەى
through anywhere collaboration capability.	ئابىنكردنى تولناى ھەماھەنگى شوينى.

Page 2 of 3

Attitude towards using	و «رنظيرای بیش کورد» ی
 Using the HC-VOBE application would be a 	1.بەكارھىتانى تەپلىكەيشنى HC–VOBE بىرۆكەيەكى باش
good idea.	دەبىت.
-	
Using the HC-VOBE application make	2.بەكارھىنانى ئەپلىكەيشنى HC-VOBE دابىنكردنى
providing healthcare more interesting.	خزمەتگۈزارى تەندروستى ئارەزوومەند تر دەكات.
Using the HC-VOBE application would be a	3.بەكارھىتانى ئەپلىكەيشنى HC–VOBE تەجروبەيەكى
pleasant experience.	ئاسودە بەخش دەييت.
 A consult like to use the use speed on lighting 	HC is dia the decision of the
 I would like to use the HC-VOBE application 	4-تاردزووم ھەيە بۇ بەكارھىتانى تەپلىكەيتىنى −∪⊓
	.VOBE
	1

Intention to use	بدگران بنگردی
	8-03- 3: 302-0-3
1. I intend to use the HC_VOBE application.	1.من ئامادەيم تيايە بۇ بەكارەيتانى ئەپلىكەيشنى –HC VOBE.
3. It is likely that I will use the MC_VORE	HC-VOBE Michael Contractor
2. It is likely that I will use the HC_VOBE	عنادري هاي له بن تاپيديسي -200 - 110
application.	بەكارىپېتىم.
I expect to use the HC_VOBE application.	3.پیشینی دهکهم دهکهم که من تهپلیکهیشنی HC-VOBE
	بەكارىھىتم.
4. I am willing to recommend other people to	4من ئامادەيم تيايە پېتشنيارى بەكارھېتاى ئەپلىكەيشنى
use the HC_VOBE application.	HC-VOBE بۆ كاسانى تر بكەم.
	, , , , , , , , , , , , , , , , , , , ,

زۇر سوپاس بۇ ھاوكارىت

Appendix C: HC-VBE-F healthcare requesters survey questionnaire

Appendix C1

Questionnaire 2

Questionnaire 2 (Prospective Patients)

The purpose of this questionnaire is to examine Technology Acceptance Model of developed VOBE-based concepts for healthcare virtual collaboration by prospective patients.

Keys: 1- Strongly Disagree 2- Quite Disagree 3- Slightly Disagree 4- Neutral 5- Slightly Agree 6- Quite Agree 7- Strongly Agree

1. Expected healthcare availability	Strongly Disagree		Neutral			Strongly Agree		
 Using the HC-VOBE application facilitate fair access to healthcare. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application provides anytime anywhere access to healthcare. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application makes more varieties of healthcare available. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application makes more specialist healthcare provider available. 	1	2	3	4	5	6	7	

2. Expected usefulness (convenience)	Strongly Disagree		N	leutral		Strongly Agree		
 Using the HC-VOBE application improves my chance to get on demand healthcare. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application saves me effort and time to find needed healthcare. 	1	2	3	4	5	6	7	
 Using the HC-VOBE application would be more convenient than visiting healthcare institutions. 	1	2	3	4	5	6	7	
 Overall, I find the HC-VOBE application useful in receiving healthcare. 	1	2	3	4	5	6	7	

Page 1 of 3

3. Expected ease of use	Strongly Disagree		e Neutral			Strong Agre		
 Learning to use the HC-VOBE application would be easy for me. 	1	2	3	4	5	6	7	
 I would find it easy to receive care using the HC-VOBE application. 	1	2	3	4	5	6	7	
 It would be easy for me to become skilful at using the HC-VOBE application. 	1	2	3	4	5	6	7	
 I would find the HC-VOBE application easy to use. 	1	2	3	4	5	6	7	

4. Expected healthcare quality	Stron Disa	gly gree		Neutral	I	2	Strongly Agree
 Using the HC-VOBE helps me find the right healthcare provider. 	1	2	3	4	5	6	7
 Using the HC-VOBE application helps me find healthcare respectful of and responsive to my preferences and needs. 	1	2	3	4	5	6	7
 Using the HC-VOBE application helps me find the right healthcare for reasonable price. 	1	2	3	4	5	6	7
 Using the HC-VOBE application helps me find the right healthcare in a reasonable time. 	1	2	3	4	5	6	7
 Using the HC-VOBE application helps me find the right healthcare from a location of my choice. 	1	2	3	4	5	6	7

Page 2 of 3

5. Attitude towards using	Strongly Disagree		Neutral			S	trongly Agree
 Using the HC-VOBE application would be a good idea. 	1	2	3	4	5	6	7
 Using the HC-VOBE application makes receiving healthcare more interesting. 	1	2	3	4	5	6	7
 Using the HC-VOBE application would be a pleasant experience. 	1	2	3	4	5	6	7
 I would like to use the HC-VOBE application. 	1	2	3	4	5	6	7

6. Intention to use	Strongly Disagree		Neutral			S	Strongly Agree	
1. I intend to use the HC-VOBE application.	1	2	3	4	5	6	7	
 It is likely that I will use the HC-VOBE application. 	1	2	3	4	5	6	7	
3. I expect to use the HC-VOBE application.	1	2	3	4	5	6	7	
 I am willing to recommend other people to use the HC-VOBE application. 	1	2	3	4	5	6	7	

Thank you for completing this questionnaire

If you have any questions regarding this questionnaire please email: hoger.mahmud@hud.ac.uk

Appendix C2

Questionnaire 2

Questionnaire 2 (Prospective Patients)

The purpose of this questionnaire is to examine Technology Acceptance Model of developed VOBE-based concepts for healthcare virtual collaboration by prospective patients.

Keys: 1= Strongly Disagree 2= Quite Disagree 3= Slightly Disagree 4= Neutral 5= Slightly Agree 6= Quite Agree 7= Strongly Agree

1 Expected healthcare availability	
1. Expected fieldfiltare availability	ومرجع إدراو بو خوردی
 Using the HC-VOBE application facilitate 	1. بەكار ھىتاتى تەشكەيتىنى HC-VOBE تاسانكار مى دەكات
fair agence to bealthcare	0
fair access to healthcare.	له بەردەستكردىنى دادپەروەرانەى خزمەتكوزارى
	تەنبروسىتى.
	-
Using the HC-VOBE application provides	2.بەكارھىتانى ئەپلىكەيشنى HC-VOBE خزمەتگوزارى
anytime anywhere access to healthcare	alfas annas a bat i li turchat
anytime anywhere access to nearthcare.	لاندروسنی دمی و سوینی باردمست دمتان.
3. Using the HC-VOBE application makes	3.بەكارھىتانى تەپلىكەيشنى HC-VOBE جۇرى زياتر لە
more variation of healthcare available	
more varieties of healthcare available.	خزمەتقورارى تەندروستى بەردەست دەكات.
 Using the HC-VOBE application makes 	Aبەكارھىتانى تەخلىكەنشنى HC-VOBE جۇرى زىلار لە
an an airlist har bhann ann idean airlibh	
more specialist healthcare provider available.	پسپۇرى تەندروستى بەردەست دەكات.

Expected usefulness (convenience)	و «رنظر درای بۆ خوره ی
 Using the HC-VOBE application improves my chance to get on demand healthcare. 	1.بەكارەيتانى ئەپلىكەيشنى HC-VOBE ئەكەرى دەستكەرتنى خزمەتگوزارى ئەندروستى پيويستم زياتر دەكات.
 Using the HC-VOBE application saves me effort and time to find needed healthcare 	2.بەكارھىتانى ئەپلىكەيشنى HC-VOBE كات و ھەرآم بۆ دەكيرىتەوە لە دۆزىيتەوەى خزمەتكوزارى تەندروستى يېويست.
 Using the HC-VOBE application would be more convenient than visiting healthcare institutions. 	3.بەكارەيتانى ئەپلىكەيشنى HC-VOBE لەبارتر دەبىت تا سەردانىكردنى دامەزراوە تەندروستى يەكان.

Page 1 of 3

Questionnaire 2					
 Overall, I find the HC-VOBE application useful in receiving healthcare. 	4.بەشىوديەكى گشتى بەكارھىتانى ئەپلىكەيشنى –HC VOBE بە بەسود دەبىنم لە بەدمىت ھىتانى خزمەتگرزارى تەندروستى.				

Expected ease of use	وهر ڪيردراي يو. ڪوره ي
1. Learning to use the HC-VOBE application	1.فيربوونى بەكارەينانى ئەپلىكەيشنى HC-VOBE بۇ من
would be easy for me.	ئاسان دەپېت.
2. I would find it easy to receive care using	2من با ئاسانی دہبینم خزمەتگوزاری تەئدروستیم پیکات له
the HC-VOBE application.	ريگەى بەكارھىتانى ئەپلىكەيشنى HC–VOBE.
3. I would be easy for me to become skilful at	3.ئالسان دەبىت كە من شارەزايم لە بەكارەيتانى تەپلىكەيشنى
using the HC-VOBE application.	.HC-VOBE
4. I would find the HC-VOBE application easy	4بەكارھىتانى تەپلىكەيشنى HC-VOBE بە ئاسان دەبىنىم.
to use.	

 Even shad be although sour liter. 	and the Bar
 Expected nearthcare quality 	والاسورادي بن موردي
1. Using the HC-VOBE helps me find the right	1.بەكارھىتانى ئەپلىكەيشنى HC-VOBE ھاوكارم دەبىت لە
healthcare provider.	دۆرىلەردى پسېۋرى ئەلدروستى دروست.
2. Using the HC-VOBE application helps me	2بەكارھىتانى ئەپلىكەيشنى HC-VOBE ھاركارم دەبىت لە
find healthcare respectful of and responsive to	دۆزىنەودى خزمەتگۈزارى تەندروسىتى بە پېي داواكارى و
my preferences and needs.	پيَرسٽي خَوْم-
3. Using the HC-VOBE application helps me	3.بەكارھىنانى ئەپلىكەيشىنى HC–VOBE ھاركارم دەبىت لە
find the right healthcare for reasonable price.	بۆزېنەودى خزمەتگوزارى ئەنبروسىتى بە نرخېكى گونجاو.
 Using the HC-VOBE application helps me 	4.بەكارھىنانى ئەپلىكەيشنى HC-VOBE ھاوكارم دەبىت لە
find the right healthcare in a reasonable time	دۆزىنەودى خزمەتگوزارى ئەندروستى بە كاتېكى گونجاو.

Using the HC-VOBE application helps me find the right healthcare from a location of my	5.بەكارەيتانى ئەپلىكەيشنى HC-VOBE ھاركارم دەبىت لە دۆزىنەرەي خزمەتگوزارى تەندروستى لە شوينى
choice.	ھەلېژارىدى خۆم.

5. Attitude towards using	و در تلپر در او. بغ خور دی
1. Using the HC-VOBE application would be a	 بەكارھىنانى ئەپئىكەيشنى HC-VOBE بىرۆكەيەكى
good idea.	باش دەبېت.
Using the HC-VOBE application makes	2.بەكارھىنانى تەپلىكەيشنى HC-VOBE دابىنكردنى
receiving healthcare more interesting.	خزمەتگوزارى تەندروستى ئارەزوومەند تر دەكات.
3. Using the HC-VOBE application would be a	3.بەكارھىنانى تەپلىكەيشنى HC-VOBE تەجروبەيەكى
pleasant experience.	ئاسودە بەخش دەبېت.
4. I would like to use the HC-VOBE application	4ئاردزووم ھەيە بۇ بەكارھينانى ئەپلىكەيشنى –HC
	.VOBE

Intention to use	ى تركيردراى بۇ كۈردى
 I intend to use the HC-VOBE application. 	1من ئامادەيم تيايە بق بەكارەيتانى تەپلىكەيشنى –HC
	VOPE
	.VOBE
a set 19 for a 19 second	
It is likely that I will use the HC-VOBE	ت کاری هایه که من تاپلیکایشنی HC–VOBE
application.	alian See
	÷ - Città
 Lexpect to use the HC-VOBE application 	BC-VOBE بىتلىكەرلىكەرلىكەرلىكە 10-3
streapeer to use the no tobe application.	the same find on the source of the
	بەكارىپىتم.
I am willing to recommend other people to	4من ئامادەيم تيايە پېتشنيارى بەكارھېتاى ئەپلىكەيشنى
use the HC-VORE application	C THE R TO VORE
use the no-voic application.	HO-VOBE بق کاسانی در بخام.
	1

زۇر سوپاس بۇ ھاوكارىت

Page 3 of 3

Appendix D: HC-VBE-M-F System Developer Questionnaire

Appendix D1

Questionnaire 4

Questionnaire 4 (System Developers)

The purpose of this questionnaire is to examine Technology Acceptance Model of the modelling language and the orchestration mechanism developed for healthcare virtual collaboration based on VOBE concept by system developers.

Keys: 1- Strongly Disagree 2- 0	Juite Disagree 3- Slightly Disagree 4-1	Neutral 5- Slightly Agree 6- Quite	Agree 7- Strongly Agree
---------------------------------	---	------------------------------------	-------------------------

1. Expected ability to illustrate	Stron; Disag	gly ree		ieutral	Strongly Agree		
 Models produced using HC-VOBE modelling package represents virtual collaboration in healthcare. 	1	2	3	4	5	6	7
 Models produced using HC-VOBE modelling package simplifies complex healthcare virtual collaboration scenarios. 	1	2	3	4	5	6	7
 Models produced using HC-VOBE modelling package represent roles of virtual healthcare stakeholders. 	1	2	3	4	5	6	7
 Models produced using HC-VOBE modelling package describes the collaboration process between virtual healthcare stakeholders. 	1	2	3	4	5	6	7

2. Expected usefulness	Strongly Disagree			Neutral			Strongly Agree		
 Using the HC-VOBE modelling package improves my healthcare virtual system development process. 	1	2	3	4	5	6	7		
 Using the HC-VOBE modelling package saves me effort and time to understand a healthcare virtual collaboration scenario. 	1	2	3	4	5	6	7		

Page 1 of 3

Question	inaire 4						
 Using the HC-VOBE modelling package would be more effective to use for modelling healthcare virtual collaboration systems than other modelling languages. 	1	2	3	4	5	6	7
 Overall, I find the HC-VOBE modelling package useful for modelling healthcare virtual collaboration systems. 	1	2	3	4	5	6	7

3. Expected ease of use	Strong Disag	ly ree	N	eutral	Strongly Agree		
 Learning to use HC-VOBE modelling package would be easy for me. 	1	2	3	4	5	6	7
 I would find it easy to model healthcare virtual collaboration systems using the HC- VOBE modelling package. 	1	2	3	4	5	6	7
 It would be easy for me to become skilful at using the HC-VOBE modelling package. 	1	2	3	4	5	6	7
 I would find the HC-VOBE modelling package easy to use. 	1	2	3	4	5	6	7

4. Attitude towards using	Strong Disag	;ly ree	Þ	leutral	Strongly Agree		
 Using the HC-VOBE modelling package would be a good idea. 	1	2	3	4	5	6	7
 Using the HC-VOBE modelling package makes modelling healthcare systems more interesting. 	1	2	3	4	5	6	7
 Using the HC-VOBE modelling package would be a pleasant experience. 	1	2	3	4	5	6	7

Page 2 of 3

Questionnaire 4							
 I would like to use the HC-VOBE modelling package. 	1	2	3	4	5	6	7

5. Intention to use	Strong Disag	;ly ree	N	leutral		S	trongly Agree
 I intend to use the HC-VOBE modelling package. 	1	2	3	4	5	6	7
 It is likely that I will use the HC-VOBE modelling package. 	1	2	3	4	5	6	7
 I expect to use the HC-VOBE modelling package. 	1	2	3	4	5	6	7
 I am willing to recommend other people to use the HC-VOBE modelling package. 	1	2	3	4	5	6	7

Thank you for completing this questionnaire

If you have any questions regarding this questionnaire please email: hoger.mahmud@hud.ac.uk

Appendix D2

Questionnaire 4

Questionnaire 4 (System Developers)

The purpose of this questionnaire is to examine Technology Acceptance Model of the modelling language and the orchestration mechanism developed for healthcare virtual collaboration based on VOBE concept by system developers.

Keys: 1= Strongly Disagree 2= Quite Disagree 3= Slightly Disagree 4= Neutral 5= Slightly Agree 6= Quite Agree 7= Strongly Agree

1. Expected ability to illustrate	وهرگيراو بۇ كۆردى
 Models produced using HC-VOBE modelling 	 ا.وینای بهرمهم هاتوو له ریگهی به کارهیتانی پاکیجی
package represents virtual collaboration in	HC-VOBE بق ويتا كردن ويتهى ههمانكي نابهرجاسته له
nearchcare.	خزمەتكوزارى تەندردوستى دەكات.
2. Models produced using HC-VOBE	2.وینای بەرھەم ھائور ئە رېگەی بەكارھېنانى پاكېچى
modelling package simplifies complex	HC-VOBE بق وينا كردن سيناروني تالوز له الهمانكي
healthcare virtual collaboration scenarios.	ئابەرجەستە ئە خزمەسگوزارى تەندريوستى سايە دەكات.
 Models produced using HC-VOBE 	3.وينای بەرھەم ھائوو لە رېكەی بەكارھېنانى پاكېچى
modelling package represent roles of virtual	HC-VOBE بق وینا کردن ویتای دهوری بهشداربوانی
healthcare stakeholders.	ئابەرجەستەي خزمەتگۈزارى تەندروستى دەكات.
 Models produced using HC-VOBE 	4.وینای بەرھەم ھائوو ئە رېگەی بەكارھينانی پاكیمی
modelling package describes the collaboration	HC-VOBE بق ویتا کردن پروسهی ههماههنگی له نیوان
process between virtual healthcare	بەشداربوانى ئابەرجەستەي خزمەتگوزارى تەندروستى
stakenoruers.	رونډهکاتهوه.

 Expected usefulness 	
2. Expected userumess	وهر حيراو يو خوردى
 Using the HC-VOBE modelling package 	 بەكارھيتانى پاكيجى HC-VOBE بو ويتا كردن كارى
improves my healthcare virtual system	پەرەپيدانى سېستەمى خابەرجەستەي خزمەتگوزارى
development process.	تەندروستى باشتر دەكات
2. Using the HC-VOBE modelling package	2بەكارھىتانى پاكىجى HC-VOBE بۇ ويتلكردن ھەول و
saves me effort and time to understand a	كاتم بۇ دەكتېرىتەرە ئە تېگەشتنى سىناربۇي خزمەتگوزارى
healthcare virtual collaboration scenario.	تەندروستى ئابەرجەستە.

 Using the HC-VOBE modelling package would be more effective to use for modelling healthcare virtual collaboration systems than other modelling languages. 	3.بەكارەيتانى پاكيىمى HC-VOBE بۇ ويتا كرىن گارىگەرترە لە ئامرازەكانى تر بۇ دروستكردنى ويتاى سېستەمى نابەرجەستەى خزمەتگوزارى تەخروستى.
 Overall, I find the HC-VOBE modelling package useful for modelling healthcare virtual collaboration systems. 	4.بەشىزەيەكى گشتى من پاكېنېى HC-VOBE بۇ ويتا كرينم بەلارە بە سويە بۇ سىيستەمى ئابەرجەستەى خزمەتگۈزارى تەندروستى.

Expected ease of use	ومرگتراور نئ کوردی
 Learning to use HC-VOBE modelling 	1.فيربوونى بەكارەيتانى پاكيمى HC-VOBE بۇ ويتا
package would be easy for me.	كردن بق من ثاسان دەييت.
I would find it easy to model healthcare	2من به ئاسانی تەبىنم كە وينای سىيىتەمى نابەرجەستەی
virtual collaboration systems using the HC-	خزمەتگوزارى تەندروستى بكېشم لە رېڭەي بەكارھېنانى
VOBE modelling package.	پاکیجی HC-VOBE بق وینا کردن.
3. It would be easy for me to become skilful at	3ىتاسان دەبېت بۇ من كە شارەزابم لە باكارەيتالى پاكېجى
using the HC-VOBE modelling package.	HC-VOBE بق ويتا كردن.
 I would find the HC-VOBE modelling 	4من بەكارھىتائى پاكىيمى HC-VOBE بۇ وينا كردن با
package easy to use.	ئاسان دەبيتم .

 Attitude towards using 	وهرکېراو بو کوردی
1. Using the HC-VOBE modelling package	1.بەكارھىتانى پاكىجى HC-VOBE بۇ ويتا كرىن
would be a good idea.	بېرۇكەيەكى باش دەبېت.
2. Using the HC-VOBE modelling package	2.بەكارەيتانى پاكيجى HC-VOBE بۇ ويتا كردن
makes modelling healthcare systems more	ويتلكردنى سيستهمى نابهرجهستهى خزمهتكوزارى
interesting.	تەندروستى ئارەزوومەند تر دەكلت.
Using the HC-VOBE modelling package	3.بەكارھىتانى پاكىجى HC-VOBE بۇ ويتا كردن
would be a pleasant experience.	تەجروبەيەكى ئاسودە بەخش دەبېت.

Questio	nnaire 4
4. I would like to use the HC-VOBE modelling	4.تاردزووم هەيە بۇ بەكارھينانى پاكيمى HC-VOBE بۇ
package	وينا كرين.

5. Intention use	وهرگيراو بۇ كۈردى
 I intend to use the HC-VOBE modelling	1ـمن ئامايميم ٿيايه بق بهکارهيٽائي پاکٽِجي HC–VOBE
package	بق ويٽا کردن.
 It is likely that I will use the HC-VOBE	2ئەكەرى ھەيە كە من پاكىجى HC-VOBE بۇ ويتا كردن
modelling package	بەكارىيىتم.
 I expect to use the HC-VOBE modelling	3.پینٹسبنی دہکەم که من پاکنیجی HC-VOBE بق وینا
package.	کردن بەکاربھینم.
 I am willing to recommend other people to	4من ئامادەيم تيايە پېتشنيارى بەكارھيتانى پاكيجى −HC
use the HC-VOBE modelling package.	VOBE بۇ ويتا كردن بۇ كاسانى تر بكەم.

زۇر سوپاس بۇ ھاوكارىت

Page 3 of 3

Appendix E: Official permissions obtained to conduct research



حكومەتى ھەريمى كوردستان حكومة اقليم كوردستان سەرۇكايەتى ئە نجومەنى وەزيران رئاسة مجلس الوزراء - وەزارەتى تەندروستى وزارة الصحه بەريوەبەرايەتى كشتى تەندروستى گەرميان المديريه العامة لصحة كرميان نەخۇشخانەي گشتى كەلار مستشفى كلار العام Kurdistan Regional Government 🔶 کارگیری ﴾ Ministry of Health ف الادار» ي Kalar General Hospital 075 10,65 باروار: >/ 5 / 14/ 5 STVIA/ (پیشمهر که سومبولی نهتهوهیی و پاریز دری کوردستانه) بِيشه : ماموستا لهزانكوى كمشميدانى مروّيى at man (act of the set To : Huddersfield University_UK This is to confirm that (Hoger Mahmud) who is a PhhD student has conducted part of his research in our health institues . He is working on research entitled (Modelling and Investigating Collaboration for Healthcare). Should you require further information please decoder contact us . +964770/066609 Yours Faithfully Y.19/0/1 بدريودبدرايدت كشتى تدندروستى كدرميان ىلەخۇشخانەي كَشْتَى كەلار . بەشى كاركَيْرى SUSA 4 i ناونیشان کهلار ریگای گشتی ت/۲۱۷۹، ۲٤۸۰

Appendix F: All data collected for the HC-VBE-M-F

1	PAM1	PAM2	PAN	ИЗ	PAM4	P	U1	PU2	PU3	PU4		PEU1	PEU2	F	PEU3	PEU4	AU1		AU2	AU3	A	U4	IU1	IU2	1	U3	IU4	-
2		5	6	7		6	5	7		5	6		7	6	6	;	7	7		6	7		6	7	6		6	6
3		5	5	5		5	6	6		6	6		5	4	5	i	5	6		6	6		5	6	5		5	5
4		6	6	7		4	6	6		4	5		6	6	6	;	6	7		5	5		6	6	6		6	6
5		5	7	6		5	7	7		6	6		7	5	6	i	7	5		6	7		5	5	6		6	7
6		5	5	6		4	5	7		6	6		7	6	5	i	5	6		6	6		6	5	5		6	5
7		5	7	6		5	7	6		7	5		7	5		,	7	7		5	7		6	5	5		6	7
8		7	7	7		6	6	7		6	5		6	5	7		7	7		6	6		5	5	5		5	7
9		6	7	5		6	7	6		6	5		5	6	5	i	6	7		6	5		6	5	6		5	4
10		6	5	6		6	5	6		6	5		6	6	6	i	6	6		5	6		6	6	6		5	6
11		5	4	5		5	4	5		5	4		5	5	7		7	5		5	6		5	5	6		5	5
12		7	6	7		7	7	7		7	7		7	6			7	7		7	7		5	6	4		5	7
13		3	2	4		5	5	5		5	4		4	4	4	L .	4	5		5	5		5	4	5		4	5
14		7	7	5		6	7	6		6	7		6	6	6	i	6	7		7	6		6	6	6		6	7
15		6	6	6		6	5	6		6	6		7	6	- 7	1	7	7		6	6		5	5	5		6	6
16		5	5	6		6	5	5		4	6		5	5	6	i	6	5		5	5		5	5	5		5	5
17		6	6	7		6	6	6		6	6		5	6	5	i i	5	6		5	6		6	6	5		5	5
18		6	4	6		6	5	6		3	4		4	5	5	i	4	6		5	6		6	5	6		6	5
19		6	5	6		5	5	6		5	6		5	6	5	i	6	6		6	6		6	5	6		5	6
20		3	4	3		3	3	3		4	3		4	3	4	ŧ.	3	3		3	3		4	4	4		4	4
21		6	5	5		5	5	5		5	5		5	4	5	i	4	5		4	4		5	5	5		5	5
22		3	4	3		4	3	3		4	4		2	3	4	ŧ.	3	4		3	4		3	4	3		4	3
23		3	3	4		3	4	4		4	3		4	4	4	ŧ.	3	2		3	2		3	4	3		4	3
24		3	4	3		4	3	4		3	4		4	4	8	1	3	3		4	3		3	4	4		3	4
25		4	3	3		4	5	5		5	4		3	4			4	3		4	3		4	3	3		4	3
26		3	2	3		2	3	4		3	2		3	4	3	1	4	4		5	3		5	4	3		4	3
27		2	3	2		2	3	2		3	3		3	3	2	1	2	2		3	3		2	4	5		4	4
28		3	4	4		3	4	4		3	4		3	3	4	4	3	3		3	2		3	3	4		3	4
29		3	3	2		3	3	3		3	4		4	4	5	i	5	3		3	3		3	4	3		3	3

Appendix G: All data collected for the HC-VBE-F from healthcare professionals

1	PHG1	PHG2	PHG3	PHG4	PU1	PU2	PU3	PU4	PEU1	PEU2	PEU3	PEU4	PCE1	PCE2	PCE3	PCE4	AU1	AU2	AU3	AU4	IU1	IU2	IU3	IU4	
2	6		7 .	7	5	5	7	6	6	5	6	6	6	7 7	7 1	6	6	3 1	3	7	6	6	5	4	5
3	5	1	6 1	6	7	4	6	6	7	4	5	5	5	6 6	5 !	5	6	7	7	5	6	6	6	6	6
4	5		3 !	5	5	7	7	6	7	4	4	5	3	4 4	L .	4	5	7 1	3	7	5	6	5	5	6
5	7	1	6 1	6	7	7	7	6	7	5	6	5	7	6		7	6	3	7	4	5	6	7	7	6
6	2		1	1	2	2	1	1	1	7	6	6	6	3 2	2 :	2	2	3	3	2	4	5	5	4	5
7	5	1	6 .	7	6	3	4	3	5	3	4	3	5	3 4	۱ I	3	5	7 1	3	5	4	5	6	4	6
8	6	:	5 9	5	5	4	6	5	6	5	6	5	4	4 5	5 1	6	5	5 1	5	4	4	5	5	6	5
9	6		5 .	7	7	6	6	5	7	5	6	4	3	56	6	4	6	7 1	3	5	7	7	6	7	6
10	6		7 1	6	6	5	6	6	6	6	6	5	7	5 7	7 1	6	5	7 1	6	7	7	7	7	7	7
11	6		7 .	7	6	7	6	6	7	5	5	6	6	7 7	? I	6	7	7 1	3	6	7	6	7	7	7
12	6		5 4	4	6	5	7	6	5	5	6	5	6	6	,	4	6	7 1	6	7	7	6	7	6	7
13	6	1	6 4	4	6	6	6	7	7	5	7	6	7	7 7	7 1	6	7	7 1	3	7	7	7	6	6	7
14	6		7 1	6	7	7	7	6	7	7	6	7	7	7		7	7	7	7	6	7	5	6	6	6
15	7		7 .	7	7	7	7	7	7	6	7	7	7	7 7	, .	7	6	7	7	7	5	6	7	7	6
16	6	1	6 1	6	7	4	6	5	6	4	4	5	5	6 5	5 !	5	4	3 !	5	6	6	7	5	5	5
17	7		7 .	7	7	7	7	7	7	5	5	7	6	7 7	, .	7	7	5 1	5	6	6	6	7	6	7
18	7		7 .	7	7	7	7	7	7	5	6	6	7	6 7		7	6	7	3	7	7	6	6	7	6
19	7		7 .	7	7	7	7	7	7	7	7	7	7	7	,	7	7	7	7	7	7	7	7	7	7
20	6		5 1	6	7	5	6	6	7	7	6	7	7	56	8	7	6	7 1	3	7	7	7	7	7	7
21	4		5 1	6	7	6	5	7	7	7	6	7	7	6 7	7 1	6	7	7	7	6	7	7	7	7	7
22	7		7 .	7	7	7	7	7	7	7	7	7	7	7 7	, ,	7	7	7	7	7	7	7	7	7	7
23	7		7 .	7	7	7	7	7	7	5	7	7	6	7 7	· ·	7	7	7	7	7	7	7	7	7	7
24	7		7 .	7	7	7	7	7	7	5	7	7	7	7 7	, .	7	7	7	7	7	7	7	7	7	7
25	6		7 .	7	7	4	5	6	6	7	6	6	6	7 7	? 1	6	6	7 1	3	6	6	5	6	6	5
26	7		7 1	6	6	6	6	6	5	5	4	5	5	76	6	4	7	5	7	6	7	7	7	7	7
27	7	1		7	7	7	7	7	7	6	7	7	7	6		7	7	7	7	7	7	7	7	6	7
28	7		7	7	7	7	7	7	7	7	7	7	7	7 7	7	7	7	7	7	7	7	7	7	7	6
29	7		7 .	7	7	7	7	7	7	7	7	7	7	7 7	, .	7	7	7	7	7	7	7	7	7	7
30	5		7 .	7	6	7	7	7	7	5	6	5	5	7 7	7	7	7	7	7	7	6	5	6	7	6
31	5		6 .	7	6	7	6	6	5	2	5	4	6	6	1	5	6	7 1	5	6	6	6	5	5	6
32	7		7 .	7	7	7	7	7	7	5	6	6	6	7 7	7	7	7	7	7	7	7	7	7	7	7
33	7		7 .	7	6	7	7	6	7	7	7	7	6	7 7	1	7	6	7	6	7	7	7	7	7	6
34	7		7 .	7	7	7	7	7	7	7	7	7	7	7	, .	7	7	7	7	7	7	7	7	7	7
35	6		7 !	5	6	4	5	7	7	4	4	6	5	7 7	· ·	7	7	5	7	6	4	7	7	7	7
36	6		5 4	4	7	7	5	6	7	6	6	7	5	7 6	6 !	5	5	3 !	5	7	5	5	5	6	6

37	7	5	6	5	6	5	6	6	4	5	5	5	6	6	5	6	6	6	5	5	5	5	6	6
38	7	7	7	7	6	7	5	7	7	6	7	7	7	7	7	7	7	7	6	7	7	7	7	6
39	6	5	6	7	6	7	6	5	4	5	5	6	5	6	5	6	6	5	5	6	4	5	5	6
40	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
41	7	7	6	7	7	7	7	7	6	6	7	7	7	6	7	7	7	7	7	7	7	7	7	7
42	7	7	6	7	7	7	7	7	7	7	6	7	7	6	6	7	7	7	6	6	7	7	7	7
43	7	7	7	7	7	7	7	7	6	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
44	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
45	4	5	5	4	4	3	4	5	3	4	6	6	5	6	5	5	6	5	5	4	5	7	6	5
46	5	6	7	6	6	5	6	6	3	5	5	4	6	7	6	5	5	6	7	6	6	7	6	7
47	6	6	5	6	4	6	3	4	6	4	7	7	6	7	6	6	4	5	6	5	5	4	4	5
48	7	7	7	7	7	7	7	7	7	6	7	7	7	7	7	7	7	7	7	6	7	6	5	6
49	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
50	5	6	7	7	5	5	6	6	7	7	7	7	7	6	6	6	6	7	6	6	7	7	7	7
51	7	6	7	6	7	6	6	6	6	6	7	6	6	6	6	7	6	7	6	5	6	6	6	7
52	6	7	5	7	5	6	6	7	7	5	7	6	7	6	5	7	7	5	6	6	7	6	6	7
53	7	7	7	7	7	6	6	7	6	6	6	6	6	5	5	6	7	7	7	6	6	5	6	6
54	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
55	7	7	7	6	7	7	7	7	7	7	7	7	7	7	7	7	7	6	6	6	7	7	7	6
56	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	6	7	7	7	6	6
57	5	5	5	6	5	5	5	4	6	6	5	4	6	6	6	7	7	7	7	7	7	7	7	7
58	7	7	7	7	7	7	7	5	6	6	6	7	7	6	7	7	7	7	6	5	7	7	7	7
59	7	7	7	7	6	6	7	7	7	7	7	6	7	7	6	7	7	6	6	7	7	6	6	7
60	7	7	7	7	7	7	6	6	7	7	6	7	7	7	7	7	7	7	7	7	7	6	6	7
61	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
62	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
63	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	3	3	3	3	4	4	4	4
64	7	7	6	5	6	7	6	6	7	7	7	7	7	6	7	5	7	6	5	6	7	6	6	7
65	6	7	7	6	5	6	6	7	7	7	7	7	6	7	6	7	7	7	7	6	6	6	6	6
66	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	5	5	5	5
67	7	7	7	7	7	7	7	7	7	5	7	4	6	6	6	7	6	5	6	6	7	7	7	7
68	7	7	7	7	7	7	7	7	7	7	7	7	7	7	6	7	7	7	7	6	6	7	7	7
69	5	7	7	7	5	7	6	4	7	6	7	7	7	7	7	6	7	6	7	7	7	6	5	6
70	5	5	6	6	7	7	6	7	4	4	3	3	7	6	7	7	6	6	7	7	5	7	5	7
71	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
72	3	6	6	7	6	7	7	7	5	7	5	7	6	7	7	7	6	5	6	7	7	7	7	6
72	7	6	7	7	4	5	6	7	7	e	7	7	5	4	7	7	7	6	7	7	7	7	7	6
74	7	7	7	7	7	7	7	6	7	7	7	7	7	7	6	7	7	7	6	7	7	6	7	6
75	7	7	7	6	7	7	7	7	7	7	7	7	7	7	5	7	7	7	7	7	7	7	7	7
70	7	7	7	7	7	7	7	7	7	7	7	6	7	7	5	7	7	7	7	7	7	r e	7	7
76			(r (0	ſ	r	0		ſ	r	ſ	ſ	(0	1	ſ

76	7	7	7	7	7	7	7	7	7	7	7	6	7	7	6	7	7	7	7	7	7	6	7	7
77	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
78	7	7	7	7	7	7	7	7	7	7	6	7	7	7	7	7	6	7	7	7	7	7	6	7
79	7	7	6	6	7	7	7	7	7	5	6	7	7	7	7	6	6	7	7	6	5	5	6	7
80	7	7	7	7	6	6	7	7	7	7	7	5	7	7	7	6	6	7	6	7	7	7	7	7
81	7	7	7	7	7	7	7	7	6	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
82	6	7	5	7	4	5	5	6	6	5	6	5	6	4	6	5	6	6	6	5	5	5	4	4
83	7	7	7	6	5	5	6	6	6	6	6	4	6	5	6	6	6	6	6	6	6	5	6	6
84	7	7	7	7	7	7	6	7	7	7	7	7	7	7	6	7	7	7	7	7	7	7	7	7
85	7	7	7	6	5	6	6	4	6	5	7	4	7	6	5	6	7	6	6	5	3	7	5	5
86	7	6	7	5	7	6	7	7	7	5	7	7	7	7	6	7	7	6	7	7	7	3	7	7
87	7	7	7	7	7	7	7	7	7	7	7	5	6	7	7	7	7	7	7	7	7	7	7	7
88	5	7	5	7	5	6	7	7	6	6	6	6	7	7	6	7	7	7	6	7	7	6	7	6
89	6	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
90	7	7	7	6	7	7	7	7	7	7	6	6	7	7	7	7	7	7	7	7	7	7	7	7
91	7	7	7	7	7	7	7	7	7	7	7	7	7	6	6	7	7	7	7	7	7	6	7	7
92	7	7	7	7	7	7	7	7	7	7	7	6	7	7	5	7	7	7	7	7	7	7	7	7
93	7	7	7	7	7	6	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
94	6	7	7	7	6	6	7	7	6	6	5	6	6	7	6	5	6	6	5	7	5	4	5	5
95	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
96	7	7	7	7	7	7	6	7	6	6	7	7	7	6	5	7	7	7	6	7	6	7	7	7
97	7	7	7	7	7	6	7	7	7	7	7	7	7	7	7	7	7	7	7	6	7	6	7	7
98	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	5	6	6	6
99	6	7	7	7	7	6	6	7	6	6	7	6	6	7	6	6	6	7	6	6	6	5	6	6
100	6	7	7	7	6	7	7	7	7	7	7	7	7	7	6	7	6	6	6	7	7	7	7	7
101	7	6	7	6	7	7	7	7	5	5	6	6	7	7	7	7	7	7	7	5	5	5	5	5

Appendix H: All data collected for the HC-VBE-F from healthcare requesters

1	PHA1	PHA2	PHA3	PHA4	PLI	PI I2	PH3	PI 14	PFI I1	PELI2	PFI I3	PFI 14	PHQ1	PHQ2	PHO3	PHQ4	PHO5 AL	11 AL 12	ALI3	AI 14	111	112	113	4	
2	7		7	5	5	4	5	5	4	5	4	5	4	4	5 5	; .	5 5	5	5	5	5	6	4	5	5
2	6		- 5	5	5	6	7	4	5	4	7	7	В	7	6 6	5 F	3 5	6	5	5	6	6	7	6	6
4	5		4	5	3	6	5	5	4	6	4	5	5	6	5 4		5 5	4	5	5	6	4	4	5	5
5	5		B	6	7	4	5	4	a	4	4	5	4	5	4 4		5 6	6	6	6	5	5	5	5	5
6	7		6	6	7	7	7	6	7	5	7	6	B	7	7 F		7 6	7	6	7	7	5	6	7	7
7	7		7	6	5	6	7	7	a	7	7	4	6	7	6 7	, .	7 7	7	7	6	6	5	7	7	7
	Ś		- 5	4	5	6	5	4	, S	7	B	Ś	6	5	5 6	5 F	3 7	4	3	4	5	6	5	5	5
9	4		6	7	7	6	7	7	5	4	7	4	3	7	7 6		7 7	6	5	6	6	5	4	5	6
10	4		7	3	6	6	6	5	4	7	B	7	7	4	4 4	í s	5 5	5	4	5	5	5	5	5	5
11	5		7	4	6	4	6	3	5	ŝ	6	7	5	6	5 6		5 6	6	6	4	6	5	5	5	6
12	5		4	5	5	4	4	3	3	4	4	4	5	4	4 9		1 4	4	4	5	4	4	3	4	4
13	6		7	7	5	7	6	7	5	7	7	7	6	6	6 9	; •	5 6	7	6	5	6	6	6	6	5
14	4		B	Б	5	4	6	6	5	6	B	B	4	5	3 6		1 4	5	6	5	5	5	5	5	4
15	5		6	5	6	5	6	6	6	5	5	6	7	5	5 5	, ; f	 5 6	7	6	6	6	5	6	6	7
16	5		5	6	5	6	6	5	5	6	6	5	B	5	5 5	; .	5 5	6	5	5	5	5	5	6	6
17	7		7	6	7	5	6	6	6	4	4	5	5	4	6 6	5 F	3 7	5	5	5	5	4	4	4	3
18	7		B	6	6	7	7	6	6	4	5	6	4	5	6 6	5 F	3 7	5	5	5	5	5	4	a	5
19	5		7	- 6	6	6	7	4	5	5	5	6	6	6	6 6	-	7 6	6	5	5	5	4	5	5	5
20	4		5	5	6	5	5	4	5	6	5	5	6	4	5 4	I f	R 4	6	5	4	6	5	6	6	6
21	4		2	4	4	5	5	4	4	3	3	5	5	4	5 5		5 4	4	5	5	6	5	6	6	6
22	5		4	6	5	5	5	5	5	5	5	5	5	6	6 6	5 f	6 6	6	6	6	6	5	5	5	5
23	5		6	6	7	6	6	6	6	7	6	7	7	6	6 5	5 6	6 6	6	6	6	6	6	6	6	6
24	5		2	3	6	7	6	7	7	7	4	7	7	5	6 2	2 5	5 4	7	7	7	7	7	7	6	7
25	4		5	5	5	5	5	4	4	3	3	4	3	5	5 5	5 6	5 5	4	6	6	5	4	5	4	4
26	5		5	6	6	6	6	6	7	6	6	7	7	6	7 7	, .	7 7	7	7	7	7	6	7	7	7
27	6		4	4	4	6	4	5	5	4	5	4	5	5	6 4	F 6	5 3	6	5	4	5	6	4	6	4
28	6		6	7	7	6	7	7	7	7	6	7	7	6	5 7	, -	7 7	7	7	7	6	5	4	4	7
29	6		7	6	6	7	6	4	6	5	5	6	5	4	6 7	, -	76	6	7	6	7	6	5	6	6
30	5		7	6	7	7	7	4	5	7	7	7	7	5	4 6	; ·	4 7	6	5	4	6	7	6	6	7
31	5		6	6	6	6	5	5	6	6	5	6	5	5	6 6	6 6	6 6	6	5	5	6	5	5	6	5
32	5		5	5	5	5	6	4	5	5	5	6	6	5	4 4	ب ا	4 5	6	5	6	6	5	5	5	6
33	6		6	5	6	5	7	6	5	6	6	6	5	5	6 5	5 6	6 6	6	5	5	7	6	6	6	6
34	5		1	3	2	4	4	4	3	5	5	4	5	5	4 4	L S	56	6	5	5	5	6	5	5	5
35	2		3	3	2	4	4	3	4	5	5	4	4	4	4 5	5 :	3 5	3	4	4	4	4	3	4	5
36	7		7	6	7	7	7	7	7	7	7	6	7	7	6 7	7 6	67	7	7	7	7	7	7	6	7
	-		-		-	-	-	-		-	-	-	-	-	-			-	-	-	-	-	-	-	

37																						
	5	6	4	5	5	6	5 4	7	5	5	8	3	5	6	5	7 7	5	8	5 5	8	7	6
			-	-	-	-	5 7	-			-	-	0	0	0			-	3 3		-	0
38	((- (4	5	6 6			1	1	- (ь	ь	ь	ь /			1 1		(ь
39	6	5	6	7	6	6	77	6	5	5	6	6	7	5	6	5 6	7	7	7 7	7	6	7
40	5	3	3	4	6	4	3 5	4	3	5	3	5	5	4	5	6 4	5	5	4 5	6	5	4
				-		-	0 0			-	-	0	~	-	-	0 7		-	7 7			
41	Б	(Б		ь	(6 6	1	ь	(1	6	ь	Б	1	6 /	6	(1 1	1	1	
42	4	5	6	6	4	4 !	5 4	3	4	6	4	4	4	5	5	5 6	6	5	5 4	4	4	4
42	6	6	6	5	5	5	8 5	4	6	7	4	6	6	5	5	5 7	6	5	6 5	5	5	4
										-	-	-	-	-	-							
44	5	5	Б	ь	ь	Б	((6	ь	(1	- (1	((((ь	ь	1 1		1	
45	6	7	5	7	7	6	7 7	6	4	6	6	6	6	5	6	6 7	7	6	7 6	7	7	7
40	6	7	7	7	7	7	7 7	6	6	6	6	7	7	7	7	7 7	7	6	7 7	7	7	6
40	0		-	-	-	-	1 1	0		0	-	-	-	-	-		1	0			1	
47	5	2	6	5	3	2	5 6	7	4	7	7	2	2	7	6	5 7	6	6	7 6	4	4	4
48	5	6	6	6	7	7 1	6 6	7	7	7	7	7	6	7	7	7 7	7	7	6 6	6	6	6
40	7	7	7	7	7	7	0 0	7	7	7	7	7	7	7	7	7 7	7	7	7 7	7	7	7
43	1	1	1	1	1	1	0 0	1	r	1	r	-	1	1	1	1 1	r	1	1 1	1	r	
50	7	7	6	6	7	7	7 7	7	7	7	7	7	7	7	7	7 7	7	6	7 7	6	6	7
51	7	6	7	7	7	7 1	6 6	7	7	6	6	7	7	7	7	6 7	7	7	7 7	7	7	7
FO	0	7	F	F	6	e 1	5 7	6	0	7	0	e	0	7	0	E 0	6	E	7 5	E	6	e
02	0	r	5	5	0	0	5 r	0	0	r	0	0	0	r	0	3 0	0	5	1 3	3	0	0
53	6	7	7	7	6	7	5 6	7	7	7	7	7	6	6	6	7 7	7	7	7 7	7	7	7
54	7	7	7	7	7	7	7 7	6	6	6	6	6	6	6	6	6 7	6	5	7 7	7	7	7
EE	E	6	7	6	6	7	7 7	7	6	7	7	7	7	7	7	7 7	7	7	7 7	- 7	7	- 7
00	5	0	r	0	0	1		r	0	r	ſ	1	1	1	1	1 (r	r	1 (((
56	5	6	5	5	5	6	5 5	7	5	7	7	5	5	6	6	6 5	5	5	5 5	5	5	5
57	5	6	6	6	6	6	6 6	7	7	7	6	6	6	6	6	6 6	6	6	6 6	6	6	6
FO	0	ě	ě	7	- 7	7	- 7	-		ė	E	7		-	7	0 7		7	e e	0	0	7
- 08	ь	ь	0	r	r	r I	0 /	5	ь	0	5	r .	ſ	r	r	0 /	Б	r	0 0	ь	6	
59	5	6	4	4	4	6	5 4	6	5	6	6	3	4	4	5	3 6	4	4	6 5	4	6	5
60	7	6	7	7	6	7	7 7	7	7	7	7	7	7	7	7	7 7	5	5	7 7	7	7	7
00	-				F			Ē		F	è	è			E .		0	F	7 0	-	-	
61	5	ь	ь	ь	2	0	5 6	5	ь	5	6	0	5	0	5	0 6	б	5	r 6	Б	5	6
62	6	5	6	6	7	7	6 7	7	6	7	6	6	6	6	6	6 6	5	6	5 5	6	5	6
63	6	7	5	6	5	7 1	6 7	4	5	6	5	7	6	7	6	5 6	7	5	6 7	6	4	7
04	0	-	0	7	0	7	7 5	- 7		7	7		0	7	7	0 7		7	0 7	0	- 7	- 7
69	0	5	0	ſ	0	ſ	1 3	ſ	0	ſ	r	0	0	r	r	0 (5	ſ	0 (0	ſ	ſ
65	5	5	6	5	4	5 1	6 6	5	6	5	6	6	6	5	5	6 5	6	5	6 6	5	6	7
66	6	7	6	7	7	7	7 7	4	5	4	4	6	6	7	4	4 7	7	7	6 6	6	6	6
07	6	e	4	E	4	- -	4 4	7	- 7	7	7	7	7	7	7	7 7	. 7	7	7 7	- 7	- 7	7
01	0	0	4	3	4	3	4 4	1		1	1	1	-	1	-	1 1	1	1	1 1		1	
68	(5	6	- (6	(6	5	5	5	4	5	6	5	3	6 5	3	6	2 5	4	4	5
69	7	5	6	4	7	7 1	6 5	5	7	6	7	5	6	3	7	5 6	7	6	4 4	3	2	4
70	ė	-	6	e	- -	e 1	- 7		- 7	-		- e	7	Ê	E	7 7		÷.	7 5	Ē	-	7
70	0	0	0	0	J	0	0 1	1		U	1	U		J	3	1 1	0	J	1 3	J	U	
71	6	7	6	7	6	7	46	7	7	7	7	6	5	6	7	7 7	6	7	6 7	6	7	6
72	5	6	5	4	6	5	4 5	4	5	4	4	5	5	5	5	5 4	5	5	4 5	4	5	5
72	8	5	5	5	5	6	4 4	5	5	5	8	4	5	5	5	5 5	5	5	8 8	8	8	6
10	0		3	3	3	·	4 4	J	3	3	0	4	3	3	3	J J	J	J	0 0	0	0	0
		_											-		_		-			-		
74		- 5			5	6 -		. 5		6	6	5	6	5	6 1		- 6		6 6	- 5		. 6
74		- 5	6	 6	5	6 !	5 6	5	. 6	6	6	5	6	5	6 6	6 6	- 6	6	6 6	5		6
74 75	 5 4	- 5 3	6	6 5	5 4	6 ! 4 :	5 6 3 6	5 2	6 5	 6 4	 6 3	5 3	6 5	5	6 6 4 5	6 6 5 4	- 6 5	 6 4	6 6 4 3	- 5	 5 5	6
74 75 76	 5 4 6	- 5 3 6	6 3 5	6 5 6	5 4 4	6 ! 4 :	5 6 3 6 4 6	5 2 4	6 5 5	6 4 5	 6 3 4	5 3 6	6 5 4	5 6 5	6 0 4 5 6 5	6 6 5 4 5 6	- 6 5 7	6 4 5	6 6 4 3 6 6	- 5 4 5	 5 5	6 3 5
74 75 76 77	 5 4 6	- 5 3 6	- 6 3 5 5	6 5 6 4	5 4 4 3	- 6 - 4 - 5 -	5 6 3 6 4 6 3 4	- 5 2 4 6	6 5 5 5	 6 4 5 5	6 3 4 5	5 3 6 3	6 5 4 4	5 6 5 5	6 6 4 5 5 5	6 6 5 4 5 6 5 4	- 6 5 7 3	6 4 5 5	6 6 4 3 6 6 4 4	5 4 5 3	 5 6 3	6 3 5 4
74 75 76 77 79	 4 6 6	- 5 3 6 6	6 3 5 5	6 5 6 4	5 4 4 3		5 6 3 6 4 6 3 4	- 2 4 6	5 5 5	6 4 5 5	6 3 4 5 2	5 3 6 3	6 5 4 4	5 6 5 5	6 (4 5 5 5	 5 4 5 6 5 4	6 5 7 3	6 4 5 5	6 6 4 3 6 6 4 4	- 4 5 3	 5 6 3	3 5 4
74 75 76 77 78	5 4 6 1	- 5 3 6 6 3	6 3 5 5 2	6 5 6 4 3	5 4 4 3 4	6 9 4 3 5 3	5 6 3 6 4 6 3 4 3 2	5 2 4 6 2	- 6 5 5 5 3	6 4 5 5 2	6 3 4 5 2	5 3 6 3 1	6 5 4 4 2	5 6 5 5 1	6 0 4 5 5 5 1 2	6 6 5 4 5 6 5 4 2 1	- 6 5 7 3 1	6 4 5 5 4	6 6 4 3 6 6 4 4 5 4	5 4 5 3	5 5 6 3 5	6 3 5 4 4
74 75 76 77 78 79	5 4 6 6 1 3	5 3 6 6 3 2	6 3 5 5 2 3	6 5 6 4 3 3	5 4 4 3 4 3	6 9 4 3 5 3 2 9	5 6 3 6 4 6 3 4 3 2 4 3	- 5 2 4 6 2 2 2	6 5 5 5 3 2	6 4 5 5 2 1	6 3 4 5 2 4	5 3 6 3 1 3	- 5 4 4 2 2	5 6 5 5 1 3	6 6 4 5 5 5 1 2	6 6 5 4 5 6 5 4 2 1 3 4	- 6 5 7 3 1 2	6 4 5 5 4 3	6 6 4 3 6 6 4 4 5 4 4 4	5 4 5 3 5 3	 5 6 3 5 4	- - - - - - - - - - - - - - - - - - -
74 75 76 77 78 79 80	 5 4 6 1 3 7	- 5 3 6 6 3 2 6	6 3 5 5 2 3 6	6 5 6 4 3 3 6	5 4 4 3 4 3 6	6 5 5 5 7 1	 5 6 3 6 4 6 3 4 3 2 4 3 5 7	5 2 4 6 2 2 4	5 5 3 2 6	6 4 5 5 2 1 5	 6 3 4 5 2 4 5 5	5 3 6 3 1 3 7	6 5 4 4 2 2 7	5 6 5 5 1 3 7	6 6 4 5 5 5 1 2 6	6 6 5 4 5 6 5 4 2 1 3 4 7 7	- 6 5 7 3 1 2 6	6 4 5 5 4 3 6	6 6 4 3 6 6 4 4 5 4 4 4 7 7	5 4 5 3 5 3 6	 5 6 3 5 4 7	- - - - - - - - - - - - - - - - - - -
74 75 76 77 78 79 80 81	 5 4 6 1 3 7 5	5 3 6 3 2 6 4	- 6 3 5 5 2 3 6 5	6 5 4 3 3 6 4	5 4 4 3 4 3 6 5	6 9 4 3 5 3 2 9 7 1	5 6 3 6 4 6 3 4 3 2 4 3 6 7 4 5	5 2 4 6 2 2 2 4 4	6 5 5 3 2 6 4		 6 3 4 5 2 4 5 4 5 4	5 3 6 3 1 3 7 5	6 5 4 4 2 2 7 4	5 6 5 5 1 3 7 5	6 6 4 5 5 5 2 5 6 5	6 6 5 4 5 6 5 4 2 1 3 4 7 7 4 5	6 5 7 3 1 2 6 5	6 4 5 5 4 3 6 4	6 6 4 3 6 6 4 4 5 4 4 4 7 7 7 4	- 5 3 5 3 6 4	 5 6 3 5 4 7 4	- - - - - - - - - - - - - - - - - - -
74 75 76 77 78 79 80 81	 4 6 1 3 7 5	- 5 3 6 6 3 2 6 4 4	6 3 5 2 3 6 5	6 5 6 4 3 3 6 4	5 4 4 3 4 3 6 5	6 9 4 3 5 3 7 1 5 4	5 6 3 6 4 6 3 4 3 2 4 3 6 7 4 5	5 2 4 6 2 2 2 4 4 4	6 5 5 3 2 6 4	6 4 5 2 1 5 4	6 3 4 5 2 4 5 4 5	5 3 6 3 1 3 7 5 5	6 5 4 4 2 2 7 4	5 5 5 1 3 7 5	6 6 4 5 5 5 7 2 5 5 6	6 6 5 4 5 6 5 4 2 1 3 4 7 7 7 7 4 5	6 5 7 3 1 2 6 5	6 4 5 5 4 3 6 4	6 6 4 3 6 6 4 4 5 4 4 4 7 7 7 4 4	5 4 5 3 5 3 6 4	5 5 6 3 5 4 7 4	6 3 5 4 4 4 7 3
74 75 76 77 78 79 80 81 81	5 4 6 1 3 7 5 7	5 3 6 3 2 6 4 7	6 3 5 2 3 6 5 7	6 5 4 3 3 6 4 7	5 4 4 3 4 3 6 5 4	6 9 4 5 5 5 7 1 5 9 4 9	5 6 3 6 4 6 3 4 3 2 4 3 6 7 4 5 5 5	5 2 4 6 2 2 4 4 4 4	6 5 5 3 2 6 4 5		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5	6 5 4 4 2 2 7 4 4	5 6 5 5 1 3 7 5 4	6 6 4 5 5 5 2 3 6 7 5 6 5 6	6 6 5 4 5 6 5 4 2 1 3 4 7 7 4 5 6 7	- 6 5 7 3 1 2 6 5 4	6 4 5 5 4 3 6 4 7	6 6 4 3 6 6 4 4 5 4 4 4 5 4 4 4 7 7 7 4 4 4 7	5 4 5 3 5 3 6 4 4	 5 6 3 5 4 7 4 6	3 5 4 4 4 7 3 5
74 75 76 77 78 79 80 81 81 82 83	5 4 6 1 3 7 5 7 4	- 5 3 6 3 2 6 4 7 5	6 3 5 2 3 6 5 7 4	6 5 6 4 3 3 6 4 7 4	5 4 4 3 4 3 6 5 4 5	6 9 4 3 5 3 7 1 5 4 4 9	5 6 3 6 4 6 3 4 3 2 4 3 6 7 5 5 5 5 4 6	5 2 4 6 2 2 2 4 4 4 4 4	6 5 5 3 2 6 4 5 6	6 4 5 2 1 5 4 4 6	6 3 4 5 2 4 5 4 5 4 4 5 5	5 3 6 3 1 3 7 5 5 5 6	6 5 4 4 2 2 7 4 4 4 6	5 6 5 5 1 3 7 5 4 6	6 6 6 5 5 5 6 5 6 5 6 5 6	6 6 5 4 5 6 5 4 2 1 3 4 7 7 4 5 6 7 6 5	- 5 7 3 1 2 6 5 4 5		6 6 4 3 6 6 4 4 5 4 4 4 5 4 4 4 7 7 7 7 4 4 4 7 5 6	5 4 5 3 5 3 6 4 4 4 6	 5 6 3 5 4 7 4 6 5	6 3 5 4 4 4 7 3 6 6
74 75 76 77 78 79 80 81 82 83 84	5 4 6 1 3 7 5 7 4 4	- 5 3 6 3 2 6 4 7 5 4	6 3 5 2 3 6 5 7 4 4	6 5 6 4 3 3 6 4 7 4 4	5 4 3 4 3 6 5 4 5 4	6 9 4 5 5 5 7 1 5 9 4 9 4 9	5 6 3 6 4 6 3 4 3 2 4 3 6 7 4 5 5 5 4 6 3 4	5 2 4 6 2 2 4 4 4 4 4 4 5	5 5 3 2 6 4 5 6 4 5 6 4	6 4 5 5 2 1 5 4 4 4 6 4	- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 5 6 3	6 5 4 4 2 2 7 4 4 6 4	5 5 5 1 3 7 5 4 6 4	6 (4 5 5 5 1 2 5 6 5 6 6 6 6 4	 5 4 5 6 5 4 2 1 3 4 7 7 7 7 4 5 6 7 6 5 4 4	- 6 5 7 3 1 2 6 5 4 5 4 5 3		6 6 4 3 6 6 6 4 4 4 4 7 7 7 4 4 4 4 7 5 6 4 5	5 4 5 3 5 3 6 4 4 4 5	 5 6 3 5 4 7 4 6 5 5	6 3 5 4 4 4 7 3 6 6 5
74 75 76 77 78 79 80 81 82 83 83 84	 4 6 1 3 7 5 7 4 4	- 5 3 6 6 3 2 6 4 7 5 4 5	6 3 5 2 3 6 5 7 4 4	6 5 6 4 3 6 4 7 4 4 5	5 4 4 3 4 3 6 5 4 5 4 5 4 5	6 9 4 5 5 5 7 1 5 4 4 9 4 9 4 9	5 6 3 6 4 6 3 4 3 2 4 3 5 7 4 5 5 5 4 6 3 4 5 5	5 2 4 6 2 2 4 4 4 4 5	5 5 3 2 6 4 5 6 4 5 6 4	6 4 5 2 1 5 4 4 4 6 4	- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 6 3 2	- 6 5 4 4 2 2 2 7 4 4 6 6 4	5 6 5 5 1 3 7 5 4 6 4 6 4	6 6 4 5 5 5 1 2 5 6 5 6 6 6 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7	 5 4 5 6 5 4 2 1 3 4 7 7 4 5 6 7 6 5 4 4	- 5 7 3 1 2 6 5 4 5 4 5	6 4 5 4 3 6 4 7 5 3 5 3	6 6 4 3 6 6 5 4 4 4 5 4 4 4 7 7 7 4 4 7 5 6 4 5 5 6 4 5	- 5 4 5 3 5 3 6 4 4 6 5	 5 6 3 5 4 7 4 6 5 5	6 3 5 4 4 4 7 3 6 6 6
74 75 76 77 78 79 80 81 82 83 83 84 85	5 4 6 6 1 3 7 7 5 7 7 4 4 4	- 5 3 6 6 3 2 6 4 7 5 4 5	6 3 5 2 3 6 5 7 4 4 4	6 5 6 4 3 3 6 4 7 4 4 5	5 4 3 4 3 6 5 4 5 4 5 4 5 5	6 4 6 5 5 5 7 1 7 1 4 9 4 9 3 1 3 1	5 6 3 6 4 6 3 4 3 2 4 3 6 7 4 3 5 5 5 5 4 6 3 4 5 5 5 5	5 2 4 6 2 2 4 4 4 4 4 5 4	6 5 5 3 2 6 4 5 6 4 4 4	6 4 5 2 1 5 4 4 6 4 4 4	- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 6 3 2	6 5 - 4 4 - 2 2 - 7 4 - 4 4 - 6 4 - 2	5 5 5 1 3 7 5 4 6 4 2	6 6 6 5 5 5 7 2 3 6 7 5 6 6 6 6 6 6 6 4 4	6 6 5 4 5 6 5 4 2 1 3 4 7 7 7 7 6 7 6 5 4 4 2 4	- 5 7 3 1 2 6 5 4 5 5 5 5		6 6 4 3 6 6 4 4 5 4 4 4 4 4 4 4 5 6 4 5 6 6 5 6 5 6	5 4 5 3 5 3 6 4 4 6 5 6	5 6 3 5 4 7 4 6 5 5 5 6	6 3 5 4 4 4 7 3 6 6 5 6
74 75 76 77 78 79 80 81 82 83 83 84 85 86	5 4 6 1 3 7 5 7 7 4 4 4 4 4	- 5 3 6 6 3 2 6 4 7 5 4 4 5 4	- 6 3 5 5 2 3 6 5 7 4 4 4 4 5	6 5 6 4 3 3 6 4 7 4 4 5 5 4	5 4 3 4 3 6 5 4 5 4 5 4 5 4	6 4 6 5 5 5 2 7 7 1 5 4 4 9 3 3 3 3 4 4 4 4	 5 6 3 6 3 4 3 4 3 4 3 2 4 3 5 7 5 5 5 5 4 6 3 4 5 5 5 5 4 6 3 4 4 6 3 4 4 6 3 4 4 6 4 6 5 5 5 5 4 6 5 5 4 6 5 5 4 6 5 5 4 6 5 5 5 4 6 7 6 7 6 7 6 7 6 7 6 7 6 7 6 7 6 7 6 7	5 2 4 6 2 2 2 4 4 4 4 4 5 4 5	- 6 5 5 3 2 6 4 5 6 4 4 4 4 4		6 3 4 5 2 4 5 4 5 4 4 5 4 4 5 4 5 5	5 3 6 3 1 3 7 5 5 6 3 3 2 3		5 5 5 1 3 7 5 4 6 4 2 3	6 6 6 5 5 5 6 5 6 5 6 6 6 6 6 6 6 2 3	6 6 5 4 5 6 5 4 2 1 7 7 7 4 5 6 7 7 4 5 6 7 6 5 4 4 3 4	- 6 5 7 3 1 2 6 5 4 5 4 5 3 5 4	6 4 5 4 3 6 4 7 5 3 5 3 5 4	6 6 4 3 6 6 4 4 5 4 4 4 5 4 4 4 7 7 7 7 4 4 4 7 5 6 4 5 5 6 4 5 5 6 4 5	5 4 5 3 5 3 6 4 4 6 5 6 4 4 4 6 5 6 4	5 5 3 5 4 7 4 6 5 5 5 5 5 5	3 5 4 4 4 7 3 6 5 5 6 4
74 75 76 77 78 79 80 81 82 83 84 85 86 87	 5 4 6 6 6 1 3 3 7 5 5 7 7 4 4 4 4 4 5	- 5 3 6 6 3 2 6 4 7 5 4 5 4 5 4 4 5	6 3 5 2 3 6 5 7 4 4 4 4 5 5	6 5 6 4 3 3 3 6 4 4 4 4 5 5	5 4 3 3 4 3 6 5 4 5 4 5 4 5 4 5 5	6 4 5 5 5 7 7 1 5 4 4 9 4 9 4 9 4 9 4 9 4 9 4 9 5 10 10 10 10 10 10 10 10 10 10 10 10 10 1	5 6 3 6 4 6 3 4 3 2 4 3 5 5 5 5 4 6 3 4 5 5 5 5 5 5 4 6 3 4 4 5 5 5 5 5 4 6 3 4 4 5	5 2 4 6 2 2 2 4 4 4 4 4 5 4 5 5 6	- 5 5 5 3 2 6 4 5 6 4 4 4 4 4 4 5		6 3 4 5 2 4 5 5 4 4 5 4 4 5 5 4 5 5 5 5 5 5	5 3 6 3 1 1 3 7 5 5 5 6 3 2 2 3 6	6 5 - 4 4 - 2 2 - 2 7 - 4 4 - 4 6 - 4 2 - 3 3 - 6	5 6 5 5 1 3 7 5 4 6 4 2 3 4	6 (4 5 5 5 5 6 5 6 6 6 4 4 2 3 3 3 5 6	6 6 6 5 4 5 6 5 4 2 1 3 4 7 7 7 4 5 6 6 7 6 5 4 4 2 4 2 4 3 4 6 5 5 4 5 6 5 4 5 6	- 5 7 3 1 2 6 5 4 5 3 5 4 6	6 4 5 4 3 6 4 7 5 3 5 3 5 4 6	6 6 4 3 6 6 4 4 5 4 4 4 5 4 4 7 7 7 4 4 4 7 7 7 4 4 5 6 4 5 5 6 4 5 5 6 4 5 6 4 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	5 4 5 3 5 3 6 4 4 4 6 5 6 4 6 6 4 6	5 5 6 3 5 4 7 4 6 5 5 6 6 6	6 3 5 4 4 4 7 3 6 6 5 6 6 6 6
74 75 76 77 78 80 81 82 83 84 85 86 87 89	 5 4 6 6 1 3 3 7 5 5 7 4 4 4 4 4 5	5 3 6 3 2 6 4 7 5 4 4 5 4 4 4 4	6 3 5 5 2 3 6 5 7 4 4 4 5 4 4 5	6 5 6 4 3 3 6 4 7 4 4 5 5 4 5 5	5 4 4 3 4 3 6 5 4 5 4 5 4 5 5 4 5 5 5 5 5 5 5 5 5 5	6 9 4 6 5 5 5 5 7 1 5 4 4 9 4 9 4 9 4 9 5 6	5 6 3 6 4 6 3 4 3 2 4 3 5 5 5 5 4 6 3 4 5 5 3 4 5 5 3 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 2 4 6 2 2 4 4 4 4 4 5 4 5 6 6	6 5 5 5 3 2 6 4 5 6 4 4 4 4 5 5	6 4 5 2 1 5 4 4 4 6 4 4 5 6 4 4 5 6		5 3 6 3 1 3 7 7 5 5 6 3 3 2 2 3 6 6 5		5 6 5 5 5 7 5 4 6 4 2 3 4 4	6 6 6 4 5 5 5 2 2 5 6 6 6 4 4 2 3 3 5 5 6	6 6 6 5 4 5 4 5 4 2 1 3 4 7 7 6 5 6 5 4 4 2 4 3 4 6 5 6 5 6 5 6 5 6 5 6 5 6 5 6 5 6 5 6 5	- 6 5 7 3 1 2 6 5 4 5 4 5 3 3 5 4 6 8	6 4 5 4 3 6 4 7 5 3 5 4 6 6	6 6 6 4 3 6 6 4 4 4 4 4 4 4 4 4 7 7 7 7 4 4 4 4 4 4 5 6 6 6 5 6 4 5 6 6 4 5 6 6	5 4 5 3 5 3 6 4 4 6 5 6 4 4 6 5	5 5 6 3 5 4 7 4 6 5 5 6 5 5 6 5 6	6 3 5 4 4 4 4 7 3 6 6 5 6 4 6
74 75 76 77 78 80 81 82 83 84 85 86 87 88 87 88	 5 4 6 6 1 3 7 7 7 4 4 4 4 4 4 5 5 5	- 5 3 6 6 3 2 6 4 7 5 4 5 4 4 5 4 4 4 4		6 5 6 4 3 3 6 4 7 4 4 5 5 5 5	5 4 3 4 3 6 5 4 5 4 5 4 5 5 4 5 5 5	6 9 4 3 5 5 2 9 4 9 4 9 4 9 4 9 5 9 5 9 6 9 6 9	5 6 3 6 4 6 3 4 3 2 4 3 5 7 4 5 5 5 4 6 3 4 3 4 3 4 3 4 4 3 5 5 5 4 6 3 4 4 5 5 5 5 6 3 4 4 5 5 5 6 6 3 6 7 7 4 5 5 5 6 6 3 6 7 7 4 5 5 5 5 6 6 6 3 6 7 7 4 6 7 7 4 5 5 5 5 6 6 6 7 7 4 6 7 7 7 6 7 7 7 6 7 7 7 7 7 7 7	5 2 4 6 2 2 4 4 4 4 4 5 4 5 6 6 4	- 6 5 5 5 3 2 6 4 5 6 4 4 4 5 5 5 5		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 1 3 7 5 5 6 3 2 2 3 6 5 5	6 5 4 4 2 2 2 7 4 4 4 6 4 2 3 6 5 5	5 6 5 5 1 3 7 5 4 6 4 2 3 4 4 4 4	6 6 6 4 5 5 5 6 6 5 6 6 6 4 4 3 3 5 6 5 6 6 7 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7	6 6 5 4 5 6 5 4 2 1 3 4 7 7 7 4 5 6 5 4 4 4 5 6 5 4 4 2 4 3 4 4 5 6 5 5 6 5 5	- 6 5 7 3 1 2 6 5 4 5 3 3 5 4 6 6 6	6 4 5 4 3 6 4 7 5 3 5 4 6 6 6 6	6 6 6 6 4 3 6 6 6 4 4 4 3 6 6 6 6 4 4 4 4	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5	55 63 55 47 46 55 65 65 66	6 3 5 4 4 4 7 3 6 6 5 6 6 4 6 6 6
74 75 76 77 78 79 80 81 82 83 84 85 86 85 86 87 88 89	5 4 6 6 1 3 7 7 5 7 4 4 4 4 4 5 5 4	5 3 6 3 2 6 4 7 5 4 4 5 4 4 4 4 4 4	6 3 5 5 2 3 6 5 7 4 4 4 5 4 4 5 4 4 6 4	6 5 6 4 3 3 6 4 7 4 4 5 5 4 5 5 4	5 4 4 3 4 3 6 5 4 5 4 5 4 5 5 5 5 5 5	6 9 5 5 5 5 7 1 5 4 4 9 4 9 4 9 4 9 5 9 6 9 6	5 6 3 6 4 6 3 2 4 3 5 7 4 5 5 5 5 5 5 5 3 4 5 5 5 5 3 4 5 5 5 5 6 5 6	5 2 4 6 2 2 4 4 4 4 4 5 5 6 6 4 5 5	- 6 5 5 5 3 2 6 4 5 6 4 4 5 5 5		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 6 3 2 3 6 5 5 5 5	- 5 4 4 2 2 7 4 4 4 5 4 4 2 2 7 4 4 4 5 5 5 5 5	5 6 5 5 1 3 7 5 4 4 6 4 2 3 4 4 5	6 6 6 5 5 2 2 3 5 6 6 5 6 6 6 4 4 2 3 5 6 5 6 5 6 5 5 6 5 5 5 6 5 5 5	6 6 6 5 4 5 6 5 4 2 1 3 4 2 1 3 4 7 7 7 7 7 7 6 5 5 5 5 5 5 5	- 6 5 3 1 2 6 5 4 5 4 5 4 5 4 6 6 5 5	6 4 5 5 4 3 6 4 7 5 5 4 6 5 4 5 5 5	$\begin{array}{c} 6 \\ 6 \\ 4 \\ 3 \\ 6 \\ 6 \\ 6 \\ 6 \\ 4 \\ 4 \\ 4 \\ 4 \\ 4 \\ 4$	5 5 5 5 5 5 5 6 4 4 6 5 6 4 6 5 5 5 5 5	5563544746556565655655655	6 3 4 4 4 4 7 3 6 5 6 6 4 6 6 6 6
74 75 76 77 78 80 81 82 83 84 85 86 85 86 87 88 88 89 90	5 4 6 6 1 3 7 5 7 4 4 4 4 4 5 5 5 4 4	- 5 3 6 6 3 2 6 4 7 5 4 4 5 4 4 4 4 4 4 4 6			5 4 4 3 4 3 6 5 4 5 4 5 5 4 5 5 5 5	6 9 4 3 5 5 3 2 9 4 9 5	5 6 6 3 6 4 3 6 4 3 4 3 2 4 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 2 4 6 2 2 4 4 4 4 4 4 5 5 6 6 4 5 5	- 6 5 5 5 5 3 2 6 4 4 4 5 5 5 6			5 3 6 3 1 3 7 5 5 6 3 2 2 3 6 5 5 5 5 5 5	6 5 4 2 2 2 7 4 4 4 6 6 4 2 3 6 5 5 6	5 6 5 5 1 3 7 5 4 6 4 2 3 4 4 4 5 5 5	6 6 6 5 5 5 1 2 2 5 6 6 5 6 6 5 6 6 6 5 6 5 6 5 6 5 6 5 6	6 6 5 4 5 6 5 4 2 1 3 4 7 7 7 7 6 5 6 5 6 5 6 5 6 5 5 5 6 7	6 5 7 3 1 2 6 5 5 4 5 5 4 5 5 4 6 6 5 4	6 4 5 5 3 6 4 7 5 3 5 4 6 6 6 6	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6	5 6 3 5 4 7 4 6 5 5 6 5 6 6 5 6 6 5 6	6 3 5 4 4 4 4 4 3 6 6 5 6 4 6 6 7 7
74 75 76 77 78 80 81 82 83 84 85 86 87 88 87 88 89 90 91	5 4 6 6 7 7 5 7 7 4 4 4 4 4 5 5 4 4 4 4	- 5 6 6 3 2 6 4 7 5 4 4 4 4 4 4 4 6	6 3 5 5 2 3 6 5 7 4 4 4 5 4 4 5 4 4 5 4 4 5	6 5 6 4 3 3 3 6 4 4 4 5 5 4 4 5 5 4 4 5 5 5 5 5 5 5 5	5 4 4 3 4 3 6 5 4 5 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5	6 9 6 9 5 5 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1	5 6 3 6 4 6 3 2 4 3 5 5 5 5 5 5 5 5 5 5 5 6 5 6 5 6 5 6 5 6	5 2 4 6 2 2 4 4 4 4 4 5 5 6 4 5 5 6 4	6 5 5 3 2 6 4 5 6 4 4 4 4 5 5 5 6 4		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 6 3 2 3 3 6 5 5 5 5 5 4	- 6 5 4 4 2 2 2 7 4 4 6 4 2 3 6 5 5 5 5 5 5 5 5	5 6 5 5 1 3 7 5 4 6 4 2 3 4 4 5 5 5 4 4 5 5 4	6 6 6 5 5 5 5 6 5 6 6 6 6 6 6 6 7 5 5 6 6 6 5 5 6 6 6 6 6 6 6 6 6 7 6 7 6 7 6 7 7 6 7 7 7 7	6 6 6 5 4 5 6 5 4 2 1 3 4 4 5 6 5 5 5 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 7 3 1 2 6 5 5 5 5 5 5 5 5 4 6 6 6 5 5 4 4 4	6 4 5 5 4 4 3 6 4 7 5 5 4 4 6 6 6 5 5 4	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	5 3 5 3 6 4 4 6 5 5 6 4 6 5 5 6 4	556354746556656656565	6 3 5 4 4 4 4 4 7 3 6 6 5 6 6 6 6 6 6 6 7 5
74 75 76 77 78 80 81 82 83 84 85 86 85 86 87 88 89 90 91	5 5 6 6 6 1 3 7 7 5 7 4 4 4 4 5 5 5 4 4 4 4 4 4	- 5 3 6 6 3 2 6 4 7 5 4 4 5 4 4 4 4 4 4 6 6	6 3 5 2 3 6 5 7 4 4 4 5 7 4 4 5 4 4 5 4		5 4 4 3 6 5 4 3 6 5 4 5 5 5 5 5 5 5 5	6 4 5 5 2 0 7 1 5 4 4 4 5 5 5 4 4 4 5 5 5 6 6 9 6 9 6 9 6 9 6 9 6 9 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1	5 6 3 6 4 6 3 4 3 4 3 2 4 3 5 5 5 5 6 6 5 5 5 6 6 5 5 5 2 4 4 5 5 4 4 5 5 5 6 6 5 5 5 4 4 5 5 4 4 5 5 5 6 4 6 7 7 4 6 6 7 4 7 5 7 5 7 6 7 6 7 7 7 7 7 7 7 7 7 7 7 7	5 2 4 6 2 2 4 4 4 4 4 4 4 5 5 6 4 5 5 6 4	6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 5 6 3 2 2 3 6 5 5 5 5 5 4	6 5 4 2 2 2 7 4 4 6 5 5 5 6 5 5 6 5	5 6 5 1 3 7 5 4 6 4 2 3 4 4 2 5 5 4 4	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	6 6 6 5 4 5 6 5 4 2 1 3 4 7 7 7 7 4 5 6 7 6 7 6 7 6 7 6 5 5 4 4 4 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 3 1 2 6 5 3 5 3 5 4 6 6 6 5 4 4 4 4 4	6 4 5 5 4 3 6 4 7 5 5 3 5 4 6 6 6 5 5 6 6 6 6 6 6 6 6 6 6 6 6 6	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4	55 63 54 74 65 56 65 65 65 65 55 65 55	6 3 5 4 4 4 4 7 3 6 6 5 6 6 5 6 6 7 5
74 75 76 77 78 80 81 82 83 84 85 86 85 86 87 88 89 90 91 92	5 4 6 6 7 3 7 5 7 4 4 4 4 4 5 5 4 4 4 4 4 4 4	- 5 6 6 3 2 6 4 7 5 4 4 4 4 4 4 6 5 5	6 3 5 2 3 5 5 7 4 4 4 4 5 4 4 5 4 4 5 4 4 4		5 4 4 3 6 5 4 5 5 4 5 5 4 5 5 5 5 5 3	6 4 4 5 2 7 1 4 4 5 6 1 6 1 6 4 4 4 4 5 5 6 1 5 6 1 5 6 1 5 6 1 5 6 1 5 5 6 1 5 5 6 6 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7	5 6 3 6 4 6 3 2 4 3 5 5 5 5 4 5 5 5 5 6 5 6 5 6 5 6 5 6 5 6 5 5 5 6 5 5 5 5	5 2 4 6 2 2 4 4 4 4 4 5 5 6 4 5 5 5 4 4 4 4	6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 3 7 5 5 5 5 6 3 2 3 6 5 5 5 5 4 4	6 5 4 2 2 7 4 4 4 6 4 2 3 6 5 5 5 4	5 5 5 5 1 3 7 5 4 6 6 4 2 3 4 4 5 5 5 4 4 4 4 4 4 4 4 4 4	6 6 6 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6	6 6 6 5 4 5 6 2 1 3 4 4 5 6 7 6 5 4 4 2 4 3 4 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 7 3 1 2 6 5 5 4 5 5 4 6 6 5 5 4 4 6 5 5 5 4 4 5 5 5 5	6 4 5 5 4 3 6 4 7 5 3 5 4 6 6 6 6 6 4 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 3 5 3 6 4 4 6 5 5 6 4 6 5 5 6 4 4 4	55 6 35 4 7 4 6 5 6 5 6 5 6 5 6 5 6 5 4 4	- 6 3 5 4 4 4 7 3 6 6 5 6 6 5 6 6 6 7 5 4
74 75 76 77 78 80 81 82 83 84 85 86 87 88 85 86 87 88 89 90 91 92 93	5 4 6 6 1 3 7 7 5 7 7 4 4 4 4 4 4 4 4 4 4 4 4 3	- 5 6 6 3 2 2 6 4 7 5 4 5 4 4 4 4 4 4 4 6 5 5 5	6 3 5 2 3 6 5 7 4 4 4 5 4 4 5 4 4 5 4 4 4 4 4		5 4 3 4 3 6 5 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 6 5 5 5 2 7 7 1 5 4 4 9 4 9 4 9 6 9 6 9 6 9 6 9 6 9 6 9 6 9 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1	5 6 3 6 4 6 3 4 3 4 3 2 4 3 5 5 5 5 5 5 6 6 5 5 5 6 5 5 6 6 5 5 2 4 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5	5 2 4 6 2 2 4 4 4 4 5 5 4 5 5 4 4 4 4 4	6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 7 5 5 5 6 3 2 2 3 6 5 5 5 5 5 5 4 4 4	- 6 5 4 4 2 2 7 4 4 6 4 2 3 6 5 5 6 5 4 3	5 6 5 5 5 5 7 5 4 6 4 2 3 4 4 5 5 4 4 5 5 5 4 5 5 5 5 5 5 5 5 5	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	6 6 6 5 4 5 6 5 6 5 6 5 6 5 4 2 1 2 1 2 1 3 4 4 7 7 7 7 7 7 7 6 6 7 6 6 5 5 5 4 4 4 4 4 2 4 4 6 5 5 5 5 5 5 5 5 5 5	6 5 7 3 3 2 6 5 4 5 5 4 5 5 4 6 5 5 4 4 5 5 4 4 5 5 4 6 5 5 6	6 4 5 5 4 3 6 4 4 7 5 3 5 4 4 6 6 5 5 4 5 5 4 5 5 5 5 5 5 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 5 6 4 4 6 5 5 6 4 4 6	5 6 3 5 4 7 4 6 5 5 6 5 6 5 6 5 5 4 5 5 6 5 5 4 5 5 5 5	6 3 5 4 4 4 4 7 3 6 6 5 6 6 6 7 7 5 5 4 5
74 75 76 77 78 79 80 81 82 83 84 85 86 85 86 87 88 89 90 91 92 93 93	5 4 6 6 1 3 7 5 7 4 4 4 4 4 4 4 4 4 4 4 4 3 3	- 5 6 6 3 2 2 6 4 7 5 4 4 4 4 4 4 4 4 4 6 6 5 5 5 5	6 3 5 2 3 5 5 7 4 4 4 4 5 4 4 5 4 4 4 5 4 4 4 4 4		5 4 4 3 6 5 4 5 5 4 5 5 5 5 5 5 5 5 3 3 3 3	6 4 4 5 5 7 1 4 4 5 6 1 6 1 6 1 5 4 4 5 5 6 1 5 6 1 5 6 1 5 6 1 5 6 1 5 5 6 6 6 7 7 1 7 7 7 1 7 7 7 7 7 7 7 7 7 7 7 7 7	5 6 3 4 6 3 4 6 3 4 3 3 4 4 3 2 4 3 2 4 3 2 4 5 7 4 5 5 5 5 6 5 6 5 6 5 6 5 6 5 6 5 5 5 6 5 6	5 2 4 6 2 2 4 4 4 4 4 5 5 6 4 5 5 5 4 4 4 4 3	6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 6 4 4 4 5 5 5 6 4 4 4 4			5 3 6 3 1 3 7 5 5 6 3 2 3 6 5 5 5 4 4 4 3 3 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 4 4 2 2 7 4 4 4 2 2 7 4 4 4 2 3 6 5 5 5 5 6 5 5 4 3 4 3	5 5 5 5 1 3 7 5 4 6 4 2 3 4 4 5 5 4 4 5 5 4 4 5 5 4 4 5 5 5 4 4 5	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	$ \begin{array}{c} $	6 5 7 3 1 2 6 5 4 5 5 4 6 6 6 5 4 4 4 5 5 4 4 4 4 4	6 4 5 5 5 4 4 3 6 4 4 7 5 3 5 5 4 6 6 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 3 5 3 6 4 4 6 6 5 5 6 4 4 6 6 5 5 6 4 4 6 6 4 4 6 6 5 5 6 4 4 6 6 5 5 6 6 4 6 6 5 5 6 6 4 6 6 6 6	5563547746555665565565565565555555555555	- - - - - - - - - - - - - - - - - - -
74 75 76 77 78 80 81 82 83 83 84 85 86 87 88 88 89 90 90 91 92 93 94 95	5 4 6 6 1 3 7 7 4 4 4 4 4 4 4 4 4 4 3 3 3	- 5 6 6 3 2 6 4 7 5 4 4 4 4 4 4 4 6 5 5 5 4	6 3 5 2 3 6 5 7 4 4 4 4 5 4 4 5 4 4 4 5 4 4 4 3 2		5 4 3 4 3 6 5 4 5 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 6 3 4 4 6 3 4 4 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 2 4 6 2 2 4 4 4 4 5 5 4 5 5 4 4 4 3 3	6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 5 5 2 1 5 4 4 4 4 6 6 5 6 6 7 4 6 5 5 5 6 7 4 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	- · · · · · · · · · · · · · · · · · · ·	5 3 6 3 1 3 7 5 5 6 3 2 3 6 5 5 5 5 5 4 4 4 3 4 4 3 4	- 6 5 4 4 2 2 7 4 4 6 4 2 2 7 4 4 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 6 5 5 6 6 6 6 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7	5 5 5 5 5 5 5 7 5 4 6 4 4 6 4 4 5 5 4 4 4 5 5 5 4 6 4 4 5 5 5 5	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	6 6 5 4 5 4 2 1 3 4 5 5 6 5 7 7 6 5 5 5 6 5 5 5 6 5 5 5 5 5 5 5 6 5 5 5 6 5 6 5 6 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	6 5 7 3 3 2 6 5 4 5 4 5 4 5 4 6 5 4 4 5 4 4 5 4 4 5 4 4 5 4 4 5 5 4 4 5 5 4 4 5 5 4 6 5 5 4 6 5 5 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	6 4 5 5 4 3 6 4 4 7 5 3 3 5 4 6 6 6 5 5 6 4 5 5 5 8 4 4 5 5 3 4	$\begin{array}{c} 6 \\ 6 \\ 4 \\ 3 \\ 6 \\ 6 \\ 4 \\ 4 \\ 4 \\ 4 \\ 7 \\ 7 \\ 7 \\ 4 \\ 4 \\ 7 \\ 7$	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 4 6	55635477465566655455 663566554555555555555555555	6 3 5 4 4 4 4 7 7 3 6 6 5 5 6 4 6 6 6 7 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6
74 75 76 77 78 79 80 81 82 83 84 85 86 85 86 87 88 89 90 91 92 92 93 94 95	5 4 6 6 1 3 7 5 7 4 4 4 4 4 5 5 4 4 4 4 4 3 3 4	- 5 6 6 3 2 2 6 4 7 5 4 5 4 4 4 4 4 4 6 6 5 5 5 5 4	6 3 5 2 3 6 5 7 4 4 4 4 5 4 4 5 4 4 4 4 4 3 3	6 5 6 4 3 3 6 4 4 5 5 5 5 5 4 4 4 5 5 4 4 4 5 5 4 4 4 4 4 4 4 4 4	5 4 3 4 3 6 5 4 5 5 5 5 5 5 5 5 5 3 3 3 4	6 4 5 5 2 4 4 5 5 7 7 1 5 4 4 4 4 5 5 5 6 8 5 5 6 8 5 5 6 8 5 5 5 5 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1	5 6 3 4 6 3 4 7 3 4 3 5 7 4 5 5 5 5 4 6 5 5 5 5 6 5 6 5 6 5 6 5 6 5 6 5 6 5 6	5 2 4 6 2 2 4 4 4 4 5 5 5 6 4 5 5 5 4 4 4 4 3 3	6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 5 5 2 1 5 4 4 4 4 5 6 6 5 6 6 7 4 6 3 3 5		5 3 6 3 1 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 4 4 2 2 7 4 4 2 2 7 4 4 4 2 2 7 4 4 4 2 5 5 5 5 5 5 5 5 4 3 3 4 5	5 5 5 5 1 3 7 5 4 6 4 2 3 4 4 5 5 5 4 4 5 5 5 4 4 5 5 5 5 5 5 5	6 6 9 6 9 5 5 9 1 2 2 6 6 9 5 6 6 6 5 7 5 7 5 7 5 7 5 7 5 7 6 8 6 9 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	$ \begin{array}{c} $	6 5 7 3 2 6 5 4 5 5 4 5 6 6 6 5 4 4 4 5 5 4 4 4 4	6 4 5 5 5 4 4 3 6 4 4 5 5 3 5 4 6 6 5 5 6 4 4 5 5 3 4	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 3 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 6 4 6 6 4 6 6 4 6 6 4 6 6 4 6 6 4 6 6 4 6	5563547465566665654555 655655654555555555555	6 3 5 4 4 4 4 4 7 3 6 6 5 6 6 6 7 5 4 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6
74 75 76 77 78 79 80 81 82 83 84 85 86 87 88 83 90 91 92 93 94 95 95 96	5 4 6 6 1 3 7 7 4 4 4 4 4 4 4 4 4 4 4 4 3 3 3 2 2	- 5 6 6 3 2 6 4 4 7 5 4 4 4 4 4 4 6 5 5 5 4 4 4 3	6 3 5 5 2 3 6 5 7 4 4 4 4 5 4 4 6 4 4 4 3 3 3	6 5 6 4 3 3 6 4 4 4 5 4 5 5 4 5 5 4 4 4 4 4 4 4 4 4	5 4 3 3 4 3 6 5 4 5 4 5 5 5 5 5 5 5 3 3 3 3 3 4 3	6 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		- 5 2 4 6 2 2 4 4 4 4 4 5 5 5 5 5 5 5 5 4 4 4 4 3 3 3 4	6 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 5 5 2 1 5 4 4 4 4 6 6 5 6 6 7 4 6 5 5 5 3 3		5 3 6 3 1 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 4 4 4 4 4	6 5 4 4 2 2 7 4 4 6 4 2 3 6 5 5 6 5 5 6 5 5 5 5 5 5	5 5 5 1 3 7 5 4 6 4 2 3 4 4 5 5 4 4 5 5 4 5 5 4 5 5 5 5 5 5 5	6 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	$ \begin{array}{c} 6 \\ 6 \\ $	- 6 5 5 7 3 1 2 2 6 6 5 5 4 4 5 5 4 4 6 6 5 5 4 4 4 5 5 4 4 4 3	6 4 5 5 4 4 3 6 4 7 5 5 3 5 5 4 6 6 6 5 5 5 6 4 5 5 5 4 4 5 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 4 6 5	5563547746556656656545555555555555555555	6 3 5 4 4 4 4 4 7 3 6 6 5 6 6 6 6 7 5 5 4 6 6 6 6 6 6 6 6 6 6 6
74 75 76 77 78 79 80 81 82 83 84 85 86 83 84 85 86 89 90 91 90 91 92 93 94 95 95 97	5 4 6 6 7 7 5 7 7 4 4 4 4 4 5 5 5 4 4 4 4 4 4 4	- 5 3 6 6 3 2 6 4 7 5 4 4 4 4 4 4 4 6 5 5 5 4 4 4 3 3	6 3 5 2 6 5 7 4 4 4 5 4 6 4 6 4 7 4 3 3 3 3 3 3	6 6 4 3 3 6 4 4 4 5 5 5 5 5 4 4 5 5 5 4 4 4 4 4 4	5 4 3 4 3 6 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5 3 3 3 3 4 3 3	6 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 6 3 6 4 6 3 4 3 2 4 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 5 5 2 1 5 4 4 4 4 4 5 6 6 5 6 6 7 7 4 6 3 3 3		5 3 6 3 1 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 4 4 2 2 7 4 4 6 4 2 3 6 5 5 5 6 5 5 5 6 5 5 5 4 4 3 6 5 5 5 5 4 4 2 2 7 4 4 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 6 5 5 1 3 7 5 4 6 4 4 5 5 4 4 5 5 4 4 5 5 4 4 5 5 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5	6 6 9 6 9 5 5 9 5 6 6 9 5 6 6 9 5 6 6 9 4 4 9 5 6 6 9 5 6 6 9 4 4 9 5 5 9 5 6 6 9 6 9 5 5 9 6 9 6 9 6 9 6 9 6 9 6 9 6 9 6	$ \begin{array}{c} 6 \\ 6 \\ $	- 6 7 3 1 2 5 5 4 5 5 4 6 6 6 6 6 4 4 4 4 3 3 3	6 4 5 5 4 3 6 4 3 6 4 7 7 5 5 5 5 4 6 6 5 5 6 4 4 5 5 5 5 4 4 4 4	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 5 6 5 5 6 5 5 5 6 5 5 5 5	5 5 6 3 5 4 7 4 6 5 5 6 5 6 6 5 6 5 4 5 5 5 5 5 5	6 3 5 4 4 4 4 7 3 6 6 5 6 6 6 6 7 5 4 5 6 6 5 5 6 6 5 5 5 4 5 5 5 6 6 5 5 5 5
74 75 76 77 78 79 80 81 82 83 83 84 85 86 87 88 88 89 90 91 92 93 94 95 96 97 90	5 4 6 6 7 7 5 7 4 4 4 4 4 4 5 5 4 4 4 4 4 4 4 4	5 3 6 3 2 6 4 7 5 4 4 4 4 4 4 4 6 5 5 5 5 5 4 4 4 3 3 4	6 3 5 5 2 3 6 5 7 4 4 4 5 5 4 4 4 5 4 4 4 3 3 3 3 3 3 3	6 5 6 4 3 3 6 4 4 5 4 4 5 5 4 4 5 5 4 4 4 4 4 4 4 4	5 4 3 4 3 6 5 4 5 4 5 5 5 5 5 5 5 3 3 3 3 4 3 3 4 3 3 4 3 4	6 4 5 5 7 7 1 6 7 7 1 7 7 1 7 7 1 7 7 7 1 7 7 7 1 7 7 7 7 1 7 7 7 7 7 1 7	5 6 3 6 3 4 6 3 4 5 7 5 5 5 5 5 5 5 5 5 6 6 6 5 5 5 6 6 6 5 5 5 6 6 6 5 5 5 6 6 6 6		- 6 5 5 5 5 3 2 6 4 5 6 4 4 4 5 5 5 6 4 4 4 4 3 3 4	6 4 5 5 2 1 5 4 4 4 4 4 4 4 5 6 6 5 6 6 7 7 4 6 3 5 3 3 3 4		5 3 6 3 1 3 7 5 5 5 6 3 2 3 6 5 5 5 5 5 4 4 4 4 3 4 4 3 4 4 3 4 4 3 4 4 3 4 4 3 4 4 3 4 4 3 4 4 3 4 5 5 5 5	6 5 4 4 2 2 7 4 4 6 4 2 3 6 5 5 6 5 5 6 5 5 4 3 4 5 5 3 4	5 5 5 5 1 3 7 5 4 6 4 2 3 4 4 5 5 4 4 5 5 4 4 5 5 4 3 7 8 4 4 5 5 5 8 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	- 6 5 5 7 3 1 2 2 6 6 5 5 4 4 5 5 4 4 5 5 4 4 4 5 6 6 6 5 4 4 4 3 3 3 3	6 4 5 5 5 4 4 3 6 4 4 5 5 5 5 5 5 5 6 6 6 5 5 5 5 5 5 4 4 5 5 5 4 4 3 3 4 4 3 3 4 4 3 3 4 4 3 3 5 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 5 4 5 5 6 5 5 5 6 5 5 5 6 5 5 5 6 5 5 5 6 5	5 5 6 3 5 4 7 4 6 5 5 6 5 6 6 5 6 5 4 5 5 5 5 5 5 5 5	6 3 5 5 4 4 4 4 7 3 6 6 5 6 6 6 6 6 6 6 6 6 6 6 6 6 5 4 5 5 4 5 6 6 6 6
74 75 76 77 78 79 80 81 82 83 84 85 83 84 85 86 87 88 86 87 88 89 90 91 92 93 94 95 95 96 97 98	5 4 6 6 7 7 5 7 7 4 4 4 4 4 5 5 4 4 4 4 3 3 4 2 3 3 7 5 7 7 5 7 7 4 4 4 5 5 5 7 7 4 5 7 7 5 7 7 5 7 7 5 7 7 4 5 8 6 7 7 7 7 5 7 7 7 7 5 7 7 7 7 7 7 7 7 7	- 5 3 6 6 3 2 6 4 7 5 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	6 3 5 5 7 3 6 5 7 7 4 4 4 4 5 4 4 5 4 4 4 3 3 3 3 3 3 3 4	6 5 6 4 3 6 4 5 5 5 5 5 4 5 4 5 4 4 4 4 4 4 4 4 3 3	5 4 3 6 5 4 3 6 5 4 5 5 5 5 5 5 5 3 3 4 3 4 3 4 3 4 3 4	6 4 5 5 5 7 7 1 6 6 5 7 7 1 7 7 1 7 7 7 1 7 7 7 7 1 7 7 7 7	5 6 3 4 6 3 3 4 3 3 3 4 5 5 5 5 5 5 5 5 5 5 5 5 5		6 5 5 5 5 5 3 2 6 4 5 5 5 5 5 2 6 4 4 4 5 5 5 5 5 5 5 2 6 4 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 4 5 5 2 1 5 4 4 6 6 5 5 6 7 7 4 6 6 5 5 3 3 3 4		5 3 6 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 5 4 4 2 2 7 4 4 2 2 7 4 4 6 5 5 5 5 6 6 5 5 5 4 3 4 5 5 5 4 4 4 5 5 5 4 4 4 2 2 7 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	$ \begin{array}{c} $	- 6 7 3 1 2 5 5 4 5 5 4 4 6 6 5 4 4 4 5 5 4 4 4 5 6 4 4 3 3 3 3	. 6 4 5 5 5 4 4 3 6 4 4 7 5 5 5 4 6 6 6 6 6 6 6 6 5 5 5 3 4 4 4 3 3 4 4 3 3 4 4 4 3 3 4 4 5 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 5 6 4 6 5 5 6 4 4 6 5 5 6 6	5 5 6 3 5 4 7 4 6 5 5 6 5 6 6 5 6 5 4 5 5 5 5 5 5 5 5	6 3 5 4 4 4 7 3 6 6 5 6 4 6 6 6 7 5 4 5 6 6 6 6 6 6 6 7 5 4 4 5 6 6 6 6 6 6 6 6 6 6 6 6 6
74 75 76 77 78 80 81 82 83 84 85 85 86 87 88 85 86 87 89 90 91 92 93 93 94 95 96 97 98 89 93	5 4 6 6 7 7 5 7 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	- 5 3 6 4 7 5 4 4 4 4 4 6 5 5 5 4 4 4 4 3 3 4	- 6 3 5 5 5 2 3 6 5 7 4 4 5 4 4 5 4 4 5 4 4 4 5 4 4 3 3 3 3 3	6 5 6 4 3 3 6 4 4 5 5 4 4 5 5 4 4 4 5 5 4 4 4 4 4 4	5 4 3 4 3 6 5 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 - 6 - 5 - 7 - 7 - 4 - 4 - 5 - 6 - 3 - 4 - 5 - 6 - 5 - 4 - 5 - 3 - 4 - 3 -	5 6 3 6 4 6 3 4 5 5 5 5 5 5 5 5 5 5 5 5 5 6 5 5 5 6 5 5 5 5	5 2 4 6 2 2 4 4 4 4 4 4 4 5 5 6 4 4 4 4 3 3 3 4 3 3 4	- 6 5 5 5 5 3 2 6 4 5 6 4 4 4 5 5 5 6 4 4 4 4 3 3 4 5	6 4 5 5 2 1 5 4 4 4 4 4 4 4 5 6 5 6 5 6 5 6 5 7 4 6 5 3 3 3 3 4 4		5 3 6 3 1 3 7 5 5 6 3 2 3 6 5 5 5 5 4 4 4 3 4 4 3 4 4	6 5 4 4 2 2 7 4 4 6 4 2 2 7 4 4 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 4 4 5	5 6 5 5 1 3 7 5 4 6 4 2 3 4 4 5 5 4 4 5 5 4 4 5 5 5 5 5 5 5 5 5	6 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	- 6 7 3 1 2 6 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	- 6 4 5 5 4 3 6 4 7 5 3 5 4 4 5 5 6 6 6 6 6 4 4 5 5 5 3 4 4 5 5 5 5 5 5 5 5 5 5 5 5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 6 6 5 5 6 6	5563547465566565656555555555555555555555	6 3 5 5 5 4 4 4 4 4 4 4 4 7 3 3 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6
74 75 76 77 78 80 80 81 82 83 84 85 86 87 88 86 87 88 89 90 91 92 93 93 94 95 95 95 95 95 95	5 4 6 6 7 5 7 7 5 7 7 4 4 4 4 4 4 4 4 4 4 4 4 4	- 5 3 6 3 2 6 4 4 4 4 4 4 4 6 5 5 5 4 4 4 4 4 3 3 4 4 3	6 3 5 5 3 6 3 6 7 4 4 4 4 5 4 4 4 5 4 4 4 3 3 3 3 3 4 5 5 2 2	6 5 6 4 3 3 8 6 4 7 7 4 4 4 5 5 5 5 5 4 4 5 5 5 5 5 5 4 4 5 5 5 5 1 1	5 4 3 6 5 4 3 6 5 4 5 5 5 5 5 5 5 5 3 3 3 4 3 4 4 3 3 4 4 3 3 4 4 3 3 4 4 3 3 5 5 5 5	6 - 5 - 5 - 5 - 7 1 5 - 4 - 4 - 5 - 6 - 6 - 6 - 5 - 5 - 5 - 5 - 5 - 5 - 3 - 4 - 3 - 4 - 3 - 4 - 3 - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 5 - 5 - 5 - 5 - 4 - 4 - <tr td=""> <th>5 6 3 4 6 3 3 4 3 3 4 3 3 4 4 5 5 5 5 5 5 5 5 5 5</th><th>2 4 6 2 2 4 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 4 4 4 3 3 4 3 3 4 3</th><th>6 5 5 5 5 5 3 2 6 4 5 5 5 5 3 2 6 4 4 4 4 5 5 5 5 5 5 2 6 4 4 4 4 4 4 4 4 4 4 3 3 4 5 5 2 2</th><th>6 4 5 5 2 1 5 4 4 4 4 5 6 6 5 7 7 4 6 5 5 3 3 4 4 4 3</th><th></th><th>5 3 6 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5</th><th>- 6 5 4 4 2 2 7 4 4 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 7 4 4 6 5 5 7 4 4 6 6 5 5 7 4 4 6 6 5 5 7 7 4 4 6 6 5 5 7 7 4 6 6 5 5 7 7 4 7 7 7 7 4 7 7 7 7 7 7 7 7 7 7</th><th>5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5</th><th>6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6</th><th>$\begin{array}{c}$</th><th>- 6 5 7 1 2 6 5 4 5 5 4 6 6 6 6 6 5 4 4 4 5 5 4 4 4 3 3 3 3 4 2</th><th>6 4 5 4 4 5 5 3 3 5 5 5 5 5 5 5 6 6 6 6 5 5 5 5 5</th><th>6 6 6 6 4 3 4 4 5 4 4 4 7 7 7 4 4 7 5 6 5 6 5 6 6 5 6 5 7 7 7 6 6 5 5 4 6 5 5 4 6 5 5 4 6 5 6 4 7 7 6 4 5 5 6 4 5 3 4 5 5 5 6 5 5 5 6 5 6 5 5 5 6 5 6 5 6 5 7 6 6 5 7 6 6 5 7 6 6 7 6<th>5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 6 5 5 6 6 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 5 5 6 6 6 5 5 5 5 5 6 6 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5</th><th>556354746556656554555555564</th><th>- 6 3 3 5 5 4 4 4 4 4 4 4 4 4 4 4 4 4 6 6 6 6 6 6 6</th></th></tr>	5 6 3 4 6 3 3 4 3 3 4 3 3 4 4 5 5 5 5 5 5 5 5 5 5	2 4 6 2 2 4 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 4 4 4 3 3 4 3 3 4 3	6 5 5 5 5 5 3 2 6 4 5 5 5 5 3 2 6 4 4 4 4 5 5 5 5 5 5 2 6 4 4 4 4 4 4 4 4 4 4 3 3 4 5 5 2 2	6 4 5 5 2 1 5 4 4 4 4 5 6 6 5 7 7 4 6 5 5 3 3 4 4 4 3		5 3 6 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	- 6 5 4 4 2 2 7 4 4 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 7 4 4 6 5 5 7 4 4 6 6 5 5 7 4 4 6 6 5 5 7 7 4 4 6 6 5 5 7 7 4 6 6 5 5 7 7 4 7 7 7 7 4 7 7 7 7 7 7 7 7 7 7	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	$ \begin{array}{c} $	- 6 5 7 1 2 6 5 4 5 5 4 6 6 6 6 6 5 4 4 4 5 5 4 4 4 3 3 3 3 4 2	6 4 5 4 4 5 5 3 3 5 5 5 5 5 5 5 6 6 6 6 5 5 5 5 5	6 6 6 6 4 3 4 4 5 4 4 4 7 7 7 4 4 7 5 6 5 6 5 6 6 5 6 5 7 7 7 6 6 5 5 4 6 5 5 4 6 5 5 4 6 5 6 4 7 7 6 4 5 5 6 4 5 3 4 5 5 5 6 5 5 5 6 5 6 5 5 5 6 5 6 5 6 5 7 6 6 5 7 6 6 5 7 6 6 7 6 <th>5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 6 5 5 6 6 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 5 5 6 6 6 5 5 5 5 5 6 6 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5</th> <th>556354746556656554555555564</th> <th>- 6 3 3 5 5 4 4 4 4 4 4 4 4 4 4 4 4 4 6 6 6 6 6 6 6</th>	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 6 5 5 6 6 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 5 5 6 6 6 5 5 5 5 5 6 6 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5	556354746556656554555555564	- 6 3 3 5 5 4 4 4 4 4 4 4 4 4 4 4 4 4 6 6 6 6 6 6 6
5 6 3 4 6 3 3 4 3 3 4 3 3 4 4 5 5 5 5 5 5 5 5 5 5	2 4 6 2 2 4 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 4 4 4 3 3 4 3 3 4 3	6 5 5 5 5 5 3 2 6 4 5 5 5 5 3 2 6 4 4 4 4 5 5 5 5 5 5 2 6 4 4 4 4 4 4 4 4 4 4 3 3 4 5 5 2 2	6 4 5 5 2 1 5 4 4 4 4 5 6 6 5 7 7 4 6 5 5 3 3 4 4 4 3		5 3 6 3 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	- 6 5 4 4 2 2 7 4 4 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 6 5 5 7 4 4 6 5 5 7 4 4 6 6 5 5 7 4 4 6 6 5 5 7 7 4 4 6 6 5 5 7 7 4 6 6 5 5 7 7 4 7 7 7 7 4 7 7 7 7 7 7 7 7 7 7	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	$ \begin{array}{c} $	- 6 5 7 1 2 6 5 4 5 5 4 6 6 6 6 6 5 4 4 4 5 5 4 4 4 3 3 3 3 4 2	6 4 5 4 4 5 5 3 3 5 5 5 5 5 5 5 6 6 6 6 5 5 5 5 5	6 6 6 6 4 3 4 4 5 4 4 4 7 7 7 4 4 7 5 6 5 6 5 6 6 5 6 5 7 7 7 6 6 5 5 4 6 5 5 4 6 5 5 4 6 5 6 4 7 7 6 4 5 5 6 4 5 3 4 5 5 5 6 5 5 5 6 5 6 5 5 5 6 5 6 5 6 5 7 6 6 5 7 6 6 5 7 6 6 7 6 <th>5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 6 5 5 6 6 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 5 5 6 6 6 5 5 5 5 5 6 6 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5</th> <th>556354746556656554555555564</th> <th>- 6 3 3 5 5 4 4 4 4 4 4 4 4 4 4 4 4 4 6 6 6 6 6 6 6</th>	5 4 5 3 5 3 6 4 4 6 5 6 4 6 5 5 6 4 4 6 5 5 6 6 5 5 6 6 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 6 6 5 5 5 5 6 6 5 5 5 5 5 5 6 6 5 5 5 5 6 6 5 5 5 6 6 5 5 5 5 5 5 6 6 6 5 5 5 5 5 6 6 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5 5 5 5 5 5 5 5 6 6 5	556354746556656554555555564	- 6 3 3 5 5 4 4 4 4 4 4 4 4 4 4 4 4 4 6 6 6 6 6 6 6							

Appendix I: HC-VBE-F mobile application screenshots and explanations

1. Home Screen	2.1 Register	2.1 Register
¹ ad [*] ad ≈ (\$73% ■) 21:20	⁶ ad %al ≈ 873% ■0 21:21	ीजी *all ॡ (8773% ■) 21:27
Healthcare Virtual Breeding Environment Username Password Password Login as	Please fill in the form below to register Register as Username Username Password Password Confirm Password Confirm Password Dualification Disabled	Qualification Disabled Address Address Email or Phone Number Email or Phone Number Register Cancel
This is the home interface of	This is the registration form of	the HC-VBE mobile application
the HC-VBE mobile	that new users will have to fill	in to register in the system and
application that includes the	includes the following functior	IS:
following functions:	1- Register : This function lets u	users submit their details into
1- Log in: This function lets	the HC-VBE system for registra	tion.
users to login into the HC-	3- Cancel: This function allow	vs users cancel registration if
VBE system.	they wish to do so.	
2- Register: This function		
allows new users to register		
to use the virtual		
environment.		
3- Information: This function		
provides some basic		
information about the HC-		
VBE.		

3. Requesters options	3.1 Manage Profile 1	3.2 New Request
	(Requester)	(Requester)
5.al %all 🖳 (872% 🖬 € 21:30	5ad %ad ≈. 8372% ■ 21:30	ૈની ⁹ ની ૬ ઉ
Poquector's Name: adam ali	Managa Personal Data	Create a pow request
Requesters Name, adam an	Manage Personal Data	Number of Pequired
Manage Profile	Username	Collaborators
New Request	adam ali	No. of Collaborators
Negotiations	Password	Service Description
Virtual Organisations	Password	Service Description
Logout	Confirm Password	Service Constraint
	Confirm Password	Service Constraint
	Qualification	Payment Amount
	Disabled	Payment Amount
	Address	Payment Type
		< ○ □
After a successful log in the	The "Manage Profile"	To make a new healthcare
APP will provide users with a	function is used to make	requester will have to use the
number of optional functions	changes to the requester's	"New request" function.
as below:	profiles. When triggered a	When clicked a request form
1- Manage Profile: Allow	form with relevant filled in	will open for the requester to
users to make changes to	detail of the requester opens	fill in The form cantures the
their personal details in the	where the requester can	details required to initiate an
	make changes and save the	SLA and ultimatoly the VPE
A New request. This	shange	SLA and ultimately the VBE
2- New request: This	change.	creates a vo based on this
function allows requester		request.
users to request a new		
healthcare service.		
3- Negotiation: Allow		
requesters to negotiate the		
offers sent to them by the		
HC-VBE.		
4- Virtual Organisation:		
Allows users to view and		
progress the Created VOs for		
them.		
5- Log out: Allows users to		
log out of the system		
iog out of the system		

3.2 New Request	3.2 New request	3.3 Negotiate
(Requester)	(Requester)	(Requester)
्रित ¹⁸ al द. (3) 71% वा ≥ 21.38	्रिती ¶ती थे. 13 71% ■ 0 21:39	5ad ™ad %. (\$70% IIII > 21.45
Visa card	Sun Sep 29 21:45:00 2019	Negotiate a Service
Payment Constraint	End Date & Time	View My Requests
None	Set End Date & Time	Back to Home
Start Date & Time	Sun Sep 29 22:30:00 2019	
Set Start Date & Time	Duration Constraint	
Sun Sep 29 21:45:00 2019	None	
End Date & Time		
Set End Date & Time	Create Request	
Thu Jan 1 03:00:00 1970	Back to Home	
Duration Constraint		
4 0 0	4 0 0	4 0 🗆
The screenshot above is a continuation of the request form	The screenshot above is a continuation of the request form where the process is finalised by clicking on the "Create Request" button. If no data is missing the request will be submitted to the HC-VBE system for processing.	The third function available to the requester is "Negotiation" when triggered two options will be presented to the requester as below: 1-View My Requests: This function enables the requester to view all the requester to view all the requests made by him/her to the system. It can all show all offers made by providers to provide the service. 2- Back to Home: This function will the requester back to the main page of the application.
3.3.1 Offers made	3.3.1 Offer details	3.3.1 Offer details
---	--	--
(Requester)	(Requester)	(Requester)
tut*at €. 1070% ■ #2140	ین کی	العامة Back Name laura smith Qualification bsc medicine Address paris, france Email or Phone Number l. s@gmail.com
4 0 0	< ○ □	< ○ □
The requester will be notified	When the offer is clicked	The screenshot above shows
of any offer made by a	several options as above will	how the requester sees the
healthcare service provider.	be displayed to the	provider profile detail as a
For more details about the	requester:	result of "Provider's Profile"
offer the requester clicks on the offer.	 1- Negotiate Change in SLA: Requester can use this option to propose a new offer after viewing the offer received by a provider. 2- Accept SLA: This option enables the requester to accept the offer made by a provider without any change. 3-Reset this SLA: This function enables the 	being clicked.

available for offers by other	
healthcare providers and do	
not accept the current	
healthcare provider.	
4- back to Negotiation:	
Enables requesters to go	
back to the negotiation page	
where all offers can be	
viewed.	
5- Provider profile: Enables	
requester view the profile of	
the healthcare provider who	
has made the offer to	
provide the service.	

3.4 Virtual Organisations	3.4.1 SLA View (Requester)	3.4.1 SLA View (Requester)
(Requester)		
Lad ™ad ♥. 0769% ₩2 2159	Lad ™ad 😤 🖉 69% 🖬 2 21:59	ेसं ⁸ .सा दि: 00.09% ≡ 1.22.00 +. 1. 1165∩01011. Эстүрсе
Virtual Organisation History	SLA Details	4.1.1: Properties 4.1.1.1: i feel depressed and need a psychologist to consult with
Running VOs	Clause 1: < <hc-vc-sla>> General 1.1: ID: 2 1.2: Creation Date: Sun Sep 29 21:58:27 2019</hc-vc-sla>	4.1.1.2: OCL Clause 5: < <hc-vc-sla>> Duration</hc-vc-sla>
i feel depresse Your Gollabor- ation	1.3: Status: Active 1.4: MasterID: 2 Clause 2: < <hc-vc-sla>> Collaborator 2.1: Participant: provider 2.1.1: Properties</hc-vc-sla>	5.1: StartDateAndTime: Sun Sep 29 21:45:00 2019 5.2: Constraint: OCL 5.3: EndDateAndTime: Sun Sep 29 22:30:00 2019 5.4: Constraint: OCL
	2.1.1.1: ID: 3 2.1.1.2: Name: laura smith 2.1.1.3: ContactDetails: provider@gmail.com 2.1.1.4: Credentials: N/A 2.1.1.5: Constraints: OCL	Clause 6: < <hc-vc-sla>> Reward 6.1: Payment 6.1.1: Properties: 6.1.1: Amount: 20 6.1.2: PaymentType: visa card</hc-vc-sla>
	Clause 3: < <hc-vc-sla>> Collaborator 3.1: Participant: requester 3.1.1: Properties</hc-vc-sla>	6.1.1.3: PaymentDateAndTime: Sun Sep 29 22:30:00 2019 6.1.1.4: Constraints: OCL
	3.1.1.1: ID: 2 3.1.1.2: Name: adam ali 3.1.1.3: ContactDetails:	Description
Back to Home	requester@gmail.com 3.1.1.3: Credentials: N/A 3.1.1.5: Constraints: OCL	i feel depressed and need a psychologist to consult with
	Clause 4: < <hc-vc-sla>> ExpectedService 4.1: HasRollin: Service 4.1.1: Properties</hc-vc-sla>	Payment Amount
After an offer is accepted by	The screenshot above is the	SLA continued
the requester as a result of	SLA view that is created	
the negotiation process the	between the healthcare	
function" Virtual	requester and provider based	
Organisation" will enable the	on which a VO has been	
requester to view the VO	created for the real-time	
created by the HC-VBE for	collaboration.	
the service as shown in the		
screenshot. Requesters can		
click the "View" option to		
view the SLA approved by the		
requester and provider. Or		
click on the "Start		
Collaboration" option to start		
the real-time VO.		

3.4.1 View SLA (Requester)	3.4.2 Start Collaboration	3.4.3 Real-time video and
	(Requester)	sound communication
5ad %ad ऌ ॐ69% ₩ 3 22:00	(।।।⊂()२२०२०२०।) ात्री [×] नी ॡ ७१२०१२	
End Date & Time	Leave Video Call	and the second second
Sun Sep 29 22:30:00 2019	Time Left: 0H:27M:51S	
Duration Constraint	Server:	
none	You have successfully joined the session	C B
Condio Filo		1700
Download Files		a march
Back to Services		
Requester's Profile		
	Text Message Send	
At the end of the view of the	When the "Start	The screenshot above shows a
SLA. several options are	Collaboration" is triggered	real-time video call
available to the requester	the requester will be	collaboration between the
before the real-time	connected to a session	healthcare requester and
collaboration takes place as	which is basically a virtual	provider after they both chose
below:	room and ready to speak to	to start a video call.
1- Send a File: Requesters	the provider. The requester	
can send a file to the	will have to trigger the Video	
provider for example an X-	call by pressing the "Video	
ray image. The file will be	Call" option, or leave the	
sent to the provider in the	session.	
SLA.		
2-Download files: This		
option allows the requester		
to download any file that		
has been sent by the		
provider.		
3- Back to Service: This		
option will take the		
requester back to the Virtual		
Organisation page.		

 Home Screen (Provider) 	1.2 Current offers (Provider)	1.2.1 offer details (Provider)
⁵ .at [™] adl ⊗. (\$70% 10) 21:49	દતા ^w તાી જે. ઉ?70% ■D 21.49	⁸ .ad ⁹⁸ .ad %. ♥ 21:50
Provider's Name: Jaura smith	Current Offers	SLA Details
Managa Drofilo		Description
	i feel depresse	Description
Service Offers	Requested By: adam ali	to consult with
Negotiations		Payment Amount
Virtual Organisations		20
Logout		Payment Constraint
		none
		Start Date & Time
	Back to Home	
		Sun Sep 29 21:45:00 2019
		End Date & Time
< ○ □		< 0 □
After a successful log in the	This is the view of the	Drovidors can click on the
After a successful log in the	requests sent into the HC-	offer to see the details of the
will have a number of options	VBE that can be seen after	request as shown above.
as below:	the "Service offer" function	
1- Manage Profile: Allows	is triggered.	
provider to make changes to		
their personal details in the		
HC-VBE system.		
2- Service Offers: Allows		
providers to view requests		
sent into the HC-VBE and		
choose to make an offer to		
provide a service.		
3- Negotiation: Allow provider		
to negotiate the offers sent to		
the HC-VBE for a service by		
requesters.		
4- Virtual Organisation: Allows		
providers to view and progress		
the Created VOs for them.		
5- Log out: Allows providers to		
log out of the system		

1.2.1 Offer details (Provider)	1.2.2 Make an Offer	1.3.1 Negotiate (Provider)
	(Provider)	
⁸ ad %al &. 8770% ■ ≥ 21:50	⁶ ad %al ≈ (370% m .) 21:50	⁸ ad %ad ≈. 8970% III ≥ 21:50
Sun Sep 29 21:45:00 2019	Sun Sep 29 21:45:00 2019	Negotiate a Service
End Date & Time	End Date & Time	i feel depresse View
Sun Sep 29 22:30:00 2019	Sun Sep 29 22:30:00 2019	Requested by adam ali
Duration Constraint	Puration Constraint	
none	Success You have successfully joined the service, please go to negotiations to accept the SL 4	
	DONE	
Make an Offer	Make an Offer	
Back to Join Services	Back to Join Services	
Requester's Profile	Requester's Profile	Back to Home
	< ○ □	< ○ □
This screenshot shows the	The above screenshot shows	Once the healthcare provider
continuation of request	the result of making an offer	triggered the "Negotiation"
details where at the end three	by the provider. The HC-VBF	function he will he provided
options are available to the	system notifies the provider	with a link to the offer he/she
	that he has successfully joint	has made in the previous
1- Make an Offer : Allows providers to make and offer	with a requester to start	step. Now the provider has
to the requester who has	service negotiation.	the option to view the details
made the request	service negotiation.	of the request and start
2- Back to Join Services:		negotiating its details
Allows providers to be		
available again to offer		
current intention to offer a		
service.		
3- Requesters Profile: Allows		
the provider to view the		
profile of the requester.		

1.3.2 Negotiate (Provider)	1.3.2 Negotiate (Provider)	1.3.3 Negotiate (Provider)
⁸	ैती ⁸ ती दे अ छि705 ■ >21:51	5 _{all} %all %. (369%) ■ 221:51
SLA Details	Change End Date & Time	Sun Sep 29 21:45:00 2019
Negotiatable Data	Thu Oct 3 23:45:00 2019	End Date & Time
Payment Amount	Duration Constraint	Sun Sep 29 22:30:00 2019
20	none	Duration Constraint
Payment Constraint		none
none	Negotiate Change in SLA	
Start Date & Time	Accept SLA	Awaiting Approval
Change Start Date & Time	Back to Negotioations	Back to Negotioations
Sun Sep 29 21:45:00 2019	Requester's Profile	Requester's Profile
End Date & Time		
		< ○ □
When the "View" is clicked the service request details will be displayed to be viewed.	At the end of viewing the request several options as above will be displayed to the requester: 1- Negotiate Change in SLA: Requester can use this option to propose a new offer after viewing the offer received by a provider. 2- Accept SLA: This option enables the requester to accept the offer made by a provider without any change. 3- back to Negotiation: Enables requesters to go back to the negotiation page where all offers can be viewed. 4- Requester Profile: Allows the provider to view the requesters profile	If the healthcare provider chooses to negotiation a different offer presented in the request, then necessary changes can be made to the request and be sent back to the requester using the "Negotiate Change in SLA" option. Once this is done the status will be changed to "Awaiting Approval" by requester.

124 VO greation	12E VO triggered to start	126 Video call started
(Provider)	(Provider)	(Provider)
Start Q0000 Virtual Organisation History Running VOs i feel depresse Vive	Leave Video Call Time Left: 0H:24M:42S Server: You have successfully joined the session Send Text Message Send	
After an offer is accepted by the requester as a result of the negotiation process the function" Virtual Organisation" will enable the provider to view the VO created by the HC-VBE for the service as shown in this screenshot. providers can click the "View" option to view the SLA approved by the requester and provider. Or click on the "Start Collaboration" option to start the real-time VO. the SLA view will be the same as the ones detailed in the requester screenshots 3.4.1	When the "Start Collaboration" is triggered the provider will be connected to a session which is basically a virtual room and ready to speak to the requester. The provider will have to trigger the Video call by pressing the "Video Call" option, or leave the session.	The screenshot above shows a real-time video call collaboration between the healthcare requester and provider after they both chose to start a video call.

1. Organiser Log in	1.2 Show the Running VOs	1.2.1 List of Running VOs
	(Organiser)	(Organiser)
5 _{ad} %ad \$	5ad ≋al \$: ③ 68% ∎0 22:06	5.ad ≈.ad ≪.(5) (3 \$95%, max 11.45
Healthcare Virtual Breeding Environment	Organiser Section	Virtual Organisation History
Username	Show Running VOs	Running VOs
Org	Logout	i havea back pa
Password		Ended VOs
		i have a bad ne View
Login as		I would like to
Register		
Information		Back to Home
4 0 0	4 0 0	< ○ □
This is the home interface of	After a successful log in the	Triggering the "Show running
the HC-VBE mobile	APP will provide the	VO" will result in showing all
application that includes the	Organisers of the HC-VBE to	the currently running VO and
following functions:	view the detail of all the	the ones that has ended.
1- Log in : This function lets	choosing to show the running	master SLA agreement and
users to login into the HC-VBE	VOs.	the SLA of a particular VO by
system.		choosing to "View" the detail.
2- Register: This function		
allows new users to register		
to use the virtual		
environment.		
provides some basic		
information about the HC-		
VBE.		

1.2.2 Master SLA of a VO	1.2.3 SLA of a VO (Organiser)	1.2.3 SLA of a VO (Organiser)
(Organiser)		
Master Service Level Agreement Clause 1: <+HC-VC-SLA>> Master Details 1.1: MasterID: 2 1.2: CreationDate: Sun Sep 29 21:58:27 2019 1.3: Status: Active 1.4: CollaborationType: Service 1.5: SlaNo 1 1.5: AllSlaID 2 Clause 2: <+HC-VC-SLA>> Changes 2.1: ChangeInSlaID: 2 2.2: ChangeInClauss: Clause Number 2.1: New Details: new details 2.2: 1.2: Affects: clause number 2.1: I New Details: new details 2.2: 1.2: Affects: clause number 2.1: I New Details: new details 2.2: 1.2: Affects: clause number 2.1: I feel depresse by provider_3	Survey Exceeds a service of the serv	All Street All Street
This is the view of the moster.	View of CLA of the chosen	View of CLA of the shares
SLA created for a particular	view of SLA of the chosen	view of SLA of the chosen
VO.	Tequest	Tequest
Here the Organiser can view		
all the SLAs contained in the		
Master SLA by clicking the		
"View" button next to the request name.		

Apendix J: HC-VBE-F-Provider AMOS Data Analysis Output

Notes for Group (Group number 1)

The model is recursive.

Sample size = 100

Variable Summary (Group number 1)

Your model contains the following variables (Group number 1)
Observed, endogenous variables
PHG4
PHG3
PHG2
PHG1
PCE4
PCE3
PCE2
PCE1
PU1
PU2
PU3
PU4
PEU4
PEU3
PEU2
PEU1
AU4
AU3
AU2

AU1	
IU1	
IU2	
IU3	
IU4	
Unobserved, endogenous variables	
Perceived_Clinical_Effectiveness	
Perceived_Usefulness	
Perceived_Ease_of_Use	
Attitude_Towards_Using	
Intention_to_Use	
Unobserved, exogenous variables	
Perceived_Healthcare_Globalisation	
e1	
e2	
e3	
e4	
e5	
e6	
e7	
e8	
e9	

e10		
e11		
e12		
e13		
e14		
e15		
e16		
e17		
e18		
e19		
e20		
e21		
e22		
e23		
e24		
EUL1		
ATL2		
IUL4		
PCEL3		
PEU5		

Variable counts (Group number 1)

Number of variables in your model:	59
Number of observed variables:	24
Number of unobserved variables:	35
Number of exogenous variables:	30
Number of endogenous variables:	29

Result (Default model)

Minimum was achieved

Chi-square = 398.70118

Degrees of freedom = 243

Probability level = .00000

Regression Weights: (Group number 1 - Default model)

			Estim	СЕ	C D	D	Lab
			ate	3.E.	С.К.	٢	el
Perceived_Clinical_E	<-	Perceived_Healthcare	1.07	.13	7.754	***	par
ffectiveness		_Globalisation	686	887	37		_26
Perceived_Ease_of_	<-	Perceived_Healthcare	.837	.15	5.577	***	par
Use		_Globalisation	31	013	20		_27
Perceived_Usefulnes	<-	Perceived_Healthcare	.260	.30	.8598	.38	par
S		_Globalisation	83	335	4	988	_19

			Estim ate	S.E.	C.R.	Р	Lab el
Perceived_Usefulnes s	<- 	Perceived_Clinical_Eff ectiveness	1.05 803	.25 682	4.119 78	***	par _20
Perceived_Usefulnes s	<- 	Perceived_Ease_of_Us e	- .108 54	.09 076	- 1.195 95	.23 171	par _24
Attitude_Towards_U sing	<- 	Perceived_Usefulness	.500 72	.06 930	7.225 51	***	par _21
Attitude_Towards_U sing	<- 	Perceived_Ease_of_Us e	.159 93	.05 532	2.890 74	.00 384	par _22
Intention_to_Use	<- 	Attitude_Towards_Usi ng	1.22 322	.39 625	3.086 96	.00 202	par _23
Intention_to_Use	<- 	Perceived_Usefulness	- .201 26	.22 853	- .8806 7	.37 850	par _25
PHG4	<- 	Perceived_Healthcare _Globalisation	1.00 000				
PHG3	<- 	Perceived_Healthcare _Globalisation	1.25 343	.15 560	8.055 54	***	par _1
PHG2	<- 	Perceived_Healthcare _Globalisation	1.37 575	.15 235	9.030 24	***	par _2
PHG1	<- 	Perceived_Healthcare _Globalisation	1.19 950	.15 514	7.731 88	***	par _3

			Estim ate	S.E.	C.R.	Р	Lab el
PCE4	<- 	Perceived_Clinical_Eff	.992 36	.09 455	10.49 573	***	par 4
PCF3	<-	Perceived_Clinical_Eff	1.04	.11	8.937	***	_ · par
		ectiveness	639	709	01		_5
PCE2	<-	Perceived_Clinical_Eff	.985	.10	9.773 76	***	par
		Perecived Clinical Eff	94	088	70		_0
PCE1	<- 	ectiveness	000				
PU1	<- 	Perceived_Usefulness	1.00 000				
PU2	<- 	Perceived_Usefulness	.882 65	.08 945	9.867 16	***	par _7
PU3	<- 	Perceived_Usefulness	.972 93	.08 594	11.32 076	***	par _8
PU4	<- 	Perceived_Usefulness	.828 39	.09 162	9.041 43	***	par _9
PEU4	<- 	Perceived_Ease_of_Us e	.936 48	.12 057	7.767 39	***	par _10
PEU3	<- 	Perceived_Ease_of_Us e	1.00 000				
PEU2	<- 	Perceived_Ease_of_Us e	.934 12	.10 493	8.902 57	***	par _11

			Estim ate	S.E.	C.R.	Ρ	Lab el
PEU1	<- 	Perceived_Ease_of_Us e	1.24 873	.12 508	9.983 78	***	par _12
AU4	<- 	Attitude_Towards_Usi ng	1.18 548	.15 202	7.798 35	***	par _13
AU3	<- 	Attitude_Towards_Usi ng	1.20 138	.14 829	8.101 48	***	par _14
AU2	<- 	Attitude_Towards_Usi ng	1.07 963	.13 001	8.304 08	***	par _15
AU1	<- 	Attitude_Towards_Usi ng	1.00 000				
IU1	<- 	Intention_to_Use	1.00 000				
IU2	<- 	Intention_to_Use	.934 14	.12 545	7.446 18	***	par _16
IU3	<- 	Intention_to_Use	1.09 384	.11 919	9.177 46	***	par _17
1U4	<- 	Intention_to_Use	.894 54	.10 188	8.780 09	***	par _18

Standardized Regression Weights: (Group number 1 - Default model)

			Estimate
Perceived_Clinical_Effectiveness	<	Perceived_Healthcare_Globalisation	.89755
Perceived_Ease_of_Use	<	Perceived_Healthcare_Globalisation	.63540
Perceived_Usefulness	<	Perceived_Healthcare_Globalisation	.17187
Perceived_Usefulness	<	Perceived_Clinical_Effectiveness	.83645
Perceived_Usefulness	<	Perceived_Ease_of_Use	09425
Attitude_Towards_Using	<	Perceived_Usefulness	.80107
Attitude_Towards_Using	<	Perceived_Ease_of_Use	.22216
Intention_to_Use	<	Attitude_Towards_Using	1.02106
Intention_to_Use	<	Perceived_Usefulness	26876
PHG4	<	Perceived_Healthcare_Globalisation	.75022
РНG3	<	Perceived_Healthcare_Globalisation	.79234
PHG2	<	Perceived_Healthcare_Globalisation	.87858
PHG1	<	Perceived_Healthcare_Globalisation	.76378
PCE4	<	Perceived_Clinical_Effectiveness	.84505
PCE3	<	Perceived_Clinical_Effectiveness	.76204
PCE2	<	Perceived_Clinical_Effectiveness	.80825
PCE1	<	Perceived_Clinical_Effectiveness	.84028
PU1	<	Perceived_Usefulness	.81504

		Estimate
PU2	< Perceived_Usefulness	.83360
PU3	< Perceived_Usefulness	.91167
PU4	< Perceived_Usefulness	.78483
PEU4	< Perceived_Ease_of_Use	.70967
PEU3	< Perceived_Ease_of_Use	.84552
PEU2	< Perceived_Ease_of_Use	.78430
PEU1	< Perceived_Ease_of_Use	.85545
AU4	< Attitude_Towards_Using	.75212
AU3	< Attitude_Towards_Using	.77697
AU2	< Attitude_Towards_Using	.79340
AU1	< Attitude_Towards_Using	.76555
IU1	< Intention_to_Use	.78496
IU2	< Intention_to_Use	.72183
IU3	< Intention_to_Use	.86332
1U4	< Intention_to_Use	.82925

Model Fit Summary

CMIN

Model	NPAR	CMIN	DF	Р	CMIN/DF
Default model	57	398.70118	243	.00000	1.64075
Saturated model	300	.00000	0		
Independence model	24	2099.36512	276	.00000	7.60640

RMR, GFI

Model	RMR	GFI	AGFI	PGFI
Default model	.05892	.76948	.71541	.62328
Saturated model	.00000	1.00000		
Independence model	.44431	.14138	.06672	.13007

Baseline Comparisons

Madal	NFI	RFI	IFI	TLI	CEL	
Model	Delta1	rho1	Delta2 rho2			
Default model	.81008	.78429	.91613	.90301	.91461	
Saturated model	1.00000		1.00000		1.00000	
Independence model	.00000	.00000	.00000	.00000	.00000	

Parsimony-Adjusted Measures

Model	PRATIO	PNFI	PCFI
Default model	.88043	.71323	.80525
Saturated model	.00000	.00000	.00000
Independence model	1.00000	.00000	.00000

NCP

Model	NCP	LO 90	HI 90
Default model	155.70118	104.84277	214.46261
Saturated model	.00000	.00000	.00000
Independence model	1823.36512	1681.33610	1972.81631

FMIN

Model	FMIN	FO	LO 90	HI 90
Default model	4.02728	1.57274	1.05902	2.16629
Saturated model	.00000	.00000	.00000	.00000
Independence model	21.20571	18.41783	16.98319	19.92744

RMSEA

Model	RMSEA	LO 90	HI 90	PCLOSE
Default model	.08045	.06602	.09442	.00055

Model	RMSEA	LO 90	HI 90	PCLOSE
Independence model	.25832	.24806	.26870	.00001

AIC

Model	AIC	BCC	BIC	CAIC
Default model	512.70118	551.21470	661.19589	718.19589
Saturated model	600.00000	802.70270	1381.55106	1681.55106
Independence model	2147.36512	2163.58133	2209.88920	2233.88920

ECVI

Model	ECVI	LO 90	HI 90	MECVI
Default model	5.17880	4.66508	5.77235	5.56783
Saturated model	6.06061	6.06061	6.06061	8.10811
Independence model	21.69056	20.25592	23.20016	21.85436

HOELTER

Madal	HOELTER	HOELTER
Wodel	.05	.01
Default model	70	74
Independence model	15	16